Capturing, exploring and understanding the experiences of undergraduate student nurses undertaking a Pre-registration nursing degree course: An Experience Based Co-design study

by

Samantha Jean Pollitt

A thesis submitted in partial fulfilment for the requirements for the degree of Doctor of Philosophy, at the University of Central Lancashire

September 2021



STUDENT DECLARATION FORM

Type of Award

Doctor of Philosophy (PhD)

School

Nursing

1. Concurrent registration for two or more academic awards

I declare that while registered as a candidate for the research degree, I have not been a registered candidate or enrolled student for another award of the University or other academic or professional institution

2. Material submitted for another award

I declare that no material contained in the thesis has been used in any other submission for an academic award and is solely my own work

3. Collaboration

Where a candidate's research programme is part of a collaborative project, the thesis must indicate in addition clearly the candidate's individual contribution and the extent of the collaboration. Please state below: N/A

4. Use of a Proof-reader

No proof-reading service was used in the compilation of this thesis.

Spallitt Signature of Candidate

Print name: Samantha Jean Pollitt

ABSTRACT

Background: Nurse education in the UK has undergone significant changes in the last few decades such as the move to an all-graduate profession, the introduction of self-funded tuition fees and the new Nursing and Midwifery Council (NMC) education standards that change the way student nurses are supervised and assessed in clinical practice. Uncertainties about how these changes may impact student recruitment, expectations and experience of their course abound. Understanding and improving the quality of the student experience may enhance retention and reduce attrition. Yet the experiences of Preregistration (Adult) nursing students' is under-researched in the UK.

Aims: To capture, explore and understand the experiences of students' undertaking the Preregistration (Adult) nursing course, and co-design potential improvements to future experiences, through the application of an Experienced Based Co-design (EBCD) methodology.

Methodology: Using EBCD, a user design methodology, semi-structured interviews were conducted with student nurses (n=8), and emotional touchpoint mapping with student nurses (n=7) and key stakeholders (n=8) to gain insight into what shapes students' experience on the nursing course. Findings were analysed using thematic analysis. EBCD aims to both understand and improve user experiences of a service by co-designing improvement strategies for that service. Therefore, two Co-design events were conducted that enabled students (n=7) and key stakeholders (n=11) to collaboratively produce priority recommendations for course improvement.

Findings: Students had varied experiences on the nursing course, both positive and negative, particularly with student support. Expectations of sufficient support was evident at the beginning, and throughout the student's journey of the nursing course and appeared to align with students' sense of learner agency. Three priority recommendations for course improvement were identified from the findings including: facilitate and support student development of independent study skills, enhance student support in the clinical practice placement environment and clarify and enhance the role of the Academic advisor.

Implications: Findings from this study adds to existing but dearth research in the field of the Pre-registration (Adult) nursing student experience. It has highlighted areas for improvement on the Pre-registration nursing course that could impact future students' experience. Furthermore, this study appears to be the first documented as using EBCD in a higher education setting with the foci on students, that enabled students and key stakeholders in the nursing course to co-design priority recommendations for course improvement.

ii

TABLE OF CONTENTS

ABSTRACT	ii
TABLE OF CONTENTS	iii
LIST OF TABLES	vii
LIST OF FIGURES	viii
ACKNOWLEDGEMENTS	ix
GLOSSARY OF TERMS	x
CHAPTER ONE: INTRODUCTION	1
1.1 BACKGROUND TO THE STUDY	1
1.2 CONCEPTUALISING THE STUDENT EXPERIENCE	
1.3 PLACE OF THE RESEARCHER WITHIN THE STUDY	6
1.4 The studies contribution to knowledge	7
1.5 Thesis structure	7
1.6 Conclusion	9
CHAPTER TWO: LITERATURE REVIEW	10
2.1 INTRODUCTION	
2.2 Literature Search Strategy	
2.3 Critiquing framework	14
2.4 The knowledge gap	17
2.5 SUMMARY OF LITERATURE CHOSEN FOR REVIEW	
2.7.1 Transition into higher education	
2.7.2 Expectations of academic learning	24
2.7.3 Adapting to the role of university student	
2.8 STUDENT NURSE EXPERIENCE OF CLINICAL PRACTICE	29
2.8.1 Transition into the clinical practice environment	
2.8.2 Experience of clinical practice supervision	
2.8.3 Experience of socialisation, acculturation and sense of belonging within the community	of practice
	33 35
CHAPTER THREE: METHODOLOGY	3/
3.1 INTRODUCTION	
3.2 THEORETICAL FRAMEWORK UNDERPINNING THIS STUDY	
3.3 RESEARCH PARADIGM	
3.3.1 Epistemology	
3.3.2 Theoretical perspective	
3.3.3 Research methodology	
3.3.4 Experienced Based Co-Design	
3.3.4.1 Touchpoints	
3.3.5 Experienced Based Co-Design model	
3.4 STAGE 1 - SETTING UP THE PROJECT	51
3.4.1 Research Aim	
3.4.2 Kesearch Objectives	

3.4.3 Ethical considerations	51
3.4.4 Equipment and facilities	54
3.4.5 Confidentiality and storage of data	54
3.5 Stage 2 - Engaging students and gathering experiences	55
3.5.1 Sampling strategy, student demographics	56
3.5.2 Student participant semi-structured interviews	61
3.5.3 Student participant emotional touchpoint mapping event	61
3.6 Stage 3 - Engaging Key stakeholders and gathering experiences	64
3.6.1 Key stakeholder emotional touchpoint mapping event	64
3.6.2 Key stakeholder demographics	65
3.7 Stage 4 - The Student and Key stakeholder Co-design event	66
3.8 Stage 5 - Small co-design working groups	67
3.9 DATA ANALYSIS	69
3.9.1 Thematic analysis	71
3.9.2 Trustworthiness	75
3.10 CHALLENGES TO USING EXPERIENCE BASED CO-DESIGN	76
3.11 CONCLUSION	77
CHAPTER FOUR: PRESENTATION OF FINDINGS FROM SEMI-STRUCTURED STUDENT INTERVIEWS	79
4.1 INTRODUCTION	79
4.2 NAVIGATING THE ACADEMIC JOURNEY	83
4.2.1 The journey to Pre-registration nursing - Course enquires and the Interview and selection proc	ess 83
4.2.2 Transition through year one - Adjusting to being a student	85
4.2.3 Transition through year two – Managing the academic workload	89
4.2.4 Transition through year three - Preparing to become a newly qualified nurse	91
4.2.5 The classroom learning experience and the role of academic staff	94
4.2.6 Engagement with academic assessment, support and feedback	98
4.3 THE CLINICAL PRACTICE LEARNING EXPERIENCE	101
4.3.1 The workplace as a learning environment	102
4.3.2 Accessing support in the clinical setting: the role of mentors and Practice Education Facilitator	rs 105
4.4 Emergent Networks of Support	109
4.4.1 Seeking support from peers (Home groups)	109
4.4.2 Expectations of the Academic advisor	112
4.4.3 The influence of social media	114
4.5 CONCLUSION	117
CHAPTER FIVE: EMOTIONAL TOUCHPOINT MAPPING	118
5.1 INTRODUCTION	118
5.2 DESCRIPTION OF THE EMOTIONAL TOUCHPOINT MAPPING EVENTS	118
5.3 FINDINGS FROM THE STUDENT EMOTIONAL TOUCHPOINT MAPPING EVENT	123
5.4 FINDINGS FROM THE KEY STAKEHOLDER EMOTIONAL TOUCHPOINT MAPPING EVENT	128
5.5 SUMMARY OF CORE THEMES AND SUBTHEMES CAPTURED FROM THE STUDENT AND KEY STAKEHOLDER EMOTIONAL	
TOUCHPOINT MAPPING EVENTS	134
CHAPTER SIX: FINDINGS FROM THE STUDENT EMOTIONAL TOUCHPOINT MAPPING EVENT	136
	126
6.2 STUDENT PERCEPTIONS OF DISPARATE CUNICAL PRACTICE LEARNING EXPERIENCES	130
6.3 THE STUDENT EXPERIENCE OF THE CLINICAL MENTORING PROCESS	139
6.3.1 The effects of a negative mentor experience	140
6.3.2 Mentors as gatekeepers to student learning in the clinical practice environment	142
6.4 SUPPORT WITH ACADEMIC STUDY - STUDENT PERCEPTIONS OF ABANDONMENT.	144
6.5 THE INFLUENCE OF ACADEMIC STAFF ON THE STUDENT EXPERIENCE	147

6.6 THE EFFECT OF DISPARATE APPROACHES TO ACADEMIC ADVISOR SUPPORT	
6.7 CONCLUSION	
CHAPTER SEVEN: PRESENTATION OF FINDINGS FROM THE KEY STAKEHOLDER EMOTIONAL TOUCH	POINT
MAPPING EVENT	
7.1 INTRODUCTION	
7.2 ACADEMIC ADVISOR SUPPORT - PERCEIVED AMBIGUITY ASSOCIATED WITH THE ROLE	
7.3 ACADEMIC SUPPORT - NAVIGATING THE TENSION BETWEEN SUPPORT AND INDEPENDENCE	
7.4 'BEING' OR 'BECOMING' PROFESSIONAL - KEY STAKEHOLDERS' EXPECTATIONS OF STUDENTS	
7.4.1 'Student' or 'Student nurse' - what is in a name?	
7.5 CONCLUSION	
CHAPTER FIGHT: PRIORITIES AND ACTIONS FOR COURSE IMPROVEMENT FROM THE STUDENT ANI) KFY
STAKEHOLDER CO-DESIGN EVENTS	
8.1 INTRODUCTION	
8.2 Overview of the student and key stakeholder Co-design event	
8.3 Co-design using the KJ brainstorming technique	
8.3.1 KJ brainstorming technique step 1	
8.3.2 KJ brainstormina techniaue step 2	
8.3.3 KJ brainstormina techniaue step 3	
8.3.4 KJ brainstormina techniaue step 4	
8.4 PRIORITY 1 - FACILITATE AND SUPPORT STUDENT DEVELOPMENT OF INDEPENDENT STUDY SKILLS	
8.5 PRIORITY 2 - ENHANCE STUDENT SUPPORT IN THE CLINICAL PRACTICE PLACEMENT ENVIRONMENT	
8.6 PRIORITY 3 - CLARIFY AND ENHANCE THE ROLE OF THE ACADEMIC ADVISOR	
8.7 CONCLUSION	
CHAPTER NINE: DISCUSSION	
	402
9.1 INTRODUCTION TO DISCUSSION	
9.2 CONTEXTUALIZING THE FINDINGS ON THE STUDENT EXPERIENCE OF THE BSC PRE-REGISTRATION (ADULT) NURSING	J COURSE
9 2 1 Framina student sunnort	182 186
9 3 COGNITIVE STUDENT SUPPORT	188
9 3 1 Support with transition onto the course	189
9 3 2 Support with transition through year two and year three of the course	193
9 3 3 Support with learning and assessment	195
9 3 3 1 Facilitating student development of independent study skills	196
9 3 4 Support with transition into the clinical practice placement environment	199
9.3.4.1 Preparing students for the clinical practice placement experience	
9.4 Relational student Support	
9.4.1 Clinical practice placement support through the mentorship model	
9.4.2 Academic advisor support	
9.5 AFFECTIVE STUDENT SUPPORT	
9.5.1 Supporting students develop learner agency	
9.5.2 Dispositional understanding of self, active, intentional, and possessing purpose	
9.5.3 Motivational beliefs, ability to act on beliefs, persisting and making choices	
9.5.4 Positional experiences within social context and how situational variables inform agency	-
interactions and negotiations	
- 9.6 Systemic student support	
9.7 IMPLICATIONS OF THIS STUDY FOR PRE-REGISTRATION NURSING EDUCATION SYSTEMIC	
9.8 REFLECTIONS ON THE USE OF EXPERIENCE BASED CO-DESIGN IN A HIGHER EDUCATION SETTING	
9.8.1 Implications of this study for Experience Based Co-design	
9.9 LIMITATIONS OF THE STUDY	227

9.10 DISSEMINATION OF FINDINGS				
9.11 CONCLUSION				
REFERENCES	232			
BIBLIOGRAPHY	256			
LIST OF APPENDICES	257			
APPENDIX 1: 12-STEP GUIDELINE FOR STRUCTURING AND DOCUMENTING A LITERATURE SEARCH STRATEGY				
APPENDIX 2: LIST OF LITERATURE IDENTIFIED AS ELIGIBLE FOR REVIEW	259			
APPENDIX 3: CRITERIA FOR APPRAISING QUALITATIVE AND MIXED METHODS RESEARCH STUDIES				
APPENDIX 4: LITERATURE SUMMARY TABLE				
APPENDIX 5: CRITICAL APPRAISAL OF THE QUANTITATIVE SURVEYS USING THE 'CRITICAL APPRAISAL OF A SURVEY' (C	EBM) TOOL			
APPENDIX 6: STEMH ETHICS COMMITTEE APPROVAL				
APPENDIX 7: PARTICIPANT INFORMATION SHEET (PIS) / EXPRESSION OF INTEREST FORM				
APPENDIX 8: STUDY FLYER FOR STUDENT PARTICIPANTS				
APPENDIX 9: STUDENT PARTICIPANT CONSENT FORM FOR SEMI-STRUCTURED INTERVIEWS				
APPENDIX 10: SEMI-STRUCTURED INTERVIEW TOPIC GUIDE				
APPENDIX 11: STUDENT PARTICIPANT CONSENT FORM FOR EMOTIONAL TOUCHPOINT MAPPING EVENT				
APPENDIX 12: KEY STAKEHOLDER PARTICIPANT INFORMATION SHEET (PIS) AND CONSENT FORM				
APPENDIX 13: STUDENT AND KEY STAKEHOLDER PARTICIPANT CONSENT FORM FOR CO-DESIGN EVENT				
APPENDIX 14: PROBLEM STATEMENTS TABULATED FROM STUDENT AND KEY STAKEHOLDER EMOTIONAL TOUCHPOIN	T MAPPING			
EVENTS				
APPENDIX 15: MAP OF THE STUDENT JOURNEY THROUGH THE COURSE				
APPENDIX 16: EXAMPLE THEMATIC ANALYSIS FROM STUDENT EMOTIONAL TOUCHPOINT MAPPING EVENT				
APPENDIX 17: POST-IT NOTE COMMENTS FROM THE STUDENT EMOTIONAL TOUCHPOINT MAPPING EVENT				
APPENDIX 18: COMMENTS' STUDENT PARTICIPANTS STRONGLY AGREE WITH				
APPENDIX 19: COMMENTS' STUDENT PARTICIPANTS STRONGLY DISAGREE WITH				
APPENDIX 20: POST-IT NOTE COMMENTS FROM THE KEY STAKEHOLDER EMOTIONAL TOUCHPOINT MAPPING EVENT .				
APPENDIX 21: COMMENTS KEY STAKEHOLDER PARTICIPANTS STRONGLY AGREED WITH				
APPENDIX 22: COMMENTS KEY STAKEHOLDER PARTICIPANTS STRONGLY DISAGREE/FELT SADDENED BY				

LIST OF TABLES

Table 1 - Quality rating (Downe et al., 2007)	15
Table 2 - Emergent themes and subthemes from literature review	16
Table 3 - Student participant inclusion/exclusion criteria	57
Table 4 - Participant sample	58
Table 5 - Student participant demographics	59
Table 6 - Key stakeholder demographics from emotional touchpoint mapping event and Co-desig	n
event	65
Table 7 - KJ brainstorming technique adapted for Co-design event	68
Table 8 - Excerpt from key stakeholder emotional touchpoint mapping event transcript	134
Table 9 - Priority 1, Facilitate and support student development of independent study skills	176
Table 10 - Priority 2, Enhance student support in the clinical practice placement environment	178

LIST OF FIGURES

Figure 1 - Literature search strategy	13
Figure 2 - The four steps of EBCD	48
Figure 3 - Adapted Experience Based Co-design process	50
Figure 4 - Data collection phases	56
Figure 5 - Representation of respondents	60
Figure 6 - Example of a subjective experience map	63
Figure 7 - Co-design event discovery stage, image 1	67
Figure 8 - Co-design event discovery stage, image 2	67
Figure 9 - Small co-design working groups recording and sharing ideas and solutions	68
Figure 10 - Grouping or Clustering ideas and solutions	69
Figure 11 - Braun and Clarke (2006) Six phases of thematic analysis	73
Figure 12 - Example Touchpoint	80
Figure 13 - Overview of core themes, subtheme and touchpoints from semi-structured interview .	81
Figure 14 - Touchpoints on the nursing students' journey through the course	82
Figure 15 - Core theme 1 'Navigating the academic journey'	83
Figure 16 - Core theme 2 'The clinical practice learning experience'	101
Figure 17 - Core theme 3 'Emergent networks of support'	109
Figure 18 - Image of student journey map	119
Figure 19 - Student emotional touchpoint mapping event	120
Figure 20 - Student strongly agree stickers	120
Figure 21 - Student strongly disagree stickers	120
Figure 22 - Example of touchpoints, verbatim quotes, post-it notes and stickers	121
Figure 23 - Key stakeholder stickers	122
Figure 24 - Post-it note comment affixed to 'mentor support' touchpoint	124
Figure 25 - Number/distribution of student positive/negative emotions	126
Figure 26 - Number/distribution of student strongly agree/disagree stickers	127
Figure 27 - Significant touchpoints identified on the student journey map	128
Figure 28 - Number/distribution of Key stakeholder positive/negative emotions	130
Figure 29 - Number/distribution of Key stakeholder strongly agree/disagree stickers	132
Figure 30 - Key stakeholder post-it notes comments	133
Figure 31 - Key stakeholder post-it notes, image 1	133
Figure 32 - Core themes/subthemes from emotional touchpoint mapping events	135
Figure 33 - Findings from the student emotional touchpoint mapping event	137
Figure 34 - Findings from the key stakeholder emotional touchpoint mapping event	153
Figure 35 - Adaption of double diamond design model for Co-design event	168
Figure 36 - Exploring and discussing the student journey map	169
Figure 37 - Small co-design working groups	170
Figure 38 - Step 1, individual brainstorming	1/1
Figure 39 - Step 2, Grouping and clustering ideas and solutions	172
Figure 40 - Forming affinity diagram, image 1	1/2
Figure 41 - Forming annity diagram, image 2	1/3
Figure 42 - Priority improvement from Co-design event 1	174
Figure 43 - Fhority improvement from Co-design event Z	1/4
Figure 44 - A model of student agency (Vaugnin, 2020)	213
rigure 45 - rramework for documenting the relationship between core themes and touchpoints	221

ACKNOWLEDGEMENTS

I would like to thank my Lord and saviour Jesus Christ for giving me strength, hope and clarity throughout the process of producing this thesis. I would like to thank my husband Mark for his patience, support and encouragement, I am only where I am because you always believe in me. Thank you to my sons Ben and Lewis who were boys when I started this process but are now young men that I am proud of. I would also like to thank my supervisors Dr Rob Monks, Professor Dawne Gurbutt and Professor Nigel Harrison for their unwavering support, guidance and time spent reading endless drafts! Many thanks to Dr Jean Duckworth for so kindly offering to supervise me towards the end of my thesis, your fresh perspective was encouraging.

Thank you Tony and Sue Woodham (Dad and Mum), your prayers have got me through (as always). A special thanks goes to my friends and colleagues Yvonne Whiteley, Tara Brookes, Dr Sabina Gerrard, Dr Sue Quayle and Gemma Spencer who have cheered me on every step of the way. You all kept me going when I was ready to give up.

I am eternally grateful to you all.

GLOSSARY OF TERMS

Academic advisor - A member of the academic staff, usually a lecturer or senior lecturer, who supports a group of students pastorally for the 3-year duration of the nursing course.

Field - A field of nursing relates to the specific part of the Nursing and Midwifery Council nursing register student nurses enter on qualification. There are four fields of nursing practice including Adult, Children, Learning disabilities and Mental health. For this study participants were recruited from the 'Adult' field of nursing.

Cohort - A group of students who commence the same field of the Pre-registration nursing course on the same date and proceed through a programme together at the same time, for example September 2017 cohort.

Mentor - A registered nurse, midwife, or health and social care professional who facilitates learning, supervises, and assesses students in a clinical practice setting.

Personal tutor - The term 'Personal tutor' is used interchangeably with 'Academic advisor' by participants in this study and denotes the same role (See Academic advisor term above).

Services - For the purposes of this study 'services' refer to any education service that forms part of the Pre-registration nursing course the course, for instance facilitated learning sessions, interaction with Blackboard Learn and undertaking a work-based learning placement

Touchpoint - Are key moments or events that stand out for students on the nursing course. Some touchpoints are intense points of contact with an aspect of the course that can positively or negatively shape a student's experience.

Clinical practice placement - Also referred to as work-based learning in some university settings, the clinical practice placement is where students spend a set period of their course in a healthcare environment to achieve learning their learning outcomes while engaging in patient care and delivery of public health services.

Supernumerary - In clinical practice placements students have a 'supernumerary' status whereby they are not included as part of the staffing numbers but have some form of supported or protected learning time (NMC, 2019).

х

CHAPTER ONE INTRODUCTION

'Education is not preparation for life; education is life itself' (Dewey, 1972)

1.1 Background to the study

This study captures the experiences of student nurses undertaking a BSc Pre-registration (Adult) nursing course and explored their journey through the course for the purpose of better understanding what shapes that experience and identify ways to improve future experiences on the course. The inspiration for this study originated from over twenty years' experience delivering the pre-registration adult nursing course and supporting students undertaking the course in a university in the Northwest of England. In that time pre-registration nurse education had undergone significant reforms with a potential to influence the overall student experience. For example, after the transfer of pre-registration nurse education from schools of nursing to Higher Education Institutions (HEI), the move towards an all-graduate profession began in 2008 following the Nursing and Midwifery Council (NMC) decision for a bachelor's degree to be the minimum academic level for future gualifying nurses (NMC, 2010). Policy drivers for these reforms in nurse education include the framework document 'Framing the Nursing and Midwifery Contribution: Driving up the Quality of Care' (Department of Health (DH), 2008) and the Willis report (2015) 'The Shape of Caring Review; Raising the Bar'. As a result of these changes nursing students became the second largest student body in universities after business study students (Willis commission, 2012; Higher Education Statistics Agency (HESA), 2021). Furthermore, a review of higher education funding and student finance resulted in a departure from student nurses tuition fees being paid in full by the National Health Service (NHS) to the introduction of self-funded tuition fees of over £9000 for student nurse education in England where the university under study is based (Browne, 2010). My interest was in understanding what current students experiences of the preregistration nursing course was like on the back of the changes described above.

A further driver for the study was the notable drop in applicants for undergraduate nursing courses of >35% since the introduction of fees and simultaneously rise in attrition rates for pre-registration nursing courses (Universities and Colleges Admissions Service (UCAS), 2018). Undergraduate nursing course appeared to be losing more students than any other academic course, with UK attrition rates for pre-registration nurses averaging 24% (Buchan et al., 2019). Reasons articulated for the high attrition rates, or the loss of students from the

nursing programme, are varied and complex, such as nursing education programmes not meeting students' expectations, a demanding academic workload, poor clinical practice placement experience and lack of academic support (Chan et al., 2019). The widening access agenda that yielded an increase in non-traditional entrants has been proposed as a further influential factor on attrition (Brimble, 2015; Heaslip, 2017). The National Audit Office (NAO) (2007) in their report on student retention suggest much of what an institution does affects the quality of the student experience, and that institutions should act to improve the student experience, ultimately contributing to a better-quality education and leading to better value for money for students. Consequently, the quality of the student experience and the services provided whilst undertaking their course will play an increasingly important role in attracting and retaining pre-registration nurses. Identifying and understanding the students' experiences of their course is a necessary step in the move towards improving the student experience. Yet student experience of higher education is unusually under-researched in the UK, particularly when compared to the United States (US) and Australia where studies of student experience have been conducted for decades (Yorke & Longdens, 2008; Rosborough, 2018). What is more, student experience research tends to focus on the general student population which sometimes includes undergraduate nurses, rather than specifically on student nurses.

Nursing is a multifaceted profession, as a result nursing courses are often complex requiring nursing students to not only accrue knowledge but become competent, skilful practitioners before graduation. This means nursing students are exposed to a diverse range of learning situations early in their course such as classroom-based teaching, skills laboratories and 'real world' clinical practice. As such their student experience, although sharing some similarities to traditional university students, tend overall to be different than non-health related students' experiences of the curriculum. For instance, pre-registration nursing courses are delivered over a full academic year (three semesters) to meet the Nursing and Midwifery Council (NMC) minimum standard of 4600 hours divided equally between theory and practice learning over the three-year programme (NMC, 2010). Whereas other courses tend to be delivered over two academic semesters. This constrains nursing student's time to fully embrace student life or find part-time jobs over a summer period to support their studies. Furthermore, in 2018 the NMC introduced significant changes to the way student nurses practice, are supervised and assessed in the workplace learning environment (NMC, 2018a). The new NMC (2018a) standards governing curriculum design and delivery of undergraduate nursing courses will directly influence the overall student experience.

During my own experience of delivering education predominantly to undergraduate adult field nurses as a senior lecturer I had observed students have different experiences of what was evidently the same course, with some students appearing to have a poorer experience overall than other students. As a result I sought to understand the experiences of student nurses on an adult field undergraduate course. Exploring the available literature on student experience of pre-registration (Adult) nursing courses revealed a paucity of research when compared with the literature on Mental health and Child pre-registration nursing. Of the studies conducted the focus was almost exclusively on classroom/learning experiences or learning in the clinical practice environment. Very few studies explored the student experience of a BSc Pre-registration nursing curriculum from pre-entry to graduation. Furthermore, most published studies identified were surveys undertaken as part of the Quality Assurance Agency (QAA) process, aimed specifically at students learning and teaching experiences such as the Higher Education Academy 'Student Academic Experience Survey' (Neves & Hillman, 2016). Similarly, much of the literature on the experiences of student nurses focused on specific, and often narrow, aspects of that experience such as student nurse identity (Grealish & Trevitt, 2005; Andrew et al., 2011; Walker et al., 2014); transition from student nurse to graduate nurse (Kumaran & Carney, 2014; Malouf & West, 2015) and the guality of clinical practice placements (Courtney-Pratt et al., 2012). However, my interest lay in understanding both academic and non-academic aspects of the student experience or the 'total student experience' as espoused by Harvey et al. (1992). Notably, there were no published studies aimed at exploring the student nurse experience using an Experienced Based Co-Design (EBCD) model. An EBCD process was chosen as it enabled me to explore the student nurse journey through the whole course from pre-entry to the point of qualification. Therefore, this study is the first documented to use elements of EBCD as a methodology to capture, explore and understand the experiences of student nurses undertaking a BSc Preregistration (Adult) nursing course, and enable student and key stakeholder participants to collaboratively identify and co-design course improvement priorities that potentially enhance future student experience of the nursing course.

1.2 Conceptualising the student experience

Articulating the concept of the 'student experience' is fraught with challenges, not least because an experience is unique to the individual describing that experience and can be influenced by a multitude of internal (personal) and external factors. Within education much of the literature on experience pertains to 'experiential learning' or the process of learning through experience as advocated by early educational theorists such as Dewey (1938/2015),

Lewin (1951) and Kolb (1984). However, 'experiential learning' theories aptly focus largely on the learning process and the experience of learning as a concrete activity, whereas a student's experience encompasses all aspects of student life (i.e. academic, social, welfare and support) with learning at the heart of it (Morgan, 2012).

The term 'student experience' was first used by Harvey et al. (1992) in their survey of HEIs analysing the 'total student experience'. Their initial report, in which they coined the term, was part of a national survey of over 4000 respondents including students, teaching, non-teaching and research staff (Harvey et al., 1992). They suggested that the 'student experience' was a key factor in the assessment of 'quality' in higher education, and the student experience was not restricted to the classroom but was a multifaceted concept (Harvey et al., 1992). Similarly, Temple et al. (2014, p.3) state the student experience is "the totality of a student's interaction with the institution". For nursing students their experience extends beyond the university and includes their experiences within the clinical practice environment predominantly within the NHS organisation.

Mukerji and Tripathi (2014) see the student experience as the value-added component that a student anticipates receiving from their course once enrolled. Since Harvey et al. (1992) initial survey multiple studies have attempted to deconstruct the term and identify those facets that make up the student experience, yet a universal definition of the student experience remains elusive and difficult to define (Benckendorff et al., 2009; Mujtaba, 2012) with some authors even suggesting it is "fruitless to attempt to define it as a single thing" (The 1994 Group, 2007, p.5). Instead, most authors accept the student experience as a wide-ranging concept meaning different things to different students and academic staff (The 1994 group, 2007; Ainley, 2008; Gibney et al., 2011; Mujtaba, 2012). In addition, because student experience is influenced by a variety of different factors an understanding of what constitutes the student experience will vary from one institution to another (Pitkethly & Prosser, 2001) and even one course to another. Jones (2018, p.1041) believes "this creates a challenge in identifying a suitable theoretical framework through which to appraise the student experience".

Bate and Robert (2007a) the architects of EBCD assert that all experience is personal and significant, how individuals understand, feel about and give meaning to an experience, as well as the context in which the experience occurs is subjective (Bate & Robert, 2007a). Although experience is an "inner subjective immaterial phenomenon" which can be difficult to access or observe directly, an individual's experience can be understood indirectly when

portrayed through the words and language used and gathered through personal narratives, semi-structured interviews and emotional mapping events (Bate & Robert, p.39, 2007a). As EBCD is the methodological approach utilised in this study the concept of 'student experience' will be framed by Bate and Robert (2007a) description above. As a user design methodology EBCD specifically focuses on capturing, exploring and understanding the subjective experiences of users of a service and enables a service user to collaboratively find ways to improve their experience of that service. Within this study the service user is represented by nursing students and the service is the BSc Pre-registration (Adult) nursing course.

Alongside the overarching methodology of EBCD guiding this study, Bate and Robert's (2007a) concept of 'experience' and Dewey's (1938/2015) 'Philosophy of experience' will form the theoretical framework for this thesis. Dewey, often considered the father of progressive education, first proposed his idea for a philosophy of experience in his 1938 paper 'Experience and education'. In it Dewey maintains that experience is more than an internal phenomenon taking place as a vacuum within an individual but is shaped by sources outside of that individual. Moreover, experience is a continuum, in that it grows out of other experiences and leads to further experiences, in particular experiences arise from the assimilation of two aspects: continuity and interaction (Dewey, 1938/2015). Continuity is the influence each person's experience has on his/her future or later experiences, for better or for worse, whereas interaction is viewed as the situational influence on an individual's current experience (Dewey, 1934/2005). In other words, a current experience as perceived by a student is a product of the interaction between their past experiences and the present situation, and the value of the experience is judged by the effect that experience has on the individual's present situation and their potential future experiences. Dewey (1938/2015) further proposes progressive, forward-thinking education is contingent on the quality of a student's experience, with a quality student experience being composed of two features; immediate agreeableness or disagreeableness, (thus whether the experience was pleasant or unpleasant) alongside its influence on future experiences. Dewey views experience having both an intellectual (cognitive) and an emotional element that give that experience significance and propels experience into 'an experience', thus "emotion is the moving and cementing force" of an experience (Dewey, 1934/2005, p.44). More specifically, "emotions are attached to events" (Dewey, 1934/2005, p.43).

Several fundamental premises of EBCD build on Dewey's concept of experience as described above and present a useful lens for understanding the student experience. For

example, EBCD similarly purports an individual's experience of a service is personal and must appeal on a cognitive and emotional level (Bate & Robert, 2007a). Likewise, Key moments, or events, within an experience generate identifiable emotions (Bate & Robert, 2007a). These Key moments, or events, are referred to as 'touchpoints' and are a central concept of EBCD (Bates & Robert, 2007a). Identifying and examining the crucial touchpoints of a service presents the opportunity for service providers to understand users experience of their service and enable re-design of those services to improve future experiences (Bates & Robert, 2007b). Since educators have a responsibility to understand what shapes students experience and to ensure the conditions conducive to shaping worthwhile experiences are established (Dewey,1938/2015) this study will capture and explore student experiences of the BSc Pre-registration (Adult) nursing course to better understand what shapes that experience and identify ways to improve future experiences on the course.

1.3 Place of the researcher within the study

The inspiration for this study originated from my twenty-year engagement with the delivery of the theoretical component and support of students undertaking an undergraduate preregistration nursing education programme in a Northwest of England university. In that time I have been a Module leader, Cohort leader, Year lead and Course leader for the BSc Preregistration (Adult) nursing course. A fulfilling aspect of my varied roles had always come from the satisfaction of feeling I had contributed to a 'good student experience' whether through my teaching, assessment support or pastoral support. What stood out to me in my various roles was the conversations with innumerable students that indicated some students were not always having a 'good student experience' that (in my naivety) I had expected all students to have. I began to wonder why some students had mostly positive comments to make about their experience on the nursing course, whilst other students described frequently poor or dissatisfying experiences. Around that time significant changes to preregistration nursing were imminent, particularly around student funding, that would potentially impact the student experience of the course. This led to me exploring the available literature on student experience of pre-registration (Adult) nursing courses. This investigation revealed a paucity of research when compared with the literature on Mental health and Child preregistration nursing. Furthermore, much of the available literature was narrowly focused on specific experiences, predominantly the clinical practice placement experience. The lack of research focused on the whole student nurse experience influenced my choice of methodology as EBCD enabled me to explore the student nurse journey through the whole course (pre-entry to the point of qualification).

1.4 The studies contribution to knowledge

This study contributes to knowledge in several ways. Firstly, this study adds to the exiguous knowledge and currently under researched concept of the student nurse experience of higher education in the UK. Secondly, this study draws together students experience of the whole journey through the nursing course, in contrast to those studies with a narrow focus on specific aspects of the student experience. The findings originating from this study contribute to discourse on student transition to higher education and student support in the academic and clinical practice placement environment. The findings from this study are timely in view of the major changes to nurse education and could impact future policy development around the quality of the student experience of pre-registration nursing education. Finally, this study presents the use of a novel methodological approach for conducting research in higher education. Notably, at the time of writing this thesis this study is the first documented draw upon elements of EBCD exclusively with student and key stakeholder participants in a university setting, making it distinct from other research studies conducted on the student experience of higher education. Therefore, this study offers rich insight into the benefits and challenges of utilising EBCD in a higher education setting,

1.5 Thesis structure

This chapter has set the context for this study by providing a background to the study, an overview of the theoretical framework underpinning this study and positioned the studies original contribution to knowledge. Chapter two presents the literature review for this study related to student experience and explores three key themes; the experience of being a nursing student, student nurse experience of the academic journey and student nurse experience of clinical practice also referred to as the work-based learning environment. The place of this study is also acknowledged in relation to the literature presented.

Chapter three introduces the methodology chosen for this study and begins by exploring the underpinning research paradigm including the epistemological and theoretical perspective that informed the research methodology. Chapter three examines the key tenets of EBCD as a methodology with a focus on the stages of EBCD, the significance of emotional touchpoint mapping and the process of co-design aimed at identifying priorities for course improvement. The study's aim and objectives are laid out alongside the data collection, analysis methods, and ethical considerations for the research undertaken.

Findings from this study are set out in chapters four, five, six and seven. Chapter four presents the findings from the semi-structured interviews with student participants. Three core themes are presented including 'navigating the academic journey', 'the clinical practice learning experience' and 'emergent networks of support'. From these three themes 18 touchpoints, or specific points of contact with the nursing course, are identified as significant for student participants within this study. These touchpoints formed a student journey map used as a visual aid to prompt discussion and encourage storytelling during the subsequent student emotional touchpoint mapping event and a key stakeholder emotional touchpoint mapping event. Chapter five provides an overview of the emotional touchpoint mapping process undertaken and a summary of the core themes and subthemes captured from the student and key stakeholder emotional touchpoint mapping events. Chapter six presents the findings from the student emotional touchpoint mapping event relating to five core themes including 'Student perceptions of disparate clinical practice learning experiences', 'the student experience of the clinical mentoring process', 'support with academic study - student perceptions of abandonment', 'the influence of academic staff on the student experience' and 'the effect of disparate approaches to Academic advisor support'. Chapter seven presents the three core themes generated from the key stakeholder emotional touchpoint mapping event including 'Academic advisor support - perceived ambiguity associated with the role', 'academic support - navigating the tension between support and independence', and "being' or 'becoming' professional - key stakeholder's expectations of students'.

The priority recommendations identified and explored during the student and key stakeholder Co-design events for implementing course improvement are presented in chapter eight. The three priorities are 'enhance student support in the clinical practice placement environment', 'facilitate and support student development of independent study skills' and 'clarify and enhance the role of the Academic advisor.

A discussion chapter (chapter nine) completes this thesis by contextualising the findings in relation to wider theoretical perspectives around the emergent core themes from the study. In particular, the discussion focuses on the mechanisms for supporting students in higher education by relating to the findings to Tait's (2000) 'Taxonomy of student support' that incorporates three interrelated elements: cognitive, affective and systemic student support. The discussion chapter also includes a reflection on the use elements of EBCD in a higher education setting, the implications of this study for EBCD, the implications of this study for pre-registration nursing education and the limitations of the study.

1.6 Conclusion

This introductory chapter has provided a background context to current pre-registration nurse education in relation to the numerous changes that have occurred during the last three decades including the transition from schools of nursing within hospitals to higher education, the move to an all-graduate profession, the introduction of student fees and the new NMC standards governing education and clinical practice. An understanding of the term 'student experience' has been established as personal and subjective but influenced by a myriad of factors outside of the learning process resulting in a 'total student experience'. The theoretical framework guiding this study has been outlined as that of EBCD and Dewey's (1938/2015) 'Philosophy of Experience'. The following chapter will present a literature review of contemporary studies exploring student experience of higher education.

CHAPTER TWO LITERATURE REVIEW

'Literature conveys the meaning of the past that is significant in present experience and is prophetic of the larger movement of the future' (Dewey, 1934/2005, p.359)

2.1 Introduction

Chapter one presented a background to pre-registration nurse education and the many recent changes to undergraduate nurse education. This chapter provides a review, critique and synthesis of the literature on the experiences of undergraduate student nurses. The rationale for completing this literature search and review is to position this study in the context of what is already known about students' experiences of pre-registration nursing courses. An outline of the structured approach used to undertake the literature search is described incorporating the search terms, inclusion/exclusion criteria and databases used to conduct the literature search. Key themes identified from the literature review are presented and explored. Throughout the literature review the proposed study is placed in context to gaps identified as arising from the literature.

2.2 Literature Search Strategy

Through the literature review researchers seek to critique and synthesise the literature on the research area being studied (Aveyard & Sharp, 2017) by providing both an overview and a critical assessment of previous research that has been undertaken (Boell & Cecez-Kecmanovic, 2014). A well-conducted literature search and review is essential to good quality research (Walsh & Downe, 2006). Charmaz (2014) argues a literature review gives the researcher a sophisticated knowledge of leading studies and theories in their field and enables the researcher to engage with this material critically and comparatively in relation to their proposed study. In addition, literature reviews enable a researcher to identify gaps in current knowledge, giving justification for a new study and focus for the research question, as well as meeting the requirements for the research governance frameworks and ethics committees (Aveyard, 2019). Consequently, completing a thorough, sharply focused literature review can strengthen an argument and credibility of a proposed study (Charmaz, 2014). Therefore, the decision was made to undertake a systematic literature search and review before gathering and analysing any data for this study with the understanding that as the research study progressed literature searching, and reviewing, would be an ongoing process.

The framework chosen to guide the literature search for this study was Kable et al.'s (2012) '12 step guideline to documenting a search strategy' who define a structured approach to searching the literature systematically and document the search strategy in a clear and articulate way (See Appendix 1 - 12-step guideline for structuring and documenting a literature search strategy). This form of research transparency allows future researchers to both evaluate and replicate the process in a similarly rigorous way, which is fundamentally important since the inception of multiple electronic databases and evidence-based search engines. Steps within this guideline include documenting the databases, policy sites and evidence-based search engines accessed, listing the search terms used (key text and freetext), listing the inclusion and exclusion criteria for the search and documenting a summary table of articles included in the literature review. Therefore, the purpose of this literature search was to identify published research studies examining the experiences of undergraduate pre-registered student nurses with their course. Due to the nature of this study involving student nurses as well as having an educational focus, it was decided to include both allied health databases and educational databases. The databases used in the search were ProQuest Education Journals, CINAHL Complete (The Cumulative Index to Nursing & Allied Health), OVID MEDLINE, EMBASE (Excerpta Medica database), and ERIC (Education Resources Information Center). In addition, there was a hand/electronic journal search of the snowball literature from the reference lists of retrieved papers. An initial search was conducted to ascertain any previous studies to use EBCD with student nurses in a higher education setting. Search terms were used in combination with database syntax language as detailed below:

Experience Based Co-design AND Preregistration OR Undergraduate AND Student* AND Nurs* AND Experience

The search was limited to research studies published from 1990 onwards following the introduction of the 'Project 2000 Framework' which ushered in the move of student nurse education from schools of nursing to higher education settings, which was a landmark moment for nurse education within the UK. Although the proposed study is UK based the literature chosen for this review was not restricted to UK literature, however English language only research studies were included. An inclusion/exclusion criteria for the literature search was applied as described below.

The inclusion criteria:

• Undergraduate pre-registration student nurses (Adult)

- Papers from 1990 to present
- English language papers
- Qualitative and quantitative empirical studies
- International research studies

The exclusion criteria:

- UK colleges/college education programmes
- Medical student programmes
- Papers Pre-1990 (Pre-project 2000 programme)
- Post-registration courses and foundation degrees
- Non-English language papers
- Literature reviews/discussion papers/pilot studies (although these were read to provide context)

The initial search yielded no papers published on the use of EBCD with student nurses at the point of conducting this study, the literature search was re-run in 2020 and confirmed this to still be the case, suggesting this study is the first documented to use elements of EBCD to capture, explore and understand the experiences of student nurses undertaking a BSc Pre-registration (Adult) nursing course. The literature search was then re-focused on published papers exploring the student nurse experience of higher education. The term EBCD was removed from this further search of the databases as detailed below:

Preregistration OR Undergraduate AND Student* AND Nurs* AND Experience

This search yielded 1496 papers. Once duplicates were removed further elimination took place through reading titles and abstracts for articles germane to research topic reducing the search yield to 188 papers (See Figure 1 - Literature search strategy). All 188 papers were reviewed for relevance, of which 172 studies were rejected whilst 16 articles were selected as eligible studies for this current literature review (See Appendix 2 - List of literature identified as eligible for review). Reasons for rejecting studies related to there being poor or no discussion of their methodological approach and articles being published in non-peer-reviewed publications. The 16 eligible published studies that form the basis of this literature review were critically evaluated using the 'Criteria for appraising qualitative research studies' (Walsh & Downe, 2006) and the 'Critical Appraisal of a Survey' (Center for Evidence-Based Management (CEBM)) framework (See Section 2.3 Critiquing framework).



Figure 1 - Literature search strategy

2.3 Critiquing framework

Two critiquing tools were identified for critically appraising the quality of research chosen for this literature review. Critical appraisal is about making balanced informed judgements on the merit of a research study, its good and bad points, and its significance to the research field (Moule, 2015). Furthermore, it is the process of examining research evidence systematically to assess its validity, results and relevance when applied to quantitative research (Heale & Twycross, 2015) and its transferability, credibility, reflexivity and dependability when assessing the quality of qualitative research (Williams et al., 2018) before using it to inform a decision. Ross (2012) warns literature has the potential to be rated as good or bad evidence depending upon whether it supports the reviewer's viewpoint or not. In other words, reviewers can be biased towards the literature they are reviewing. Preventing bias requires reviewers to explicitly disclose the process applied when identifying and including studies during a literature review (Gough et al., 2012). Without this level of explicitness it is impossible to assess the appropriateness of such decisions, or whether those decisions were applied in a consistent and rigorous manner (Gough et al., 2012). Hence, reviewing the literature in a systematic manner is a key activity in critical appraisal as well as being able to demonstrate the systematic process that took place (Gough et al., 2012). Thus, the manner for demonstrating an open, transparent and systematic decision-making approach to assessing the validity of the studies chosen for this literature review was through the clear application of two critical appraisal tools. Although no consensus currently exists regarding the most appropriate critical appraisal tool, or criteria, to use when evaluating research, most critical appraisal tools share similar components aimed at answering three important questions: Is the research Valid? What outcomes should be expected following the implementation of this research? Is this research useable by its target audience? (Rychetnik et al., 2002).

Choosing an appropriate critical appraisal tool is dependent upon its specificity to determine the relevance, strengths and limitations of each paper and enable the reviewer to address the research question (Aveyard, 2019). Overall, the critical appraisal tool should be specific to the type of literature being reviewed (Aveyard, 2019), peer-reviewed, well designed, simple to implement and include guidelines for use (Crowe & Sheppard, 2011). The Downe et al. (2007) adaptation of the Walsh and Downe (2006) 'Criteria for appraising qualitative and mixed methods research studies' was chosen for this literature review as is met the criteria above (See Appendix 3 - Criteria for appraising qualitative and mixed methods research studies). Designed and published as a critiquing checklist for qualitative literature and mixed method studies positioned within an interpretive research paradigm, this tool was relevant for assessing the value of 13 out of the 16 studies identified during the literature search. The

original Walsh and Downe (2006) critiquing framework assessed the appropriateness and coherence of a studies scope and purpose, design, sampling strategy, analysis, interpretation, researcher reflexivity, ethical dimensions, relevance and transferability. However, Downe et al. (2007) simplified the criteria to include scope and purpose, design methods, sampling strategy and participants, analytical strategy, quality rating and emerging themes. Interestingly, Downe et al. (2007) added a 'quality rating' system, whereby studies are allocated a quality rating from A to D based on the quantity and significance of flaws, or weaknesses, within that study (See Table 1 - Quality rating (Downe et al., 2007)).

A: No, or few flaws. The study credibility, transferability, dependability and confirmability is high.

B: Some flaws, unlikely to affect the credibility, transferability, dependability and /or confirmability of the study.

C: Some flaws that may affect the credibility, transferability, dependability and /or confirmability of the study.

D: Significant flaws that are very likely to affect the credibility, transferability, dependability, and /or

confirmability of the study.

Table 1 - Quality rating (Downe et al., 2007)

All qualitative studies in this literature review scored a C or above with most scoring B, thus having some flaws, that were unlikely to affect the study credibility, transferability, dependability and confirmability when applying the Downe et al. (2007) quality rating (See Literature summary table). Although four Appendix 4 studies scored a -C suggesting they contained some flaws/weaknesses that may affect the credibility, transferability, dependability and/or confirmability of the study, these flaws were not considered significant or noteworthy enough to warrant exclusion from this review. As the Walsh and Downe (2006) critical appraisal tool was designed for critiquing qualitative literature and mixed method studies, literature that utilised a quantitative survey methodology are included in the literature summary table but reviewed using the 'Critical Appraisal of a Survey' (Center for Evidence-Based Management (CEBM)) framework. The CEBM framework asks 12 appraisal questions related to study design, sampling and statistical analysis (See Appendix 5 - Critical appraisal of the guantitative surveys using the 'Critical Appraisal of a Survey' (CEBM) tool). Through this process of assessing and scoring the studies, common and reoccurring themes began to emerge. Similar themes were then grouped together until eventual three distinct themes related to student experience emerged from the literature and six subthemes (See Table 2 - Emergent themes and subthemes from literature review).

Theme	Subtheme	Literature					
The experience of being a nursing student		Magnussen and Amundson (2003)					
Student nurse experience of the academic journey	Transition into higher education	Andrew et al. (2009)	Birks et al. (2013)	Mckendry et al. (2014)	Porteous and Machin (2018)	Smith et al. (2018)	Mills et al. (2020)
	Expectations of academic learning	Last and Fulbrook (2003)	Magnussen and Amundson (2003)	Andrew et al. (2009)	Hamshire et al. (2013)	Mckendry et al. (2014)	Hamshire et al. (2017)
		Ten Hoeve et al. (2017)	Porteous and Machin (2018)	Smith et al. (2018)			
	Adapting to the role of university student	Birks et al. (2013)	Mckendry et al. (2014)	Ten Hoeve et al., 2017;	Porteous and Machin (2018)	Mills et al. (2020)	
Student nurse experience of clinical practice	Transition into the clinical practice environment	Andrew et al. (2009)	James and Chapman (2010)	Hamshire et al. (2013)	Liljedahl et al. (2016)	Hamshire et al. (2017)	Ten Hoeve et al. (2017)
		Donough and Van der Heever (2018)					
	Experience of clinical practice supervision	Last and Fulbrook (2003)	Andrew et al. (2009)	James and Chapman (2010)	Hamshire et al. (2013)	Papastavrou et al. (2016)	Hamshire et al. (2017)
		Ten Hoeve et al. (2017)	Donough and Van der Heever (2018)				
	Experience of socialisation, acculturation and sense of belonging within the community of practice	Andrew et al. (2009)	Mckendry et al. (2014)	Grobecker (2016)	Liljedahl et al. (2016)	Smith et al. (2018)	

Table 2 - Emergent themes and subthemes from literature review

2.4 The knowledge gap

Following an extensive search of the literature no study was found exploring the student experience of the BSc Pre-registration (Adult) nursing course using elements of EBCD as a methodology. Therefore, a distinct aspect of the study presented in this thesis is its use of EBCD to capture, explore and understand student nurses' experiences of the BSc Preregistration (Adult) nursing course. Furthermore, analysis of the emergent themes displayed in Table 2 indicate only one study directly examines the experience of being an undergraduate nursing student (Magnussen & Amundson, 2003), whilst the remaining 15 explored specific and narrow aspects of the student experience, including student nurse experience of the academic journey and of clinical practice. For example, the pre-registration student nurses' experience of transition into higher education was considered by just six studies in this literature review, and none of the literature explicitly investigated transition into year two and year three. Many significant touchpoints in the latter years of the pre-registration nursing course can directly influence the student experience, yet this aspect of student experience remains unexplored. This suggests there is a gap in current understanding of the BSc pre-registration adult nurses' whole experience of the course. Since EBCD enables experiences to be mapped across users (students) journey through a service (the BSc Preregistration (Adult) nursing course) the current study contributes to knowledge of the whole three-year course from pre-entry to qualification. The following literature review will reveal how the above gaps in knowledge were identified and how the current research study proposes to address some of these gaps.

2.5 Summary of literature chosen for review

The papers within this literature review used different approaches to conducting their research. Three studies applied a quantitative approach (Grobecker, 2016; Papastavrou et al., 2016; Hamshire et al., 2017). Ten studies used qualitative methodologies (Magnussen & Amundson, 2003; James & Chapman, 2010; Birks et al., 2013; Mckendry et al., 2014; Liljedahl et al., 2016; Ten Hoev et al., 2017; Donough & Van der Heever, 2018; Porteous & Machin, 2018; Smith et al., 2018; Mills et al., 2020) and three studies utilised a mixed methods approach (Last & Fulbrook, 2003; Andrew et al., 2009; Hamshire et al., 2013). Several of the studies in this literature review employed a post-positivist method. This philosophy follows similar principles to positivism but engages quasi-experimental research designs, namely a nonrandomised intervention study approach (Harris et al., 2006) whilst at the same time enabling more interaction between the researcher and the research participants (Willis, 2007). Post-positivist methods use wider criteria for data acceptability than is the case for

positivist research enabling large amounts of qualitative data to be categorized in such a way as to produce quantitative data for analysis using statistical methods (Gasson, 2009). Postpositivist methods frequently use 'Survey research' as their methodological approach (Creswell, 2008), as was the case with six of the studies for this literature review (Andrew et al., 2009; Birks et al., 2013; Hamshire et al., 2013; Grobecker, 2016; Papastavrou et al., 2016; Hamshire et al., 2017).

Within this review ten studies utilized alternative methods to survey research. Specifically, several papers utilised an interpretivist epistemology (James & Chapman, 2010; Hamshire et al., 2013; Liljedahl et al., 2016; Porteous & Machin, 2018; Mills et al., 2020), two papers used a constructivist epistemology (Mckendry et al., 2014; Smith et al., 2018), whilst three studies applied a descriptive epistemology (Magnussen & Amundson, 2003; Ten Hoeve et al., 2017; Donough & Van der Heever, 2018). Last and Fulbrook (2003) were the only authors to describe use of a Delphi study approach to explore attrition on a pre-registration nursing course, where three-rounds of questionnaires were administered to an 'expert' panel of student nurses until a consensus level of 75% was reached to explain why student nurses left the course. Mckendry et al. (2014) also addressed attrition issues in pre-registration nursing programmes with a focus on retention strategies, such as support mechanisms in the first year of study using a grounded theory methodology. Likewise, Smith et al. (2018) used a constructivist grounded theory approach to explore 17 undergraduate nursing students' satisfaction with learning in higher education.

Three studies explored different aspects of the lived experience of student nurses in higher education, with James and Chapman (2010) using Heideggerian phenomenology to explore the experiences of six undergraduate nursing students undertaking their first acute clinical practice placement. Ten Hoeve et al. (2017) used descriptive phenomenology to describe 17 nursing students experience of completing a Bachelor of Nursing programme. Whilst Porteous and Machin (2018) explored ten nursing students experiences of transition into higher education using Hermeneutic phenomenology.

Although Magnussen and Amundson (2003) state the use of a qualitative descriptive approach to describe and explicate the experience of being a nursing student in their study, they did not explain the tenets of the qualitative descriptive methodology used. Similarly, Donough and Van der Heever (2018) also employed a qualitative descriptive approach to their study of 36 undergraduate nursing students' experience of clinical supervision. Although Donough and Van der Heever (2018) describe using focus groups interviews to gather their

data, the principles underpinning their qualitative descriptive approach were not identified in their paper. Finally, Liljedahl et al. (2016) did describe their use of an ethnographic approach to conducting an observational study of the influences on the student nurse experience of the clinical learning environment in their research.

The studies above were conducted with various numbers of pre-registration nursing students in different years of study on the course. Four papers recruited participants from either year 1, or from all three years of the nursing programme. One study recruited from year 2, whilst two studies recruited from year three of the nursing course and five papers did not indicate their participants year of study (See Appendix 4 - Literature summary table). The smallest number of student recruits was for James and Chapman's (2010) phenomenological study which recruited six participants. The small participant numbers for this study can be explained by phenomenology's emphasis on quality of data rather than quantity, with the focus on individual stories of a lived experience. The highest number of study participants was for Hamshire et al.'s (2017) quantitative survey study, with a reported 1983 students from professional undergraduate courses taking part, including nursing students. A summary of the overall findings from these studies within this review can be found in Appendix 4. The overall themes identified from the studies in this literature review will be explored in relation to their impact on 'student experience' in the following sections. In addition to the 16 key studies used to form the themes for this literature review wider literature will be cited to add context and depth to some of the core themes discussed.

2.6 The experience of being a nursing student

Only one study within this literature review directly explored the experience of being a nursing student. Using a qualitative descriptive approach to their UK study Magnussen and Amundson (2003) interviewed a convenience sample of 12 undergraduate nursing students. Overall, four main themes were identified from the study: meeting conflicting demands, feeling overworked, feeling unprepared and seeking respect and support from one's faculty (Magnussen & Amundson, 2003). Student participants specifically described finding it difficult to maintain their work/life balance and this resulted in them struggling with the conflicting demands of work, study and family (Magnussen & Amundson, 2003). An interesting finding was that some students felt frustrated by the amount of 'busy work' designated by academics that they perceived as not necessary for their professional practice development which contributed to them feeling overworked (Magnussen & Amundson, 2003). A solution recommended by Magnussen and Amundson (2003) was to help students find strategies for improving their time management, such as introducing time management strategies in the

Leadership and Management module of the nursing programme. However, Leadership and Management modules tend to be delivered at the end of pre-registration nursing courses to prepare students for their graduate nursing roles, whereas issues with poor time management occur earlier in the course resulting in students feeling stressed and overwhelmed.

Another interesting finding related to how unprepared students felt for qualification as a registered nurse. Participants in Magnussen and Amundson's (2003) study expressed having insufficient knowledge for the demands of the profession. Wider literature shares Magnussen and Amundson (2003) conclusions around student nurses' sense of feeling unprepared and ill-equipped for their graduate roles (Newhouse et al., 2007; Milton-Wildey et al., 2014; Moreton et al., 2017). Magnussen and Amundson (2003) advocate embedding Benner's (1984) '*From Novice to Expert*' concept as a way to assist students understand the process of working towards an expert practitioner and feel more prepared for their professional roles. Yet, a pragmatic approach to managing anxiety around this significant point of transition is not offered in Magnussen and Amundson (2003) study. Transition from student to registered nurse may be established as a significant touchpoint by participants in the proposed study. If so, deeper understanding of students' experiences of this transition point through use of EBCD could enable the co-design of clearer approaches to supporting students feel better prepared for qualification.

Finally, Magnussen and Amundson (2003) found the lecturer/student relationship to be a major source of stress for many students, particularly during clinical skills sessions where some students describe feeling criticised, disrespected and 'put on the spot'. Several themes identified by Magnussen and Amundson (2003) emerged in some of the later studies presented in this literature review. Notably, Magnussen and Amundson (2003) original study is now over 17 years old and undertaken in the USA, since then no study has directly researched the pre-registration student nurse experience which is the aim of the study presented in this thesis. Instead, successive studies have focused on specific aspects of the nursing student experience.

2.7 Student nurse experience of the academic journey

Student's experience of the academic journey was the second most explored theme within the literature after student nurse experience of clinical practice, with a total of 11 studies directly investigating one or more aspect of the academic journey. Six studies focused on the student experience of transition into higher education and the factors that influenced a smooth or challenging transition (Andrew et al., 2009; Birks et al., 2013; Mckendry et al., 2014; Porteous & Machin, 2018; Smith et al., 2018; Mills et al., 2020). Nine papers explored student expectations of the academic learning experience in relation to the course content and experiences with teaching staff (Last & Fulbrook, 2003; Magnussen & Amundson, 2003; Andrew et al., 2009; Hamshire et al., 2013; Mckendry et al., 2014; Hamshire et al., 2017; Ten Hoeve et al., 2017; Porteous & Machin, 2018; Smith et al., 2018). Whilst five studies examined the strategies and mechanisms students utilised when adapting to the role of university student (Birks et al., 2013; Mckendry et al., 2014; Ten Hoeve et al., 2017; Porteous & Machin, 2018; Machin, 2014; Ten Hoeve et al., 2017; Porteous & Machin, 2014; Ten Hoeve et al., 2017; Porteous & Machin, 2014; Ten Hoeve et al., 2017; Porteous & Machin, 2014; Ten Hoeve et al., 2017; Porteous & Machin, 2014; Ten Hoeve et al., 2017; Porteous & Machin, 2014; Ten Hoeve et al., 2017; Porteous & Machin, 2014; Ten Hoeve et al., 2017; Porteous & Machin, 2014; Ten Hoeve et al., 2017; Porteous & Machin, 2014; Ten Hoeve et al., 2017; Porteous & Machin, 2014; Ten Hoeve et al., 2017; Porteous & Machin, 2018; Mills et al., 2020).

2.7.1 Transition into higher education

Within the literature the move into higher education and year one of the pre-registration nursing course was shown to be a pivotal period of transition. Students often transition from school or college to university, from living at home to living independently, and from paid employment to a full-time student role (Mills et al., 2020). Most students in the Mckendry et al. (2014) study appeared very satisfied with their move onto the course, whilst 10% of students in Birks et al.'s (2013) study reported a problem free transition. For those students with prior study experience at higher education level the transition appeared easier than for those starting a degree course for the first time (Birks et al., 2013; Smith et al., 2018). Factors associated with an easy transition included early adaptation to university life (Birks et al., 2013), finding a balance between personal/family commitments and university learning and study (Mckendry et al., 2014), interacting with student peers (Smith et al., 2018) and accessing available support early in the course (Porteous & Machin, 2018). Factors associated with a more challenging transition include struggling to adapt to a new learning/social environment (Porteous & Machin, 2018), perceived lack of support (Porteous & Machin, 2018; Mills et al., 2020), perceived excessive course demands (Ten Hoeve et al., 2017; Mills et al., 2020) and practical problems navigating the university systems (Birks et al., 2013; Mckendry et al., 2014).

Adapting to university life was a struggle for students who lacked basic study skills such as time management and computer skills (Birks et al., 2013). However, these skills are not inherently present in all students and may require learning and development in the early stages of their course. Transition problems are exacerbated as students attempt to juggle family life, childcare, shift work and academic study (Magnussen & Amundson, 2003; Birks et al., 2013; Mills et al., 2020). This is particularly evident on pre-registration nursing course that recruit large numbers of mature and diverse students (Birks et al., 2013; Mckendry et al.,

2014; Mills et al., 2020). The first few months of a nursing course can feel like a difficult balancing act for some students. An interesting finding from Mills et al.'s (2020) study was that those students who struggled to 'switch off' from the demands of the course found the balance negatively tipped, with the increasing pressure resulting in exhaustion and stress.

Stress and anxiety was a common theme within several studies exploring student transition to higher education. As a potential stressor a difficult transition can negatively impact students learning experience (Mills et al., 2020). For example, problems associated with timetabling and accessing the virtual learning environment had all been successfully resolved in Mckendry et al.'s (2014) study yet had caused unnecessary anxiety for the student participants. Increased levels of anxiety was also evident in Porteous and Machin's (2018) study during the first four months of the course. This anxiety related to what Porteous and Machin (2018, p.60) themed 'uncertainty' related to the "process of transition and adjustment" around the unknown aspects of the course, fears around to adapting to the new ways of learning and apprehension around their professional transformation. Mckendry et al. (2014) suggests student anxiety arises out of a sense of unfamiliarity, particularly when students feel unprepared for university pre-entry.

Andrew et al.'s (2009) study explored how professionally, and socially prepared students were for the first year of the nursing programme. Their findings indicate first year student nurses need additional support mechanisms to help them adapt to the academic and clinical environment, failure to adapt quickly during the early stages of the course can lead to attrition and problems with progression through the course (Andrew et al., 2009). Additional support mechanisms included extra support from mentors, more on-line learning resources and better understanding of the first-year student experience through the involvement of all stakeholders (Andrew et al., 2009). The proposed study aims to recruit both students and key stakeholders in the pre-registration nursing course to gain better understanding of the student experience student recommendations that may help support students better adapt to the clinical and academic environment.

In both the Birks et al. (2013) and Ten Hoeve et al. (2017) studies students specifically expressed feeling unprepared and lacking the skills to adjust to their new undergraduate roles. Participants in Hamshire et al.'s (2017) study expressed feeling unprepared for the academic workload, resulting in a sense of feeling 'out of their depth'. Both Birks et al. (2013) and Smith et al. (2018) suggest students must adjust to the intensified academic demands. Birks et al. (2013) report 'issues faced in adapting to the role of a university student in the

first year of study' as the strongest theme to emerge from their study of 112 student nurses. Several of these issues included "'the initial shock of study', 'the actual task of studying', 'juggling study commitments and meeting deadlines', and understanding 'what was expected of me as a student'" (Birks et al., 2013, p.29). The type and amount of material to be learned over the time allocated seemed overwhelming, particularly to first year students.

Wider research such as Andrew et al. (2008) found it was during the first year of study that most of the attrition from nursing courses occurred, and interestingly students who left in the first semester described themselves as feeling unprepared for university and deciding earlyon that the course was unsuitable for them (Ten Hoeve et al., 2017). In contrast, students who left in the second semester claimed they wanted to remain on the course but crisis events in their life left them unable to manage their studies. Furthermore, students who left in the second semester were more likely to eventually return to nursing, whereas students who left in the first semester were unlikely to consider returning (Andrew et al., 2008). Hamshire et al. (2017) report as many as 42% of their 1983 study participants had considered leaving their undergraduate course. Several studies suggest student nurses face discipline related challenges alongside the challenges of studying at university level, making the transition even more significant (Andrew et al., 2009; Birks et al., 2013; Ten Hoeve et al., 2017; Porteous & Machin, 2018; Smith et al., 2018; Mills et al., 2020). Pressure related to additional professional responsibilities can contribute to students struggling with some aspects of the course (Porteous & Machin, 2018). Barton (2007, p. 345) describes the move into nursing as akin to a "rite of passage", involving a transition from one way of life to another; whereby students need to learn to behave like nurses. In professions such as nursing, much of the learning is centred on a way of being, and tacit knowledge (Eraut, 2000; Booth et al., 2007). In other words, student nurses not only face the same challenges as other students in learning to manage their time and learning effectively but have an added pressure of moving from novice to expert, a term infamously coined by Benner (1984), and developing clinical competence. Students in one study within this literature review expressed feeling unprepared and overwhelmed by the professional and academic expectations of them (Magnussen & Amundson, 2003).

It is evident from the literature above that transition onto the pre-registration nursing course can be a critical period of stress and anxiety for many students. Feeling unprepared for university can exacerbate difficulty adapting to the role of a university student in the first year of study and was shown to be associated with higher student attrition. Identifying emotional touchpoints associated with transition to the pre-registration nursing course could offer some understanding into how students' experience transition. Furthermore, effective measures to support students through their transition could be devised during the student and key stakeholder Co-design event proposed in this EBCD study.

2.7.2 Expectations of academic learning

The literature revealed that students' expectations of the pre-registration nursing course was largely commensurate with their experience. Most participants in studies by Hamshire et al. (2013) and Mckendry et al. (2014) report their expectations being met and at times exceeded, resulting in a generally positive experience. Positive experiences were aligned with having realistic perceptions of the course (Mckendry et al., 2014) and appeared to be influenced by students' former educational experiences and how well-informed they were about the course (Hamshire et al., 2013). Both studies acknowledge some students' expectations were not met and for those students' disparity between their expectations and the reality of higher education study resulted in dissatisfaction (Hamshire et al., 2013; Mckendry et al., 2014). In particular, student dissatisfaction with the course related to the academic content (Hamshire et al., 2017), workload and level of academic writing (Hamshire et al., 2013) and structure and organisation of the nursing course (Mckendry et al., 2014; Ten Hoeve et al. 2017).

Early work by Last and Fulbrook (2003) indicate students had anticipated more practical i.e. clinical skills, teaching on the nursing course than they received with 91% of students expressing there was insufficient clinical skills teaching, and 97% suggesting too much emphasis was placed on the theoretical components of the nursing course. For many students in this study the perception of inadequate practical teaching left them feeling underconfident in their nursing skills and struggling to relate theory to practice, particularly in the first year of the course (Last & Fulbrook, 2003). Later evidence from the literature suggest students continue to expect nursing courses to be more hands-on. For example, participants in Mckendry et al. (2014) study expressed disappointment with the paucity of clinical skills teaching received on their nursing course. Ten Hoeve et al. (2017, p.30) report students feeling "stressed and unprepared" for the theoretical learning on the course and wanting more practical nursing skills teaching. Smith et al. (2018) suggests students assess and judge the worth of their learning, giving attention to the information they perceive as most relevant to their goal of nursing (Smith et al., 2018). Student satisfaction occurs when students view the course content as appropriate and valuable for shaping their knowledge and learning journey (Smith et al., 2018). Conversely, dissatisfaction occurs where they perceive discrepancies between their expectations and actual experiences, and where they feel unprepared for the academic rigors of the nursing course.

Within the literature it is noted that some student nurses feel unprepared and out of their depth with the academic demands of the course (Birks et al., 2013; Last & Fulbrook, 2003; Hamshire et al., 2013; Hamshire et al., 2017; Ten Hoeve et al., 2017). Specifically students in Hamshire et al.'s (2017) study reported feeling 'shocked' by the academic workload and 42% expressed feeling unprepared for the extent of work required. Comments made by students in the study by Andrew et al. (2008) suggest the type and depth of the content of the course, alongside the amount of work and academic skills required influenced many students' decision to leave their nursing course (Andrew et al., 2008). Students who had left the course described feeling that they were not prepared for the amount of academic work required and they had struggled with skills such as essay writing and in-text referencing (Andrew et al., 2009). Magnussen and Amundson's (2003, p.264) study indicates many student nurses feel pressure to strive for excellence knowing that "their future success as professionals' rests on a thorough understanding of the material they are learning". Consequently, the type and amount of material to be learned over the time allocated can seem overwhelming, particularly to first year students as one student from Magnussen and Amundson's (2003, p.264) study comments, "the first semester was very tough for me. I had no life. I have no life. My parenting suffers, which I feel guilty about. But I've done well in nursing school. You know, I want to be a nurse". The comments above suggest some students face the dilemma of having to choose between commitment to the course at the expense of their personal life or a life outside nursing at a cost to their success on the course. This can particularly be the case for some mature students who find childcare responsibilities. family commitments, and financial difficulties an issue that can affect their studies (Birks et al., 2013).

The Higher Education Statistics Agency (HESA) (2015) define a mature student as those who are aged 21 or over. According to Universities UK (2014) 52% (713,955) of all undergraduate enrolments at HEIs in England were by mature students, with nursing courses accounting for a significant proportion (20%) of those enrolments. This figure is much higher in the area for the proposed study with mature students representing 67% of the total adult field of the nursing course (Smith, 2015). Through the process of widening access mature students have been encouraged into nurse education over the last decade. However O'Brien, et al. (2009) believe older students are expected to fit into structures that already exist, and that were originally formulated with 'traditional' (18 - 21 year) students in mind. Notably, for many mature students it may have been several years since they were last in full time education.
Wider literature suggests mature students are disadvantaged by the fact they may not have been involved in education for a number of years, as result they tend not to 'perform' as well as the rest of their cohort and need time to adjust to being full time students (Fleming & McKee, 2005). O'Brien et al. (2009) concur that mature students tend to struggle with the academic components of their course and have difficulty balancing home and university life. However, in a study by El Ansari (2002) of 460 students attending various health care modules 'mature' students performed better and showed higher satisfaction than 'traditional' students. Interestingly, the highest satisfaction was amongst the 'older' mature students (>25 years) followed by the mature student participants (21-25 years), with the lowest satisfaction being amongst the 'traditional' students (<21 years) (El Ansari, 2002). El Ansari (2002) assert the idea of mature students being at a disadvantage when studying at university as a myth, what is more younger students are more likely than older students to leave their nursing programme (Pitt et al., 2012). This position is supported further through research by Donaldson et al. (2010) who found in their study of 638 adult nursing students, at a Scottish HEI, that mature students were more likely to successfully pass year one modules/assessments on their nursing course than their younger counterparts. Interestingly, they even suggest age could be the most reliable predictor of success at recruitment interviews (Donaldson et al., 2010). Much of the research on mature students focuses on attrition and success on their courses rather than enabling them to voice or share their unique experiences.

Although not specifically intending to focus on mature students the proposed study could enable the researcher to capture and understand older students' experiences since mature students represent a large proposition of students on undergraduate nursing courses. Furthermore, the proposed study would enable the researcher to work closely with students, through the co-design element of the process, to identify those disparities between students' expectations and experiences.

2.7.3 Adapting to the role of university student

Adapting to their new role as student nurse on the pre-registration nursing course requires students to develop new skills and adopt effective coping strategies. Several studies within the literature describe students developing what they termed 'juggling' skills (Birks et al., 2013; Mckendry et al., 2014; Mills et al., 2020). The nursing course necessitates a need to juggle multiple demands on time and resources including home and family demands alongside social, financial, academic and clinical practice pressures (Birks et al., 2013; Mills

et al., 2020). Managing all these demands becomes a "careful balancing act" (Mckendry et al., 2014, p.875) which for some students can have a direct effect on their physical, mental and emotional health and wellbeing (Mills et al., 2020). Strategies that emerged from the literature as fundamental to helping students find equilibrium consisted of seeking support from family, peers and academics.

Relying on support from family was evident at the beginning of the nursing course, but as students progressed through the course more support was sought from peers (Ten Hoeve et al., 2017; Porteous & Machin, 2018). Some students turn to their peers for support as they often share a sense of belonging with other students. The concept of peer support is nothing new amongst student groups, but for nursing students peer support fosters a sense of belonging and can have a positive impact on the socialisation process (Chapman & Orb, 2001). Peer support networks have been shown to play a positive role in enhancing student's health and wellbeing on the course (Ten Hoeve et al., 2017; Mills et al., 2020). Peer support allows for sharing experiences, reinforcing knowledge and enhancing confidence. Knowing peers face similar challenges on their journey through the first year improves students' self-confidence (Porteous & Machin, 2018). Porteous and Machin (2018) acknowledge peer support networks foster a sense of community in students and contributes to development of resilience.

Peers on the same course are viewed as a 'strong resource' to turn to for ideas and support (Birks et al., 2013) as participants feel the experience of being a student nurse can only be understood by other nursing students (Chapman & Orb, 2001). As a result, an informal student grapevine often operates in undergraduate nursing courses (Chapman & Orb, 2001). Through this grapevine students communicate their fears and worries with others deemed 'in the same boat', putting things into perspective and helping them realise they are not alone. However, the grapevine could also have a negative effect and is not always viewed as positive. For instances, some students in a study by Chapman and Orb (2001) reported feeling more anxious and fearful prior to commencing a placement area due to information they received on the grapevine about a clinical practice area. Chapman and Orb (2001) argue being 'forewarned' is not inevitably 'forearmed' for some students.

Support from academic staff was also strongly indicated in the literature as valuable for helping students adjust to their new role (Mckendry et al., 2014; Ten Hoeve et al., 2017). Two studies highlighted communication with approachable and professional teaching staff as an effective source of academic support (Birks et al., 2013; Mckendry et al., 2014). Notably, it

was considered important for academics to facilitate communication with students and foster a sense of connection (Birks et al., 2013). Porteous and Machin (2018, p.60) proposes an "integrated partnership approach between university, practice staff and the student" as fundamental to helping students develop the coping skills necessary for the challenges of academic life and the clinical practice setting. However, Porteous and Machin (2018) do not indicate what that partnership should incorporate. Participants in Birks et al. (2013) study wanted more personal interactions and engagement with lecturers. Where students sense a lack of support from teaching staff, dissatisfaction with the nursing course was noted in the literature (Ten Hoeve et al., 2017). The importance of student-lecturer interaction and relationships in higher education is unsurprising given that academic staff are the main point of contact between student and institution.

Although the literature indicates some students wanted more contact with lecturers, Magnussen and Amundson (2003) signifies student-lecturer interaction needs to be positive to be effective. A negative interaction with their lecturers was one of the most common sources of anxiety for students in Magnussen and Amundson's (2003) qualitative descriptive study of the experience of being a nursing student. Interestingly, making mistakes during nursing procedures came only second as a source of anxiety to negative student-Lecturer interactions (Magnussen & Amundson, 2003). Study results suggest lecturers are viewed as pivotal in student's learning experiences. Many of the negative experiences for students on their courses appeared to be either caused or characterized by inattentive and unapproachable staff for whom students appeared not to be a priority (Magnussen & Amundson, 2003). Porteous and Machin (2018) report some participants in their study describing a negative experience with staff making them feel undervalued, unable to seek their support and difficulty to communicate with. Inattentive staff were those classed as frequently unavailable, unresponsive or unapproachable. By contrast, approachability was identified as a highly valued trait and key characteristic of a good relationship between lecturer and student by Birks et al. (2013). Participants in Porteous and Machin (2018, p.58) study similarly valued "helpful, approachable, caring" academic staff who made themselves available, and listened to student's concerns.

Through the EBCD process this study will provide insight into student participants experiences of support from academic staff on the BSc Pre-registration nursing course and make some contribution to existing knowledge. Furthermore, through recruiting both student and key stakeholder participants for the Co-deign event this study creates the opportunity for

dialogue to occur between students and lecturers/senior lecturers around the student-lecturer relationship/interaction.

2.8 Student nurse experience of clinical practice

Clinical placements offer valuable practice learning experiences for students (Hamshire et al., 2013). Students studying BSc Pre-registration nursing spend half of the course in the clinical practice placement environment. In the UK this is represented by a minimum of 2,300 hours compared to around only a 1000 hours in Canada, Australia and the USA (Mills et al., 2020). In addition, like nurses employed by the NHS, students are required to work various shift patterns including day and night duty, weekends and bank holidays (Mills et al., 2020). Unlike NHS staff, student nurses in the UK are not paid for their attendance on clinical placements.

Within the literature students' experience of the clinical practice environment was the most explored theme with a total of 13 studies directly investigating one or more aspect of the clinical practice placement experience. Seven studies identified facets of the transition into the clinical practice environment as significant to the student experience such as expectations of and preparation for the first placement (Andrew et al., 2009; James & Chapman, 2010; Hamshire et al., 2013; Liljedahl et al., 2016; Hamshire et al., 2017; Ten Hoeve et al., 2017; Donough & Van der Heever, 2018). Eight papers explored student Experience of clinical practice supervision (Last & Fulbrook, 2003; Andrew et al., 2009; James & Chapman, 2010; Hamshire et al., 2013; Papastavrou et al., 2016; Hamshire et al., 2017; Ten Hoeve et al., 2017; Donough & Van der Heever, 2018). Whilst five studies examined the Experience of socialisation, acculturation and sense of belonging within the community of practice (Andrew et al., 2009; Mckendry et al., 2014; Grobecker, 2016; Liljedahl et al., 2016; Smith et al., 2018).

2.8.1 Transition into the clinical practice environment

The clinical practice learning environment is distinct from a higher education environment and requires some adjustment from student nurses. Notably, clinical placement areas where direct patient care occurs can be physically, mentally and emotionally demanding, requiring students to work effectively as part of the wider clinical team. Successful transition into clinical practice can be influenced by students' expectations and prior experiences of clinical practice. In particular, the first clinical practice placement was shown by two studies to significantly influence student nurses' success in the first year of study on the course (Andrew et al., 2009; Ten Hoeve et al., 2017). What is more, initial exposure to the nursing profession is a significant influence on students' impetus to continue the course (Andrew et al., 2009). Overall findings from the literature indicate most student nurses looked forward to their first clinical placement and largely found placements to be a positive experience. Within Andrew et al.'s (2009) study 76% of participants described looking forward to their first clinical placement, and students in Hamshire et al. (2013) and Hamshire et al. (2017) reported enjoying placements. A successful first clinical practice placement appeared to influence student retention and overall satisfaction on the course. A positive first clinical placement experience influenced student's decision to continue with the nursing course in three studies (Andrew et al., 2009; Hamshire et al., 2017; Ten Hoeve et al., 2017). Furthermore, the first few weeks of clinical practice appeared to be a major benchmark for students' practice learning experience. By the end of this period many students feel they have begun to understand the nature of nursing and what is expected of them (Andrew et al., 2009). Although most students were excited to experience clinical practice, 19% articulated having anxiety about placement and 3% expressed dreading their first placement in Andrew et al. (2009) study. Similar results have been reported in subsequent studies (James & Chapman, 2010; Hamshire et al., 2013; Donough & Van der Heever, 2018).

First year student's fears and anxieties relate to the 'unknown' aspects of the clinical practice placement environment (Donough & Van der Heever, 2018). Participants in James and Chapman's (2010) study described feeling overwhelmed and disorientated by the unfamiliar and busy environments they experienced in some clinical practice placement areas. This was particularly the case for participants who had minimal exposure to a clinical practice environment prior to commencing the nursing course (James & Chapman, 2010). Student nurses in Liljedahl et al.'s (2016) study felt overwhelmed by the responsibility of patient care, with pressure to deliver patient care safely and competently. A perceived lack of sufficient knowledge to deliver good care was a theme identified in Ten Hoeve et al.'s (2017) study with negative experiences of clinical placements related to feeling unprepared for practice.

Feeling unprepared for clinical practice emerged as a concern in Andrew et al.'s (2009) study with only 41% of students reporting an understanding of what was expected of them at the beginning of their placement with this figure rising to 84% by the second week of practice. Feeling unprepared for their first clinical placement was shown to reduce confidence for some students in James and Chapman's (2010) study. Unlike more confident students who actively sought out learning opportunities early in their placement, students who lacked confidence tended to be less involved with conducting clinical procedures and were perceived as disinterested and lacking initiative (James & Chapman, 2010). James and Chapman (2010)

conclude a student's level of confidence directly influences their clinical placement experience.

Overall, the findings above suggest that although most students look forward to their first clinical placement some students experience anxiety and fear prior to commencing the placement. This anxiety appears to arise from fear of the unknown, with less than half of the students in the studies reviewed having a clear understanding of their role as student nurses. Feeling unprepared for the first clinical placement reduces student's confidence overall and can result in a negative clinical practice experience. Evidently a negative first clinical practice experience influences student's decision to leave the course. Conversely, students who have positive experiences once in the clinical practice area tend to express more satisfaction with the nursing course than students who have negative experiences in practice. Clearly a positive first clinical practice experience influences attrition rates in the first year of the BSc nursing course. Therefore, the proposed study seeks to use the EBCD approach to identify and explore those touchpoints related to clinical practice placements that positively or negatively affect the student experience on the BSc Pre-registration nursing course.

2.8.2 Experience of clinical practice supervision

Several studies within the literature identified the mentor-student relationship to be the most influential factor on the student experience and satisfaction with their clinical practice placement (Last & Fulbrook, 2003; Andrew et al., 2009; James & Chapman, 2010; Hamshire et al., 2013; Papastavrou et al., 2016; Hamshire et al., 2017; Ten Hoeve et al., 2017). The NMC definition of a mentor is a person who "facilitates learning and supervises and assesses students in a practice setting" (NMC, 2008, p.45). At the time of this study the NMC required all student nurses to be mentored by an NMC registered nurse, midwife, nursing associate, or other registered health and social care professional for a minimum of 40% of their time in the placement area (NMC, 2008). The remaining 60% of students time is spent working with other members of the nursing team, including qualified and unqualified staff such as Health Care Assistants (HCA). Within the literature satisfaction with the mentor-student relationship was based on the clinical practice setting, the mentor's approach to supervision and the frequency of contact between student and mentor. The quality of mentoring was considered a "make or break component of the first placement" by Andrew et al. (2009, p.17).

This was evident in Papastavrou et al.'s (2016) findings where greater satisfaction was noted in students with a mentor who offered regular one-to-one supervision i.e. three times per week. Furthermore, good mentors were perceived as helping students develop clinical competence and socialisation into the clinical practice setting (Papastavrou et al., 2016). This corroborates earlier findings by Last and Fulbrook (2003) where student satisfaction correlated with mentors being committed to helping them meet their learning needs. By contrast, a negative placement experience was associated with mentors lacking the time to support student learning (Hamshire et al., 2013).

Hamshire et al. (2017) report 54% of students perceiving mentors having insufficient time to support students, yet 99% of their participants had noted mentor support as critical to a positive clinical placement experience. First-year students viewed the support of mentors or clinical supervisors as essential to alleviating their anxieties around clinical practice (Donough & Van der Heever, 2018; Hamshire et al., 2017). Whereas a perceived lack of mentor support contributed to student attrition early on the nursing course (Ten Hoeve et al., 2017). Lack of mentor supervision results in reduced enthusiasm, deterioration in clinical practice performance and subsequent disillusionment with nursing in general for students who feel unsupported in the clinical environment (Last & Fulbrook, 2003).

Heavy workloads and time constraints on mentors undoubtedly affects the amount of time they can spend with students. Within the literature some mentors report disillusionment and lack of enthusiasm due to lacking the time to properly teach students (Last & Fulbrook, 2003). Due to workload demands some mentors feel disinclined to invest what time and energy they do have in supporting students (Last & Fulbrook, 2003). One study attributed this to a lack of 'ownership' or 'duty' towards students since nurse education had moved to universities (Last & Fulbrook, 2003). This results in some students feeling like a 'burden' to their mentor (James & Chapman, 2010) and 'unwelcomed' in the clinical area (Ten Hoeve et al., 2017). Mentors lacking time to support students' often impacts on students meeting their learning goals and subsequently resulted in a negative clinical placement experience for participants in Ten Hoeve et al.'s (2017) study. An interesting finding from one study was that student satisfaction with clinical supervision decreased as they progressed through the course. Papastavrou et al. (2016) noted the highest satisfaction to be amongst first year students who had positive clinical experiences and lowest amongst final year students. Papastavrou et al. (2016) conclude as students' climb the competence ladder their supervision and learning requirements change, but not all mentors successfully adapt to these changing needs.

It is evident that mentor support has an influential impact on student nurses, and a lack of support inhibits student learning and results in a negative student experience (James & Chapman, 2010; Hamshire et al., 2013). Similarly, the clinical practice placement is key to a

good experience and a successful first year (Andrew et al., 2009). The proposed study will provide insight into student participants experiences with mentors in the clinical practice placement by identifying those emotional touchpoints that are pivotal in either positively or negatively affecting the BSc Pre-registration nurse's journey through the course. Furthermore, through the process of co-designing experiences revealed during emotional touchpoint mapping can help compose recommendations for course improvement.

2.8.3 Experience of socialisation, acculturation and sense of belonging within the community of practice

It has clearly been established that some undergraduate students' experience anxiety whilst transitioning onto their course. Alongside coping with academic issues, students also identify concerns around personal and social issues such as being away from family for the first time, making friends and fitting-in. But for student nurses there is the added pressure of professional socialisation and acculturation. It is already understood that students' socialisation within nursing is an important part of their education. Professional socialisation is a concept that has been documented in nursing as far back as Becker and Geer (1958), with socialisation within nursing seen as a complex and continuous interactive process through which professional identity, and role, is learned and established. Student nurses begin to align their professional identity with their community of practice within the first year of the nursing course (Porteous & Machin, 2018). Since nurses operate collectively in an interactive social system, student nurses must then adopt the norms, values and rules that characterize that 'collective working group' during the professional socialization process (Randle, 2003). Several papers within this review revealed student nurses start to feel part of the profession by developing a sense of professional identity and 'belonging' (Andrew et al., 2009; Mckendry et al., 2014; Grobecker, 2016; Liljedahl et al., 2016; Smith et al., 2018).

Within the literature students' sense of belonging had a direct influence on their placement experience. Liljedahl et al. (2016) argue a prerequisite for successful student learning in the practice setting is a sense of belonging. A sense of belonging positively or negatively influences students' confidence, motivation and learning within the clinical practice environment (Grobecker, 2016). Students' knowledge, attitudes and ideas towards nursing are shaped and reshaped throughout the course by their interactions with other students, clinical practice staff and nurse academics (Smith et al., 2018). It is specifically shaped by participation and observation in the clinical practice environment a student nurse is exposed to (Smith et al., 2018). As such student nurses want to 'fit' into the clinical practice environment and feel part of their clinical practice placement area and team. When placement

areas facilitate 'belongingness', students feel safe, comfortable, satisfied and happy, and were more willing and motivated to engage in clinical practice learning (Levett-Jones & Lathlean, 2008). They were also less likely to feel stressed and anxious (Grobecker, 2016). Furthermore, having a sense of belonging results in acceptance of the profession leading to more motivation and commitment to the course (Mckendry et al., 2014). In other words, when students feel like they 'fit in' it may be perceived as validating their skills and competence as 'real nurses' which in turn fosters in them an affinity or sense of comradeship with their colleagues.

On the other hand, students who sense a lack of belongingness express greater stress, anxiety and dissatisfaction with their clinical practice experiences. Using the Belongingness Scale-Clinical Placement Experience (BES-CPE) Grobecker (2016) found an inverse relationship between a sense of belonging and perceived stress. Meaning where students felt a sense of belonging, they reported less stress. Conversely, stress increased where students perceived a low sense of belonging. Early research by Levett-Jones and Lathlean, (2008) revealed an absence of belongingness resulted in a negative and long-lasting impact on students' confidence, anxiety levels and learning. Students were seen to 'sacrifice' their supernumerary status and learning opportunities so as to become 'an extra pair of hands' so they would be seen as more useful and accepted into the nursing team (Levett-Jones & Lathlean, 2008). Thus, some student nurses may internalise rejection and lack of acceptance as related to how hard they appear to work. Ten Hoeve et al. (2017) suggests feeling welcomed by the clinical practice team results in a more positive experience which then directly influences some students' decision to remain on the nursing course.

With 50% of the pre-registration nursing course taking place in the clinical practice environment it is evident from the literature above that students need to feel welcomed and perceive a sense of belonging in order to start developing their professional identities early in the course. Where students lack a sense of belonging their clinical experiences are perceived as more negative which appears to be a contributing factor to student attrition. The proposed study could provide insight into student participants experiences of socialisation and acculturation by identifying those emotional touchpoints that contribute to students' sense of belonging in the clinical practice placement environment and to the wider nursing profession.

2.9 Conclusion

In summary, this literature review has identified and examined the research on the undergraduate nursing students' experiences of higher education. Overall, 16 studies met the inclusion criteria for this review. Three major themes and six subthemes emerged from the chosen literature including; the experience of being a nursing student, student nurse experience of the academic learning journey (transition into higher education, expectations of the academic learning, and adapting to the role of university student), and student nurse experience of clinical practice (transition into the clinical practice environment, experience of clinical practice supervision and experience of socialisation, acculturation and sense of belonging within the community of practice). Findings from the literature suggest student nurses often feel overwhelmed and unprepared for the nursing course, particularly in relation to the academic demands of the course. In addition, many students feel stressed and anxious during their transition to the clinical practice learning environment. Where students experience good support from mentors and feel welcomed into the clinical environment, they are more likely to perceive a sense of belonging which reduces their stress and anxiety.

Although all the studies included in this literature review were important contributors to the discourse on student experience much of the literature focused on specific, and often narrow, aspects of that experience rather than enabling student nurses to identify those aspects of their experience that are significant to them. Equally there were no published studies uncovered for this review aimed at capturing, exploring and understanding the key moments or events (touchpoints) that form and shape the student nurse experience using an Experienced Based Co-Design model. Although most universities employ student feedback tools such as Module Evaluation Questionnaires (MEQ) and Course Evaluation Questionnaires (CEQ) to support some changes in their processes, these mechanisms have limited value in enabling key stakeholders (e.g. students, academic staff, admin staff etc.) to work together to improve course provision. EBCD offers a unique opportunity for service users, in this case student nurses, to be involved in the change, or design process, itself (Bate & Robert, 2007b).

Therefore, based on this review it is clear the experiences of undergraduate student nurses have been inadequately explored in the nursing education literature, suggesting there are gaps in our knowledge about students experiences of their pre-registration nursing course. Student insights of the higher education environment are invaluable and as such student nurses need to be included in the process of exploring that environment through their own experience and re-designing those experiences. Consequently, this current study hopes to plug this knowledge gap by capturing, exploring and gaining some understanding of the experiences of undergraduate student nurses undertaking a BSc Pre-registration (Adult) nursing degree course using an Experience Based Co-design approach. Furthermore, the proposed study will also explore the student experience across the whole course, rather than at an isolated point, by capturing first, second and third-year student nurses' experiences.

CHAPTER THREE METHODOLOGY

'Scientific method is the only authentic means at our command for getting at the significance of our everyday experiences of the world in which we live' (Dewey, 1938/2015, p.88)

3.1 Introduction

The previous chapter presented a review of the literature and research available focused on undergraduate nursing students' experiences of higher education. Gaps in knowledge of students experience of pre-registration nursing courses were identified and the need for the current study outlined. This chapter presents the methodology utilised to conduct the current study including an exploration of the stages of Experienced Based Co-Design (EBCD) approach adopted to explore the student experience of the BSc Pre-registration (Adult) nursing course. Particular attention is given to the research paradigm and theoretical perspective that informed the research methodology and study design choices employed throughout the research. At the beginning of this thesis 'experience' was theoretically framed by Dewey's 'Philosophy of experience', therefore this chapter will begin by establishing the principles of Dewey's philosophy that underpin this study.

3.2 Theoretical framework underpinning this study

Establishing a theoretical framework for a study helps align a researcher's epistemology, and theoretical perspective as well as directing the study aim, objectives and methodology. Dewey's (1938/2015) 'Philosophy of experience' where he presents experience as a subjective personal phenomenon, shaped by events and conditions outside individuals, formed the theoretical framework for this thesis. Dewey (1938/2015) purports current experience grows out of past experiences (Continuum/Interaction) and potentially influences future experiences based on how pleasant or unpleasant (agreeableness or disagreeableness) the experience was. In other words, a current experience as perceived by an individual is a product of the interaction between their past experiences and the present situation, and the value of the experience is judged by the effect that experience has on the individual's present situation and their potential future experiences. Furthermore, Dewey views experience having both a cognitive and an emotional component that give that experience significance, with the more memorable experiences having specific emotions attached to them (Dewey, 1934/2005, p.43). This concept of viewing experience as having an emotional element offers a lens for which to explore subjective experiences, identify those

satisfactory experiences, and potentially understand what may constitute those less satisfactory experiences.

Although the emphasis of Dewey's work was on human experience (Simpson, 2006), his theory of inquiry is often viewed as "the centrepiece of his philosophy, shaping his conception of science, knowledge, values, practice and society" (Willower, 1994, p.18). Underlying assumptions of Dewey's philosophy have been viewed as having a significant influence on design research (Dixon, 2019). Most notably, Steen (2013) reports two key themes to emerge from Dewey's philosophy as its focus on individual's concrete knowledge of their own personal experiences and its endeavour to empower people to improve their situations or experiences. Both Steen (2013) and Dixon (2019) see the Deweyan process of inquiry as holding similar principles to co-design. Specifically, Dewey describes a process of inquiry comprising of five stages, "Perceiving a situation as 'questionable'; Exploring and formulating the problem provisionally; Simultaneously exploring and further developing the problem ('perceptual') and possible solutions ('conceptual'); Exploring the relations between problems (ends) and solutions (means); and Trying-out solutions in practice to find out whether or how they can help to solve the problem" (Steen, 2009, p.1). Dixon and French (2020) further denote a Deweyan approach to inquiry as compelling the inquirer to define a problem, explore solutions, implement those solutions (experimentation) resulting in workable conclusions. Tenets of this process of inquiry are recognisable in design research, and particularly in EBCD, the approach chosen for conducting the current study as will be explored in the following sections.

3.3 Research paradigm

A research paradigm or worldview is "a basic set of beliefs that guides action" (Guba, 1990, p.17). Thus, a researcher's belief and how they see the world is reflected in their choice of paradigm, with this worldview directing the research process and impacting a researcher's choice of methodology and study methods (Kivunja & Kuyini, 2017). Different worldviews influence a researcher's orientation towards either a qualitative, quantitative or mixed methods approach in their research (Creswell, 2009). Identifying and documenting the theoretical assumptions of a researcher can justify choices made during the research process and ensure the soundness of a study (Crotty, 1998). Crotty (1998) suggest researchers address questions around four distinct elements that formulate their theoretical assumptions at the start of the research process relating to epistemology, theoretical perspective, methodology and methods (Crotty, 1998):

What epistemology informs our perspective?

What is our theoretical perspective? What methodology governs our choice of methods? What methods do we propose to use? (Crotty, 1998, p.2)

For Crotty (1998) each element informs the next, with continuity evident in a researcher's decision-making process originating from their epistemological standpoint and ending in their choice of methods for conducting their study. The distinction between qualitative and quantitative research is determined at the methodology/methods level, whereas decisions related to the philosophical worldview, for example positivism versus constructivism, occurs at the higher epistemology and theoretical perspective levels (Crotty, 1998). The following sections explore the four elements of epistemology, theoretical perspective, methodology and methods as they relate to the research process utilised within the study presented in this thesis.

3.3.1 Epistemology

A researcher's epistemological position informs their assumptions about knowledge, what it is and how it is acquired. The term epistemology was introduced to the English language by Scottish philosopher James Frederick Ferrier (1808-1864), but its origin is from the Greek word 'episteme' meaning knowledge and 'logos' meaning study or science (Truncellito, 2016). Therefore, epistemology is considered as the study (or science) of knowledge and is a way of understanding and explaining "how I know what I know" (Crotty, 1998, p.3). As a science of knowledge epistemology is concerned with the analysis of knowledge, its relationship to belief and truth (Ladyman, 2007). A researcher's position on what constitutes knowledge can result in them falling into one of several different epistemological camps: objectivism, constructionism or subjectivism (Crotty, 1998).

Objectivism is the belief that reality exists as a distinct, independent meaningful entity outside of human consciousness and experience (Crotty, 1998). Humans merely discover an object's meaning by studying it (Crotty, 1998). Therefore, the essence of what is being observed is not influenced by human bias and vice versa, resulting in universally applicable knowledge (Levers, 2013). Objectivist researchers view phenomena as external to the individual and traditionally favour experimental ways of examining phenomena (Cohen et al., 2011). Hence, objectivists assume different researchers will obtain the same results if they conduct an experiment using the exact same process in comparable circumstances (Parahoo, 2014). Hence, from an epistemological standpoint objectivist research studies are often used to explain, predict, and control (Grant & Giddings, 2002). Research within the social sciences has seen a move away from objectivism towards subjectivism (Guba, 1990).

Subjectivist assumptions regard knowledge to be dependent on individual's perception and understanding of reality (Moon & Blackman, 2014) Furthermore, subjectivist knowledge is "always filtered through the lenses of language, gender, social class, race, and ethnicity" (Denzin & Lincoln, 2005, p. 21). Thus, there is no universal knowledge beyond an individual's reflections and interpretations (Levers, 2013). For subjectivists, the subject and object, observer and observed, or mind and world cannot be separated (Moon & Blackman, 2014). Researchers who hold a subjectivist epistemology regard the knower and respondent as cocreators of knowledge and understanding (Denzin & Lincoln, 2011). The value of subjectivist research is in revealing how an individual's experience shapes their perception of the world (Moon & Blackman, 2014). Bate and Robert (2007a) present the 'subjective pathway' as a major focus of the EBCD approach to user design. In contrast to an 'objective pathway' with its focus on the different steps in a process, for example how a student logs-on and navigates the university intranet, the 'subjective pathway' is concerned with how the process is experienced, particularly "...the events, people, and issues along that pathway which shaped the experience" (Bate & Robert, 2007a, p. 32).

Constructionism is the view that knowledge, or reality, is constructed through human interaction with their world (Crotty, 1998), though a 'real world' exists outside of human knowledge individuals construct understanding based only on their experiences of that world (Nola & Irzik, 2006). Objects and events in the world are without meaning until humans through conscious awareness, engagement and experience of those objects and events attach meaning, hence "meaning is not discovered but constructed" (Crotty, 1998, p.42) and "Knowledge and truth are created by the mind" (Petit & Huault, 2008, p.4). Fundamentally, "God alone can know the real world because He knows how and of what He has created it. In contrast, the human knower can only know what the human knower has constructed" (Von Glasersfeld, 1989, p.123). A central tenet of constructionism is the idea that meaning can only be constructed when there is an interaction between a subject (the interpreter) and the world (the interpreted) (Crotty, 1998), and those constructs are determined by the interpreter's previous experiences, beliefs and background knowledge (Ultanır, 2012). For Crotty (1998) our view of the world is already shaped by a cultural lens where meaning is often already socially ascribed, thus 'construction of meaning' cannot take place in isolation but is 'socially constructed'.

Constructionism differs from constructivism in that the former refers to the social processing of collective experiences and generation of meaning, whereas the latter focuses on the individual cognitive processing of experiences (Crotty, 1998). In other words, constructivists make sense of the world by mentally attaching unique meaning to their individual experiences, whilst constructionists emphasise the interaction between the individual, the experience and the (social) context of that experience. Social constructions of reality frequently form the bases of qualitative research inquires (Gorman & Clayton, 1997) where an individual's construction of their own reality lays at the heart of the inquiry (Pickard & Dixon, 2004). Although more commonly associated with pragmatism, aspects of Dewey's philosophy can be observed in constructionism where he states "in an experience, interaction occurs between an individual, objects, and other people. The experience becomes what it is because of this transaction between an individual and what constitutes his or her environment. The environment consists of whatever conditions (objects or people) interact with an individual's internal personal needs, desires, capacities, and purposes that create the resulting experience" (Dewey, 1938, pp. 43-44).

Constructionism as a significant epistemology has become rooted in the study and design of many educational processes (Kynigos, 2012) with EBCD as a form of Participatory design (PD) and Action Research (AR) holding a constructionist epistemology which maintains knowledge is inherently co-constructed (Frauenberger et al., 2015) through individuals' amalgamation of already held beliefs and knowledge, and interaction with new experiences, events and activities (Ultanır, 2012), such as those experienced when students enter a higher education setting for the first time. Therefore, the epistemological assumption of the current study would fit with constructionism, as a key aim of this research is to capture, explore and understand the student experience of the nursing course including the social processes (social context of the experience) and the interactions students engage in (or touchpoints), in order to co-design priority improvements on the nursing course that enhance the student experience. Research driven by constructionist epistemologies tend to by driven by an interpretivist theoretical perspective and adopt qualitative methodologies in the research process.

3.3.2 Theoretical perspective

The second of Crotty's (1998) four elements is the theoretical perspective which provides the "philosophical stance informing the methodology" (p.3) which subsequently determines the methods adopted within a research study. Each theoretical perspective contains assumptions about how people view the world and know what they know. Key theoretical perspectives in

research include positivism and interpretivism (Crotty's, 1998). Positivism was originally conceived by Auguste Comte (1822) within the modern era of science as the development and systematic re-construction of scientific knowledge (Halfpenny, 2014). Comte (1830, as cited in Halfpenny, 2014, p.14) believed a "fundamental character of positivist philosophy is to consider all phenomena as subject to invariable natural laws". In other words, nature is ordered and regular, rather than haphazard and random, as such all-natural phenomena can be rationally and empirically observed and experimented on (Polit & Beck, 2009; Cohen et al., 2011). Positivism quickly became associated with 'the experimental' sciences (Denzin & Lincoln, 2011). Thus, a researcher who holds a positivist theoretical perspective tends to conduct research aimed at proving or disproving hypothesis, using quantitative methodologies to objectively study, discover and interpret facts.

Fundamentally, positivists only accept true knowledge as arriving from the objective observation of phenomena. Consequently, positivism is not the epistemological position that informs the assumptions about knowledge, what it is and how it is acquired within the current study. In particular, human perceptions, behaviour and interactions within the social and organisational setting of a university are not ordered regular phenomena, or subject to invariable natural laws that are easy to observe, as required from a positivist understanding within the physical sciences. In particular, the social world cannot be studied objectively and in the same way as can objects in the natural world (Bhaskar 1978; Allsop, 2013). As Ormston et al. (2014) discuss, the social world and natural world are very different since people have 'agency' and can choose what they do, whereas the natural world is governed by immutable laws. Thus, the experiences of the students within the current study take place within the social world of a university and NHS organization, with their own established culture, values, norms and traditions which guide student behaviour, potentially affecting their experiences. Today positivism has come to be recognized as the dominant paradigm within health, social science and nursing research (Butler-Kisber, 2010; Polit & Beck, 2009; Allsop, 2013). Ross (2012) even suggests positivism in the form of quantitative research holds more prestige and has made a greater impact on healthcare interventions and policy than other forms of research. However, an alternative theoretical perspective in social science research is that of interpretivism.

Interpretivism is seen as a postmodern way of understanding knowledge and how it is formed. Snape and Spencer (2003) see interpretivism as a way of overcoming some of the perceived limitations of positivism. The central purpose of interpretivism is "to understand the subjective world of human experience" (Cohen et al., 2011, p.17). Interpretivism stresses the importance of understanding the social world as it is interpreted by those being studied (Ormston, et al., 2014). For interpretivists "truth and knowledge are subjective, as well as culturally and historically situated, based on people's experiences and their understanding of them" (Ryan, 2018). Therefore, interpretivism as a theoretical perspective enables a researcher to gain a deeper understanding of a phenomenon (Creswell, 2014). Interpretivists tend to conduct research in natural settings utilising qualitative methodologies such as grounded theory and methods such as interviews (Pham, 2018). Interpretivist researchers own values and beliefs naturally influence the way they gather, analyse and interpret data (Ryan, 2018). As such interpretivist research is concerned with exploring the world using "both participants and the researchers understanding" (Snape & Spencer 2003, p.17). Interpretivist principles and values align with the aim and objectives of the current study which is to capture, explore and understand the student experience of the nursing course, and the qualitative methodology utilised for analysing those experiences. Qualitative methodologies are generally accepted as the predominant approach to studying human experiences and interactions through an interpretivist lens.

Aspects of the Deweyan framework align with the application of a qualitative approach to conducting the current study. In his paper '*Qualitative thought*' Dewey (1930/1984) expresses a view that qualitative factors and constructions are as vital a characteristic of our theory of knowledge as scientific thinking expressed in physical science. To ignore the reality of qualitative existence is to indirectly misconceive "the world in which the ordinary experience of the common man is lived" (Dewey,1930/1984, p.116). Dewey's theory of inquiry is often viewed as emerging as a compelling alternative to both scientific positivism and subjectivist relativism (Willower, 1994, p.20).

There is no fixed definition and no single accepted way of undertaking qualitative research (Snape & Spencer 2003; Denzin & Lincoln, 2011; Creswell, 2013). However, Snape and Spencer (2003, p.3) have proposed "qualitative research is a naturalist, interpretative approach concerned with understanding the meanings which people attach to phenomena within their social worlds". Therefore, "qualitative researchers study things in their natural settings, attempting to make sense of, or to interpret, phenomena in terms of the meanings people bring to them" (Denzin & Lincoln, 2011, p.3). There is no specific theory or paradigm that uniquely belongs to qualitative research, instead it claims use of multiple theoretical paradigms (Denzin & Lincoln, 2011). Creswell (2013) further suggests qualitative research shares several characteristics, firstly it takes place in the natural setting, secondly it employs multiple methods of data collection and analyses, and unlike positivist approaches the

researcher plays an instrumental part in the process. There are many methodologies that fit within a qualitative research paradigm such as ethnography, phenomenology, grounded theory, Participatory Action Research (PAR) and Experience Based Co-design (EBCD).

3.3.3 Research methodology

The research methodology forms the third domain of the four elements of a researcher's theoretical assumptions (Crotty, 1998). A research methodology is the design or strategy for meeting the aim and objectives of a study (Parahoo, 2014). Crotty (1998, p.3) differentiates between the research 'methodology' and research 'methods' by defining methodology as the "strategy, plan of action, process or design" that lays behind our choice and use of particular research methods, whereas research methods are simply procedures and techniques used to collate and analyse generated data. Research 'methodology' and research 'methods' are often confused (Grix, 2002) or used interchangeably by researchers (Parahoo, 2014). Although considered a broad term for describing the strategy used to answer a research question or test an hypothesis, a methodology must consist of the following elements; a method for collecting data, source of data, time and place of data collection and a method of data analysis (Parahoo, 2014). Stating the research methodology enables transparency around the decision-making processes adding rigour and reliability to a research study. As this study aims to capture, explore and understand the undergraduate nursing student experience, as well as recommending potential improvements to the course, a methodology that incorporated qualitative and user design elements was considered as an appropriate approach to conducting the research. EBCD was chosen as the research methodology as it is a qualitative inquiry approach to exploring phenomena, in this case student experience, that does not fit within a quantitative research paradigm. In addition, the user design elements of EBCD can be situated within the Deweyan framework (See section 3.2 Theoretical framework underpinning this study). Tenets of the EBCD research methodology utilised with the current study is presented in the following sections.

3.3.4 Experienced Based Co-Design

As the purpose of this study was to firstly capture, explore and understand the experiences of student nurses undertaking the BSc Pre-registration (Adult) nursing course and secondly to work collaboratively with study participants to identify and co-design priority improvements on the nursing course that enhance the student experience, an Experienced Based Co-Design (EBCD) methodology was utilised. EBCD is a 'user focused' design methodology, in which users of a service are integral to the process of improving the service (Bate & Robert, 2006). As the architects of EBCD Bate and Robert (2007a) believe service users possess

unique knowledge and insight into how a service can be improved due to personal experience of that service. Therefore, a key aim of EBCD is to make the service better for the user, and for the user to be involved in the design process itself (Bate & Robert, 2007a). For this study the 'user' denotes student nurses', and the 'service' refers to aspects of the BSc Preregistration (Adult) nursing course that affects the student experience.

EBCD is a relatively new approach to improving experiences and was espoused for the first time in healthcare settings by Bate and Robert (2006). Subsequently EBCD has evolved as a methodological approach to undertaking change or experience research in over 80 projects worldwide (Donetto et al., 2014). Historically, EBCD has been influenced by what Robert (2013) describes as four overlapping strands: Participate action research, User centred design, Learning theory, and Narrative, all of which have historically been used in various educational and organisational settings as distinct methodological approaches.

Originally termed Experienced Based Design (EBD) the name was later changed to Experienced Based Co-Design (EBCD) with the 'Co' prefix placing a greater emphasis on the collaboration between the user and the service provider (Donetto et al., 2014). Fundamentally, users of a service are invited to directly contribute to the design of that service (Bate & Robert, 2007a; Donetto et al., 2014). Over the past 20 years there has been a move towards more patient-centric services as part of wider healthcare reforms. However, the frameworks used to provide insight to, and improve, patient care has relied on traditional approaches such as surveys and complaints processes (Dewar et al., 2010; Robert, 2013). Bate and Robert (2006) believe those existing methods, approaches and perspectives, particularly the underlying theories that drove them, could not be relied on to deliver the required patient-centric change in a timely manner and on the scale required. In particular, they felt the need to find a 'better' more effective approach to large scale change within the NHS that specifically made the users "integral to the design process itself, focusing on their experience moving through the service and interacting with its various parts" (Bate & Robert, 2006, p.307).

Engagement is viewed as a central element to improving service experiences, with EBCD emerging as a fitting approach (Palmer et al., 2015). As a result, "the traditional view of the user as a passive recipient of a product or service gives way to the new view of users as the Co-designers of that product or service, and integral to the improvement and innovation process" (Bate & Robert, 2006, p.308). This co-design approach involves the service user and service provider reflecting on their experiences of the service and jointly identify priorities

for change (Donetto et al., 2014). Approaches to capturing and understanding student's experiences within higher education settings have relied heavily on surveys (i.e. NSS), or more informal module and course evaluations. This study invites students to be Co-designers in potentially improving student experience of their nursing course (as will be discussed in section 3.7 of this chapter and in chapter 8).

Bate and Robert (2006, p.308) define experience of a service as relating to "how well people understand it, how they feel about it while they are using it, how well it serves its purpose, and how well it fits into the context in which they are using it". EBCD makes "someone else's personal experience the object and focus of attention" with the researcher becoming "mindful of experience from the first-person point of view" (Bate & Robert, 2007b, p.48). Likewise, EBCD enables a deeper understanding of how users perceive, experience and interact with all aspects of an organisation, particularly the look, feel, processes and structures of the service (Palmer et al., 2015). In other words, there is much more to a service than just its external delivery or provision, on the contrary the internal experience plays as significant a part as the quality and usability of the service. An efficient service can still result in a poor experience; equally a poor-quality service can provide a good experience as perceived by the user of that service (Bate & Robert, 2006). For example, a lecture could be delivered on the designated date, at the allotted time and by a knowledgeable and experienced teacher, in an appropriate comfortable classroom, but some students may still perceive their experience of that session as poor. Bate and Robert (2006) suggest the design of the human experience should be targeted and considered as important as the service, environment, interactions and processes themselves. Fundamentally, the focus of EBCD is to create a whole experience that both appeals and works on a cognitive and emotional level, not just a service (Bate & Robert, 2007a). EBCD enables the service user and service providers to come together and share their experiences of that service (Tollyfield, 2014). However, experience is difficult to access or observe directly, as it is an inner, subjective, immaterial phenomenon; often a reconstruction of something lived through (Bate & Robert, 2006). Furthermore, expressing fully the meaning of an experience can be hard for research participants' (Dewar et al., 2010). Arguably, this can make the study of experience difficult. However, Bate and Robert (2006) suggest a way of capturing, exploring and understanding experience can be through the identification of key moments and places, termed 'touchpoints', where individuals come into direct contact with a service and therefore where their subjective experience is shaped.

3.3.4.1 Touchpoints

Touchpoints are a central concept of EBCD and involve identifying the key moments or events that stand out for users as crucial to their experience of a service (Bates & Robert, 2007a). Temkin (2010) refers to these touchpoints as 'moments of truth' as experienced by the participants' themselves, and it is those touchpoints that represent the most intense points of contact within a service that shape a user's experience of that service both good and bad (Locock et al., 2014). As well as being literal points of contact with the course touchpoints are also considered as "intensely personal points on the journey where one recalls being touched emotionally (feelings) or cognitively (deep and lasting memories) in some kind of way" (Bates & Robert, 2007b, p.49). For Bate and Robert (2007a) touchpoints are the 'big moments' users of any service will reflect back on and tell stories about years after their experience of that service. Thus, EBCD places particular emphasis on the emotional aspects of those touchpoints, not least because an individual's emotions are often an appraisal of and response to personally meaningful events (Mauss & Robinson, 2009). Therefore, for student nurses there will be personal touchpoints during their three-year course where they will develop emotional and cognitive associations, for example two significant touchpoints on the nursing course that emerged from the current study included the first day of course and the first clinical practice placement experience. By mapping those emotions and analysing those touchpoints a deeper understanding of the student journey through their undergraduate course can be formed. Once these interaction points or 'touchpoints' are identified and made sense of, a process of co-designing those touchpoints on the course can begin with users of, and key stakeholders in, the service.

Although previous EBCD projects were all based in healthcare settings, the tenets of EBCD can be useful for 'experience research' in higher education settings, for example within the current study. At the time of writing this thesis this study appears to be the first documented as using EBCD in a higher education setting with a focus on students as 'users' of higher education services. For the purposes of this study 'services' refer to any point of direct contact (touchpoint) with the course, for instance facilitated learning sessions, interaction with Blackboard learning and undertaking a clinical practice placement. Furthermore, studies in student experience have traditionally relied on survey approaches in a similar way to those utilised in healthcare settings as indicated by Dewar et al. (2010), which seldom engage participants in an active way in the research process to generate effective change. However, the current study aims to capture, explore, understand the student experience, and involve students and key stakeholders in the co-design of potential improvements to future experiences, through the application of the EBCD process. Users of EBCD recognise the

approach as both a methodological philosophy and a method for conducting research (Donetto et al., 2015), therefore the following section will describe the EBCD methods used.

3.3.5 Experienced Based Co-Design model

The final of Crotty's (1998) four elements is the research methods. Research methods are denoted as those techniques and procedures employed to gather and analyse data (Crotty, 1998). EBCD is viewed as a methodology that offers a set of methods or tools for conducting user design research. An EBCD model for gathering experience related data from service users (patients and carers), staff and the project management team was proposed by Bate and Robert (2007b) during their pilot study exploring the experiences of head and neck cancer patients then simplified to a four-step approach aimed at supporting users of EBCD through the process (See Figure 2 - The four steps of EBCD) (NHS Institute for innovation and improvement (NHSIII), 2009).



Figure 2 - The four steps of EBCD

By 2014 Donetto et al. (2014) had created an EBCD cycle broken down into six distinct stages: setting up the project, engaging staff and gathering experiences, engaging patients and gathering experiences, bringing patients and staff together for a Co-design meeting, forming small Co-design teams and holding a celebration event. This model has become the most frequently used approach to conducting an EBCD study and was adapted for use in the current study (See Figure 3 - Adapted Experience Based Co-design process). For the purposes of the current study the first five stages of the six-stage EBCD process was adapted for the higher education setting and applied to the current study. Notably, stage two and stage three were reversed to enable student experiences to be gathered prior to key stakeholder experiences. This was a significant change to the original EBCD cycle as this enabled a student journey map to be devised and utilised during the preceding student and key stakeholder emotional touchpoint mapping events. Donetto (2015) acknowledge the application of the EBCD process is inherently flexible and has been tailored by many projects to the contexts and needs of different organizations. As a result, "the EBCD approach proposed by Bate and Robert (2007a) has undergone a variety of adaptations in response to

a variety of local contingencies and organizational circumstances" (Donetto, 2015, p.235). The rationale for swapping stage two and stage three was so that individual semi-structured interviews with current undergraduate Pre-registration (Adult) nursing students could be conducted and their accounts of their experiences used to systematically identify touchpoints on the course.

Within EBCD users of a service are considered as best placed to understand how they experience and interact with a service, how using that service makes them feel and how well a service works for them. Therefore, these initial interviews gave early insight into student participants experiences of the course and the key touchpoints within the nursing course. Touchpoints identified from the semi-structured interviews were used to form an objective 'process map', known for the current study a student journey map across the three-year course from pre-entry to qualification, charting specific points of contact between students and aspects of the course. This 'student journey map' was then used during the emotional touchpoint mapping events to capture key stakeholder (staff) experiences and more student experiences. Similar adaptations to the EBCD approach have been previously reported by Donetto (2014) who identified researchers not conducting staff interviews as they sought to understand the experiences of patients only, and some occasions eliminating specific phases of the EBCD approach altogether (Donetto, 2015). During stage five small co-design working groups coproduced potential improvements for the BSc Pre-registration (Adult) nursing course (See Chapter 8). Recommendations were made for implementation of those priority course improvements to be completed as post-doctoral work, resulting in this model no longer being cyclical for the current study. The five stages of the EBCD process utilised for the current study are presented in more detail in the following sections.



Figure 3 - Adapted Experience Based Co-design process

3.4 Stage 1 - Setting up the project

To set up the project the research site, course and participant population was identified for the study at a northwest university where the researcher is employed as a Senior lecturer. A research ethics application was submitted with the aim and objectives of the study established before the commencement of any data collection (See Appendix 6 - STEMH Ethics Committee approval).

3.4.1 Research Aim

The aim of the research is to capture, explore and understand the experiences of undergraduate nursing students on the BSc Pre-registration (Adult) nursing course, and codesign potential improvements to future experiences, through the application of the EBCD process.

3.4.2 Research Objectives

- Capture and explore the student experience of a BSc Pre-registration (Adult) nursing course in one Northwest university
- Identify touchpoints (direct points of interaction on/with the course) experienced by student participants across the three-year BSc Pre-registration (Adult) nursing course
- Devise a student journey map through the three-year BSc Pre-registration (Adult) nursing course incorporating the touchpoints
- Explore and clarify previously identified touchpoints and map emotional highs and lows along the student journey through the nursing course
- Explore student emotional touchpoints from the key stakeholder perspective
- Conduct an Experience Based Co-design event with students and key stakeholders
- Produce recommendations and priorities for improving the student experience

3.4.3 Ethical considerations

As participants for the study would be recruited from students currently undertaking the Preregistration (Adult) nursing course, and key stakeholders in the course, particular care was taken to negotiate research governance and ethics. All ethics documentation was completed and submitted to the University Ethics Committee for Science, Technology, Engineering, Medicine and Health (STEMH). As the current study involved interviewing students whilst they were enrolled on the nursing course key ethical considerations included issues of informed consent, confidentiality and anonymity, potential risk to participants, and perceived coercion and power differential between student and researcher. Any research involving human respondents invariably raises ethical issues requiring the researcher to strike a balance between the needs of the researcher in the pursuit of truth, and the rights of the participants (Cohen et al., 2011). This is particularly the case when lecturers, involve their students as participants in their research studies as was the case in the current study. This dual role of teacher/researcher is referred to as double agency (Edwards & Chalmers, 2002), which Nutt and Bell (2002) believe can result in divided loyalties and conflicts of interest within the student/lecturer relationship, particularly in instances where the researcher potentially has direct teaching responsibilities for those same students they are recruiting from. Double agency can lead students to perceive a power differential (unequal relationship) resulting in some students feeling under pressure to participate in a study.

To avoid students feeling obligated, or under pressure, to take part in the study participants were recruited only from modules on which the researcher had no direct teaching, supervision or assessment responsibilities during the period of the study. At the time of the study the university employed a system of anonymous marking which ensures there was anonymity within the student assessment process. In addition, group assignment support had replaced individual supervision for all students on the BSc Pre-registration (Adult) nursing course, however at the point of recruitment potential participants were advised they could opt to request a different module supervisor (via the module leader) if they found themselves allocated to the researcher's supervision group in future modules. When recruiting key stakeholders for the emotional touchpoint mapping and Co-design event an agent or third party was used to recruit potential candidates and avoid potential coercion, as some of these key stakeholders were peers, managers and team members.

Additional ethical concerns related to informed consent, confidentiality and anonymity of participants, and the risks involved in participating in the study. Students were reassured, via the Participant Information Sheet (PIS) and consent form, that there was no risk to future grades or their place on the course based on anything they disclosed during the semistructured interviews or emotional touchpoint mapping event. Furthermore, non-participation was an option at any stage in the process, and refusal to participate would not result in discrimination or negative consequences in relation to their position or grades on the course. Written consent and verbal consent was sought from the student participants prior to the semi-structured interviews, emotional touchpoint mapping and Co-design events, to give participants the option to withdraw if they no longer wanted to take part in the study, as Webster et al. (2013) acknowledge, consent should be a process and not a single event. A final ethical consideration related to the use of 'trigger films' during the Co-design events. Within some EBCD studies initial participant interviews are recorded and edited to produce short clips that act as triggers for discussion during emotional touchpoint mapping and Codesign events. Filmed patient narratives have frequently been used to trigger discussion in Co-design events (Adams et al., 2014). However, for this study filming student interviews was carefully considered in relation to ethical issues and possible future repercussions on study participants. Filming student interviews and using the film clips in emotional touchpoint mapping and Co-design events, can result in 'deductive disclosure' as it risks participants being recognised by current peers and academics responsible for the teaching and assessment of student participants. Notably, many previous EBCD research was conducted with participants at the end of their patient journey who reflected on their past experiences. By contrast, student participants within this study were reflecting on both past and current experiences as they were still at various points of their journey through the pre-registration nursing course. Therefore, the use of 'trigger films' had the potential to negatively influence student participants future experiences of the course if their narratives were based on events that had involved a key stakeholder recruited to the Co-design event. To avoid exposing participants to any future ramification of taking part in the study and protect their interests' steps were taken to retain anonymity and confidentiality. Thus, the semi-structured interviews and emotional touchpoint mapping events were audio recorded on handheld portable recording devices following consent being given. Once the audio interviews were transcribed verbatim quotes from the interviews related to each touchpoint were pre-affixed to a student journey map depicting touchpoints across the three-year nursing course. In the absence of trigger films, these pre-affixed verbatim comments served as prompts, provoking discussion and reflection on personal experience of the touchpoints.

Arguably some authors view 'trigger films' as critical for generating discussion, however many other EBCD studies have also chosen not to film study participant interviews due to time and resource issues (Donetto et al., 2014) or in response to staff reluctance and patient nervousness (Cooper et al., 2016). Although the decision not to use trigger films is not in keeping with most EBCD studies a review by Donetto et al (2014) of 57 EBCD studies conducted international revealed only 53% used trigger films, whilst 49% audio recorded their patient interviews. As researchers have a great responsibility to act in the best interest of their study participants both during and after study completion the decision not to film interviews felt like the right decision.

3.4.4 Equipment and facilities

All semi-structured student interviews, the emotional touchpoint mapping events and Codesign events were conducted on-site at the university where the study was being conducted, in available private meeting/classrooms, away from researchers own office. They were arranged at a mutually convenient time and scheduled for when student participants were on site where possible to avoid students incurring extra travel time and costs whilst participating in the study. As discussed above within the EBCD approach filmed narratives from participant interviews have frequently been used to trigger discussion in the Co-design events (Adams, et al., 2014). Within this study film recording was not conducted during the interviews, emotional touchpoint mapping events or the Co-design event to retain participants anonymity and confidentiality, as filmed interviews potentially make individuals identifiable by current peers and academics responsible for the teaching and assessment of student participants. Instead, the semi-structured interviews and emotional touchpoint mapping events were audio recorded on handheld portable recording devices following consent being given. Other equipment utilised during the emotional touchpoint mapping events included sticky notes, bubble graphics and a map of the student nurses' journey through their three-year course developed following the semi-structured interviews. Refreshments were also provided at the emotional touchpoint mapping events and Co-design event, funded by the lead researcher as some students and key stakeholders attended these events during lunch breaks (in between scheduled classroom sessions).

3.4.5 Confidentiality and storage of data

Unique participant codes were allocated to all transcribed data, to retain anonymity and protect confidentiality. Transcription was conducted by the researcher and all transcribed data was stored electronically on a password protected database in accordance with the Data Protection Act (DPA) (1998), the UCLan Data Protection Policy (UCLan, 2015) and the General Data Protection Regulation (Information Commissioner's Office (ICO), 2018). This data will be kept for 5 years from the end of the study in accordance with the University's policy on Academic Integrity, after which it will be deleted and destroyed. The consent forms, sticky notes, bubble graphics and map of the student nurses' journey are stored in a secure "locked filing cabinet", in a locked office, on-site at the University where the study was conducted in accordance with university policy (UCLan, 2015). Data from the audio digital recordings was transferred to a university secured computer and stored within password-protected files, these will also be stored electronically for 5 years following

completion of the study in accordance with the UCLan Data Protection Policy (2015) after which it will be deleted and destroyed.

3.5 Stage 2 - Engaging students and gathering experiences

Engaging students with the research study involved a series of briefing and recruitment events. Student participants were drawn from cohorts across all three years of the course and were enrolled as undergraduate Pre-registration (Adult) nurses at the time of the study. Recruitment took place over three years between January 2017 and January 2020. Students were recruited from six separate cohorts of students to include participants from different points of the course: September 2014, September 2015, September 2016, September 2017, September 2018 and March 2015. There were no respondents from two cohorts: March 2014 and March 2016. Reasons for this are unclear. However, the March 2014 cohort was a much smaller intake of only 64 students compared to the student numbers in the September cohorts (up to 480 students). In addition, March 2014 cohort were on their final placement from December 2017 until the beginning of March 2017, before the finish of their course, with study days each Friday of placement (mostly off campus), therefore many of them may have been feeling the pressure of completing the course.

Briefing and recruitment took place at the beginning of scheduled whole group taught sessions (key lectures) to capture whole cohorts. Taught sessions where 10 minutes could be spared to discuss the study without negatively affecting student learning were identified and used. An agent, or third party, not involved with the study delivered each briefing. The briefing consisted of an introduction to the study and study aims, why the students had been approached and invited to participate in the study, and what would happen if they agreed to take part in this study. A Participant Information Sheet (PIS) and expression of interest form (See Appendix 7) was left with the students at each briefing event with the intention of giving students time to read the information provided and make an informed decision about their willingness to participate. Crucially, during briefing students were advised that nonparticipation was an option at any stage in the process without consequences and would not affect their place on the course or impact any future marks or assessments. Students were also advised that any data gathered during the semi-structured interviews and emotional touchpoint mapping would be anonymised at the time of transcription, thus if they chose to withdraw from the study after the data collection and transcription their data may still have been used as part of the study. In addition, flyers were placed around the university during the second data collection phase for the emotional touchpoint mapping event to capture the interest of those students who may have missed the briefing sessions (See Appendix 8 -

Study flyer for student participants). Data collection for the study took place over four phases using a combination of data collection methods including semi-structured interviews, emotional touchpoint mapping events and Co-design events (See Figure 4 – Data collection phases).

Phase one: Semi-structured interviews

Conduct individual interviews with undergraduate Pre-registration (Adult) nursing students Capture student experience on their course. Generate touchpoints for student journey map to be used during phase two data collection n=8 students

> Phase two: Student emotional touchpoint mapping event Conduct an emotional touchpoint mapping event with student participants n=7 students

Phase three: Key stakeholder experience event incorporating emotional touchpoint mapping Conduct emotional touchpoint mapping event with key stakeholders in service provision, including course administrators, course year leads, module leaders, lecturers, practice education facilitators and mentors n=8 key stakeholders

Phase four: Student and key stakeholder Co-design events Conduct Co-design event with students and key stakeholders n=7 students / n=11 key stakeholders

Figure 4 - Data collection phases

3.5.1 Sampling strategy, student demographics

Within the current study purposive sampling, a purposeful non-random strategy undertaken when researchers intentionally choose cases or participants' according to specific characteristics, criteria or features was used, (Polit & Beck, 2009; Silverman, 2010; Davis et al., 2013). Purposive sampling can be used when there is a need to access people who have in-depth knowledge about specific issues or phenomena, therefore within the current study those perceived as having in-depth knowledge about their experiences on the adult field of the pre-registration nursing course are undergraduate student nurses currently enrolled on the course and key stakeholders who manage and deliver the programme. Within phase one, two and four of the data collection process the target population for the purposive sample was student participants enrolled at different stages of progression across the three years of the BSc Pre-registration (Adult) nursing course who volunteered to take part in the study. Although female students still represent the majority population, Pre-registration nurses are a diverse non-homogenous group which includes a growing number of Black, Asian and

Minority Ethnic (BAME) and male students (NMC, 2015). In addition, mature students, defined as those aged over 21, represent 67% of the total adult field of the nursing course at the university currently under study (Smith, 2015). Although maximum variation sampling (a type of purposive sampling) of the respondent sample was not intentionally employed the student participant sample table below indicates a variety of participants in terms of age, gender, ethnicity and year of study were recruited for this study (See Table 4 - Participant sample). Similarly, an inclusion/exclusion criterion was applied to students recruited for the study (See Table 3 - Student participant inclusion/exclusion criteria). Student participant demographics from the semi-structured interviews, student emotional touchpoint mapping and Co-design event are summarised in table 5 and figure 5 in relation to cohort, year of study, age, gender and ethnicity (See Table 5 – Student participant demographics and Figure 5 - Representation of respondents). In total 22 student participants are identified by a unique participant code for example, SP01 for student participant 1.

Inclusion criteria	Participants enrolled as a student on the undergraduate Pre-registration (Adult) nursing
	course within the university where the study was being conducted
Exclusion criteria	Students who have transferred to the Adult field of the Pre-registration nursing course
	from another field of the Pre-registration course (i.e. Child / Mental Health fields)
	Students who have transferred to the Adult field of the Pre-registration nursing course
	from another university's Pre-registration nursing course
	Students currently undertaking the MSc route of the Pre-registration nursing course

Table 3 - Student participant inclusion/exclusion criteria

Data Collection	Study Participant Total	Maximum Variat	ion	Total
		Gender	Male	N=1
			Female	N=7
Phase one	N=8 students	Ethnicity	BAME	N=1
Semi-structured interviews			White	N=7
		Age	18 - 21	N=1
			> 21	N=7
		Year of study	Year one	N=3
			Year two	N=2
			Year three	N=3
		Gender	Male	N=0
			Female	N=7
Phase two	N=7 students	Ethnicity	BAME	N=1
event			White	N=6
		Age	18 - 21	N=0
			> 21	N=7
		Year of study	Year one	N=1
			Year two	N=0
			Year three	N=6
Phase three Key stakeholder emotional touchpoint mapping event	N=8 key stakeholders	N/A	N/A	N/A
		Gender	Male	N=2
	N=7 students		Female	N=5
Dhase four		Ethnicity	BAME	N=1
Student and key stakeholder Co-design			White	N=6
events		Age	18 - 21	N=0
			> 21	N=7
		Year of study	Year one	N=4
			Year two	N=4
			Year three	N=4
	N=11 key stakeholders	N/A	N/A	N/A

Table 4 - Participant sample

Id - Code	Year of study	Gender	Event
01	Year 2	Female	Semi-structured Interview
02	Year 1	Female	Semi-structured Interview
03	Year 1	Female	Semi-structured Interview
04	Year 1	Female	Semi-structured Interview
05	Year 3	Female	Semi-structured Interview
06	Year 2	Female	Semi-structured Interview
07	Year 3	Female	Semi-structured Interview
08	Year 3	Male	Semi-structured Interview
09	Year 3	Female	Emotional touchpoint mapping event
10	Year 3	Female	Emotional touchpoint mapping event
11	Year 3	Female	Emotional touchpoint mapping event
12	Year 3	Female	Emotional touchpoint mapping event
13	Year 3	Female	Emotional touchpoint mapping event
14	Year 3	Female	Emotional touchpoint mapping event
15	Year 1	Female	Emotional touchpoint mapping event
16	Year 3	Female	Co-Design event
17	Year 3	Female	Co-Design event
18	Year 3	Male	Co-Design event
19	Year 3	Female	Co-Design event
20	Year 3	Male	Co-Design event
21	Year 2	Female	Co-Design event
22	Year 2	Female	Co-Design event

Table 5 - Student participant demographics



Figure 5 - Representation of respondents

3.5.2 Student participant semi-structured interviews

During phase one data collection individual semi-structured interviews with undergraduate Pre-registration (Adult) nursing students were conducted to capture stories, or accounts, of their experiences on their course and use their experiences to systematically identify touchpoints. Within EBCD these initial interviews enable the researcher to gain early insight into participants experiences of a service and the key touchpoints within that service (Bates & Robert, 2007b). Touchpoints identified from the semi-structured interviews were used to form an objective 'process map', known as a 'clinical care pathway' within healthcare projects (Bates & Robert, 2007b). For the current study this was a map of the student journey across the three-year course from pre-entry to qualification, charting specific points of contact between students and aspects of the course. It is these touchpoints that represent points of contact within the course and form the 'objective pathway' that are crucial to student's experience of that course, for example the first day of the course. The 'student journey map' was used during the student emotional touchpoint mapping event to enable participants' to determine which were particularly emotional touchpoints from their experience of the nursing course, alongside acting as a trigger for discussion in place of the trigger films.

Prior to each semi-structured interview students were reminded of the purpose of the study and that they could withdraw from the study at any time up to the anonymisation of data and asked to complete a consent form if they chose to take part in the research (See Appendix 9 - Student participant consent form for semi-structured interviews). All semi-structured interviews were conducted on-site at the university main campus at each participants' convenience and recorded using a handheld portable recording device with written consent from the participant. Interviews ranged from 24 minutes to 58 minutes in length, with most interviews lasting around 40 minutes. A standard one-page topic guide which had received ethics approval formed the bases of each interview (See Appendix 10 - Semi-structured interview topic guide). However, participants were encouraged to discuss their own personal experiences even when that meant deviating from the topic guide. Once transcribed by the researcher all semi-structured interviews were analysed using Braun and Clarke (2006) 'Six phases of thematic analysis' for touchpoints with the course (See Section 3.8 Data analysis).

3.5.3 Student participant emotional touchpoint mapping event

The objective of the emotional touchpoint mapping event was to capture student participant's emotional highs and lows along their journey through the nursing course. During an emotional touchpoint mapping event participants are asked to map their emotions as they relate to the touchpoints most poignant to them on the student journey map produced from the previous
semi-structured interviews (Bates & Robert, 2007b). The process began with participants being reminded of the purpose of the study and that they could withdraw from the study at any time up to the anonymisation of data. They were then asked to complete a consent form if they still wished to take part in the research (See Appendix 11 - Student participant consent form for emotional touchpoint mapping event). After this student participants began exploring the student journey map containing touchpoints generated from the semi-structured student interviews. Participants were then asked to add their own experiences and comments to the journey map in the form of post-it note comments and stickers, with a specific focus on describing their feelings and emotions as they linked to the touchpoints that most related to them. 'Emotional mapping' is a significant process in EBCD as this transforms an 'objective pathway' into a 'subjective experience map' with these "emotionally charged touchpoints" providing an informed vantage point into the bigger picture of a service users whole experience and how the different processes fit together to form the whole (Bates & Robert, 2007a, p.53). Notably it is the positive and negative subjective experiences described by students that adds meaning to the student journey. Using verbatim quotes from student participants in the current study figure 6 demonstrates how the subjective experiences of some students relate to specific touchpoints early in the BSc Pre-registration (Adult) nursing course (See Figure 6 - Example of a subjective experience map).

During the emotional touchpoint mapping event an audio recorded group interview, similar to a focus group, took place with all student participants once emotional mapping was completed. The objective of the audio recorded interview was to gain an in-depth understanding of some of the stories and experiences identified on the post-it notes and explore participants thoughts and feelings whilst conducting the emotional mapping of the touchpoints. The audio recording was then transcribed and analysed for core themes and subthemes using the Braun and Clarke (2006) 'Six phases of thematic analysis framework'.



Figure 6 - Example of a subjective experience map

3.6 Stage 3 - Engaging Key stakeholders and gathering experiences

Within this study key stakeholders are considered as 'knowledgeable people' in relation to the provision of the course, i.e. the education providers. During phase three data collection purposive sampling was utilised to recruit key stakeholders based on their key role in relation to the BSc Pre-registration (Adult) nursing course. Thus, specifically identified course administrators, the course leader, each year leader, module leaders, lecturers and senior lecturers who teach an assess on the programme, work-based learning facilitators, nurse educators and clinical placement co-ordinators were emailed information about the study, the study aims, why they had been approached and invited to participate in the study, and what would happen if they agreed to take part in this study. A Participant Information Sheet (PIS) and consent form (See Appendix 12 - Key stakeholder Participant Information Sheet (PIS) and consent form for emotional touchpoint mapping event) was included in the email to give them time to carefully read the information provided and make an informed decision about their willingness to participate in the study. In total 16 key stakeholders were recruited to the study, eight for the key stakeholder touchpoint mapping event and 11 for the Co-design event. Notably, three key stakeholders attend both the touchpoint mapping event and the Co-design event. To maintain their anonymity throughout this thesis key stakeholder participants are identified by a unique participant code for example, KS01 for Key stakeholder 1 (See Table 6 - Key stakeholder demographics from emotional touchpoint mapping event and Co-design event). The three key stakeholders who attend both events were allocated a sperate code for each event, again to retain a level of anonymity, resulting in 19 key stakeholder id codes overall.

3.6.1 Key stakeholder emotional touchpoint mapping event

The objective of the key stakeholder emotional touchpoint mapping event was to establish their thoughts and feelings in relation to the student participants experience of various touchpoints with the nursing course. After exploring the student journey map key stakeholders were encouraged to add their own comments, feelings and emotions in the form of post-it notes. This was followed by an audio recorded group interview similar to the focus group that took place during the student emotional touchpoint mapping event. The objective of the audio recorded interview was to enable the researcher to explore some of the key stakeholders post-it notes comments in more depth. Subsequent to the event the audio recorded interview was transcribed and analysed for core themes and subthemes using the Braun and Clarke (2006) Six phases of thematic analysis framework.

3.6.2 Key stakeholder demographics

ld - Code	Designation	Event			
KS01	Work-based learning lead	Emotional touchpoint mapping event			
KS02	Principal lecturer Emotional touchpoint mapping e				
KS03	Pre-registration nursing course administrator	Emotional touchpoint mapping event			
KS04	Pre-registration nursing course Year lead Emotional touchpoint mapping e				
KS05	Nurse Educator	Emotional touchpoint mapping event			
KS06	Lecturer/Module leader Emotional touchpoint mapping of				
KS07	Lecturer/Academic advisor	Emotional touchpoint mapping event			
KS08	Senior lecturer/Academic advisor Emotional touchpoint mapping				
KS09	Senior lecturer/Academic advisor	Co-design event			
KS10	Senior lecturer/Module leader	Co-Design event			
KS11	Senior lecturer/Course leader Co-design event				
KS12	Lecturer/Module leader	Co-design event			
KS13	Lecturer/Academic advisor	Co-design event			
KS14	Nurse Educator	Co-design event			
KS15	Pre-registration nursing course Year lead	Co-design event			
KS16	Pre-registration nursing Course leader	Co-design event			
KS17	Pre-registration nursing Admissions tutor/Senior lecturer	Co-design event			
KS18	Lecturer/Module leader	Co-design event			
KS19	Lecturer/Academic advisor	Co-design event			

Table 6 - Key stakeholder demographics from emotional touchpoint mapping event and Codesign event

3.7 Stage 4 - The Student and Key stakeholder Co-design event

The overall objective of the Experience Based Co-design event was for students and key stakeholders to set mutually agreed priorities for co-designing and implementing course improvements to future educational provision. Fundamentally, the co-design process within EBCD requires staff and service users to meet, reflect on their experience of a service (whether receiving or providing that service) (Bate & Robert, 2006) and through collaboration identify improvement priorities, then devise and implement changes to that service (Donetto et al., 2014). Co-design events deliberately engage service users and service providers in dialogue. This enables both groups to share and understand each other's perspectives (Boyd et al., 2012). Within this study the co-design process involved students working alongside course leaders, module leaders, lecturers, senior lecturers and practice educators (key stakeholders) to identify priority areas of the BSc Pre-registration (Adult) nursing course for co-design and improvement, and together devise plans for implementing those changes.

Two Co-design events were held on site at the University's main campus. The first Co-design event recruited seven students and seven key stakeholders, however on the day of the event only the key stakeholders attended. Four students emailed their apologies and one student 'mixed up the days' and missed the event. Therefore, the first event went ahead with just key stakeholders and a second event was organised for a few weeks later with both student and key stakeholder participants. For the second Co-design event seven more students were recruited and attended the event alongside four different key stakeholders. In total 11 participants attended the second Co-design event. Consent was gained from both student and key stakeholder participants, including consent to photograph the event (See Appendix 13 - Student and key stakeholder participant Co-design event with both students and key stakeholder participants by Donetto et al. (2014) reveals a more usual approach is to hold a separate user/staff event followed by a joint event. Therefore, the process described above was deemed not to have a detrimental effect on the overall results of this study.

The format of the Co-design event was based around the 'double diamond design model' (British Design Council, 2005) selected because participants work through a process of exploring issues more widely (divergent thinking) before conducting a more defined focused exploration of the issues (convergent thinking). The Double Diamond framework is a useful Co-design tool and can be specifically tailored to different user groups (West et al., 2017). The double diamond model is divided into four stages; discover, define, develop and deliver (Huq & Gilbert, 2017). The discovery stage of the Co-design event involved all participants

exploring the student journey map and sharing their own thoughts and feelings around the comments and touchpoints (See Figure 7 and 8 - Co-design event discovery stage, image 1 and 2).



Figure 7 - Co-design event discovery stage, image 1



Figure 8 - Co-design event discovery stage, image 2

3.8 Stage 5 - Small co-design working groups

In stage 5 of the EBCD process a summary of the core themes in the form of (defined) problem statements that emerged from the emotional touchpoint mapping events were explored in small co-design working groups using the 'KJ brainstorming technique' (See Appendix 14 - Problem statements tabulated from student and key stakeholder emotional

touchpoint mapping events). The KJ brainstorming technique is a group consensus approach to prioritising problems and solutions (Spool, 2004) originating from the early work of Japanese anthropologists Jiro Kawakita in the 1950s (Hunt et al., 2016a) (See Table 7 - KJ brainstorming technique adapted for Co-design event). Using the KJ brainstorming technique each small co-design working group developed possible solutions at a wider level before collaborating as a group to agree priority improvements to the future delivery of the nursing course (See Figure 9 - Small co-design working groups recording and sharing ideas and solutions, Figure 10 - Grouping or clustering ideas and solutions). The priority improvements from the Co-design events are discussed in chapter 8 of this thesis.

Steps in the process	KJ method	Application to Co-design event
Step 1	Recording ideas	Student and key stakeholder participants were divided into small working groups of 3-4 participants. After reading an allocated problem statement each individual silently writes down ideas/solutions to the stated problems on separate post-it notes
Step 2	Sharing ideas	Individuals within each working group takes turns to share all their ideas whilst others within the group listen. Those listening can develop their ideas or add new ideas if inspired to do so.
Step 3	Grouping or Clustering	Working together participants group similar ideas or concepts together. Each group of concepts/ideas is named with a single word or short phrase that best captures the core concept underlying the ideas in that group.
Step 4VotingThe participants then vote on which solution a set number of votes (or "dots") that they want most prioritised. The solutions that g priorities for action		The participants then vote on which solutions matter most. Each person gets a set number of votes (or "dots") that they can place on the solutions they want most prioritised. The solutions that get the most dots represent the priorities for action

Table 7 - KJ brainstorming technique adapted for Co-design event



Figure 9 - Small co-design working groups recording and sharing ideas and solutions



Figure 10 - Grouping or Clustering ideas and solutions

A traditional approach to conducting Co-design events within EBCD is to have a joint Codesign event (stage 4) where staff and service users reflect on and share their experiences, followed by the small Co-design working groups (stage 5) where small groups work redesigning services to address the agreed priorities (Fylan et al., 2021). However, a variety of approaches has been noted in several EBCD studies (Donetto et al., 2014). Previous adaptations to the EBCD approach include holding one joint co-design event with no followup co-design meetings to shorten the co-design process (Donetto et al., 2014). Instead, those projects placed a stronger emphasis on journey mapping and identifying priorities together (Donetto, 2015). For this study, the decision was taken to conduct stage 4 (the student and key stakeholder Co-design event) and stage 5 (the small co-design working groups) simultaneously as part of the one event. Reasons for this included the time and logistics of arranging several smaller co-design events for students and staff who may have had concerns about the demand on their time. A further reason was to ensure the momentum gained during stage 4, the sharing of experience and identifying priorities for change, was continued into stage 5 (working on solutions), as appeared to be this case in this study.

3.9 Data analysis

The purpose of qualitative data analysis is to understand and make sense of the data in a way that enables meaningful interpretation that will ultimately represent the data (Creswell, 2009). Data analysis is the active and generative process of "deep and prolonged data immersion, thoughtfulness and reflection" (Braun & Clarke, 2019, p.591). Both narrative analysis and thematic analysis methods were reviewed for analysing the data in this study. Narratives in the form of stories and storytelling, have developed as a key method of

capturing touchpoints within EBCD (Locock et al., 2014), suggesting narrative analysis may have been an appropriate approach for analysing the data in the current study.

Several narrative approaches were explored including structural narrative analysis, dialogic narrative analysis and thematic narrative analysis (Riessman, 2008). Dialogic narrative analysis focuses on context, social/cultural settings and interactions (Riessman, 2008), it considers how the story connects people and centres on the relationship between a story, storyteller and listener (Frank, 2010). Whereas structural narrative analysis as the name suggests focuses on how a narrative is structured or put together. In other words, how the story is told, and the way language is used to tell the story holds more significance than the story being told (Frank, 2012). The Labovian model is a frequently used structural narrative analysis approach (Labov & Waletzky, 1967) within many social research studies. Labov and Waletzky (1967) present a framework for both identifying and analysing narratives that include orientation (the characters, time, place, and initial behaviour), complication (the action), evaluation (the point of the story), resolution (result of the action), and coda (ends the story for the listener and returns them to the current moment). In other words, what is the story about, who was involved, where and when did it take place, what happened and why is it important? Notably, fully formed stories contain all the above elements (Hyvarinen, 2008). Labov's model is described as 'event-centred' in that it expresses narrative in terms of its representation of events (Patterson, 2013). A central element of the current study is its use of EBCD with a focus on touchpoints, with many of these touchpoints likely to be presented as unfolding 'events' during the semi-structured interviews and emotional touchpoint mapping events, making this a fitting method for analysing the data generated. Similarly, dialogical narrative analysis gives a voice to the student participants in the current study as individual storytellers, suggesting dialogic narrative analysis as a potential approach to data analysis.

However, after careful review both a dialogical narrative analysis and structural narrative analysis were disregarded for use in this study. Dialogic analysis was eliminated as it gives prominence to the standpoint of the storyteller (Frank, 2012). However, as EBCD is a collaborative approach many other interacting stories and voices from key stakeholders such as mentors, lecturers, administrative staff etc. need to be included in the analysis. Structural narrative analysis was rejected as it can be 'narrowly structuralist' (Holstein & Gubrium, 2012), thus the focus of structural analysis becomes how the story is told (structured), the language used, and the sequence of events rather than the story being told. Riessman (2008, p.77) sees this as a shift in focus from the "narrators experience to the narrative itself". Furthermore, Labov (1997) views true narrative as having a well-formed beginning, middle,

and end. Therefore, within a strictly Labovian analysis, allowance is not made for any partially constructed account of personal experience (Patterson, 2013). Instead, each story must be considered as a whole rather than fragments of stories, which can result in relatively few participants stories being used in the research (Frank, 2012). Whilst transcribing the semistructured interviews and emotional touchpoint mapping events for the current study it became clear that many of the students and key stakeholder stories were not discrete events or fully formed narratives, based on Labov's classification of a narrative. Therefore, thematic analysis was explored as an alternative approach to structural and dialogic narrative analysis for analysing the data gathered in this study.

Interestingly, although Bate and Robert (2007b) used narrative interviews as part of their data collection in their original EBCD study, they identified thematic analysis as their method to analysing the transcribed data rather than narrative analysis. Furthermore, a review of 11 EBCD studies dated from 2006 and published in national or international journals indicate there is no consensus around a data analysis approach within EBCD. Although all studies identified EBCD as their methodological approach, six studies did not state their process for data analysis including the seminal work of Bates and Robert (See Bate & Robert, 2006; Bate & Robert, 2007b; Pickles et al., 2008; Dewar et al., 2010; Leeman & Wardekker, 2011; Carr et al., 2011). One study described their analysis involving two researchers comparing and contrasting touchpoints (Tsianakas et al., 2012a). A further study used statistical analysis to test a co-design technique aimed at optimizing psychosocial recovery outcomes in those affected by mental illness (Palmer et al., 2015). Only three articles identified a specific qualitative data analysis approach to their study, and in all three cases this was thematic analysis (See Tsianakas et al., 2012b; Locock et al., 2014; Tollyfield, 2014). Based on the review of narrative analysis methods described above, and the exploration of prior EBCD research which predominantly signifies the use of 'thematic analysis' within their studies, a thematic analysis approach was chosen for analysing the data in this study as described in the following section.

3.9.1 Thematic analysis

Thematic analysis is a process for discovering patterns within qualitative research data by analysing, organizing, describing, and reporting those patterns as themes (Braun & Clarke, 2006). Thematic analysis is one of the most frequently used forms of data analysis in qualitative research (Guest et al., 2012). Although the intention of a thematic analysis is to identify themes or patterns in data sets, a good thematic analysis should also interpret and make sense of those themes rather than simply summarising the data (Maguire & Delahunt,

2017). Thus, thematic analysis is most suitable for research studies that seek to discover through interpretation (Alhojailan, 2012) such as in the current EBCD study. Although a significant amount of data analysis within qualitative research is essentially thematic, Braun and Clarke (2006, p.79) argue thematic analysis as an analysis method is "poorly branded", often given a different identity such as content or discourse analysis and rarely receiving the recognition it deserves. Several reasons for this abound, firstly there is insufficient literature outlining the process for thematic analysis, and therefore no clear agreement about how to use this method to conduct a rigorous and relevant analysis (Nowell et al., 2017). Secondly, thematic analysis is portrayed as one of the easiest research tools to use, and therefore too simple to produce high quality data analysis (Vaismoradi et al., 2013). As such it is the most recommended approach for researchers at the beginning of their research careers (Vaismoradi et al., 2013). Furthermore, some research students are advised thematic analysis "is not sophisticated enough for a doctoral project" (Braun & Clarke, 2014, p.1). Vaismoradi et al. (2013) disagree, believing the use of thematic analysis does not necessarily produce simple or low-quality findings (Vaismoradi et al., 2013). Braun and Clarke (2014) go on to defend the use of thematic analysis as a robust method for coding datasets, and for identifying patterns across qualitative data, even suggesting it should be the first qualitative analysis method researchers learn to use, as it develops core skills that lay the foundation for conducting many other kinds of research analysis.

Clarke and Braun (2013, p.2) identify thematic analysis as an "analytic method, rather than a methodology". Maguire and Delahunt (2017) believe this makes thematic analysis a more flexible approach for analysing data as it is not tied to any particular theoretical perspective or epistemological. Instead, it is possible to apply thematic analysis to a wide range of theoretical frameworks or qualitative approaches (Clarke & Braun, 2013), such as EBCD. Many authors use thematic analysis as a tool for analysing data within other qualitative methods, although Nowell et al. (2017) consider thematic analysis as a method in its own right. Thematic analysis provides a well-structured approach for examining different research participant perspectives, highlighting similarities and differences, and generating unanticipated insights (King, 2004). Braun and Clarke (2006) identify two approaches to thematic analysis: deductive (or theoretical) thematic analysis and inductive thematic analysis. Researchers who use theoretical thematic analysis examine data via the lens of pre-existing theoretical frameworks whereas an inductive approach is data-driven and reflexive (Braun & Clarke, 2006). Cited over 73,000 times Braun & Clarke (2006) present a sophisticated and systematic step-by-step approach to using thematic analysis that has been widely adopted by many subsequent qualitative research studies (See Figure 11 - Braun and Clarke (2006) Six phases of thematic analysis). Though this six-phase model is presented as consecutive steps, a researcher will find themselves reflectively and iteratively moving back and forth between each phase as their analysis develops (Nowell et al., 2017). For the current study an inductive approach using Braun and Clarke (2006) six phases of thematic analysis was utilised for analysing the research data.

- Phase 1 Familiarising yourself with your data
- Phase 2 Generating initial codes
- Phase 3 Searching for themes
- Phase 4 Reviewing themes
- Phase 5 Defining and naming theme
- Phase 6 Producing the report

Figure 11 - Braun and Clarke (2006) Six phases of thematic analysis

The first phase requires researchers to familiarise themselves with their data through a process of listening to audio data and reading datasets until they become so immersed they "learn the content of the dataset inside out" (Clark & Braun, 2014, p.2). During this phase researchers must start identifying interesting features, and possible patterns relevant to the research question (Braun & Clarke, 2006). Within the current study the researcher became familiar with the data by personally transcribing all the semi-structured interviews, the emotional touchpoint mapping events (students and key stakeholders), and the final experienced based Co-design events. During transcription the researcher was able to listen to the audio recordings and read/re-read transcripts for accuracy, thus becoming fully immersed in the data. During phase two of Braun and Clarke's (2006) six phase model the researcher becomes familiarised with the data and begins to generate initial codes. A code is simply a way of labelling data. Any interesting feature of the data can be identified as a code (Braun & Clarke, 2006), but the real purpose of coding within thematic analysis lays in its ability to form connections, or patterns, across the data (Alhojailan, 2012). In this way coding data becomes a systematic process during which specific statements, or data sets, begin to form categorized themes (Creswell, 2014). Some initial codes form main themes, whilst some codes form subthemes (Braun & Clarke, 2006). Vaismoradi et al. (2013) describes subthemes as sharing the central idea of a theme but with more detailed focus on one prominent aspect of that theme. Searching for themes (phase 3) and reviewing themes (Phase 4) requires a researcher to understand what constitutes a theme as opposed to a code.

A theme organizes codes into unified groups of ideas with shared attributes, common points of reference and generalisability (Vaismoradi et al., 2013). A theme should "capture the key idea about the data", or "something important", as it relates to the research question (Braun & Clarke, 2006, p.82), and contain repeating ideas (Vaismoradi et al., 2013). Thus, within this phase analysis shifts to a much wider focus than the coded statements (Clark & Braun, 2014). Themes should link to form a consistent picture, or explanatory model, that relates to the overall study question (Bazeley, 2009). Furthermore, themes do not simply 'emerge' from data and are not waiting to be discovered, instead the process of identifying themes is an active one (Braun & Clark, 2006), requiring the researcher to be consistent and transparent throughout the whole process (Jugder, 2016). Themes can be analysed at two levels, a semantic or latent level. A semantic approach uses the direct statements, or words, of participants to formulate themes, or what Braun and Clarke (2006, p.84) describe as "the explicit or surface meaning of the data". Thus, semantic content relates to the literal meaning of data and description of the patterns forming. Whereas a latent approach goes beyond the obvious surface meaning, and instead begins to "examine the underlying ideas, assumptions, conceptualisations and ideologies that are shaping or informing the semantic content of the data" (Braun & Clarke, 2006, p.84), or hidden meaning. Javadi and Zarea (2016) describe this as a more analytical approach than a semantic method, requiring a researcher to move from description to interpretation of the patterns forming, and discussion on how findings fit with a wider meaning or frameworks. Thematic analyses tend to draw on both a semantic and latent approach to identifying themes, as the manifest content of semantic coding signpost a researcher to a more latent level of meaning (Joffe, 2012).

Within the current study initial codes formed semantic themes generated from the first eight semi-structured interviews, based on instances where study participants described an experience or emotional touchpoint on pre-registration nursing course. These themes, and verbatim quotes from the interview transcripts, were used to create a visual map of the course with initial touchpoints identified across that map (See Appendix 15 - Map of the student journey through the course). This map was mounted on the wall during the emotional touchpoint mapping event to enable different study participants to generate discussion, review the initial themes, explore and develop possible new codes and themes in accordance with phase 4 of Braun and Clarke (2006) model. Once the emotional touchpoint mapping events were transcribed, thematic analysis was undertaken with more focus on the latent content (See Appendix 16 - Example thematic analysis from student emotional touchpoint mapping event). Defining and naming themes (Phase 5) is a process of continuing data

analysis in order to refine each theme, generate detailed definitions and theme names as they relate to the overall story of the data analysis (Braun & Clarke, 2006). Clark and Braun (2014, p.3) describe this phase as where "the most substantive, interpretive analytic work is done", with detailed and complex definitions of each theme being produced, and Braun and Clarke (2006, p.92) believe this where a researcher identifies "the essence of what each theme is about". Following this phase a researcher will begin selecting data extracts (verbatim quotes) and narratives to be used in the overall analysis and final report (Phase 6).

Computer Assisted/Aided Qualitative Data Analysis (CAQDAS), in other words the utilisation of computer software to assist with qualitative research analysis was used during the analysis phases of this study. Bold (2012) notes that "software packages do not do the analysis", instead the benefit for the researcher is that they assist with the practicalities of sorting and classifying data. NVivo was specifically chosen as this was both the software the researcher had access to and had extensive training in the use of prior to analysing any data.

3.9.2 Trustworthiness

The validity and reliability of a research method can be determined by the level it "measures what it sets out to measure" (Parahoo, 2014, p.38). The strength of a research studies data is defined by its reliability and validity (Lewis & Ritchie, 2003). However, concepts of rigor, validity and reliability are often shunned in qualitative research due to their long association with empirical research (Polit & Beck, 2009). Consequently, qualitative research such as the current study tend to favour a "parallel perspective" related to a study's "trustworthiness" (Polit & Beck, 2009, p.490). Lincoln and Guba (1985) first posited the idea of trustworthiness to parallel quantitative criteria for ensuring scientific rigour. The criterion for trustworthiness is framed by credibility, transferability, dependability and confirmability (Guba, 1990).

Credibility relates to a study's internal validity (Lincoln & Guba, 1986) or the extent of the relationships between different factors established in a study. Lincoln and Guba (1986) suggest credibility can be establish in a study through prolonged researcher engagement in the field of study, persistent observation of the concept understudy, triangulation or cross-checking of the data, peer debriefing, negative case analysis and member checks. As evident from the current study prolonged engagement and persistent observation with the participants under study occurred over the researchers 20-year experience teaching and supporting students and key stakeholders on the pre-registration nursing course as a Senior lecturer, module leader, cohort lead and Academic advisor. Triangulation and member checks and the

process of emotional touchpoint mapping and co-design where study participants themselves identified the experiences that best reflected their own on the student journey map, and collaboratively decided the themes and concepts that would form the priority recommendations for course improvement alongside the key stakeholders. Debriefing took place during monthly meetings with a PhD supervisory team who were experienced researchers with many years' experience in nurse education.

Dependability relates to the reliability of the study findings (Lincoln & Guba, 1986). The monthly supervision meetings and ensuing 'record of supervisory meeting notes' provided an audit trail of the research process employed and offered an opportunity to ensure dependability of the study results. Pandey and Patnaik (2014) perceive the feedback from researchers external to a study as invaluable for critiquing decisions made during the research process. Furthermore, clearly outlining the stages of the EBCD process adapted for use in the current study enables the reader to appraise dependability of this study. A measurable and transparent audit trail of the data analysis process (as indicated in section 3.8 Data analysis) facilitates confirmability of study results by exposing the level of objectivity achieved during the process (Lincoln & Guba, 1986).

Transferability of a study's findings is perceived as analogous to generalisability (Polit & Beck, 2009), assessed by the reader of that research and determined by the "thick description of the situation studied" (Guba, 1990, p.113). Within the current study detailed descriptions of the semi-structured interviews, emotional touchpoint mapping events and Co-design events alongside the rich data collated on the student journey map and presented in the finding's chapters enable the reader to judge the features of this study that hold application for both similar student populations on pre-registration nursing courses as well as the general student population.

3.10 Challenges to using Experience Based Co-design

EBCD has been successfully established as an effective process for improving patient experience in a health care setting, however this was the first study documented to use elements of EBCD exclusively in a university setting with student and staff participants. Although the use of aspects of EBCD contributed to the uniqueness of the current study, it also posed several challenges. Firstly, there were no comparative studies exploring the student experience across the whole undergraduate course using EBCD in the nursing or general student population. Similarly, there were no published studies examining touchpoints along the student journey through the course. This initially made the literature review difficult

for this thesis, therefore a broader systematic search using the search terms 'Pre-registration undergraduate student nurse experience' was applied to situate the current study. The lack of prior research studies using an EBCD approach in a university setting with undergraduate nursing students offered no opportunity to ensure methodological congruence with similar research studies. Consequently, the current study begins the process of filling the gap in the literature for research focused on touchpoints within pre-registration nursing course, and the applied use of elements of EBCD within HEI settings.

Adopting elements of an EBCD approach as part of a PhD generated practical challenges with managing a realistic timeline and conducting co-design. Previous EBCD projects in the NHS tended to have a short timeline with most projects completed in under 18 months (Donetto et al., 2014; Locock et al., 2014). Unlike those EBCD projects that map a few hours, a few days, a few weeks or even months of a patient's journey through the health care system the current study explored the student journey across the entire three-year course. As a result, the whole project took over six-years to complete and generated a multitude of touchpoints along that three-year journey. Consequently, the longitudinal timeline impacted the timing of the data collection phases. Therefore, the semi-structured interviews conducted as part of phase 1 data collection began in January 2017, phase 2 and phase 3 data collection (emotional touchpoint mapping events) took place from November 2017 completing in May 2019. In total data collection took over three-years, this meant when the Co-design event was conducted in January 2020 all but one of the student participants who had contributed to phase 1 and phase 2 data collection had graduated from the course and were unable to take part in the Co-design event. Other EBCD studies based in the NHS have reported similar problems that resulted in three EBCD studies utilizing staff members only to work on service improvements, and one study excluding the Co-design event entirely from their project (Donetto et al., 2014).

3.11 Conclusion

This chapter has explored the research paradigm underpinning the current study. EBCD was introduced as the methodological approach for conducting the study using qualitative methods. Through application of Crotty's (1998) 'four elements' the study was positioned within an interpretivist theoretical perspective shaped by a constructionist epistemology. This meant participants subjective lived experience was the focus for data collection using qualitative techniques including semi-structured interviews and emotional touchpoint mapping events, and data was analysed through an interpretivist lens using a thematic

analysis approach. Therefore, by understanding students' subjective experiences of the BSc Pre-registration (Adult) nursing course, the mechanisms that generate those experiences could potentially be re-designed to improve future student experiences.

CHAPTER FOUR PRESENTATION OF FINDINGS FROM SEMI-STRUCTURED STUDENT INTERVIEWS

'We always live at the time we live and not at some other time, and only by extracting at each present time the full meaning of each present experience are we prepared for doing the same thing in the future' (Dewey, 1938/2015, p.49)

4.1 Introduction

The following chapter presents the findings of the semi-structured student interviews from phase one data collection of this study (See Figure 4 - Data collection phases). Core themes and subthemes that emerged from the student participant semi-structured interviews are presented alongside touchpoints. Three core themes, and eleven subthemes make up the Pre-registration students experience. The three core themes include 'navigating the academic journey', 'the clinical practice learning experience'' and the 'emergent networks of support'. Overall, 18 touchpoints were generated and formulate a student journey map used as the focus of the student and key stakeholder emotional touchpoint mapping events in phase two and three data collection for this study (See Figure 13 - Overview of core themes, subtheme and touchpoints from semi-structured interview). Touchpoints are organised in chronological order to form the student journey map through the course and are colour coordinated to the core theme each touchpoint emerged from (See Figure 14 - Touchpoints on the nursing students' journey through the course). Although some touchpoints occur several times across the course for instance 'Module assessments' and 'Module Assessment Group Support', these are plotted only once on the student journey map. Decisions about the sequential order of the touchpoints was determined by when students were most likely to have contact with the service on the course, for instance students are most likely to have contact with 'Administrative support' during the application process before going onto their 'First clinical practice placement'.

Throughout this chapter touchpoints are signposted using an arrow text box and a hand symbol representing a participant coming into direct contact with the nursing course where subjective experience was potentially shaped, for example during the course enquiry, interview and selection process (See Figure 12 - Example Touchpoint).

79



Figure 12 - Example Touchpoint

Core themes and subthemes are supported by verbatim quotes from participants' transcripts to evidence links between the raw data and its interpretation using the 'Long data excerpts approach'.



Figure 13 - Overview of core themes, subtheme and touchpoints from semi-structured interview

through ration rse	Pre-entry	Enrolment	Pre-registratio	Pre-registration Nursing course year-two	Pre-registration Nursing course year-three	Graduation	
Student journey the Pre-registı nursing cou	*						
Touchpoint	Course enquiries, interview and selection process	Administrative support Using the University Facebook pages	Transition into year-one / Welcome week Facilitated learning sessions First clinical placement / subsequent clinical placements Module Assessment Group Support (MAGS)	Home group contact/meetingsAcademic advisor supportAcademic staffVirtual learning environmentPractice education facilitator support (PEF)Mentor supportModule assessmentsAssessment feedback	Transition into year-two	Transition into year-three	Qualifying



4.2 Navigating the academic journey



Figure 15 - Core theme 1 'Navigating the academic journey'

Within the data participants experience of academic study is portrayed as a journey incorporating key milestones and pivotal moments that both shape the journey and students' experience of that journey. The participants journey incorporates several sequential transitional junctures such as moving from year one to year two or transitioning from the academic environment to the clinical practice placement environment. Participants responses during the semi-structured interviews indicate the core theme 'navigating the academic journey' is influenced by six subthemes including 'the journey to Pre-registration nursing - course enquires and the Interview and selection process', 'transition through year one – adjusting to being a student', 'transition through year two - managing the academic workload', 'transition through year three - preparing to become a newly qualified nurse', the 'The classroom learning experience and the role of academic staff' and 'engagement with academic assessment and support'.

4.2.1 The journey to Pre-registration nursing - Course enquires and the Interview and selection process

University open days and personal recommendation from previous students were two influential factors for some participants choosing to study on the BSc Pre-registration (Adult) nursing course. Two participants found meeting and discussing the course with current nursing students' useful preparation for the course. Students currently studying on the course

were perceived as reliable sources of information and most likely to convey an open and honest portrayal of their experience of the course:

We were well prepared for the course because I had been to open days and they were a really good indication of what was expected and what to expect. I was talking to students which I found helpful. You were getting their perspective on things not just staff members, because students are going to tell you how it is aren't they? They're not going to sugar-coat it, so that was really helpful. And I still came (SP06)

Some students value help with the process of applying for the course challenging, particularly when it came to establishing the qualifications necessary for the nursing course. For one participant the first point of contact with the university was through the university website, where she struggled with unfamiliar terms and confusing jargon such as 'credits', suggesting the institution's website may not be user-friendly for all potential candidates. Being able to speak directly to Course Enquiries about applying for the nursing course offered some relief and reassurance. Despite the move towards an automated and technology-driven service the excerpt below suggests accessing an individual who could clarify and simplify information was clearly more conducive to meeting the needs of some students:

I didn't have a clue what to do. When I'd gone onto the website they were saying about credits and I had no idea what credits were! I rang the university Course Enquiries and they were really helpful. They told me what I needed to do and told me how to go about it, so I quickly rang my local College and asked to do my Functional Skills Level 2, and everything took off from there (SP05)

Unlike some HEI courses all Pre-registration nursing students must first undergo an interview and pass the selection process before enrolling on the course. At the time of this study the interview and selection process involved a 20-minute numeracy test requiring a 70% pass rate, a 20-minute written test involving a short essay, a group debate based on provocative images or statements provided by the interviewer, and a one-to-one interview with an academic member of staff. Mixed feelings were evident about the interview and selection approach with participants describing the process as '*scary*', '*intense*' and '*nerve wracking*'. Waiting for the results of the interview was just as difficult for one participant:

It was very scary in the interview because it's so nerve-wracking... then it was just a waiting game. That was the hardest thing as it's quite a long waiting game, about eight or nine weeks which seems a really long time (SP05)

The data above suggests the interview and selection process evokes emotions of anxiety and fear, thus 'Course Enquiries' and 'Interview and selection' were identified as the first touchpoint on the student journey map for this study (See Figure 14 - Touchpoints on the nursing students' journey through the course).



Following application, interview and enrolment on the course participants then begin the transition through year-one, year two and year three of the course.

4.2.2 Transition through year one - Adjusting to being a student

Although the student journey through the course spans three years it incorporates several distinct stages of transition including the move from year-one to year two, and year two to year three. The first day of welcome week emerged as a significant emotional experience for all participants, with many students in this study stating they found it '*scary*', '*exciting*', '*busy*', '*hectic*', '*chaotic*' and '*overwhelming*'. During welcome week students have their first opportunity to meet some key stakeholders in the course such as Course leader, Year lead, Module leaders, Home group leaders (also known as Academic advisors) and Practice Education Facilitators (PEF). Welcome week is when first impressions of the nursing course are formed. Some study participants had felt overwhelmed by the large number of students on the course:

The first day was overwhelming...I just didn't expect that many people to be in one year on one course you know (SP07)

The first day it's very scary, exciting but scary at the same time, because it's a huge new experience. It was good after the first 10 minutes of pure fear, wondering if I'd done the right thing giving up a full-time job. That's a big worry for a lot of us (SP05)

During welcome week key information about the course is relayed to all students who are together for the first time as a cohort. For some students there was too much information given during welcome week making it difficult to distinguish essential information from the interesting but non-essential information:

I knew it would be a big week, loads of information, loads of it that is and isn't relevant, but it has to be said, so it's like you have to come because if you didn't come you gonna miss the stuff that you need (SP03)

The transition into higher education was likened to stepping into a different world by some students. Adjusting to this new world was considered one of the most difficult aspects of the course by one participant who felt like more support during this period of transition should have been available, although the form of extra support was not specified:

I felt like there was a lack of support in some areas. Like if I didn't understand something you couldn't really contact somebody, you were a bit out on a limb really. I know its independent learning but especially when coming into university err it's a big step, a different world and I just didn't feel like there was necessary support. The first year was the hardest... the adjustments! (SP01)

The comment above related to feeling '*out on a limb*' suggest some students can feel isolated during the initial transition into higher education. Although students are introduced to their course leader, Academic advisor and module supervisors early in the course some students struggled with remembering useful points of contact and support in this first year. Some of the earliest memories of the course for several students related to it being chaotic and stressful. For one participant who had not been allocated a permanent Home group those feelings were exacerbated. Home groups are field specific small groups of 25 students who share an Academic advisor as their Home group leader for the 3-year duration of the nursing course. The narrative below suggests Home groups can perform the function of facilitating early friendships on the course, with some of these friendships forming useful support networks as the course progresses:

Welcome week was quite chaotic because I didn't have a home tutor group to start off with, I got put with a random group just so I had somewhere to go which didn't end up being my group. You start to make friends and then I got taken away from that group and put in a different group it got me stressed in my first week. But I'm lucky in that my Home group are a good group, we all do look out for each other and go to each other for help (SP06)

The relationship between students' transition into year-one of the nursing course and several significant emotions described by student participants in this study (*fear, scary, overwhelming, exciting*) suggests 'Transition into year one - welcome week' is a touchpoint to be incorporated on the student journey map (See Figure 14 - Touchpoints on the nursing students' journey through the course).



As previously noted, many participants have been surprised by the large class sizes on the nursing course. From the beginning of the Pre-registration nursing course, and throughout year one, all three fields of nursing (Adult, Child and Mental Health) learn together in large group lectures incorporating the whole cohort, and in smaller Home groups with numbers up to 25 students, for group work sessions, MAGS and practical clinical skills sessions. For most

of the welcome week all three cohorts are taught together with numbers between 300 and 500 students in one large lecture theatre depending on the cohort. For some participants the unexpectedly large numbers of students on the nursing course was a shock and appeared to contribute to them feeling overwhelmed:

It's quite overwhelming at first how many people are actually in your year (SP06)

My first degree was very, very different to this degree. What stood out on the course is...it is HUGE! Yeah, the volume of students it's just like Whoa!!! I remember coming in the lecture theatre and I was like "oh my God there are hundreds of us!" (SP03)

The impact of such large student numbers on learning was a concern for SP03 who had learned some nursing skills in September (for example giving injections), but only started the clinical practice placement the following January. Current resources at the university include ten hi-specification clinical skills laboratories representing a realistic clinical environment where students can learn and develop their practical skills. SP03 was part of a large cohort of 460 students separated into 18 field-specific Home groups, scheduled at different points during semester one, two or three to attend clinical skills sessions in one of the skills laboratories. This had resulted in SP03's Home group having their clinical skills sessions four or five weeks earlier than other students in their cohort, and several months before starting their clinical practice placement. Consequently, SP03 worried about carrying out the clinical skills learned at the university on a real patient five months later:

We've too many on the course and that impacts on our learning. We've got 460! I had clinical skills sessions in September, but we don't go on placement until January, you've then not touched a needle since. It seems completely redundant to me to do it so far away from placement. You just forget everything. You might as well have not done it...if there were fewer students you could do those skills more frequently or closer to placement (SP03)

The extract above infers that fewer students on the nursing course could see more timely delivery of these important skills and enable consolidation of those skills if students given the opportunity to practice those skills more frequently. Balancing the academic demands of higher education, clinical practice and personal life was a challenge for some participants, particularly at the beginning of the course. This appeared to be more difficult for some of the mature students in this study including those who had studied a previous degree at higher education level as indicated in the narrative below:

I'm a mature student this time as well which is different, I was a freshman, school kid last time, so being older I feel it (SP03)

All but one participant (04) were mature students at the point of interview for this study. Mature students represent 67% of the total adult field of the nursing course where this study was conducted (Smith, 2015). Typically a mature demographic is most likely to have a family and mortgage to manage whilst in full-time study and may not have been in full-time education for many years. For example, SP02 had not studied for 20 years before starting the pre-registration nursing course and SP05 had been '*30 years out of education*'. Being so long out of education meant some participants had to adapt different approaches to help them deal with the requirements of the course alongside their personal and family commitments such as being organised and well prepared to study:

I've got three children...my eldest son is in year six, so he needs support with his SATs, they (my family) need me as well. I come on public transport, so I use all that time. I've always got books with me that I'm reading and notetaking I use the time well, that's the thing that I've done, I don't waste the time that I've got free (SP02)

For one mature student choosing to study in later life was not perceived as a disadvantage. Instead, it was the additional and sometimes competing responsibilities associated with some mature students, such as having young children, that made it harder for them to study on the nursing course:

I have enjoyed it. It's worth it coming into later life and I think I'm lucky because the kids are grown up and not a worry. Some of the girls I feel so sorry for as they've got young kids and childcare issues and kids being-off school, they feel they are neglecting their kids. It must be so hard to fully concentrate, so hats off to them I think they are amazing (SP05)

Undertaking independent learning was a recurring theme for many student participants in this study, with several participants stating they had struggled with the self-directed study on the course. SP03 recounts frustration at not knowing how to do self-directed learning in her narrative below. Making the transition to a process of independent learning was difficult for some participants. Many student nurses had come from Further Education (F.E.) colleges, or other learning environments, which delivered highly structured courses, containing large amounts of taught contact time. For all higher education degree courses engaging in self-directed learning is a fundamental component of the programme, however the participants below perceive this as a neglected aspect of their experience, where there is little or no support. This appears to be in direct contrast to classroom learning on the course, where students are most likely to receive information, guidance and support:

There is so much self-directed study, but if you don't know how to be doing that, that's a big thing! A lot of people don't know how to learn; I don't know how to learn! It lowers your confidence because I keep struggling with the same things because I don't know how to improve (SP03)

The use of a university supported virtual learning environment (VLE) platform, such as Blackboard Learn, is designed to help students with their independent learning through the provision of on-line learning resources, course/module materials and information, and selfselecting activities for students to work through. All modules on the nursing course have their own Blackboard space where they must provide the minimum baseline content of module handbook, module descriptor, online reading list, lecture notes/slides, assessment submission portals for student feedback, and module leader contact details. However, study participants noted inconsistencies amongst modules with how sites were structured, and materials organised making it difficult for them to use Blackboard. Backing confidence with computer skills alongside the frequent Blackboard upgrades and changes, led to some students feeling confused, frustrated and sometimes overwhelmed:

It's quite difficult that not every lecturer uses it in the same way, and that different things are in different sections depending on which lecturer is using it! If you are not computer savvy it could be quite overwhelming (SP06)

Based on the narratives above the transition into year one of the nursing course was challenging and overwhelming at times. Participants seemed surprised by the amount of independent learning required for the course. Virtual learning environments such as Blackboard, was a point of contact with the University where participants were able to access resources to help with their independent learning. However, inconsistencies were noted by participants affecting their overall experience of using Blackboard. Therefore the 'Virtual learning environment', was identified as a touchpoint along the student journey (See Figure 14 - Touchpoints on the nursing students' journey through the course).



4.2.3 Transition through year two – Managing the academic workload

A fundamental transition point for many students was the move from year one to year two. The familiarity of year one was replaced by a period of flux where expectations tend to have been influenced by other students' experiences of year two. Subsequently, some participants described looking forward to their second year whilst other participants anticipated year two being much harder, or challenging, than year one. As previously noted, some participants had found the large class sizes a little overwhelming in year one. In year two the numbers were much smaller averaging 25 students; this was a welcome change for SP07 who viewed this as an opportunity to form better connections with students studying the same (adult) field of nursing:

Coming into the second year it made me feel quite good having a smaller group that you could get to know rather than trying to get to know masses of people (SP07)

A perceived increased workload and increased academic level of study from level 4 to 5, resulted in some participants feeling overwhelmed during their transition into year two. Being pre-warned about the increased workload by other students and academic staff was a common theme amongst participants. However, for participants in this study this information created unnecessary anxiety and apprehension during the transition process. Four participants (01, 05, 06 and 07) used terms such as '*scared*', '*afraid*' and '*wary*' of the increased academic expectations of year two students:

Everyone had warned me, they've all said the same, fellow nursing students that are in the year above me, everybody has said the second year is the hardest. You get through the first year and you feel like you're on top of the world, and then the second year knocks you back down because it is really hard (SP05)

Anxiety intensified for those students who had failed assessments in year one. Students often retake/resubmit failed assessments alongside subsequent module assessments, resulting in double the academic workload of successful students. The extra workload demands encountered after failing several assessments prompted one participant to considered leaving the course during in year two. SP01 describes being so focused on keeping-apace her studies that she felt encumbered by the extra workload failing previous assessments had resulted in. Reaching a halfway point on the course seemed a significant and potentially defining moment for SP01 and appears to have given her the impetus to continue with her studies as completing the course now appeared achievable:

In second year it was like 'I'm just going to walk away'. It wasn't until half way through the second year I could see the goal post, and that's why I'm still here...because the way I looked at it before was I've failed this, and I failed that and I've got to keep-up with this and do that, but once I got to this stage I've gone 'well I've got me OSCE and one more assignment to do and that's the end of me second year!' (SP01)

From the discussions above participants experienced feelings of both excitement and anxiety during their transition into year two. Anxiety was frequently fuelled by information shared by peers and academics. In addition, students who had struggled academically in year one were most likely to be overwhelmed by the requirements of year two. A sense of achievement created from passing assessments appeared motivating. Thus 'Transition into year two' emerged as a touchpoint on the student journey map (See Figure 14 - Touchpoints on the nursing students' journey through the course), for participants in this study.



4.2.4 Transition through year three - Preparing to become a newly qualified nurse

At the time of their semi-structured interview's only three participants were in year three. SP05 was at the beginning of year three, both SP07 and SP08 were a few weeks from qualifying as registered nurses. Initial discussions suggest transition into year three caused less anxiety than the transition into year two, with most participants in year three describing their transition more positively than that of year two. Although participants acknowledged some modules in year three would be difficult as again there is an increased level of academic study from level 5 to 6, in contrast to their feelings about year two participants were looking forward to the year three:

It's exciting to think that I've done two years and I'm coming to the end because I know it goes really quickly. Everyone says, third year you blink and you miss it. I am looking forward to it, and looking forward to most of the modules, there is going to be some tough modules ... I think I am going to enjoy it (SP05)

As all students studying Pre-registration (Adult) nursing must complete year three before qualifying, and most participants described excitement about this transitional period, thus 'Transition into year three' was identified as a touchpoint on the student journey map (See Figure 14 - Touchpoints on the nursing students' journey through the course).



A fundamental part of students transitioning through year three, and the pinnacle of three years' work, is preparing for registering as a newly qualified nurse. Paradoxically this transition from student to registered nurse can feel double-edged, creating both positive and negative feelings in participants. According to SP05 academic staff had helped students prepare for what to expect as qualified nurses, this included portraying the positives and negatives of a newly registered nurse's role, such as how hard they would have to work. Such candour from academic staff potentially runs the risk of poor attrition, yet SP05 found this approach effective preparation for what was to come:

I feel prepared for qualification; the lecturers have made it clear that it is hard work at the end of it. It's not "you'll be qualified, and you'll get a bunch of keys and life is going to be brilliant". They are realistic and tell you how it is. They don't sugar coat things, it's very much "This is what it's all about" (SP05)

Although some participants described feeling prepared for qualification, others were concerned they might not be ready. They frequently described feeling nervous. Though most people typically experience nerves when starting any new job, the participant's nerves appear to relate directly to their sudden responsibility and accountability as qualified nurses. In particular SP08 felt nervous about being 'included in the numbers' and thus a fundamental part of the clinical team required to function at the level of a registered nurse, rather than in the supernumerary role student nurses had been used to:

It's like a seesaw of feeling nervous and excited at the same time. Being included in the numbers straight away always makes me feel nervous...I'm not 100% prepared for it (SP08)

Although nervous, participants recognised how nerves can be used positively to enhance vigilance and focus, ensuring they check and re-check they are undertaking tasks correctly. Functioning with raised anxiety levels can lead to chronic stress and burnout for nurses in the long-term, even having serious consequences on their health status (Jones et al., 2018). SP07 acknowledge raised anxiety levels but rationalises these fears as both natural and common place amongst newly qualified staff. Nevertheless, she still held deep-rooted angst about her impending future, perceiving a career in nursing as precarious. These fears seemed exacerbated by stories of nurses being '*struck-off*, in other words removed from the NMC nurses register and thus unable to practice as a registered nurse, for making a mistake whilst on duty. The context of these mistakes was akin to the environment and circumstances participants were likely to find themselves working in as qualified nurses:

You hear all the stories about people getting struck-off...you see someone that's given 22 years and then they have one bad night-shift where they didn't take blood pressures, they were short staff, they were tired, they'd worked too many hours, and they've been struck-off...your careers over! Being on a busy ward where you're trying to do your best, and you're asking for help and you're not getting it, that does scare me! (SP07)

Participants apprehension about their future as qualified nurses related to other people's expectations of them; their personal responsibility and accountability for patient care; lack of support from the organisation in relation to poor staffing levels and long working hours; and finding themselves in circumstances where their actions and decisions may be career ending, even though these circumstances may be beyond their control. SP07 appears to have experienced these concerns throughout the course:

I love nursing and I just think it's a brilliant career, but it does worry me just about the future and where it's going to go and whatever happens if that happens to me you know if I made a mistake, and then that's it...everything's gone! Having that constant worry throughout the course (SP07)

Student's anxieties about the transition from nursing student to qualified registered nurse is a concept known as 'transition shock' (Duchscher & Windey, 2018) and relates to the process of transitioning from a familiar comfortable role and adjusting to the realities of a new professional position. For some students' transition shock starts just after qualifying and commencing their new roles in clinical practice, however SP07 fears around qualifying started much earlier on the course. SP07 had found the prospect of being a qualified nurse so scary she had contemplated quitting the course with only a couple of months left on the programme to complete:

It's been a struggle for sure. There's been a few times where I've thought of quitting that's just honest. I still don't know if I want to be a nurse in terms of I love doing what I do but it's so scary (SP07)

SP07 reveals fear about making mistakes that inadvertently causes patients harm and puts her in danger of 'getting into trouble'. This fear may expose self-doubt about participants perceived competence, and ability to cope with the responsibilities. However, SP07 narrative poses several interesting questions such as 'what about my safety?', 'how can I protect myself?', 'how can I ensure I don't get into trouble?', suggesting these fears may derive from a perceived lack of wider protection or support in clinical practice, consequently SP07 feels she needs to find ways to protect herself but does not yet know how. Furthermore, the language SP07 and 08 use (worry, scary, safety, protect myself, get myself into trouble, doing the right thing, horror stories, put under so much pressure, really tough) suggests they perceive nursing as a dangerous job, with an element of risk attached to it:

I feel like I'm just going into the unknown, but when it actually comes to me, my safety as a nurse, I suppose that's where I'm a little bit worried. As in how I protect myself as a nurse and ensure that I'm not going to get myself into trouble because I'm doing the right thing, I'm following the right course (of action) (SP 07)

It's like your safety nets completely gone! (SP08)

Participants who anticipate receiving good support in clinical practice when they first qualify seemed more optimistic about their future. Having clear support available to newly qualified staff was perceived as making the transition process smoother for student nurses:

If I do manage to get on the ward that I want to then at least I know I'll be alright because they are really good with the newly qualified and they still go through everything for weeks, so they are up to scratch with everything. That gives me a little more confidence (SP05)

Although participants in the sections above are discussing their anxieties around being a registered adult nurse and functioning in the role as a qualified member of staff, this study is focused on the whole Pre-registration nursing experience beginning with the journey to Pre-registration nursing to the point of qualification. Therefore 'Qualifying' was considered as the final touchpoint on the Pre-registration nursing students' journey (See Figure 14 - Touchpoints on the nursing students' journey through the course).



4.2.5 The classroom learning experience and the role of academic staff

The NMC standards require students on the Pre-registration nursing programme to complete a total of 4,600 hours of learning across three years, with 2300 of those hours (50%) completed in clinical practice placements and 2300 hours (50%) in theory. Much of that theory time incorporates facilitated learning in a classroom environment, whether those sessions are delivered as lectures, group work sessions, seminars, Home group meetings or clinical skills sessions. Findings in this study suggest participants valued the development of practical skills which helped them feel better prepared for the clinical practice placement environment. Several participants anticipated much more time spent learning practical skills in the clinical skills laboratories at the university with one participant inferring an element of prevarication by the university about the amount of clinical skills teaching included on the programme when advertising the course to potential students. SP05 believes the reason there are fewer than expected taught clinical skills sessions is because of lack of resources, specifically available skills laboratories. This has implications for future Pre-registration nurses who are required to pay course fees like other HEI students (as indicated in the background chapter of this study) and who may feel their experience should not be influenced by resourcing issues:

The skills labs get boasted about and promoted, and it's made out like we'll spend our lives in them, but we've had three maybe four sessions in them, so few! (SP03)

I think more Clinical Skills (but) I know it's not really do-able because you're limited to the amount of live skills labs (SP05)

Worry about making mistakes in clinical practice was a recurring pattern amongst participants. Unsurprisingly participants were most concerned about mistakes associated with greater severity of consequence. Therefore, to make an error in clinical practice can potentially have far reaching outcomes for patients, colleagues and themselves. This could explain why participants valued clinical skills sessions as they offer the security of a safety net, where students could learn and practice their nursing skills without repercussions. Having a 'dry run' in a classroom setting helped participants gain confidence, particularly when given constructive formative feedback on their performance. If students made mistakes they had the opportunity to learn from those experiences, without risk to patients, colleagues or themselves:

It's nice to have the hands-on experience before you go into placement, and to have like lecturers being able to say to you "this is what you're doing wrong, but you don't have to worry because..." (SP04)

Three participants within this study had no previous nursing care experience but noted a presumption by some academic staff delivering taught sessions that all students have some prior knowledge, or experience, of nursing care before starting the course. This was most notable in practical sessions early in the course, where students were most likely to encounter complicated medical terms, jargon, unfamiliar equipment and complex procedures for the first time. Although students are encouraged to query unfamiliar terminology and techniques during clinical skills sessions, to do this repeatedly can make students feel their knowledge is not on a par with their peers, potentially affecting their confidence. One participant's inexperience made her 'feel simple':

When it came to things like the skills sessions, I felt there was definitely an expectation that people had worked in care before, and the terminology that lecturers were using was as if we'd already worked in care, a lot of the time I really didn't have a clue, and it made feel a bit like...well it made me feel a bit simple sometimes because I hadn't had that experience that some people had had (SP06)

Several participants (04, 05, 06 and 07) reported other student's behaviour in the classroom environment, particularly incivility, having an impact on their experience of the course. Incivility is any form of disruptive, inappropriate or irresponsible behaviour, attitude or speech by an individual that affects other people around them. Participants in this study identified specific behaviour by classmates that had a direct effect on their learning such as chatting in lectures, using mobile phones during class, packing-up belongings before the end of taught sessions and being noisy and disruptive. Several participants used the word 'rude' and 'annoying' to describe how they found this type of behaviour and how it made them feel. Disturbing other students was perceived as inconsiderate and disrespectful by some participants:

The talking that annoys me...I just can't see why you come into nursing and you just sit at the back and chat, or you're on your phone. I can't understand why people want to be nurses if they're just going to sit there and chat! It literally does my head in (SP04)

Err (student sighs), I've enjoyed lectures, I haven't enjoyed people talking in lectures at the back! Other students yeah, I've found it really rude at times (SP06)

Strategies employed to tackle incivility varied amongst lecturers. Some lecturers responded immediately by asking noisy students to quiet down or "shush". Although SP05 finds this an effective course of action, this approach has the potential to disrupt the flow of a lecture, or discussion. SP08 claims several lecturers move around the classroom then single out specific students to answer their question. This approach tended to cause fear among some students but at the same time kept them alert to the possibility they could be targeted next. Other lecturers avoided challenging incivility as it occurred, appearing instead to ignore bad behaviour:

I think the stricter lecturers are probably seen as better lecturers because they're stricter, and people have that respect not to talk and to get involved, and not get their phones out because they know they'd be pulled on it as opposed to a lecturer that would just stand there and talk and not bother about what anyone's doing (SP07)

Some lecturers walk up and down the lecture theatre and always scare you because you think they're going to pinpoint you (SP08)

Several participants identify those who take a strict approach to tackling incivility as better more respected lecturers, most likely to ensure a good learning experience. When students perceive there will be consequences for bad behaviour this behaviour improves. When lecturers failed to tackle incivility, they were seen as losing control over the classroom setting leading to feelings of frustration by other students who then take matters into their own hands:

The talking! I've done it quite a few times where I've put my hand up and said you know "I can't hear" and they just still chat (SP04)

How lecturers approach classroom management has a direct effect on students learning, suggesting academic staff have a role to play in student's experiences of the course. From the data above, it is evident facilitated learning sessions where incivility was not managed effectively left participants feeling frustrated and irritated. Therefore, 'Facilitated learning sessions' are identified as a touchpoint on the student journey through the course.



Although numerous stakeholders performing various roles contribute to the overall student experience, student's principal point of direct contact with the university is through their interactions with academic staff such as Lecturers, Module leaders, Course leaders etc. Participants within this study expressed mostly positive experiences of their interactions with academic staff and felt well supported by lecturers they described as approachable, helpful and willing to invest time in students. How lecturers facilitate learning affected the quality of teaching for three participants (03, 05 and 07). Some lecturers were particularly engaging when they were willing to share their own clinical practice experiences:

The tutors were really professional, just really enthusiastic about the topic the subject. The love of nursing and I think the quality that was the thing that stood out for me, the care of quality. Even though you're not seeing the tutors in clinical practice you can tell what kind of care they have come from, you know. It doesn't feel like it's being taught because that's the syllabus, it feels like it's being taught because they've lived it and they understand it, and that's what comes across (SP02)

Most lecturers do try to put things from their own practice in – rather than just reading off PowerPoint slides – they are putting a bit of real life into it from their own experience which is really good (SP05)

Some lecturers struggled to motivate and engage students in a stimulating way. Sessions were lecturers read straight from PowerPoint were perceived as a waste of time. As previously identified many participants within this study were mature students with family, work and financial responsibilities outside of the course. Therefore, participants had greater expectations of those taught sessions where they had made the effort to attend classes on-campus:

Sometimes lectures do go on and on, and the lecturer just sticks to the PowerPoint, that's why some people just scan in and leave. It's on the Facebook page...because they think the lecture's so boring they just scan in and go...all they do is read off the PowerPoint. We could just do that at home and read the PowerPoint ourselves (SP04)

Lecturers who delivered sessions with passion and enthusiasm inspired and motivated student participants. On the other hand lack of consistency amongst academics' delivery of sessions and management of incivility can influence student's experience of learning and potentially affect their engagement with academic assessments which will be discussed in the section below. Consequently, contact with 'Academic staff' was identified as a touchpoint that had the potential to shape a student's experience of their journey through the course.


4.2.6 Engagement with academic assessment, support and feedback

The pre-registration nursing course utilises two types of assessments: formative and summative, in both clinical practice and theory. The format for summative theoretical assessments varies across modules with students undertaking written assignments, written and Multiple-Choice Question exams (MCQs) and practical exams such as an Objective Structured Clinical Examination (OSCE). Students with an expectation of more practical learning were surprised by the amount of academic study required on the nursing programme. This resulted in some students struggling with the theoretical demands of the course:

I knew nursing had changed to be more academic but I don't think I realised how, like with the essay writing and things like that, I was still under the illusion that it was more like a vocational course. And I have got that quite wrong (student laughs) in the sense of essay writing and things like that (SP06)

I think year two was quite a challenge for me as I found it was a lot of academic work and nearly all assignments (SP05)

Whilst participants did not indicate any dissatisfaction with the module assessments themselves, participants did indicate they found the assessments on the nursing course more challenging than they had expected. As such 'Module assessments' was identified as a touchpoint on the student journey through the nursing course.



For participants in this study feedback on assessments caused more dissatisfaction than the assessment itself, with most experiencing variable and inconsistent feedback. These inconsistencies were noted when students discussed and compared their feedback with each other. To adopt and act on feedback students must be able to understand academics comments. However, participants who struggled to make sense of the comments reported seeking guidance from their peers rather than an academic about their feedback. This may have exposed the participant to misinterpretation and misinformation from peers at a similar level of learning as themselves:

When we got our essay back, people were saying "well I got this feedback what does it mean?" or "I got that". Lots of people from my understanding of the comments had not been given marks for referencing, and some tutors were saying you haven't referenced in your conclusion, but you don't reference in a conclusion so can you downgrade someone? (SP03)

Although the school of nursing employs a standard approach to providing feedback on written assignments across all Pre-registration modules, participants indicated the quality and quantity of feedback varied between modules and tutors. In the excerpt below one participant believed she was performing at a satisfactory level after receiving a pass grade and a few lines of generic comments on a first assignment. Consequently, she was shocked when more detailed feedback on a subsequent assessment suggested she was not performing as well as the feedback from her first assignment had indicated:

On my first assignment I didn't really get much feedback, minimal like a sentence or two, so I thought oh well I'm doing alright here. Then when I got my second one back there was SO much feedback on it, it was a shock. It literally felt like 'how have I even managed to write the title?' It was that bad! I was like how have I got this so wrong after getting my first one back and it supposedly being alright? (SP06)

The quality and quantity of feedback for the first assessment appears to have allowed scope for SP06 to misconstrue the feedback as more positive than intended. Poor or ambiguous feedback can lead to students advancing through the course with an unrealistic perception of their academic ability, potentially setting them up to fail as the level of study increases. Assignment feedback is linked to the module learning outcomes and the marking criteria, alongside at least three annotated feedback comments per page on the assignment/script. University guidance suggests critical feedback on the script should be written sensitively and constructively and should include 'feed forward' which specifically identifies the key strengths of the work and how work can be improved. Rather than receiving sensitive and constructive feedback SP06 was shocked and upset by the comments from her assessment resulting in self-doubt about her academic ability and place on the course:

It made me question if I should still be doing the course if my academic writing was that bad that I was getting everything wrong from grammar to structure to referencing. All these different things wrong? The feedback made me question whether it was the right course for me and if I should be doing it. It was 'this is bad, that's bad, this isn't right', which on my first one I'd had no comments about! (Long pause – voice rocky sounds upset), it made me cry, it made me cry (SP06)

From the data above participants have described positive and negative experiences of assessment feedback. The emotions associated with negative feedback triggered feelings of self-doubt and physical upset for some participants who even began to question their place on the course. Thus 'Assessment feedback' was identified as an emotional touchpoint where subjective experience was potentially influenced on the student journey through the course.



Variance amongst academics when advising students about assessments was raised as a concern by two participants (01 and 04). Currently students receive academic support with module assessments through a system of MAGS sessions, rather than individual one-to-one supervision. All Pre-registration modules schedule a minimum of three MAGS within their timetables prior to the submission date of all assessments. These first three MAGS are delivered to each individual Home group, thus up to 25 students will attend each session. Due to workload commitments and staff availability MAGS sessions are not consistently delivered by the same lecturer to the same Home group. SP04 feels this can result in inconsistent and sometimes contradictory information being given in MAGS sessions:

Lots of lecturers are telling us different things. Like one lecturer telling us one thing and we go into another session and then someone else is telling us another thing, so I think for consistency they need to be on the same page (as each other) (SP04)

SP03 objected to the introduction of MAGS which had replaced individual supervision part way through the course. Previously she had been able to access one-to-one support from an allocated academic, or module supervisor, on up to three occasions during a module. The module supervisor would comment, or feedback, on up to 50% of a student's draft assignment. The new MAGS system promotes peer support and feedback on only 400 words (or equivalent) of a student's assignment, which can be less than 20% of a typical 2500-word essay. Some participants were unhappy about this change from individual supervision to a MAGS approach, and those who struggled with essay writing felt disadvantaged by this new approach, preferring the one-to-one support and guidance the previous supervision approach had offered them. SP03 expressed strong negative emotions relating to MAGS:

Group supervision I hate with a passion! I don't see how it was justified, I feel it really disadvantages our learning...I really struggle writing essays, and I feel right now like I've got no support (SP03)

Participants who were not struggling with their assessments also felt disadvantaged by the new MAGS approach. SP08 experience suggests lecturers facilitating the MAGS sessions tended to spend much of their time with those struggling students such as SP03, leaving little or no time to address the questions of the more capable students. This can result in some students feeling their needs were neglected:

Group supervision I don't think they work. You'll get one group that's really struggling so the lecturer will tend to see them for the entire session, but you come away feeling you've learned nothing, and you've got more questions unanswered (SP08)

Only one student described having a positive experience of MAGS. SP06 had particularly enjoyed the peer support element of the sessions where students were able to discuss the assessments with each other and share ideas. Listening to other people express their perspectives can broaden students thinking about their assignments, as they reflect on aspects not necessarily considered before the MAGS session:

Group supervision overall I've had a really positive experience, I've enjoyed them. I find it supportive because you're bouncing things off each other, and people think of things you might not have necessarily thought of yourself. I think they're a good idea (SP06)

From the discussion above MAGS sessions left some participants feeling disadvantaged and struggling, whilst one participant enjoyed the sessions, therefore 'Module Assessment Group Support (MAGS)' was identified as a touchpoint on the course (See Figure 4- Touchpoints on the nursing students' journey through the course).



4.3 The clinical practice learning experience



Figure 16 - Core theme 2 'The clinical practice learning experience'

The clinical practice placement experience emerged as a core theme from participants' narratives within this study, with all participants identifying clinical practice placements as the high point of their whole learning experience. Subthemes that influenced participants' experiences in the clinical environment include 'the workplace as a learning environment' and 'accessing support in the clinical setting: the role of mentors and Practice Education

Facilitators'. Clinical practice placement constitutes 50% of the Pre-registration nursing programme with students completing six clinical practice placements in a health care setting across the three-year programme. Clinical practice placements can last from 8 to 12 weeks duration. Clinical practice was perceived as having a more important influence on the student learning journey by several participants who felt more learning took place in the clinical practice environment than in an academic setting. The theoretical classroom experience appeared to have less impact on learning when compared with the clinical practice placement experience.

In the first couple of weeks I learned as much on my placement as what I'd done those first few months at Uni! (SP06)

I think you tend to learn a lot more doing it than being told how to do it (SP08)

For several participants parallels were drawn between the clinical practice placement experience and what they will eventually do as registered nurses. Therefore, clinical practice placements replicates the 'reality' of nursing more than their university academic experiences. As such many student participants appeared to place greater value on the clinical practice placement experience than on their theoretical experiences on the course:

My highs have got to be on clinical placement because that's what you want to be, that's what you want to do (SP01)

The placements have all been massive highs for me, it just kind of makes you realise why you're doing it (the course) ... as much as I've enjoyed theory I just can't wait to get back to clinical work (SP07)

4.3.1 The workplace as a learning environment

Three participants (01, 06 and 07) had no previous care experience prior to starting the course, unlike the remaining participants who had previous care experience. However, former care experience left some participants no less apprehensive than students with no prior experience when transitioning from the academic environment to the clinical practice placement environment, particularly for their first clinical practice placement. This was evident from SP08's narrative, who despite having previous carer experience before commencing the nursing course, describes how daunting the extra responsibility is as a student nurse:

At first it was really daunting going from a carer to student nurse because it's a big step up really, getting a lot more responsibility (SP08)

For five participants (01, 02, 04, 05 and 06), including two who had no previous clinical practice experience, their first clinical practice placement experience was positively described as 'good', 'brilliant', 'fab' and 'amazing' and even 'gold standard learning'. During placement one SP01 had worked in a supernumerary capacity and was therefore not considered part of

the daily workforce, enabling her to work more closely with her supervising mentor. With no previous nursing experience, SP01 appeared to value the close supervision, benefitting from the guidance and direction the supernumerary role offered:

First placement was fab for me because I didn't come from a nursing background. For me that was gold standard learning...they say is your gonna be supernumerary, the person that's going to be looking after you is supernumerary (SP01)

For SP05 the first clinical practice placement experience was the best of the two placements undertaken at the point of interview. In contrast to SP01, SP05 worked in a non-supernumerary capacity from just two weeks into the first clinical practice placement, with the ward staff expecting her to work at the level of a more experienced student nurse (e.g. second or third year). Consequently, being encouraged to get involved in all aspects of patient care so early on in the course resulted in SP05's knowledge and skills developing exponentially. Although SP05 and SP01 had contrasting experiences both students had developed more confidence and felt better prepared for subsequent placements. Notably, SP01 who lacked previous care experience, had valued the close supervision she had experienced in a nurturing environment, whereas SP05's previous care experience may have resulted in her welcoming the challenge of a potentially more difficult and demanding first clinical practice placement:

First placement was really amazing and such a good experience. They made sure I was in the thick of things and after the first couple of weeks when they realised that I knew what I was doing. By the end of placement, I felt very confident, and very much like I could deal with an awful lot more...my mentor was a ward Sister, she was very busy but made sure I had two associate mentors as well if I wasn't with her (SP05)

Two participants (03 and 07) described feeling unprepared for their first clinical practice placement. Both participants were shocked by the busyness, long shift patterns and travel requirements when undertaking their first clinical practice placement. SP07 describes a similar experience to SP05 where she appears to have been working in a non-supernumerary capacity, involved in all aspects of patient care without close support or supervision. Frequently being left to manage poorly patients alone, with little help from qualified staff despite having no previous care experience had not fostered the same level of confidence in SP07 as it had in SP05. Instead SP07 felt being given '*too much responsibility*' was '*scary*', fostering a sense of helplessness and abandonment. This suggests a supernumerary status should be employed for those students with no prior nursing care experience during their first clinical practice placement:

My first placement was quite a shock for me going into a general medical ward that was extremely busy, that was quite scary. I was doing long shifts and commuting. Quite a few times I was left with bays with staff nurses that weren't helping, I was left on my own with these quite

poorly patients, and I did everything that I could at that point! Had I had that in my fourth, fifth or sixth placement I probably would have dealt with that better (SP07)

Within this study clinical practice placement gave participants an opportunity to consolidate theoretical knowledge gained in a university classroom setting with 'real world' clinical practice experience. Encountering, and actively participating in, a variety of diverse learning experiences in the clinical environment contributed to effective learning and development of clinical competence for several students. However, one participant reported feeling confused and unclear about the learning opportunities and practical clinical skills student nurses were permitted to engage in. Notably, SP03 was worried and hesitant about undertaking some nursing skills and practices in the clinical environment that had not been taught and assessed at the university. Unable to ascertain definite responses to her queries about approved nursing skills, SP03 felt she had missed-out on learning opportunities that her peers were engaging in:

I missed a lot of learning opportunities. You'd see people on Facebook saying "I did this today or I did that today" and I'm thinking I don't know if I can do those things, and I don't know if you're (they're) supposed to be doing those things. In my head it was "did the NMC say we can or can't do this? Do you (university) say we can or can't do this? Do the hospital trust say we can or can't do this?" No one would tell me... I asked my PEF she bounced me to Uni, or I'd ask the ward, I just took peoples answers like you know (SP03)

One implication of missing out on specific learning opportunities is that students may perceive themselves as less proficient than their peers in some of their nursing practice. SP03 had previously identified the large volume of students on the course as impacting her learning, particularly in relation to the clinical skills sessions. Now that she had the opportunity to develop those clinical skills in the practice environment, lack of clarity and guidance appeared to impact that learning further. A further issue that appeared to impact learning in clinical practice related to student's supernumerary status. The narrative below indicates some students benefit from working in a more non-supernumerary status the further into the nursing course they were. For several students a non-supernumerary status had fostered the development of deeper knowledge and understanding of their professional practice and skills, as they had more responsibility and active involvement in direct patient care. However, SP01 felt a non-supernumerary approached would not work as well for first year students:

I was allowed to do more, I was allowed free reign, instead of my mentor going do this, do that, I was able to do more things like discussing things with Drs, so you got a better understanding of an holistic approach to the patient, but I don't think it worked for first years, they felt a bit overwhelmed (SP01)

This observation about first year students having too much responsibility followed a similar pattern to SP07 experience of her first clinical practice placement where she had been

working in a non-supernumerary capacity, involved in all aspects of patient care, often left alone and unsupervised without close support or direction. For SP07 this experience had influenced her opinion about working in a ward environment negatively as she had felt it had been too much responsibility for her:

I was thinking this was too much responsibility for me as a student really...I just felt the pressure...and it did kind of put me off ward nursing (SP07)

For all participants in this study their first clinical practice placement, and for some students' subsequent clinical placements, appeared to be emotional and often daunting times for students. Therefore, 'First clinical practice placement / subsequent clinical placements' is identified as a touchpoint on the student journey through the course (See Figure 14 - Touchpoints on the nursing students' journey through the course).



Clearly the risks of overwhelming students with too much responsibility has to be balanced with the benefits of exposing students to relevant clinical practice placement experiences early enough in their nurse training to develop their professional knowledge, understanding and confidence in practice. Mentors in the clinical environment may have a significant role to play in determining the types of clinical practice placement experiences and the level of responsibility individual students are exposed to whilst under their guidance and supervision.

4.3.2 Accessing support in the clinical setting: the role of mentors and Practice Education Facilitators

Mentors appeared to play a pivotal role in the student experience with six of the eight participants discussing their experiences with different mentors. A mentor is an NMC registered nurse, midwife, nursing associate, or other registered health and social care professional who is responsible for supporting learning and development and assessing individually allocated students in the clinical environment (NMC, 2018). Participants described their experiences with mentors as mostly positive, using terms such as brilliant, lovely, nice, good, amazing and fantastic. SP07 identified a good mentor as one who made time available to teach students about various patient conditions and management of those conditions in the clinical environment. Many clinical areas are known to be often busy and

short staffed, mentors who prioritise their students learning needs in a busy environment appear to make their students feel valued and important:

My mentor was fab, she was absolutely brilliant, because I've learned tons...she'd go through conditions with me and taught me a lot (SP07)

Unlike the experience of SP07 who had described her mentor using a didactic approach to teaching, three participants (02, 06 and 08) identified their mentors as having a less instructive and more 'challenging' approach to facilitating their learning. Their mentors had prompted participants to actively research and explore new concepts for themselves, encouraging the students to become more accountable for developing their knowledge and understanding. SP06 describes her mentor challenging her to research conditions she came across in her own time. This had resulted in her feeling more motivated and interested in engaging in independent learning:

My mentor was fantastic, she was always challenging me, not in a nasty way, it made me want to go and learn. So I would spend a couple of hours in-between dinner and just go and logon to a computer and research things. I would go home and do the research because I was really genuinely interested in it (SP06)

Comments from a mentor challenged SP02 to stand out from the other students, changing her perspective towards learning, resulting in a desire not just to learn but to have '*an experience of learning*', suggesting the process of learning is as important to her as the information learned:

My mentor said, "you'll all qualify in the end, just what have you done differently?" and that stuck with me. I have changed my perspective...I want it to be an experience of learning rather than just being told what to do (SP02)

SP08 described a similar experience to SP06 where the mentor had expected the student to engage with independent learning and research before participating in certain nursing practices. However, SP08 had only become cognizant of the depth of learning that had occurred following a moment of reflection. This suggests students do not always recognise learning taking place until they are given opportunity to reflect on their newly acquired knowledge and skills:

I had one mentor in theatres and he wanted me to do quite a lot of research before he'd let me scrub-up. At first I'd feel like I was not really learning from it, but when you take a step back you'd realise what you had actually learned from it (SP08)

One participant compared an experience of a mentor who 'challenged' her learning with a far 'laxer' mentor on her next clinical practice placement. Although SP06 described herself as less engaged and less interested in learning due to her mentor's more casual approach to supporting the student with her learning needs, she believed learning still occurred. This supports the assumption that a variety of different teaching styles utilised in clinical practice by mentors is likely to result in some learning taking place, however the approach adopted by individual mentors appears to influence some students' level of engagement with, and experience of, the learning process:

My mentor was really nice, really lovely, she was very lax though a completely different way of working compared to my other mentor. I still learned but I felt that that interest and that drive wasn't there as much. I just wasn't as engaged as I was on the other one (SP06)

Evidently SP06 felt a lack of interest and motivation to engage in independent learning when contrasted with her previous clinical learning experience. Furthermore, the mentors lack of interest in meeting the students learning appears to have resulted in SP06 being utilised as an HCA rather than being socialised into her future professional role. HCAs perform many of the fundamental patient care without the responsibility and accountability of registered nurses. Paradoxically, SP06 who had no previous care experience before starting the nursing course found herself enjoying the extra direct patient contact time she was afforded working as an HCA. This left her questioning her desire to become a qualified nurse:

I started panicking because I was like well I'd quite happily just work as an HCA in hospitals because you get to talk to patients more and things like that, and then it made me start questioning do I actually want to do nursing, or should I just do HCA-ing? (SP06)

SP08 also described being engage in a more HCA oriented role for the duration of one placement when being mentored by a senior member of staff. Senior staff nurses and Junior ward sisters frequently assume the role of clinical area manager with responsibility for large numbers of patients and nursing staff. With so much responsibility senior members of staff can find themselves too busy to mentor students alone and regularly rely on the help of associate mentors or other qualified nursing staff. Consequently, SP08 reports being '*pushed from person to person*' with no '*stable (mentor) figure*'. Instead of fostering a sense of belonging the comments below imply the inconsistent support strategy engendered a sense of distance between the student and the community of staff:

I've enjoyed most of my placements overall apart from one where I kind of just felt I was more of a health care assistant for 10 weeks rather than a student nurse...My mentor was a band six, so I got limited time with her, so I was always being pushed from person to person, so I never really had a stable figure to mentor me really (SP08)

SP05 similarly describes working more with associate mentors than with her allocated mentor who as a Ward sister encountered conflict between the competing demands of mentoring and her management responsibilities. In contrast to SP08, SP05 had found this a positive experience, describing her associate mentors as brilliant, suggesting relinquishment of responsibility by her allocated mentor had not impinged her learning experience as appeared to be the case with SP08. Evidently being supervised by an individual who has the capacity to devout time to a student's learning needs supersedes the opportunity to work alongside more experienced managers governed by time-pressures:

It was a difficult one because it was a Sister - and I think Sisters and Ward Coordinators – I'm not sure they should be mentors because they are so busy trying to keep everything in charge. She made sure I had two associate mentors so if I wasn't with her I was with my associate mentors and all the nurses on that unit were brilliant (SP05)

As so many participants acknowledged mentors as pivotal to their learning experience in practice, 'Mentor support' was identified as a touchpoint along the student journey through the course (See Figure 14 - Touchpoints on the nursing students' journey through the course).



Only one participant (03) mentioned PEF's during their interview. PEFs are registered nurses employed to support mentors and student nurses with education in the clinical practice placement setting in England and Scotland (NHS Education for Scotland, 2013). Since their initial introduction in 2003 the PEF role has expanded to incorporate several nationally agreed core functions that include supporting mentors effectively supervise and assess learners in practice and ensuring effective learner experiences. Before going onto placement all student nurses are advised to contact their resident PEF for support if they have any issues or concerns whilst in the clinical practice placement area. SP03 had spoken to her PEF over concerns she had with some clinical nursing practice. She had specifically sought advice about which clinical skills she could perform as a first-year student nurse. On this occasion the PEF was unable to address her query. Instead of receiving an answer to her query SP03 describes feeling '*fobbed-off*' suggesting the PEF may not have had the answer but tried to appease the student anyway. This response may have influenced the student's perception of the PEFs credibility and usefulness:

Any problems with the student nurses we were just told to go to the link nurse or PEF, if you have any questions ask the PEF... she (PEF) just kind of fobbed me off (SP03)

Although PEFs were only mentioned by one study participant 'Practice Education facilitator support' has still been included as a touchpoint within the course as they perform an important

role in the education of student nurses throughout the three years directly and behind the scenes.



4.4 Emergent networks of support



Figure 17 - Core theme 3 'Emergent networks of support'

Participants in this study identified several networks they accessed for support during their time on the course, including peers within their Home group, their Academic advisor and the universities social media sites. Some participants also described seeking administrative support at various points during the course. These networks played a role in providing academic, practical and emotional support for participants. Hence these networks emerged as a core theme labelled 'emergent networks of support'. Three subthemes will be presented in the following section, they are 'seeking support from peers (Home groups)', 'expectations of the Academic advisor' and 'the influence of social media'.

4.4.1 Seeking support from peers (Home groups)

Although the whole cohort is comprised of student peers, only one participant described turning to a peer outside of their Home group for help and advice, thus within this section the term peers and Home group is used interchangeably referring to only those students who make up a participants Home group. Once enrolled on the course all students are allocated to a field specific Home group of up to 25 students, with an Academic advisor, known in some universities as a Personal tutor, who supports them academically and pastorally for the duration of their three-year study. This Home group system is designed to foster a peer

support network for students, and a sense of belonging early in the course. This is especially important as Pre-registration Adult nursing courses tend to recruit particularly high numbers (between 300 and 500 students per intake at the time of this study). For some participants this large volume of students within each cohort had the potential to impede them forming social connections, or friendship groups, whilst on the course:

There's 500 of us...it's not the best course for helping you make friends (SP03)

Home groups appeared to resolve this potential predicament by positioning students in an environment in which they were more likely to feel comfortable chatting to others, thus facilitating early friendships and initial connections. This is evident from SP06 narrative where she describes only making friends with peers within her Home group, and finding it more difficult to make friends within the larger cohort, suggesting Home groups are a useful vehicle for developing friendship networks for students studying as part of a large group:

I don't think I've really made any friends outside of my Home group. I mean it's difficult to just go up to somebody and start chatting away. I guess that's what's nice about your Home group, because it narrows the field down (SP06)

A benefit of formulating these early friendships in the smaller Home groups is that it appeared to have helped participants 'settle-in' easier in the first few weeks of the course. For example, SP05 describes building herself up to go into class on the first day but meeting her Home group and Academic advisor early on made everything seem easier, suggesting she had initially prepared herself for things to be harder. She acknowledges how meeting both her Home group and Academic advisor had been a '*nice experience*' that had not just made that first day, or first week better but had made 'everything' easier during that transition to onto the course. Getting to know peers to share this new experience with, helped participants more quickly acclimatise to the university environment, and fostered a sense of belonging:

That was a nice experience (meeting her Home group). After building up to going in (to university) we met our Home group in the first week, and our Academic advisor, that made everything seem a little bit easier (SP05)

When you got into small lecture rooms it wasn't crowded, it was nice, and you introduce yourself and it was nice to get to know everyone at first. You got to sit next to someone and just talk to people rather than just going straight to lectures. We had that time to settle ourselves in (SP04)

As well as Home groups facilitating the development of friendships, participants describe turning to peers within their Home groups for help and advice at various points across the course. As is evident from later extracts from participants' narratives this help was in the form of academic support through resource sharing, practical help e.g. through car sharing, and emotional support during times of struggle on the course. SP06 suggested this help was their way of '*looking out for each other*'. Several participants indicate these relationships are symbiotic and reciprocal in nature, with individuals at times seeking help from their peers whilst at other times being the one that helped their peers:

So this is my little group the Uni girls, we're the goose squad, because Geese stick together and they have one at the front and when that one gets tired they go to the back so that they all take the weight over the course of the journey, so we kind of had a bit of a joke about it and call each other the Goose squad...and they've just been brilliant. I couldn't have done it without them, definitely not, and they'd say the same (SP07)

Some Home groups appear to support each other academically by forming collaborations where they share information and materials and peer review each other. Peer observation and feedback is encouraged within the nursing curriculum, for instance as part of a MAGS session, although this is usually under the direction and lead of an academic. Yet peers within SP07's Home group chose to complete a mock OSCE exam '*in front of each other*', without an academic present, potentially opening themselves up to criticism or exposure to poor and inaccurate advice and information. However, SP07's experience suggests this form of peer support had been helpful and illustrates the teamwork and collaboration:

We do send each other reams of information, and if we ever get articles that we know are going to be helpful for whatever we're doing at the time we'll send it to each other. You know there's no competition there. We did like a fake OSCE, we all booked a room together, and we went in and we did it in front of each other, and we had the questions, a different one for all of us, so we couldn't get the same answers (SP07)

Although some participants were able to form bonds with peers in their Home group SP03 felt that insufficient time spent with her Home group impeded her ability to truly form bonds with her peers. Although students undertake clinical skills and small group sessions in the Home groups for SP03 that time was spent in discussion, receiving information and practising nursing skills. This left little time for social engagement, such as getting to know each other and forming bonds. This may have resulted in SP03 missing out on the support from peers' other participants in this study had experienced:

Although we have our little sessions together, we don't spend enough time in the Home groups for us to bond as a group. Like when you're in school you've got your group you spend every morning and every afternoon in registration together, so you bond quite well...but Home group sessions we've had three of them all year, that's not enough to actually spend time together talking to each other and stuff, I think that's something I'd like more of (SP03)

From the findings presented above academic and emotional support from their student peers appeared to be of great importance to many of the student participants, and Home groups play an important role in helping students form those peer support networks. Therefore, 'Home group contact / meetings' was identified as a touchpoint along the student journey through the nursing course.



4.4.2 Expectations of the Academic advisor

Although all participants valued the support and friendship of peers within their Home groups, not all students had felt supported by their Academic advisor. The role of the Academic advisor is to provide academic, professional and pastoral support throughout their allocated student's three-year course. As previously stated approximately 25 students are allocated to each Home group, with Academic advisors potentially managing up to three Home groups at a time, a first year, second year and third year group. This can result in Academic advisors personally supporting as many as 75 students alongside their other roles as module leaders and course leaders for several hundred students. Several participants reported their Academic advisor did not know their name. A consequence of not knowing a personal students name is that an Academic advisor can appear indifferent, resulting in students feeling inconsequential and unimportant. Academic advisors are responsible for writing students' personal reference at the end of the course, influencing them securing future employment in nursing. SP06 felt that if the Academic advisor did not really know her, then this important document would not truly represent her or her capabilities:

I don't think she really knows any of us by name...they don't know anything about us and they're going to end up writing a personal thing...I just find it a bit...they're going to be writing things about us in references and they don't really know us (SP06)

Alongside reviewing their students' progress and engagement, Academic advisors monitor student's well-being, offering pastoral support where necessary. In the previous statement SP06 describes feeling like her Academic advisor did not really know her enough, in much the same way she did not know her Academic advisor according to the narrative below. Evidently to feel comfortable going to an Academic advisor with a problem SP06 needs to have developed more connection with that individual. For that connection to develop both student and Academic advisor need the opportunity to invest time into this relationship:

I would feel a lot more comfortable going to her if I had a problem if I knew her a bit more, if I saw her a few more times in the year (SP06)

SP07 describes a different experience to that of SP06. SP07 had felt comfortable approaching her Academic advisor for help when she had been struggling to sustain her studies whilst endeavouring to cope with aspects of her personal life outside of the course. Her personal circumstances had left her feeling overwhelmed and questioning if she could continue with the course. Her Academic advisor had offered her good support at this pivotal time. For SP07, the help she received from her Academic advisor came in the form of practical support and advice as well as emotional support. Having the one-on-one relationship with her Academic advisor enabled her to feel comfortable seeking help when she needed it, supporting SP06 account that for students to feel at ease seeking help from their Academic advisor, and potentially disclosing personal information, a one-on-one relationship is essential. Had SP07 not had a relationship with her Academic advisor she believed she may have found it difficult to approach other tutors for help:

My Academic advisor supported me when I went through a bit of a struggle at one point. I was thinking 'I don't think I can continue all of this at one time' and he was really good. He sat me down and asked if I needed any extensions on things and stuff, he was saying this is your future and it's affecting it so you need to take that opportunity, and he really helped me out at that point. If I hadn't had that Home group then I don't think I would have just gone to any lecturer and just said "I'm struggling", so the Home group is perfect for that really, just having that one-on-one relationship with a tutor (SP07)

Based on the discussion above the significance of timely Academic advisor support can significantly influence a student's experience of the course, and even their decision to continue with course when faced with difficult circumstances. Therefore 'Academic advisor support' was identified as a touchpoint on the nursing course.



The University utilises administrative support in the form of Course Administration Service (CAS) Hubs. CAS Hubs are aligned to different schools within each faculty. Although CAS support plays an integral role in co-ordinating enrolment and induction activities, managing student data (e.g. attendance and module grades) and managing student progression through the course, participants seem to have little understanding about the purpose of the CAS team, stating instead they have little need of them:

I've not had much experience with Brook Hub apart from just basically calling in sick when I had my accident a few years ago (SP08)

One participant describes accessing CAS support on several occasions at the beginning of the course when some paperwork needed sorting out, and again whilst on clinical practice placement when a mentor needed support to access the clinical paperwork. On one further occasion SP07 had contacted placement unit, who form part of the CAS team, to find out how many nightshifts she was required to do. Like SP08, SP07 states she has not needed CAS support often despite accessing them on the occasions described above. This may signify participants perceive occasional use of a service as not significantly shaping their experience on the course. Though often working behind-the-scenes to ensure seamless delivery of the curriculum, some participants presumed the CAS team were only there for students who occasionally had a problem that needed sorting:

I don't think I've had that much of a need for them. They did help me out on my last placement, none of the mentors had logons, they said they'd sort it and they did...before starting nursing I went into Brook hub to sort out my DBS and everything... Placement unit have been good, I message them and ask how many nights etc. and they've always got back to me, but I haven't had that much dealing with them really (SP07)

Although some participants perceive themselves as not needing CAS support very often, the narratives below suggest they still had expectations of the service. In particular, when the CAS team responded in a timely manner to their requests, this appeared to meet some participant's expectations of the service:

Overall, they're pretty good. They usually get back to me the same day (SP08)

Although the CAS team work predominantly behind-the-scenes and students within this study seemed unclear about their role, much of what they do can have a direct influence on the student experience. Therefore 'Administrative support' was identified as a touchpoint along the nursing.



4.4.3 The influence of social media

There are many forms of social media within the university environment for example Twitter, WhatsApp groups, and TEAMS etc. Participants from this study only expressed using the social networking service 'Facebook' in their interviews. Seven participants described using the School of Nursing Facebook page specific to their cohort. Newly enrolled students receive access to the university Facebook pages several weeks prior to induction and teaching begins. As previously established transition to a university setting can feel overwhelming for some students, with the first day of the course described as particularly scary. Several participants were concerned about not knowing anyone on the course. However, accessing the university Facebook page early helped them make connections with other students and build friendships before their studies began. During her narrative SP05 uses the pronoun 'we' to describe how she and several previously unacquainted students arranged to meet on day one of the course. This may signify that even at this early-stage SP05 was beginning to form a sense of connectedness and belonging. By actively arranging to meet previously unknown individuals, Facebook had enabled her to generate a shared experience (first day of study) with other students, and alleviate some of the fears associated with this new venture:

I think it's a brilliant thing creating the Facebook groups with the intake, I think that's fab because I didn't know anybody whatsoever, I wasn't starting with any friends. Thankfully we had already built up friends on the Facebook group. We'd never actually met each other, but we all arranged to meet in a certain place... because on the first day it's very scary, because it's a huge new experience (SP05)

The university hosts several Facebook groups for students, this includes a general noncourse specific Facebook page for the mature student population, which SP06 had joined. However, in her experience few people used the site. Evidently students preferred to attach themselves to their course specific Facebook pages. Connecting to a specific community, for example peers on the same course, may foster a sense of identity and belonging for students early in the course:

I had joined the UCLan mature students' thing, but nobody ever goes on that, nobody posts on it. It's not specifically for nursing it's just in general (SP06)

Alongside using Facebook to make connections with peers, participants also described using Facebook as a resource for information and advice. SP04 described using Facebook at the beginning of the course for practical advice and support with transport to and from the University campus, whereas SP08 tended to use Facebook more for support with assignments, particularly towards the end of the course as the academic level of study became more difficult:

I like the Facebook page because other students comment on stuff and maybe you've got the issues as well... it was interesting because you got other peoples' perspective, like buses, it was nice to see who was getting buses, who was driving in and who you could catch a lift with, very helpful (SP04)

I tend to use Facebook more often than I did. Because if you're struggling with one of your assignments and you want a bit of advice there's always somebody whose either gone through the same problem or similar problem and they've got some way of resolving it or they can point you to a person to ask (SP08)

Two participants who had found Facebook useful at the beginning of the course by the second and third year of study, had started finding comments on the Facebook pages irritating. For SP06, posts from other students were less about supporting each other and information sharing and more about '*moaning and bitching*'. In particular, students seemed to lack constraint when sharing their dissatisfaction with elements of the course, such as the lectures, even when knowing that academic staff and other students could view these comments. Evidently, some students use virtual worlds such as Facebook as a platform to vent their feelings and opinions seemingly without consequence. This apparent level of unprofessionalism had resulted in SP06 leaving the Facebook group:

Well I've left the Facebook group now because everything was just irritating me. It went through a big thing of people moaning and bitching about people talking in lectures, and I just thought it's a bit unprofessional for people to be talking that way on a group where lecturers and other people on your course can see. I left it months ago (SP06)

Participants approaching the end of their third year and close to qualifying as a registered nurse, had begun to make changes to their own use of Facebook, becoming acutely aware they now had a professional standard to maintain. From the extract below, it is clear the blurring of the lines between professional role and personal life had begun to take place. SP07 understood comments she posted on Facebook were representative of her as a nurse and as a respectful citizen. Consequently, she changed her privacy settings, and deleted many of her previous comments, so as to take control over who could view her posts:

I ended up going on Facebook and taking a lot of things off, and changing my privacy settings and things like that, because it did just make you think you know you're not just being a nurse, now you're having to kind of be a respectful citizen if that makes sense (SP07)

From the discussion above student's appeared to view, and use, Facebook differently depending on how far along the journey through the course they are. At the beginning of the course, students appear to use social media for guidance, help and advice, as well as for making friends and forming social networks. As students' progress through the course they appear to use Facebook as a medium for voicing their feelings and opinions, particularly if disgruntled with specific elements of the course. Those students approaching the final transition to qualified nurse began to demonstrate qualities of professionalism by 'putting their house in order' through editing or deleting many of their previous comments and altering their approach to using social media. For the student participants, Facebook had clearly had an influence on their experience of the course. Therefore, 'Using the University Facebook pages' is identified as a touchpoint along the student journey.



4.5 Conclusion

In summary, this chapter has presented the findings of the semi-structured student interviews from phase one data collection of this study. Overall, three core themes made up the BSc Pre-registration (Adult) nurse experience including 'navigating the academic journey', 'the clinical practice learning experience' and the 'emergent networks of support'. Evidently, as students commence and progress through the nursing course they experience multiple points of contact with the course that stand out for students as key moments (touchpoints) that positively or negatively shape a student's experience. The data revealed 18 specific touchpoints on the nursing course. Touchpoints captured from the findings have been used to devise a student journey map through the pre-registration nursing course for use during the subsequent emotional touchpoint mapping events described in the following sections.

CHAPTER FIVE

EMOTIONAL TOUCHPOINT MAPPING

'We do not learn from experience... we learn from reflecting on experience' (Dewey, 1923)

5.1 Introduction

In the previous chapter the findings from the semi-structured interviews were presented and touchpoints captured from those findings were used to devise a student journey map through the BSc Pre-registration (Adult) nursing course for use during the subsequent emotional touchpoint mapping events. This chapter presents an overview of the two separate emotional touchpoint mapping events conducted as part of phase two and phase three data collection for this study (See Figure 4 - Data collection phases). To ensure a clear demarcation between the student emotional touchpoint mapping event, the findings from the two events will be presented in separate chapters (chapter six and seven). The current chapter begins with a brief description of the two emotional touchpoint mapping events, with a focus on the similarities and differences between both events. This is followed by an overview of the core themes and subthemes captured from the student and key stakeholder participants recruited for the events is available in the methodology chapter and study design chapter of this thesis.

5.2 Description of the emotional touchpoint mapping events

The objective of the emotional touchpoint mapping event was to capture student participant's emotional highs and lows along their journey through the nursing course. During the student emotional touchpoint mapping event participants were asked to spend time exploring the student journey map containing touchpoints and verbatim comments relating to those touchpoints gathered from the semi-structured student interviews during phase one data collection of this study (See Figure 18 – Image of student journey map and Figure 19 - Student emotional touchpoint mapping event). In the absence of trigger films which are traditionally used during an emotional touchpoint mapping event (See Section 3.3.3), these pre-affixed verbatim comments were to serve as prompts, provoking discussion and reflection on personal experience of the touchpoints. Students were then asked to add their own experiences and comments to the journey map in the form of colour coded post-it notes, with a specific focus on their feelings and emotions as they linked to the touchpoints. Each

student participant was allocated a different coloured set of post-it notes so that verbatim comments could be attributed to specific individuals after the event. Participants were also advised they could add any new touchpoints they felt were missing from the student journey map, however no additional touchpoints were added during the event. Alongside adding post-it note comments to the student journey map student participants were asked to affix green tick stickers to the semi-structured interview comments they strongly agree with, and red cross stickers to verbatim comments they strongly disagree with, so that those comments/touchpoints that evoked the most emotional responses could be captured (See Figure 20 - Student strongly agree stickers and Figure 21 - Student strongly disagree stickers).

Once students had finished applying stickers to the student journey map an audio recorded interview took place with all participants to explore their thoughts and feelings whilst conducting the emotional touchpoint mapping. In addition, student participants were asked to reflect on some of their post-it note comments and share their own stories and experiences, particularly their emotional highs and lows on the course. This audio recording was then transcribed and analysed using the Braun and Clarke (2006) Six phases of thematic analysis framework (the core themes and subthemes that emerged from the analysis are presented in the following section of this chapter).



Figure 18 - Image of student journey map



I've also found as well the group supervision for the essays, I've found that supportive as well.

Figure 21 - Student strongly disagree stickers

Verbatim quotes from the student participant transcript were then attached to the student journey map as they related to the touchpoints thus adding a further layer of student experiences and perspectives for key stakeholders to explore during their separate emotional touchpoint mapping event (See Figure 22 - Example of touchpoints, verbatim quotes, post-it notes and stickers).



Figure 22 - Example of touchpoints, verbatim quotes, post-it notes and stickers

During the Key Stakeholder emotional touchpoint mapping event participants were also asked to explore the student journey map containing touchpoints and verbatim comments captured during the semi-structured interviews, however on this occasion they were also asked to reflect on the post-it notes affixed by the student participants during the student emotional touchpoint mapping event. Next, key stakeholders were asked to affix green smiley faces and red sad face stickers to verbatim comments and post-it notes they strongly agreed and strongly disagree with from both the semi-structured interviews and the previous student emotional touchpoint mapping event (See Figure 23 - Key stakeholder stickers). Additionally, key stakeholders added their own comments, feelings and emotions to the journey map in the form of colour coded post-it notes allocated at the beginning of the event. These post-it notes were distinct from the student post-it notes and each other's. Notably, green smiley face and red sad face stickers were used during the key stakeholder event to ensure they were dissimilar to the student tick and cross stickers used in the student emotional touchpoint mapping event, however on reflection red sad faces may have carried connotations for some participants, for instance anger, sadness or disapproval, and plain green and red stickers may have been a preferable option. On the other hand 'smiley face feedback' based on the Kirkpatrick model has been used since the 1950s as a common approach to evaluating learning and development initiates by gathering participants' thoughts and feelings about a programme, facilitator, content and environment (Bennington & Laffoley, 2012). Within user experience research such as EBCD the smiley face feedback approach has been adopted as a useful method for capturing participants experiences, for instance a study by Cano et al. (2018) assessing the user experience of children with hearing impairment utilised smiley/sad faces to capture positive and negative responses to playing computer games designed to aid rehabilitation.

In compansion to other interview OCLAN offed a for better personal I only got an interview with 008 Uclar as it was my first charce. experience. Took away some of the Funds pleased. I'd chaped course onkiely due to the Graup session. last ninute, so was happy I This made me feel relaced and be 807 onto course I wated ceble la conhibite 'as me', easier. when I come for my interview, they were 't expecting me any my y none work't on the entrances of the 012 I didn't come to welcome week, I suppose in my eyes I thought it was just for people that drink, and I don't really drink, so I just thought I wouldn't be able to get involved, so I just thought I found it quite intimidating in those big groups! I couldn't hear 'well there's no point! So when it came to the large group properly, I didn't know where to sit. It was quite scary, and it lecture days I did fell like 'Oh God' I wish I had come to was like 'how am I going to learn in this environment?' so welcome week really! (Participant 008) you're almost doubting that you're going to learn ok here. But you do get used to it and you adjust. But them first few weeks were (participant makes a mind blown sign and sound) (Participant 012)

Figure 23 - Key stakeholder stickers

Although both events were delivered in similar ways with both students and key stakeholders being asked to explore the student journey map then build on the map with their own thoughts and experiences, there were notable differences. For example, the student participant event lasted a total of 1 hour 8 minutes, with 25 minutes spent exploring and contributing to the

student journey map. Whereas the key stakeholder event lasted 1 hour 32 minutes, with 36 minutes spent exploring and posting comments on the student journey map. Furthermore, during the student emotional touchpoint mapping event participants worked in silence exploring the student journey map and adding their stickers and post-it notes with comments. The student participants quickly became engrossed in the comments on the student journey map, and there appeared to be an atmosphere of calm and quietude. Throughout the emotional touchpoint mapping students dispersed themselves along the journey map avoiding contact with each other. If students bumped into each other whilst exploring the map they would whisper an apology. By contrast the key stakeholders clustered around the map in small groups, openly sharing their feelings and emotional responses to some of the comments, for instance stating when and why they felt angry at some of the comments, demonstrating outward displays of emotion. They called other participants over to read specific comments with them that stood out and discussed those comments in detail with each other and the rest of the group. Thus, the atmosphere during the key stakeholder event appeared to be one of collaboration. Following both events eight core themes and two subthemes were identified, as will be presented in the subsequent sections.

5.3 Findings from the student emotional touchpoint mapping event

Following the student emotional touchpoint mapping event the post-it note comments clustered around specific touchpoints were analysed for emotions portrayed by words/phrases in the comments. For example, on one post-it note SP14 used emotion words/phrases such as '*I felt intimidated and bullied*', therefore two negative emotions were isolated from the comments associated with 'mentor support' (See Figure 24 – Post-it note comment affixed to 'mentor support' touchpoint). This enabled the touchpoints most significant to students along their journey through the nursing course to be established for prioritisation and consideration during the Co-design event, specifically those touchpoints highlighted as emotional highs and lows.

I had a strong clash with my mentor whilst on placement to the point where I felt intimidated and bullied! When I informed the PEF at the end of placement review, they were shocked and said they would "sort it". When I returned to placement my mentor then made it evident that they had been informed about what I had said! Confidentiality? (Participant 014)

Figure 24 - Post-it note comment affixed to 'mentor support' touchpoint

The total number of student post-it notes posted under each touchpoint, the verbatim comments and emotion words/phrases identified are displayed in table format (See Appendix 17 - Post-it note comments from the student emotional touchpoint mapping event). The touchpoint 'mentor support' received the largest quantity of post-it note comments (6 in total) from student participants and the joint highest number of negative emotion comments (7) when compared with other touchpoints on the student journey map. This is followed by 'academic staff' and 'facilitated learning sessions' which received five post-it note comments each, with a mix of positive and negative comments. Although 'Module Assessment Group Support (MAGS)' only received four post-it note comments it amassed the joint highest number of negative emotion words/phrases within the post-it note comments alongside 'mentor support'. Interestingly, 'Module Assessment Group Support (MAGS)' was one of only four touchpoints ('assessment feedback', 'transition into year two' and 'first clinical practice placement (subsequent clinical practice placements)') to receive entirely negative comments. Whilst 'Academic advisor support' (5 in total) and 'Home group contact/meetings' (2) received entirely positive comments. Analyses of the post-it note comments suggests overall 'mentor support', 'Module Assessment Group Support (MAGS)' and 'Academic advisor support' were identified as the most significant and emotionally charged touchpoints for student participants in this study. Notably only one touchpoint, 'Practice Education Facilitator (PEF) support' received no post-it note comments containing emotion words or phrases during the emotional touchpoint mapping event. This may indicate for many student participants in this study

contact with a PEF was not an emotional touchpoint on the course. Figure 25 is a visual representation of the 'mean' distribution of skewed positive and negative emotions connected to each touchpoint identified on the student journey map (See Figure 25 - Number/distribution of student positive/negative emotions). For example, from the Post-it note comments affixed to the touchpoint 'using the University Facebook pages', five emotion words/phrases were identified, four were negative emotions words/phrases and one was positive, therefore 'using the University Facebook pages' was plotted as a 'mean' of negative three. This information offers insight into the touchpoints most associated with positive emotions and those touchpoints most associated with negative emotions.

Following analysis of the post-it note comments the total number of strongly agree/strongly disagree stickers student participants affixed to the comments on the student journey map were noted and tallied. In total 112 stickers were affixed to the student journey map. Overall, most student participants appeared to strongly agree with the verbatim comments from the semi-structured interviews, applying 94 strongly agree stickers and 18 strongly disagree stickers, indicating their perspective and feelings were strongly aligned with the comments from the student semi-structured interviews. Although most comments received at least one sticker during the event, student participants strongly agreed with nine comments related to eight touchpoints overall (See Appendix 18 - Comments student participants strongly agree with). By contrast verbatim comments student participants strongly disagree with were associated with just three touchpoints (See Appendix 19 - Comments student participants strongly disagree with). Figure 26 is a visual representation of the distribution of strongly agree/strongly disagree stickers mapped against touchpoints on the student journey map (See Figure 26 - Number/distribution of student strongly agree/disagree stickers).



Figure 25 - Number/distribution of student positive/negative emotions



Figure 26 - Number/distribution of student strongly agree/disagree stickers

Overall, most student participants strongly agreed with the comments relating to the touchpoint 'qualifying' with a total of eight stickers affixed, suggesting many students feel excited but nervous about qualifying as registered nurses. The touchpoint 'academic staff' received the second highest number of strongly agree stickers (six stickers in total), signifying students prefer 'strict' lecturers who manage incivility in the classroom effectively. Only three verbatim comments received more than one strongly disagree sticker including 'academic staff', 'mentor support and 'Module Assessment Group Support', indicating their experience may not have aligned with the experience described in the verbatim comment. In particular, some student participants strongly disagreed that some lecturers behaved badly, that mentors were fantastic and inspired students to learn, or that MAGS was a supportive form of supervision. Overall the same three touchpoints were identified from the verbatim comments and affixed stickers as the most significant to student participants during the emotional touchpoint mapping event (See Figure 27 – Significant touchpoints identified on the student journey map).

	Touchpoint
	Module Assessment Group Support (MAGS)
Verbatim comments	Mentor support
	Academic advisor support
	Module Assessment Group Support (MAGS)
Strongly agree/strongly	Mentor support
disagree stickers	Academic advisor support

Figure 27 - Significant touchpoints identified on the student journey map

After plotting the total number of strongly agree/strongly disagree stickers, the audio recorded interview from the student emotional touchpoint mapping event was transcribed and analysed for themes and core themes (See Section 5.5 - Summary of core themes and subtheme captured from the student and key stakeholder emotional touchpoint mapping events).

5.4 Findings from the key stakeholder Emotional touchpoint mapping event

During the emotional touchpoint mapping event key stakeholders were asked to affix post-it note comments to the student journey map that described their thoughts and feelings when reading the student comments clustered around each touchpoint. These key stakeholders post-it note comments were then analysed for any emotion words or phrases portrayed in the

comments, similar to the analysis of the student post-it notes. This enabled the touchpoints that stood out to key stakeholders as significant to the student journey to be captured, compared and contrasted with the findings from the post-it note comments and core themes generated from the student emotional touchpoint mapping event. The total number of key stakeholder post-it notes posted under each touchpoint, the verbatim comments and emotion words/phrases identified are displayed in table format (See Appendix 20 - Post-it note comments from the key stakeholder emotional touchpoint mapping event). Figure 28 presents the 'mean' distribution of skewed positive and negative emotions related to each touchpoint identified on the student journey map (See Figure 28 - Number/distribution of Key stakeholder positive/negative emotions).

From the key stakeholder post-it note comments in Appendix 20 it is evident the highest number of emotion words or phrases within the key stakeholder post-it note comments was associated with three touchpoints, 'Transition into year one - Welcome week', 'Academic advisor support' and 'Mentor support'. 'Transition into year one - Welcome week' had a total of five emotion words/phrases four of which were negative emotions and one positive emotion. 'Academic advisor support' and 'Mentor support' also had a total of five emotion words/phrases each, however all those emotion words were associated with negative emotions. Recurring emotion words used in different post-it notes comments include 'sad' expressed eleven times by several different key stakeholders, 'disappointing/disappointed' used five times, and 'concerned' stated three times. Only two positive emotion words/phrases were described more than once which include 'good' and 'glad'. 'Home group contact/ meetings' was the only touchpoint to receive no key stakeholder post-it note comments.



Figure 28 - Number/distribution of Key stakeholder positive/negative emotions

During the emotional touchpoint mapping event Key stakeholders were also asked to affix stickers next to verbatim comments on the student journey map they either strongly agreed or strongly disagreed with. In total 106 stickers were affixed to the student journey map. Unlike the student participants who appeared to strongly agree with the verbatim comments from the semi-structured student interviews, key stakeholders' stickers were more evenly spread across the comments, with 51 strongly agree stickers and 55 strongly disagree stickers affixed to the student journey map. In addition, where key stakeholders had placed strongly agree stickers, they did not appear to represent the majority of key stakeholder participant opinions. For example, there was a maximum of three strongly agree stickers affixed by student participants around one comment (See Appendix 21 - Comments key stakeholder participants strongly agree stickers mapped against touchpoints on the student journey map (See Figure 29 - Number/distribution of Key stakeholder strongly agree/disagree stickers).

In contrast to the evenly spread strongly agree stickers, wherever the student journey map did contain a cluster of key stakeholder stickers this appeared to be where participants strongly disagreed with the comments. Most key stakeholders strongly disagreed with three student comments related to two touchpoints; transition into year three and mentor support (See Appendix 22 – Comments key stakeholder participants strongly disagree/felt saddened by). However, it must be noted that as the emotional touchpoint mapping event progressed it became apparent that key stakeholders had placed red sad face stickers around student comments that had made them feel sad or upset rather than just comments that they strongly disagreed with as advised at the start of the emotional touchpoint mapping event. Several key stakeholder post-it note comments made it clear key stakeholders were not agreeing with students 'feeling abandoned during their transition into year three', rather they were mostly sad that students felt this way despite the support that was available to them (See Figure 30 - Key stakeholder post-it note comments), and the red sad face stickers most represented their emotions at that time (See Figure 31 - Key stakeholder post-it notes, image 1). This is reiterated further in the excerpt below taken from the audio recorded interview during the key stakeholder emotional touchpoint mapping event where several participants articulate feeling sad when reading some of the student comments (See Table 8 - Excerpt from key stakeholder emotional touchpoint mapping event transcript).



Figure 29 - Number/distribution of Key stakeholder strongly agree/disagree stickers

Sad that student's feelings about support and quitting in terms of transition into year three are as such (Key Stakeholder 01)



Abandoned in year three! Very sad to hear this, too much directed study and workbooks and a lack of timetabled sessions isn't good! (Key Stakeholder 04)



Really sad to hear students felt alone and unsupported (Key Stakeholder 02)

Figure 30 - Key stakeholder post-it notes comments



Figure 31 - Key stakeholder post-it notes, image 1
Interviewer – one of the comments that has the most red stickers around it is the comment "I feel abandoned in year three" can you explain why that is to me?

KS07 – it's just really sad because we don't... we have this idea that we want to make them into independent practitioners and I feel we are doing that, and we are supporting them, but they just feel like they're being abandoned so. Again, is it about managing expectations? Have we not done... it's that perception of roles isn't it, it's really sad!

Table 8 - Excerpt from key stakeholder emotional touchpoint mapping event transcript

5.5 Summary of core themes and subthemes captured from the student and key stakeholder emotional touchpoint mapping events

The post-it note comments from the student emotional touchpoint mapping event revealed 'mentor support', 'Module Assessment Group Support (MAGS)' and 'Academic advisor support' as the three most significant touchpoints for student participants. Whereas 'facilitated learning sessions', 'clinical practice placement support' and 'mentor support' received the most emotion words/phrases from students during the audio recorded interview. Therefore, five core themes and one subtheme emerged from the student emotional touchpoint mapping event (See Figure 32 - Core themes/subthemes from emotional touchpoint mapping events).

Similarly, 'mentor support', 'Module Assessment Group Support (MAGS)' and 'Academic advisor support' were the touchpoints to receive the most Post-it note comments from the key stakeholder. Whereas 'Academic advisor support', 'Module Assessment Group Support (MAGS)' and 'Transition into year one' received the most emotion words/phrases from key stakeholders during the audio recorded interview. Therefore, three core themes and one subtheme was identified from the key stakeholder emotional touchpoint mapping event (See Figure 32 - Core themes/subthemes from emotional touchpoint mapping events).



Figure 32 - Core themes/subthemes from emotional touchpoint mapping events

CHAPTER SIX

PRESENTATION OF FINDINGS FROM THE STUDENT EMOTIONAL TOUCHPOINT MAPPING EVENT

'Except in dealing with commonplaces and catch phrases one has to assimilate, imaginatively, something of another's experience in order to tell him intelligently of one's own experience' (Dewey, 1923)

6.1 Introduction

The previous chapter has described the student and key stakeholder emotional touchpoint mapping events and identified the core themes and subthemes from both events. This chapter presents findings from the student emotional touchpoint mapping event as they relate to the core themes and subthemes identified from the audio recorded interview undertaken during the emotional touchpoint mapping event, and post-it notes comments. The touchpoints that seemed to hold the most significance was the first clinical practice placement, mentor support, facilitated learning sessions, Academic advisor support and Module Assessment Group support, although this final touchpoint also incorporated issues around support with independent study skills (See Figure 33 - Findings from the student emotional touchpoint mapping event). Throughout this chapter verbatim quotes used to support core themes and subthemes are followed by participants' identification code e.g. SP01, whereas post-it note comments are indicated by participants' identification code and the acronym PINC (Post-It Note Comments) e.g. SP01 PINC.



Figure 33 - Findings from the student emotional touchpoint mapping event

6.2 Student perceptions of disparate clinical practice learning experiences

Findings from the emotional touchpoint mapping event around the clinical practice placement experience was similar to the results from the semi-structured interviews, with participants consistently placing great emphasis on the clinical practice component of the nursing course. Participants in this study describe a mix of clinical practice experience with most students having some positive and some negative experiences, however two students described only positive experiences in the clinical placement area:

Good points (on the course) definitely placements (SP13) ---Loved the placements...I've had positive experiences on placement (SP12)

Of those participants who had a mix of positive and negative experiences several indicated concern around the equitableness of placement allocations. Students appeared to align clinical competence with access to a wide variety of learning opportunities. A sense of unfairness resulted when students were not given the same opportunities to participate in clinical specialisms their peers appeared to have multiple experiences of:

There should be some way of making placements more consistent...I've heard reports that there are people having their second respiratory placement! (SP14)

Having limited experience in areas of patient care their peers have lots of experience in, can result in anxiety and under-confidence in some students about their own knowledge and competence. However, participants who had multiple experiences of the same clinical practice area had similar concerns related to the disproportionate amount of experience and specialist knowledge they had acquired in just one area of patient care:

I feel like I've just had a lot of the same (placements). I had orthopaedics twice and then rehabilitation which was orthopaedic patients (participant laughs), so I felt like I've had a lot of orthopaedics (SP11)

Having good placement experiences was attributed to being quite *'lucky'* by one participant, suggesting an element of serendipity involved in having a good placement experience and conversely misfortune for those who experience a poor placement. Thus, placement allocation can seem like a gamble and something outside of students' control:

I've had good placements I feel...I've had a variety of placements. I've had lots of experiences and lots of opportunities. I think some people haven't so I feel I've been quite lucky (SP12)

Some students assumed more pragmatic reasons for any disparate clinical practice learning experiences. One participant felt the local NHS trust where their clinical practice placements were allocated lacked the specialist areas other NHS trusts had, resulting in missed learning opportunities. On the current pre-registration nursing course students are allocated to local NHS Trusts based on their postcode of residence. Although some students expressed a willingness to travel to different NHS trusts in order to have access to the same learning experiences as other student on the same course, the large student numbers and limited clinical placement areas prevent this being an option:

I don't feel like we've got the specialist areas around our way, whereas if you're in another (NHS) Trust you've got more specialised areas. The hospice in my local Trust don't accept UCLan students, so I feel that it is the area we're falling down on, whereas if you're in another Trust they're all there, they've got lots of specialisms. I don't mind traveling to a different area or Trust but we don't get that opportunity (SP12)

Disparate clinical practice experiences were recognised as commonplace for most students in this study. For SP12 discussing positive placement experiences with some of her peers who had had negative experiences felt *'wrong'*, and appeared to trigger feelings of guilt:

We've spoken about that (referring to a peer), so I feel a bit wrong to say sometimes I've had a good experience because I know she hasn't. (SP12)

Alongside contrasting clinical practice experiences, participants in this study acknowledged the effects of different approaches to the clinical supervision (mentoring) process.

6.3 The student experience of the clinical mentoring process

Findings from the audio recorded interview and the student post-it notes indicate mentors, and support from mentors, was amongst the most significant touchpoints that shaped students' experiences on the course. Student participants reported many emotional highs and lows that related to their experiences with mentors in the clinical practice area. Overall participants in this study had mixed experiences with their mentors; two student participants describe having only positive experiences, two participants describe having negative experiences with their various mentors throughout the course and a further two participants describe having a mix of positive and negative experiences with their mentors overall. One participant (SP15) had not yet been on a clinical practice placement.

Four out of the seven student participants describe a positive mentor experience in at least one of their clinical practice placement areas. A good mentor experience was viewed as one where their mentor guides and supports them, offers lots of opportunities to meet their learning needs and communicates well with the student. Mentors who were able to effectively support students in clinical practice through skilful communication and a conscientious approach to meeting student's learning needs, were an inspiration and enhanced participant's overall learning experience. Being '*advised*' as indicated in the extract below can be a one-off activity but the use of the term '*guided*' suggests the student had opportunities to work closely alongside their mentor in a supervisory capacity:

I feel I've been strongly advised and guided all of the times where you're kind of on your own and things (SP13)

My mentors have been a huge inspiration and improved part of my learning. They communicated well, checking all areas needed to learn were covered and supported (SP12 PINC)

Mentors who demonstrated competence in their role as mentors, as described above, contributed to students having a good experience. Consequently, a good experience with a mentor can influence students' future career choice with some students deciding they would even like to work in the clinical practice placement area they were once a student and alongside those mentors:

I've had a good experience with mentors, so much so that I might even want to go back there for jobs (SP13)

Evidently mentors who role modelled behaviour and characteristics participants perceived as desirable qualities in future colleagues. However, not all participants in this study had good experiences with their mentors.

6.3.1 The effects of a negative mentor experience

At the beginning of the audio recorded interview students were asked how the process of emotional touchpoint mapping had made them feel. SP10 was the first to respond and appeared visibly upset whilst explaining how reading the positive comments other students had written about their clinical practice placement areas and mentors had made her feel, as she reflected on her own negative experiences. From the narrative below there appeared to be a stark difference between participants experiences, and a degree of unfairness around this disparity:

I've got a lump in my throat, I feel upset. Because it highlights the difference between a good experience and a bad experience. When you discuss placements and you discuss how you got on with your mentor with other people, I get quite jealous that everybody's having this really, really good time (SP10)

Negative clinical practice experiences were associated with mentors having little or no time for student. For example, some participants found it difficult to have their competencies signed off (a mandatory requirement of the nursing course), with some mentors leaving the mandatory documentation until the day before a student is due to leave the clinical area. Statements such as '*what date did you start?*' potentially devalues the participants contribution to the clinical environment as the students start date holds no significance to her mentor. Furthermore, completing the participants documentation at the latest possible occasion implies opportunities were missed to give the participant critical constructive feedback early enough for them to act on that feedback:

It's like the day before you leave they say "oh come on we'll fill your book in now", "let's do this date, what date did you start?" (SP10)

This can leave a student feeling a low priority for the mentor and their clinical experience depersonalised. This type of mentoring is described as '*impersonal*', suggesting a disconnect between the student and their designated mentor. The use of the word '*burden*' in the extract below signifies the mentor by their actions and behaviours made the student feel like something difficult or unpleasant to be dealt with:

It's very impersonal, you're made to feel like a burden (SP14)

Many of the participants describe struggling to have their clinical practice documentation completed in the practice placement area. One student uses 'battle metaphors' such as 'fighting' to describe the process of having clinical competencies signed off. Using battle metaphors implies a negative associating with students learning needs being met in the clinical practice placement area, although this battle appears to be with the time constraints placed on mentors in practice. Participants recognised some mentors were under pressure and too busy to facilitate the student's learning needs. Thus, mentors lack the time to help rather than the willingness to help. As a result, students can find themselves facing a difficult dilemma, either to '*intrude*' on busy and overwhelmed mentors or compromise their own learning needs:

You're not on your own, I've had 2 out of 4 of my placements I've had a mentor that hasn't been as helpful as they could have been. But I think it's difficult because obviously hospitals are just so overwhelmed at the moment. But it's just difficult for the student because you don't want to intruded when they're busy with something. You're fighting to get your competencies signed off, or you just want to have a chat with them sometimes and it's just hard to find that time (SP09)

The concept of fighting, or being in a conflict, in the clinical practice placement environment was not restricted to the facilitation of learning or signing of clinical documentation, some student participants in this study had experienced direct personal conflicts with their mentors in practice. Two participants felt their integrity was being questioned by their mentors:

On the placement that I was on I was made to feel I had to have days off for health reasons, and I had surgery because of it. But my mentors exact words were "I'm not calling you a liar but you need to verify it, you need to get a staff nurse with you to witness that" (SP14 PINC)

I've hardly ever had any days off whilst I've been here (on the course) up to my third year, or in placement, but I had two valid days off and I was made to feel really bad that I had missed two days, by my mentor (SP12)

In the extract below one participant describes feeling intimidated and bullied. Although the incident was reported to the PEF the situation was unresolved and subsequent comments from the mentor indicate a breach in confidentiality, resulting in this participant feeling in a more vulnerable position with her mentor:

I had a strong clash with my mentor whilst on placement to the point where I felt intimidated and bullied! When I informed the PEF at the end of placement review they were shocked and said they would "sort it". When I returned to placement my mentor then made it evident that they had been informed about what I had said! Confidentiality? (SP14 PINC)

Experiences described above can potentially deter students from speaking-out when problems occur in the clinical practice placement environment. On this occasion the support

system in place for students in the clinical practice environment appeared to fail. As indicated in the excerpt below, when support systems are not aligned with the needs of the student conflicts encountered in the clinical practice environment can leave some nursing students feeling unable to continue with the rest of that placement, possibly contributing to attrition on the nursing course:

I had a mentor that was talking about me behind my back to a patient whilst I was in the room. I was so close to the end of the placement that I didn't raise it with her...the rest of the placement became untenable (SP10 PINC)

The impact of such negative interactions with mentors can affect students on a personal level by undermining their confidence. One participant acknowledges experiencing fear and low confidence in the role of student nurse. Wanting a way to '*manage this fear*' she chose to share her concerns with a mentor. However, the mentor's response connotes an element of not wanting to deal with this issue as the student explains '*I've had it thrown back at me*'. Moreover, an ill placed 'pep talk' from the mentor further serves to undermine the student's confidence further:

I've explained that I do have low confidence in the role, I've had it thrown back at me, "well you're going to be a nurse, and you're going to be qualified", well this is what I'm worried about...that's just not working you saying that! I know, I'm not stupid, I know that in XYZ I'm going to be qualified, you're just stating the obvious. Give me a way of managing this fear (SP10)

Student's comments suggest some mentor's '*can't be bothered*' because they do not really want to be mentors but are '*forced into it*'. There is the suggestion there should be flexibility around who should become a mentor, thus avoiding the need for coercion. SP14 argues the inception of a better mentorship system, whereby nurses are selected to be mentors, rather than the current system in which all registered nurses are expected to become mentors usually within two years of qualifying:

I feel like the mentors, they don't want to be mentors, they're forced into it for progression purposes... I wondered if there could be more leeway about forcing people to be mentors if they clearly don't want to take on that role (SP10)

I feel that there should be a better system in place to monitor mentorship (SP14)

6.3.2 Mentors as gatekeepers to student learning in the clinical practice environment

Within this study several participants perceived mentors as the gatekeepers to learning opportunities and have an expectation mentors will engage them in teaching and instruction

in the clinical environment. Mentors who fulfilled students' expectations were described as *on the ball*' in other words competent, attentive and effective in their mentor role:

My mentors have always been on the ball and their guidance has been fab (SP13)

For some students effective learning in clinical practice was dependent on the availability of meaningful learning opportunities and support from clinical supervisors. The narratives below suggest students perceived themselves lacking the underpinning knowledge base of a clinical area to pursue their learning needs. Consequently, mentors were deemed as having better knowledge and understanding of the learning opportunities available in their own clinical areas. Where mentors appeared disinterested in teaching students, participants felt this resulted in missed opportunities for learning:

Mixed mentors, sadly some of them have no interest in showing you anything or teaching you (SP11 PINC)

That's difficult because you don't know what's available so you can't go with that mentality of 'well I want to learn this, this and this' because sometimes you're not aware of what's out there are you (SP14)

If they don't tell you, you're not going to know (about available learning opportunities) (SP09)

The discourse indicates some students may rely on mentors to determine their learning goals. Several participants described seeking out their own learning opportunities when mentors did not meet their expectations around facilitating learning experiences. However, employing self-directed study approaches to learning in clinical practice appeared to be viewed as additional to their student nurse role, and an activity that encroached on their '*own time*':

It takes some of your own time to find something new that you might want to do (SP09) ---I spent a whole placement finding my own things to see, learn and partake in (SP11 PINC)

Taking control and proactively seeking out their own learning opportunities was viewed as a final course of action for students when mentors were perceived as struggling to fulfil the role due to competing workload demands. This suggests self-directed learning is not considered an integral component of the clinical practice placement experience for some students:

You know they'll say "we'll have to try and find time to fit you in here or there" but you almost have to go off your own back and say "right, well I want to do this so I'm going to contact that person and make time to learn more" (SP09)

As gatekeepers to knowledge and learning in the clinical practice environment some mentors were perceived as a constraining factor for some students learning experience. For example,

in the narrative below the mentor's response to one participant's eagerness to engage with multiple learning opportunities appears obstructive. Although the mentor offers a rationale for suppressing the student's enthusiasm, this is interpreted as unsupportive behaviour by the participant:

When you do say "oh I really want to do this, and I really want to do that, I want to learn as much as possible", they're like "oh well you're only in second year" which I find quite bizarre. They don't really support you as much as you might like (SP09)

The findings presented above indicate some student participants perceive mentors as the gatekeepers to student learning in the clinical practice environment. Mentors are felt to have a better understanding of their own clinical practice placement environment than students new to the environment. Consequently, students expected mentors to facilitate learning opportunities as part of the mentor's role. Mentors who insufficiently engaged students in learning often had a negative impact on the student experience overall.

6.4 Support with academic study - student perceptions of abandonment

Several touchpoints are linked to the core theme 'Academic support - student perceptions of abandonment' these include 'Module Assessment Group Support (MAGS), 'Transition into year two', 'Transition into year three' and 'Academic advisor support'. 'Module Assessment Group Support (MAGS)' and 'Transition into year two' both received the largest number of negative comments exclusively, and no positive comments, whereas 'Transition into year three' and 'Academic advisor support' received a mix of positive and negative comments and was discussed extensively in the audio recorded interview that followed the student emotional touchpoint mapping event. Within this theme the term 'academic support' does not relate to support that affects students' academic performance, instead this theme relates to participants' sense of how support received, or perceived lack of support, influenced their experience of the course. Overall, most students in this study indicated they had struggled with the academic elements of the course at times. Points of transition during the course seemed poignant times where participants describe experiencing most difficulty, particularly for those students moving into year two:

I found the beginning of year 2 quite a shock! Year 1 seemed so easy (SP09 PINC)

Second year is a huge jump compared to year 1, not prepared for it either. 3rd year I do feel very much left on your own. I'm disappointed with this as this is when you need additional support (SP12 PINC)

Two participants describe feeling '*let down*', '*left on your own*' and '*abandoned*' since transitioning into year 3. One student reveals a feeling of isolation, as though she has been separated from her cohort and left to 'fend for herself'. Several participants believe these feelings are related to the increase in self-directed study required in year 3, where students are expected to grasp those crucial concepts central to nursing knowledge and understanding aside from their peers:

Important sessions are self-taught! Feel isolated and sometimes like I'm at an open university (SP13 PINC)

This feeling appears exacerbated when the skill of independent learning is not established whilst in year one, resulting in some students struggling in consequent years. SP12 explains how she has struggled to understand much of the material during the first two years of the course, subsequently in year three she is now having to relearn things she should have understood in year one. If she had established a better level of independent learning skills in year one this would have positioned her better for learning in year two and year three:

One of my biggest difficulties in year 1 I found was the anatomy and physiology. I don't know whether it's me or...but I found the exams we did in year 1 blew me away. I don't know how I passed them! I didn't understand half of it, and I think if I'd been in a better position in year 1 it would have helped me in year 2 and 3. But there are still things which I'm still getting my head around because year 1 didn't...The 'learn to learn' I found that a complete waste of time! (SP12)

One participant indicates the theoretical content on the course is at a higher level than she feels is necessary to become a nurse, being more akin with '*training to be a doctor*'. Furthermore, the narrative below implies far more is expected of current student nurses than was expected of previous trainee nurses, but it is the amount of theory that has left SP11 feeling overwhelmed:

The amount of theory is overwhelming. I understand we need to know, and nurses are trained to do a lot more now, but sometimes I think I'm training to be a doctor (SP11 PINC)

SP14 concurs suggesting theoretical content on the course is delivered at a higher level than current students can understand, without consideration for the level new students' may be positioned at. Rather than this being a consequence of studying at a higher educational level SP14 contends this is idiosyncratic to the pre-registration nursing course as she, and peers she had spoken to, had a very different experience when studying on other higher education courses at the current University:

We joked about it in our college group. I did the access to allied health professionals' course before coming here and we were saying we actually felt like we'd learnt more on that course in the 9 months we were there than in the first year (on this course). Because it was delivered to us (here) above a level that we could understand or than we were at. And if you had any question, you were told you need to read around the subject. Well no, give us something to learn. Give us something. We just felt like we weren't learning (SP14)

From the extract above it appears those students who sought answers from academics with things they were struggling to understand were advised to research the answers for themselves. SP14 does not indicate if they received guidance around how to undertake that research, however from her next response it seems this participant was not given further advice to help her learn how to learn. Her experience is in line with many other student participants in this study from both the emotional touchpoint mapping event and the semi-structured interviews who struggle with the concept of independent learning. This is indicated by five out of seven student participants from the emotional touchpoint mapping event strongly agreeing with the verbatim comment below captured during the student semi-structured interviews:

There's been a few times where I've thought of quitting and that's just honest. I still don't know if I want to be a nurse in terms of I love doing what I do but it's so scary (SP07)

One participant (13) used the term '*Open University course*' analogously to depict the amount of independent learning she was expected to conduct in year three, in other words her current studies resembled a form of off-campus distance or on-line learning course. SP13 signifies she had not anticipated this aspect of the course, and if she had she may have chosen to study elsewhere:

It does feel like an open university at the moment, that's what I'll say. It feels like an open university, self-taught teaching right now. And if I wanted to do an open university (course) I would have applied (for one). That's our opinion from studying at college to this. I do feel...we are older adults aren't we...but we do need extra knowledge and understanding and more depth in what we're doing (SP13)

By contrast SP12 had expected to conduct independent learning as part of the course, yet she was still shocked by the amount of independent learning required. However, she had expected far more support to conduct independent learning that she had received. Now in year 3 SP12 describes feeling '*let down*' and unsupported:

I came into it knowing I needed to do a lot of my own learning, I just think to me in my opinion it's changed over the years. So in year 3 we've been left far too much without any support, and that's been a massive erm...and actually year 3 is the time we need support... I feel like we've been left to it in year 3 and I feel quite let down (SP12)

One system established to support students on the pre-registration nursing course is the process of module group assignment support (MAGS). Consistent with the findings from the student participant semi-structured interviews, participants from the emotional touchpoint mapping event indicated MAGS was ineffectual for supporting their learning and assessment needs, even suggesting it was a 'waste of time':

The group supervision I just think it's a big waste of time for me (SP10)

I feel the Group Assignment supervision are a waste of time. Difficult to get questions over as too many in the group. I feel this hasn't helped my development in assignments (SP12 PINC)

Participants 09 and 14 regarded group supervision as less beneficial than one-to-one supervision. For SP09 MAGS was even counter-productive, leaving her feeling more confused at times:

I don't find Group Assignment supervision as supportive as individual supervision, sometimes I leave feeling more confused! (SP09 PINC)

I don't feel that group supervision is as supportive as individual supervision (SP14 PINC)

The findings presented above indicate some student participants within this study felt unsupported and even abandoned as they progressed through the course. Reasons for this include a perceived increase in self-directed learning linked with insufficient independent learning skills and decreased classroom-based contact time. For some students' theoretical content on the course was delivered at a higher level than expected. Overall student participants indicated a need for more support and direction from academics with independent learning particularly in year three as the level of study became more challenging.

6.5 The influence of academic staff on the student experience

During the student emotional touchpoint mapping event the touchpoint 'Academic staff' received the second highest number of post-it notes comments after 'Mentor support' and the highest number of strongly agree and strongly disagree stickers affixed. Notably, 'Academic staff' had a more evenly balanced proportion of emotions described, with four negative and three positive emotions identified, indicating some polarisation amongst the student views., Consequently 'Academic staff' is considered a significant emotional touchpoint within this study due to the amount of focus it received in both the student touchpoint mapping event and the student audio recorded interview that followed. As demonstrated by the mixed emotions described in the student post-it note comments different aspects of the student's experiences with academic staff was perceived as both positive and

negative. For example, several student participants described finding academic staff as approachable, and their knowledge and guidance helpful to students at various points on the course. Specifically, academic staff who shared their own personal experiences helped them to understand different professional aspects of the course and their future roles. Those academic staff who assisted students at various points on the course were held in esteem by those students:

Lecture staff are very well respected. Their knowledge and guidance allow us to embrace this challenging emotional journey (SP113 PINC)

Lecturers have mostly been approachable. However (lecturers) reading off a PowerPoint I feel it isn't as easy to take in the info. The lecturers who bring in experiences help me gain the understanding of the area (SP12 PINC)

Although one participant described some academic staff as '*awful*' (SP10 post-it note comment), by contrast all student participants, including SP10, indicate the behaviour or actions of some staff rather than the academic staff themselves affected the student experience in a negative way, particularly in the classroom setting. Frequently this related to how academic staff responded to incivility in the classroom setting. Student participants from the emotional touchpoint mapping event shared similar stories and frustrations to the students in the semi-structured interviews for this study. They describe feeling frustrated by the amount of poor behaviour from other students that disrupted their own learning and made it difficult to concentrate. SP12 purports incivility in the classroom is disrespectful and identifies disrespectful behaviour as talking, using mobile phones and being generally disruptive during lectures:

Some of the students are actually very disrespectful to the lecturers and are talking, and aren't listening to lectures, rustling, phones going off, and you know alarms going off, and then I think that isn't right and I think they should be told (SP12)

This was exacerbated further by some lecturers' approach to alleviate the situation. In the extract below the lecturer appears to misread the situation, interpreting the rise in noise levels as students feeling ready for a break in the session, but for SP15 this action interrupts her flow of learning and potentially acts as a reward system for disruptive students. This leaves the participant feeling even more annoyed and frustrated. SP14 states '*I get angry*' at this behaviour:

It's frustrating, really frustrating. You're coming here to learn and they're disrupting it, a lot of it's at the back so I generally sit at the front. You just can't hear anything, it's just continuous. And if you have a lecturer that just doesn't address it you'll end up hearing other people saying "shush" and then you hear more mumbling because they're getting annoyed and saying "don't shush me" or whatever. So it is annoying, because then they're (the lecturer) saying "oh should

we have a break then" but it's like we've not learnt anything. I do find that they have too many breaks just to shush the other people at the back (SP15)

SP15 argues lecturers should 'take more hold of the room' rather than ignoring the incivility and continuing to deliver the session. Unabated noise and disruption make it difficult for her to learn. One participant implies undealt with noise and chatter in the classroom can redirect student's attention risking an escalation in noise and chatter amongst other students resulting in a cumulative effect. Once this happens other people in the classroom become inattentive:

Sometimes I wish lecturers would take more hold (control) of the room and speak up, because it is hard to hear and they sometimes just continue lectures, but I feel it is hard to learn in those circumstances (SP15 PINC)

I find sometimes people drift, and when they're talking amongst themselves it kind of diverts everyone's attention (SP09)

Four student participants recommend academic staff adopt what they perceive as an effective strategy to deal with incivility that would enable them to 'take control' in the classroom setting, that is to be '*stricter*'. SP11 acknowledges it must be difficult for lecturers, particularly on nursing courses where class sizes can be large, yet she still feels it is essential for academic staff to be actively stricter, telling students to '*shut up*' when necessary. On the other hand, SP11 appears torn between the idea of '*stricter*' lecturers and lecturers being '*nice*':

It must be hard talking to a room full of people, but lecturers need to be stricter. Tell us to shut up if you need to, you're all too nice...which is a good thing (SP11 PINC)

From the extracts below several students clearly agreed with SP11 stance as they discussed one academic with a reputation for being strict in class. They suggest this particular academic had set boundaries early in the sessions that result in students knowing and understanding what was and what was not acceptable behaviour in the classroom. Students had responded accordingly and had automatically exhibited different behaviour with this particular lecturer compared with others. SP10 verifies this opinion:

I completely agree, (lecturer name omitted) is dam strict, put them all in with her, fantastic (SP13)

There's a level of his strictness almost, at the beginning formed that...you knew your boundaries, you knew where you could you know...and people did put their phones away naturally (SP12)

You know your boundaries don't you (SP10)

Although all participants agree that academic staff need to be stricter in the classroom setting SP10 raises concerns about lecturers being too strict. These fears appear to relate accounts of lecturers insisting students leave the session following a perceived episode of incivility:

..... (lecturer name omitted) he threatens to throw people out (SP10)

One such episode is recounted by several of the participants involving a young student nurse near the beginning of the course, whose mobile phone had rung during a classroom session. This resulted in the student being asked to leave the class. However, from the narratives below the student experienced humiliation in front of the entire cohort (several hundred students) rather than a simple reprimand:

Someone's phone went off and he went "RIGHT OUT!" in the middle of Harrington lecture theatre (SP14)

It was the first month and he made everyone look at this girl (whose phone had gone off) - I'd be surprised if she's still here – and she was walking like the 'walk of shame' and I thought...And it wasn't a phone call that she'd made, it was the phone that had gone off! (SP13)

That was too much, that was humiliation (SP10)

The narrative above suggests a fine line exists between academics taking control of the classroom by being firm with students and academics humiliating students. When the line is crossed academic's own behaviour may be perceived as a form of incivility, or even bullying, by some students. One student implies a condescending element to the interaction between the academic and student, with the student being treated like a child, not an adult learner:

It's was like bullying wasn't it (SP12)

Even the youngest student they're still an adult aren't they? (SP14)

Treating adult student like children, especially in front of their peers, shows a level of disrespect from academics. Correspondingly students may struggle to respond in an adult way when made to feel like a child to peers. Though the repercussions of the above student/lecturer interaction are unknown, the extract below suggests the young student in the narrative left the room dumbstruck, and may possibly have left the course:

Being treated like adults rather than children shows a level of respect - I think if he'd have said that to me I'd have come back with something, whereas she was actually quite you know...because we are all adult learners and I'd have really apologized...but she was just scared and walked out. We could have lost her, we don't know. But I actually did respect his lectures as well (SP12)

6.6 The effect of disparate approaches to Academic advisor support

Overall students within this study consistently reported Home groups as essential for facilitating supportive relationships developed with peers. For some students, the support received from peers in their Home group played a fundamental role in motivating them through the nursing course:

Good point of the course – home groups, just because of the people you meet and they become like a little family. They're dead supportive of each other (SP11)

I think without our home group and support from our friends there's no way I would have got through this, there's no chance, no (SP13)

It's only because of good friends who are like "come-on, come-on" that you stay, and actually the determination I suppose (SP12)

At the time of this study Academic advisors subsumed the role of Home group leader for the duration of a students' three-year course. In keeping with the findings from the semistructured interviews student participants in the emotional touchpoint mapping event described mixed experiences with their Academic advisors who led the Home group. Several participants recount positive experiences with their Academic advisors during Home group meetings. Notably, students who describe positive experiences appeared to value Academic advisors who showed concern for their student's welfare and progression:

When you've got Home group meetings they'll (Academic advisor) say "are you having any problems, are you having any this?" I don't feel the Home group is necessary for educational purposes but for meetings, and for asking how are you doing, are you on track? They're always there as guidance... in the Home group meetings if you have any personal problems you'll address it directly (SP13)

By contrast some participants depict their Academic advisors showing no interest in them or their welfare. Through establishing a personal relationship with the students in their Home group Academic advisors can provide a consistent point of contact, support and guidance to their allocated students. The comment below suggests some students experience a more impersonal relationship with their Academic advisors:

I've never had a Personal tutor (Academic advisor) say, "is everything alright?" (SP10)

An impersonal connection with the Academic advisor is evident in the excerpt below where one participant believes their Academic advisor did not even know their name. Feeling nameless can result in a sense of detachment for students. During the semi structured interviews SP06 had also indicated that her Academic advisor had not known her name, suggesting this experience was not atypical but commonplace. Evidently both participants raised concerns about their Academic advisors having responsibility for writing there end of course employment when they appeared unfamiliar with the student:

I don't even think my Personal tutor (Academic advisor) knows my name, (SP12 then turns to participant next to her and asks...Does your Personal tutor know your name?), but she's meant to write our references isn't she, but she doesn't know us! (SP12)

The findings presented above indicate some student participants within this study felt they had experienced good Academic advisor support, whilst others had less satisfactory experiences. Students who had a good relationship with their Academic advisor felt more able to seek support with problems associated with the course or personal circumstances. Overall student participants indicated a need for more support from academics with independent learning particularly in year three as the level of study became more challenging.

6.7 Conclusion

In summary, this chapter has presented the findings from the student emotional touchpoint mapping event. Overall, five core themes and one subtheme emerged from the data including: student perceptions of disparate clinical practice learning experiences, the student experience of the clinical mentoring process, Support with academic study - student perceptions of abandonment, the influence of academic staff on the student experience and the effect of disparate approaches to Academic advisor support. Evidently, students acknowledged the significance of the clinical practice placement and mentor support as fundamental to their overall student experience. Where students perceived unfairness in the placement allocation, less access to learning opportunities than their peers and lack of support from their mentors their overall experience was negative. By contrast, the overall experience was more positive for students who appeared to have a variety of learning opportunities and access to a supportive, welcoming mentor who made them feel part of the team. Findings also show students had mostly positive experiences with academic staff but indicate a need for more support and direction with independent learning study throughout the three-year course. The core themes that emerged from the student emotional touchpoint mapping event are tabulated with the findings from the key stakeholder emotional touchpoint mapping event, discussed in the following chapter, to produce problem statements that formed the bases of the small co-design working group discussions in student and key stakeholder Co-design event (See Appendix 14 - Problem statements tabulated from student and key stakeholder emotional touchpoint mapping events).

CHAPTER SEVEN

PRESENTATION OF FINDINGS FROM THE KEY STAKEHOLDER EMOTIONAL TOUCHPOINT MAPPING EVENT

'Every experience is a moving force. Its value can be judged only on the ground of what it moves toward and into' (Dewey, 1938/2015, p.38)

7.1 Introduction

This chapter presents findings from the key stakeholder emotional touchpoint mapping event. During the event participants were able to explore the student journey map comprised of touchpoints and verbatim comments devised from the student semi-structured interviews, alongside post-it notes comments from the student emotional touchpoint mapping event. Following the touchpoint mapping an audio recorded group interview with all key stakeholder participants was conducted. The findings from that interview, alongside key stakeholder post-it note comments added to the student map, are presented in the current chapter. Three core themes and one subtheme emerged from the key stakeholder emotional touchpoint mapping event; 'Academic advisor support – perceived ambiguity associated with the role', "Academic support - navigating the tension between support and independence' and "Being' or 'becoming' professional - key stakeholders' expectations of students (See Figure 34 - Findings from the key stakeholder emotional touchpoint).



Figure 34 - Findings from the key stakeholder emotional touchpoint mapping event

The findings from the emotional mapping and audio recorded group interview with the key stakeholders are presented in the following chapter. Verbatim quotes from the interview transcript used to support core themes and subthemes are followed by the key stakeholder identification code e.g. KS01. Post-it note comments are indicated by participants identification code and the acronym PINC (**P**ost-It **N**ote **C**omment) e.g. KS01 PINC.

7.2 Academic advisor support - perceived ambiguity associated with the role

During the emotional touchpoint mapping event key stakeholders voiced their concerns around student's experiences of Academic advisor support. Many key stakeholders described feeling sad when reading comments made by student participants who felt 'unknown' and 'nameless' to their Academic advisors. Academic advisors provide key support for students across the whole three-year course. Building a good relationship with their Academic advisor can be a prerequisite for students engaging in dialogue and seeking vital support when necessary. In the statement below key stakeholder KS07 portrays the Academic advisor in a two-way relationship requiring involvement and connection between a student the academic. Becoming acquainted with students underpins an honest and open connection, enabling more effective support:

The comment about the Personal tutor (Academic advisor) not knowing the student's name really stood out to me and it really got me! I don't know why, because I make such an effort. I feel I can make an effort so that my students know me, that they know how I can support them. I want them to know they can have that honest and open relationship with me. And I feel really sad that a student hasn't been able to access that (KS07)

Disparity amongst academics in their approach to undertaking the role of Academic advisor prompted some key stakeholders to contemplate their own approach to the role. One participant reflected on her current position as Academic advisor for 75 students, recognising how competing demands for her time had resulted in the Academic advisor role taking less precedence than it should do. All Key stakeholders acknowledged the role needs to be given more priority.

I know because that triggered me, because running three home groups with potentially 75 students and everything is coming in and you've got some difficult problems to sort out, and it's trying to find the time! So I did write a comment that we need to prioritise the time with our home groups (KS06)

The narratives above suggest some Academic advisors do not know their students by name, but in the excerpt below KS03 implies some students appear to not know who their designated Academic advisor is despite this information being disseminated at the beginning of the nursing course. Students seeking academic or pastoral support and advice during the course are referred to their Academic advisors as a first point of contact. A breakdown in that support process occurs when students do not know who to access for support, specifically their Academic advisor. This can have implications for both the student and other key stakeholders such as administrative staff endeavouring to signpost students to appropriate support networks. KS03 believes students share responsibility for acquainting themselves with their Academic advisor and establishing a firm relationship, yet even when directed to contact their Academic advisor some students chose not to access the available help:

We get students saying to us "oh I don't know who my Personal tutor (Academic advisor) is" and I just think you've been here nearly 2 years how can you not know who that person is? And from our point of view that's difficult because that's the person we advise them to see, because we know there's an issue with this person, we know they've had this problem and that problem and we contact them and say this is the problem they've had, but they're not connecting with that tutor so they're not going to get the help, and they are not going to think the helps there (KS03)

The excerpt above suggests key stakeholders perceive '*helping*' students as an aspect of the Academic advisor role. This perception is supported in comments by KS02 Who purports a way Academic advisors help students is predominantly through '*pastoral support*'. However, the use of the term 'Academic advisor' may misrepresent the purpose of the role and cause misunderstanding for some students who may then expect support and guidance with '*academically linked*' activities such as: module assessment advice, help with assignment writing and study skills. However, this sphere of activity is traditionally associated with specific module supervisors. KS02 suggests lack of clarity concerning the role results in some students not accessing their Academic advisor and benefitting from vital personal support when required:

They don't like the term Academic advisor because it suggests it's academically linked, and they think you're going to help them with their assignments, but actually you're that pastoral support, and if they knew that they might feel more supported (KS02)

Confusion around the purpose of the Academic advisor role may be compounded further by the recent change in designated names. On the pre-registration nursing course an Academic advisor has previously been referred to as a Personal tutor or a Home group tutor. The title was recently amended to align the school of nursing with the wider university's use of the term 'Academic advisor'. KS03 describes three name changes occurring over a four-year period, with three different cohorts of students using three different names during the transition phase. According to KS05 some documents and on-line resources presently being

used and accessed by staff and students still contain the previous titles designated, demonstrating disparity with the terminology used on the nursing course remains several years after transition to the term Academic advisor:

I think the terminology as well, because some people refer to the Academic advisor as Personal tutor, that's on the PARE, and they know you as their Home group tutor or their Academic advisor! Even when students know you and you've got a good rapport with them, I've still got PARE requests saying I don't know who my Personal tutor is! It's the changing terms we use (KS05)

It's the terminology, and I've only been here four years and that (Academic advisor name) has changed three times in those four years hence all the different terminology. You've got all the cohorts that have been running throughout that time and they've been use to one term, but a different cohort is use to another term! I think that's partly our fault, but it's because the change happens just like that (KS03)

Several key stakeholders themselves used the terms Academic advisor, Personal tutor and Home group tutor interchangeably during the key stakeholder touchpoint mapping event, signifying disparity around the terminology designated to the role of Academic advisor even amongst the key stakeholders conversant with the role:

I've got Home group students that I'm dealing with as a Home group tutor, Academic advisor, Personal tutor whatever! (KS05)

From the data presented above key stakeholders perceive the role of the Academic advisor as 'pastoral support', 'helper' and 'problem solver' for their students. Yet students can become confused about the role of the Academic advisor, in part due to the current title ascribed to the role, but also through a lack of relationship with their allocated Academic advisor. Ambiguity associated with the role may hinder some students seeking vital support from their Academic advisors as they transition through the course.

7.3 Academic support - navigating the tension between support and

independence

Several key stakeholders acknowledged some students struggle with the transition into higher education, specifically with the directed study elements of the course. Some participants suggest more support was needed with prospective students to prepare them for academic study prior to commencing the nursing course. Although participant KS02 recognises partnerships already exist between academics at the University and Local Further Education Colleges, concerns were raised around the function of those partnerships. Currently links between universities and colleges allow academics to commence dialogue with potential students and address questions related to available courses and entry

requirements. Several key stakeholders question whether those partnerships foster successful transition to higher education, by adequately preparing potential students for academic study at university level prior to enrolment on the nursing course. Missed opportunities to do this can result in some students who enter university via a Further Education College route not developing those study skills required at higher education level:

More work may be needed before students transition to higher education around what it will be like (KS02 PINC)

But I think before they get here we need to pre-work. I know we go in and do talks in partner colleges, but actually do we do any transition to higher education? Because if they've been spoon-fed education before they get here they've still got that expectation that we'll continue to spoon-feed them...then they get none of that developing independence, you know and that's what we should be doing shouldn't we. They kind of think I'll come to university and its DIY...there's lots of comments about Open University on there! (KS02)

For many key stakeholders, the development of independent study skills is a continuous process that should start before students' transition into higher education and continue throughout the course. Participant KS04 argues students are not currently prepared for independent study before the transition to the nursing course. The transition point can be difficult to determine as individual students may begin and end their transition at different points in time. Gauging where and when to assist students through their transition is challenging and can result in missed opportunities for improving the student experience. Participant KS04 perceives academics having absolute responsibility to prepare students, but terms such as 'we don't', 'we miss' and 'we're not doing that right' indicate academics may be failing in that task:

Makes you think about year one again and question are we really embedding the skills needed to study independently! But we don't prepare them for independent study do we in that way. We miss that transition don't we, and you can't pinpoint where it begins or where it ends, or where we should be inputting. We're not doing that right! (KS04)

In the narrative below participant KS07 echoes other key stakeholders concerns around building partnerships with local colleges that better manage the transition of future students to higher education, particularly in relation to the students' expectations and understanding of higher education. KS07 emphasises the obligation academics have to instruct and support students with how to conduct directed study:

I was going to say is it us not managing their expectations? And I know we go and do talks in FE about this course but are we managing that well enough? I don't know whether we are! And I know like you say they don't know what directed study is. We say, "you've got directed study" but no one's being told like okay "what does that mean" (KS07)

Omitting to support students clearly to engage with directed study early in the course had resulted in some student participants commenting on how abandoned they had felt in year three of the course. During the key stakeholder audio recorded interview participant KS04 suggested changes to the curriculum which has seen an increase in amount of directed study along with a decrease in facilitated classroom-based sessions in year three had resulted in some of the negative comments from the students in this study, particularly around feeling abandoned. The nursing curriculum is underpinned by an 'active learning' philosophy characterized by student led interactive learning that requires student participation and engagement prior to, during and after facilitated learning sessions, hence necessitating more independent learning had appeared to reduce the efficiency of the curriculum for some key stakeholders. Participant KS04 believes it is the responsibility of academics to prepare and support students for the real world of nursing but the current curriculum forces students to ready themselves. Furthermore, the current course was perceived as not always meeting students' needs as future graduate nurses:

I think there's a lot of workbooks...the curriculum has changed and it feels like I said before diluted, but we have we've weakened it, so there's a lot of online learning, directed study. I think in year three they really want to feel that we are preparing them for the real world of nursing, not just coming in for OSCE prep or clinical skills or medicines management sessions. And we've diluted that year, but I do feel this last seven years it's gone weaker and weaker and weaker. So it doesn't surprise me to see comments like that written down in black and white that they feel abandoned, because I can see that, fully see that (KS04)

Several key stakeholders suggest academics are supporting students to become independent learners, but student's assumption around this support may culminate in some students feeling '*unsupported*' and even '*abandoned*'. Furthermore, students' expectations of the nursing course may be influenced by their previous educational experiences. KS07 suggests perhaps it is the student's perceptions around the role of academics and their own role as independent practitioners that could be better managed:

It's just really sad because we have this idea that we want to make them into independent practitioners and I feel we are doing that, and we are supporting them, but they just feel like they're being abandoned. Again, is it about managing expectations? Have we not done... it's that perception of roles isn't it, it's really sad (KS07)

Really sad to hear students felt alone and unsupported, maybe students don't always see support until its really needed rather than knowing it's there. Also, this could be around their expectations, especially previous education (KS02)

Several key stakeholders described feeling disappointed that some students appeared to presuppose independent study skills could be learned via taught session times with

academics rather than developed and refined for themselves throughout the course. Key stakeholder KS01 felt some students' expectations were of an unreasonable standard, not in accordance with practical realities, with some students expecting more support and less self-direction as the course progress. Despite a few key stakeholders acknowledging '*significant efforts*' had been made by staff to support students, discrepancy still exists between staff perceptions and student's perceptions of the support possible and available:

Some of the expectations could never be delivered consistently. Like the ones with the third years wanting more support. It just seems that there's some lack of understanding of that build to independence. So actually, there's something missing in there, and that is really disappointing that despite significant efforts by everyone to do that, that it's not working, we're still getting people disappointed at that point, and I think that's a real shame that there is a real mismatch between staff perceptions and perception of students (KS01)

In response to comments from one student participant during the student emotional touchpoint mapping event suggesting '*I'm disappointed as this is when you need additional support in 3rd year*' (SP12), some key stakeholders declared year three of the course is when students should least require support with independent learning, instead this is the time they should begin to thrive in their studies. For those students who are still struggling in year three lots of support mechanisms exist at the University and students may just need reminding of that support and even helped to access relevant services:

Year three...there are lots of support mechanisms in place. This is when you should be coming into your own! Perhaps a reminder pre-year three of who is there to help (KS05 PINC)

Whilst ruminating over the idea of expecting 'additional support' as they progress through the course, a post-it note comment from KS07 questions what would constitute as '*enough*' support for students. Consistency among staff could help manage student expectations of support. However, academics need to strike a balance between supporting students and empowering students to support themselves. Empowering students to become independent learners is particularly significant for those studying on the nursing course who will be expected in their future roles as graduate nurses to make autonomous critical decisions relating to patients and patient care. According to participant KS06 overreliance on academics, particularly Academic advisors, for support may affect student's resilience and preparedness for the role:

Consistency of support from staff is disappointing, over delivering can raise expectations beyond what can be delivered. We need to take care that we get the balance between support and empowerment (KS01 PINC)

It's very hard isn't it. You want them to progress into that independent learner, they're going to be making some life changing decisions for patients, but how do you make the transition that one moment they're critically thinking, and they're sort of independent and they're not relying on...and then they think oh I'll just ring my personal Academic advisor for advice about this and that. They actually have to be able to think for themselves or look for that information (KS06)

The findings presented above indicate key stakeholders recognise some students feel unprepared for studying at higher education level, and more could be done to prepare potential students prior to their enrolment on the nursing course. Students' expectations around the extent of self-directed learning required on the course could be managed better, and key stakeholders need to navigate the tension between supporting students and encouraging students to engage those essential independent learning skills directly.

7.4 'Being' or 'becoming' professional - key stakeholders' expectations of students

During the touchpoint mapping event several key stakeholders described feeling sad, upset and even angry whilst reading some of the student comments on the journey map. One comment that caused much discussion amongst the key stakeholders was from a student who stated, '*I didn't go to much of Welcome week, and so when it came to the big lecture days I did feel like 'Oh God' I wish I had come to welcome week really!*' (SP15). Within the nursing course welcome week is a pivotal time when new students meet key stakeholders for the first time and form initial bonds with their Academic advisors and peers potentially lasting the next three years. Participant KS01 suggests there could be misunderstanding around the purpose of welcome week for some students:

If students can't commit to week one, it's disappointing! As a professional in training they need to commit to that mindset from day one. However, is this more widely perceived? Do we need to change the labelling of the welcome week to reflect the importance? (KS01 PINC)

Misunderstanding may emerge from how information about welcome week is labelled and communicated to new students enrolling on the course. SP015 comments suggests an assumption Welcome week within the nursing course was similar to Fresher's weeks that takes place across most universities. A major aspect of Freshers week is associated with many social events as indicated by the following narrative '*I suppose in my eyes I thought it was just for people that drink, and I don't drink, so I just thought I wouldn't be able to get involved, so I just thought 'well there's no point going!*' (SP15)'. However, as a non-drinker SP015 had believed attending would be a waste of her time, yet for some key stakeholders' lack of engagement so early in the nursing course was an early sign of unprofessionalism:

I'm feeling really angry, as if you don't turn up in the first week! That's made me really, really angry. I have put 'do we need to change the labelling of the welcome week to reflect its importance' on a post-it note because it's made me angry that they haven't recognised (it's importance), because they're training to be professionals so why would you just not turn up on day one? Why would you not do that? (KS01)

The narrative above reveals an expectation nursing students should be equipped with a professional attitude prior to starting the course. In the excerpt below KS01 suggests a desire to undertake the pre-registration nursing course emanates from an enthusiastic interest in nursing as a profession. Although training to be professionals, students commencing the course should commit to exhibiting enthusiasm and a professional attitude from the outset. KS01 felt uneasy about those students whose actions were contrary to her established viewpoint of professionalism:

If you're entering a profession I believe you should be passionate about why you are doing it, and it sits uncomfortably with me that somebody would think it was all right not turn up on day one if they are training to be a professional... what's the starting point to be able to get those professional behaviours? (KS01)

This lack of distinction between 'educating professionals' and training individuals to 'become professionals' was noted in several key stakeholder's narratives. For example, key stakeholder KS01 indicates students are '*training to be a professional*' in the comments above, but subsequently declares students should already 'be professional', without acknowledging any disparity between the two concepts. KS06 also implies academics are educating 'professionals' not 'pre-professional' students on the pre-registration nursing course, whereby students retain a pre-professional identity until the point of registration on completion of the course:

Be professional! Professional training... and if they were professionals they wouldn't be acting in that way saying, "we didn't know about that and we didn't know about that" (KS01)

We're there trying to educate professionals to be thinking about early intervention and prevention and its linked so much because actually if they were doing that throughout university and throughout their academic studies then they wouldn't have these outcomes in year three would they (KS06)

Participant KS02 believes professional values are something students do not start the course with but learn in the classroom through the influence of academic staff, suggesting professionalism is a process that must be learned and therefore takes time. This contradicts many key stakeholders' perceptions that student nurses must immediately '*be professional*'. Staff in the academic environment were perceived as an influence on students developing

professional values before the transition to a clinical practice placement environment. However, during the emotional touchpoint mapping event all key stakeholders stated students 'don't recognise us as professionals':

It is learning professional values that transfer to the ward environment (KS02 PINC)

I would like to deliver key lectures with my uniform on, because I had some feedback recently that they want professionals to come in and teach them (KS06)

They don't see us as nurses. We say we are still a registered nurse, and we tell them our background but...(KS07)

For participant KS01 professional values and behaviour can be role modelled in the academic environment through the activities and behaviour of staff. Knowledgeable academic staff who remain conversant of their field of nursing, and share their experiences with students are perceived as professionals:

If you got some lecturers like it says in some of the comments that are just reading off the PowerPoint, I can well believe that they don't believe we are professionals! Being a professional is more than a uniform, it's how you conduct yourself. They can tell when you say things like "let me give you a real-world experience, let me tell you about etc" it's that clinical currency, kudos, they can see that (KS02)

Some key stakeholders relate unprofessional behaviour to the specific 'labels' ascribed to undergraduate nurses such as referring to them as 'students' rather than 'student nurse', 'nurse' or 'trainee professional'. For participant KS01 the term 'student' alludes exclusively to being a learner or pupil and ceases to encompass any professional connotation. Consequently, Pre-registration students remain incognizant of any distinction between their identity as a student and their role as nurses:

One of the things I put on there in my angry post-it note, was that they're trainee professionals and actually maybe that's what we need to be thinking of their labelling of some things, they're trainee professionals...Talking to them as students isn't that different to being a pupil it's not that big a transition...'Professional trainee', I thought maybe that's what we need to change it to instead of 'student'! (KS01)

The findings above indicate the transition from non-professional to pre-professional to professional nurse is blurred for key stakeholders, and the term 'trainee professional' meant different things to different participants for this study. Ambiguity around the concepts of 'being' or 'becoming' a professional affected key stakeholder's expectations of students and student behaviour on the nursing course. Some key stakeholders felt students' level of professionalism could be linked with their identification as either 'students' or 'student nurses'.

7.4.1 'Student' or 'Student nurse' - what is in a name?

Using the name 'Student nurse' may confer a notion of professionalism to students and key stakeholders alike. During the key stakeholder audio recorded interview participant KS03 recalled hearing an academic refer to a student as 'nurse' for the first time. Though accustomed to the term 'student', the actuality of students becoming 'nurses' impacted her perception of nursing students on the course. Henceforth she believed terminology can affect how student nurses view themselves, precipitating them to develop a sense of identity as professionals. Incorporating the term 'nurse' in their title serves as a reminder of the critical nature of their role as they come to see themselves as not only students but as nurses:

I think the fact that they are called students has an impact. Because when I sit in the appeals one particular academic always calls them nurse, they never ever call them student. They always call them nurse. The first time I heard it I went "Oh God they are actually going to be a nurse in a few weeks if this goes through" but they saw themselves as a student who'd failed an assignment, failed a written piece of work, not somebody who's going to go out and deal with life and death and all that comes with it. And I think terminology has an impact on how they perceive themselves (KS03)

Participant KS03 surmises that using the term 'student' in lieu of 'student nurse' in the academic environment encourages students to view their role as nurses in isolation from their role as students. Participant KS04 describes students as having a '*different mindset*' whereby the context of a university is existentially different to the context of the clinical practice placement area, suggesting a dissociation between the two environments. One potential consequence of this perception is an adjustment in behaviour; thus, students conduct themselves more professionally when in the clinical practice placement environment and conversely less professionally in the academic environment:

(Students think) ... "I'm a student when I'm here and I'm a nurse when I'm out there", and I think that impacts on how they act when they're here as well. You should look after me when I'm here, you should lead me and guide me and do this for me, whereas on placement I'm having to think for myself! And I think that's where they see completely different things (KS03)

I think they see it separately; Uni is Uni and placement is placement, there's a different mindset! (KS04)

Being referred to as 'student' in the academic environment instead of 'student nurse' potentially strips students' of their professional identity and causes anxiety for students when entering the clinical practice placement environment. Participant KS05 believes cultural norms associated with the clinical practice environment requires students to be managed closely following strict rules and codes of conduct around behaviour, time management, learning activities and practice. Paradoxically the culture of university encourages students

to self-manage, determining their own learning schedules via independent study and selfregulation relative to their learning goals and activities. This stark difference between the two environments can create anxiety for some students as they transition between the two contrasting cultures:

I've had a lot of experience mentoring students in practice where I had to keep them on their toes, so they acted professionally, there were very strict rules for them to follow. The academic environment is very different, and I'm wondering if there's a big change. They've got used to a culture of university and being a student, coming in and going out, and then they go into practice and what a shock it is! The actual anxiety going into placement and stepping into that role is causing! (KS06)

From the extract above the clinical practice placement environment is perceived as a shock for some students. To adapt to culturally different environments some students may feel compelled to assume different identities. This tendency may be linked to a perceived difference between the role of 'student' and the role of a 'student nurse'. KS05 noted how treating students consistently as 'student nurses' throughout their course could help some students reconcile the different roles and resolve some of the challenges student nurses encounter when transitioning from academic environment to clinical practice environment:

Do you think there's too much divide in their roles like student versus student nurse? Should we be treating them as student nurses throughout? And I know that sounds a little bit ambiguous saying that but it's that coming in and saying your students here whilst you're learning, but when you going into practice suddenly you've got all these very strict rules and expectations. How do you make that role consistent throughout? (KS05)

Key stakeholder KS05 endorses the assumption above further as she reflects on her own experience training to be a nurse. From the beginning of the course KS05 describes a fixed identity as a 'student nurse' to the point of registration on completion of the course. In keeping with the changes described in the introduction chapter of this thesis, KS05 describes her nurse training as different ('*the old* way') to the current nursing course. Although the transformation of pre-registration nurse training is considered to have advanced nursing as a profession, key stakeholders recognise a consequence of this progress may be a loss of emphasis on the 'nurse' element of the course:

I trained the old way originally and you were very much a student nurse and that was it. From the minute you signed your contract and stepped through the door until you got your PIN as you qualified. It was very different. And we do want to move forward as a profession, but I wonder if we need to emphasise the point that they are student nurses (KS05)

From the findings above key stakeholders appear to refer to students on the pre-registration nursing course using interchangeable terms such as 'student', 'student nurse', 'nurse',

'professional' and 'trainee professional'. Such inconsistent and sometimes incompatible terminology can result in uncertainty for some student nurses around their identity and role, potentially making it more difficult for them to navigate transitions between the academic and the clinical practice placement environment. Establishing a clear identity as a 'student nurse' may have a positive impact on a student's transition through the nursing course and on the overall student experience.

7.5 Conclusion

In summary, this chapter has presented the findings from the Key Stakeholder emotional touchpoint mapping event. Overall, three core themes emerged from the data; 'Academic advisor support - perceived ambiguity associated with the role', 'Academic support navigating the tension between support and independence', "Being' or 'becoming' professional - key stakeholders' expectations of students' and one subtheme "Student' or "Student' or 'Student nurse' - what is in a name? Evidently, key stakeholders acknowledged some students commence the pre-registration nursing course feeling unprepared for the academic level of study and amount of self-directed learning required. Some key stakeholders felt there was lots of mechanisms in place to support students with their academic study in higher education but were concerned students may not know how, or where, to access the support. Inconsistent approaches to the Academic advisor role appeared to influence how supported students felt, overall key stakeholders felt this was an area for improvement. Findings from this study also indicate key stakeholders have an expectation student nurses exhibit professional behaviours and attitudes from the start of the course and continue to develop their professionalism throughout the course. This expectation may contrast with student's self-perception as students whilst in a university setting and student nurses in the clinical practice placement environment. Having a clearer designation as a 'student nurse' may help students establish a distinct identity early in the course precipitating the development of professionalism. Results from the key stakeholder emotional touchpoint mapping event are tabulated with the core themes that emerged from the student emotional touchpoint mapping event to produce problem statements that will form the bases of the small co-design working group discussions discussed in the following chapter (See Appendix 14 - Problem statements tabulated from student and key stakeholder emotional touchpoint mapping events).

CHAPTER EIGHT

PRIORITIES AND ACTIONS FOR COURSE IMPROVEMENT FROM THE STUDENT AND KEY STAKEHOLDER CO-DESIGN EVENTS

'If we teach today's students as we taught yesterday's, we rob them of tomorrow' (Dewey, 1923)

8.1 Introduction

The findings from the student and key stakeholder emotional touchpoint mapping events were presented in chapters four, six and seven. This chapter describes the two Co-design events conducted as part of this EBCD study, where participants included students and key stakeholders who collaboratively coproduced priority improvements for the BSc Pre-registration (Adult) nursing course. The chapter begins with an overview of the Co-design events, including an outline of the double diamond design model adapted for use in each event, and exploration of the KJ brainstorming technique implemented during the events. The three priority recommendations for course improvement students and key stakeholders agreed will be examined including: to facilitate and support student development of independent study skills, enhance student support in the clinical practice placement environment and clarify and enhance the role of the Academic advisor.

8.2 Overview of the student and key stakeholder Co-design event

The overall objective of the Co-design event was for students and key stakeholders to collaboratively identify and agree ways to redesign aspects of the BSc Pre-registration (Adult) nursing course that could improve experience of the course for future students. Although the original plan was for just one Co-design event to be held on site at the University's main campus, a second Co-design event was subsequently held as none of the six students recruited attended the first event on the day. Instead, four students emailed their apologies, a further student had missed the event after mixing up their dates and no explanation was received, or contact made, with the final student. Students expressed different reasons for non-attendance including childcare issues, poorly children and a sudden change in working hours on clinical placement. Seven key stakeholders did attend the first Co-design event. Although Co-design with student participants and key stakeholders was a key objective of this study, the opportunity to conduct Co-design with the seven attending key stakeholders was perceived as valuable for testing the KJ brainstorming technique before use with

students' and key stakeholders in a second Co-event, and for gathering the varying perspectives of the initial seven key stakeholders who had a mix of designations and experiences. The priority recommendations from this key stakeholder Co-design event are subsequently presented and discussed alongside the recommendations from the second Co-design event in this chapter.

A second Co-design event took place with both student and key stakeholder participants several weeks after the first event. For the second Co-design event seven different students were recruited and attended the meeting alongside four different key stakeholders than the first event. The intention was not necessarily to recruit different students than the first event, however when the study was re-advertised to all students enrolled on the BSc Pre-registration (Adult) nursing course none of the previously recruited students responded. This may have been a result of their workload and the inconvenience of attending the university campus, as at the time of the second Co-design event most of those students were scheduled to be in the clinical practice placement environment.

The format of the Co-design event was based on an adaption of the 'double diamond design model' (British Design Council, 2005) which is divided into a four-stage process designated as discover, define, develop and deliver (Huq & Gilbert, 2017). Application of the double diamond design model required the study participants to work collaboratively through a sequence of divergent thinking (exploring and discovering issues widely) and convergent thinking (more focused shared defining of the issues). Followed by a further process of divergent thinking (collective development of ideas for improving experience on the nursing course) and finally convergent thinking where cooperative focused solutions to redesigning aspects of the nursing course are agreed (priority recommendations for course improvements) (See Figure 35 - Adaption of double diamond design model for Co-design event).



Figure 35 - Adaption of double diamond design model for Co-design event

8.3 Co-design using the KJ brainstorming technique

Both Co-design events began with participants jointly exploring and discussing the student journey map and the problem statements that emerged from the findings of the student emotional touchpoint mapping and key stakeholder emotional touchpoint mapping event (See Figure 36 - Exploring and discussing the student journey map, and Appendix 14 -Problem statements tabulated from student and key stakeholder emotional touchpoint mapping events). This was followed by the larger group being divided into smaller co-design working groups who were each given one problem statement to explore in a more focused way using the KJ brainstorming technique. The groups were divided into three or four people, with consideration given to ensuring a mix of designation (for key stakeholders) and year of study (for student participants) where possible. For example, one small co-design working group in the second Co-design event contained a year two and year three student, and a Lecturer/Module leader (dual role). However, in the first Co-design event with only key stakeholders the seven participants were divided into two small co-design working groups based on designation and experience (See Figure 37 - Small co-design working groups). Each small co-design working group was then able to produce more focused solutions to the challenges described in the problem statements using the KJ brainstorming technique, before deciding as a whole group which solutions should become a priority action for course improvement from this study (Table 7 - KJ brainstorming technique adapted for Co-design

event). The KJ brainstorming technique enables groups of individuals to objectively reach a consensus when prioritising problems and solutions (Spool, 2004).



Figure 36 - Exploring and discussing the student journey map


Figure 37 - Small co-design working groups

8.3.1 KJ brainstorming technique step 1

The aim of step 1 of the KJ brainstorming technique is to generate and collect multiple ideas and solutions to a problem. Wilson (2013) describes this as the "divergent phase" where multiple ideas and solutions can be produced without criticism or judgement. Notably, each member of the small co-design working group is considered to have their own individual experience and expertise, making them able to view problems and solutions from a unique perspective than others in the group, resulting in large quantities of creative ideas. For this study, each small co-design working group of students and key stakeholders spent time individually, and silently, writing down ideas and solutions to the problem statement. The idea behind silent brainstorming is for individuals to have comparable opportunity to disclose their own thoughts, opinions and ideas, as the more vocal participants in a group. Each idea or solution was documented on a separate post-it note, with each participant asked to document as many as they could think of (See Figure 38 - Step 1, individual brainstorming).



Figure 38 - Step 1, individual brainstorming

8.3.2 KJ brainstorming technique step 2

Once multiple ideas and solutions have been documented each person within the small codesign working group took turns to present their ideas and solutions with the rest of the small group, without interruption from the other group members. This process serves several purposes, it encourages engagement from all members of the co-design working group, it enables the quieter group members to pitch their ideas and solutions without the more confident or forceful individuals monopolising discussion with their own ideas. Finally, it elicits more creative thinking in listeners within the group who may choose to further develop or build on their own ideas and solutions. Once all ideas and solutions had been presented, the small co-design working groups moved onto step 3 of the KJ brainstorming technique.

8.3.3 KJ brainstorming technique step 3

During step 3 individual ideas and solutions were then grouped and clustered by each small co-design working group into a collection of ideas that began to form affinity diagrams. Affinity diagramming enables groups to organise large amounts of data in a meaningful way (Plain, 2007). Therefore, students and key stakeholders discussed how each idea or solution related to another, before jointly agreeing how similar ideas should be grouped (See Figure 39 - Step 2, Grouping and clustering ideas and solutions). A unified consensus was reached by each small co-design working group around naming each cluster of ideas and solutions in a way that best captured the core concept underlying the ideas in that group (Figure 40 and Figure 41 - Forming affinity diagram, image 1 and image 2).



Figure 39 - Step 2, Grouping and clustering ideas and solutions



Figure 40 - Forming affinity diagram, image 1



Figure 41 - Forming affinity diagram, image 2

8.3.4 KJ brainstorming technique step 4

Finally, step 4 of the KJ brainstorming technique requires a group to identify and prioritise those issues, ideas and solutions they feel are the most important for improving a service. Therefore, individuals within each small co-design working group were asked to vote for the group of ideas they wanted to prioritise for course improvements using green tick stickers. Individuals were allocated two votes (two stickers) each. Two groups of ideas and solutions emerged from each Co-design event as the most important priorities for course improvement, receiving the most participant votes overall. Key stakeholders from Co-design event 1 identified 'clarify and enhance the role of the Academic advisor' and 'additional academic support' as their priorities for course improvement. Participants from Co-design event 2 identified 'study skills' and 'support in clinical practice' as their priorities for course improvement. Notably, key stakeholders in Co-design event 1, and student and key stakeholder participants from Co-design event 2, identified and voted for a similar group of ideas and solutions as a priority for course improvement, but labelled this priority differently. In Co-design event 1 this priority was labelled 'additional academic support' whereas participants in Co-design event 2 named the priority 'study skills' (See Figure 42 - Priority improvement from Co-design event 1 and Figure 43 - Priority improvement from Co-design event 2). As both priorities related to 'support with independent study skills' the ideas and solutions were combined to form one priority recommendation from this study with the overall title: 'facilitate and support student development of independent study skills'.



Figure 42 - Priority improvement from Co-design event 1



Figure 43 - Priority improvement from Co-design event 2

In summary, both Co-design events enabled students and key stakeholders to coproduce three priority recommendations for course improvements on the BSc Pre-registration (Adult) nursing course to be discussed in the following sections:

- Facilitate and support student development of independent study skills
- Enhance student support in the clinical practice placement environment
- Clarify and enhance the role of the Academic advisor

8.4 Priority 1 - Facilitate and support student development of independent study skills

Supporting students to develop independent study skills was identified as a key priority in both Co-design events. Overall, participants felt students should be introduced to the concept of independent study prior to starting the nursing course, with that information being built on from day one of the course and throughout the three years. Suggested solutions included, encouraging greater collaboration between the university and partner colleges/schools focused on developing independent study in future higher education students prior to commencing their university study. The university facilitating two-day summer camps focused on study skills and independent learning prior to students starting the nursing course. Once on the nursing course, provide clear resources to help students develop their study skills such as video guides to undertaking study skills/independent learning, 'students top tips for academic study/independent learning' devised by previous/current students for new students and more guidance during induction week about independent study skills. Finally, as students' progress through the course and different academic level, establish a student led support network focused on study skills and independent learning, provide academic study skills drop-in sessions, an academic mentor (staff member or other student) to support students with academic studies and make more use of the Academic advisor/Home group meetings to help support students with study skills and independent learning. Post-it note comments related to priority 1 and recommended solutions are summarised in table 9 (See Table 9 - Priority 1, Facilitate and support student development of independent study skills).

	Priority 1 - Facilitate and support student development of independent study skills					
Post-it note comments	Teach students what independent learning means	Ask prevíous students to develop top típs for índependent study based on theír experíence	We already have clinical skills and numerous drop-in sessions so let us have academic skills drop-in sessions too	Have more independent learning throughout the three years so when it comes to 3rd you it's not such a shock	Students are told they are adult learners, but they may not know what this means, include more about this in the 'Welcome week'	
	Put on my library study skills sessions and independent learning specifically for nursing	Facilitate learning skills by having more sessions on independent study skills in year one e.g. how to write academically, as this was only touched on at the end of year two	Have study skills sessions timetabled into first year and onwards to adapt and improve academic writing and literature reviewing skills. Make them mandatory throughout the years in home groups	Have sessions with Home group tutors to review learning and come up with an individualised learning plan	Provide more information as to the benefits of independent learning i.e. how will this help students with their future studies	
	Work with colleges to prepare students for University study	Include learning outer learn in the curriculum	Use Home group sessions to discuss what we have learned during our independent study	University to provide a definition of 'self-directed study'	Early identification of student's expectations (of independent study)	
	Offer one to one drop-ín support for academíc study skílls	Make more use of academic advisors to help support students study skills	Offer a week's 'pre course' sessions to go through referencing, academic writing and analytical/critical thinking	University to put on student sessions to promote independent learning	Provide students with academic mentors as a specific point of contact	
Summary of Solutions	 Implement supplementary study skills Devise a step-by-step video guides to undertaking study skills/independent learning similar to YouTube clips e.g. on database searching Previous/current students to create 'students top tips for academic study/independent learning' for new students Establish academic study skills drop-in sessions specific for student nurses alongside clinical skills drop-in and numeracy drop-in sessions (setting up academic study skills drop-in sessions where together students and staff can explore effective methods and practices for undertaking independent learning) Make more use of the academic advisor/Home group meetings to help support students with study skills and independent learning Introduce an academic mentor, or specific person (staff member or other student), to support students with academic studies Establish student led support network focused on study skills and independent learning Introduce greater collaboration with partner colleges and schools focused on future Higher Education students developing independent study skills prior to commencing their university study Undertake a two-day summer camp focused on study skills/independent learning prior to students commencing their university studies 					

Table 9 - Priority 1, Facilitate and support student development of independent study skills

8.5 Priority 2 - Enhance student support in the clinical practice placement environment

Enhancing student support in the clinical practice placement environment was a further priority identified in the student and key stakeholder event (Co-design event 2). Participants specifically suggested mentors, clinical practice supervisors and assessors should be better educated about their roles in managing the student clinical practice experience. Continuity of supervision and support was considered a high priority with participants agreeing a specific clinical practice supervisor should be identified as a point of contact on students first day in the clinical practice area (although this does happen in many clinical areas it is not always the case for all students). Continuity of supervision and support was considered particularly important for students identified as having neurodiversity, who should have an opportunity to work alongside the same clinical practice supervisor throughout the clinical practice placement where possible. Participants also felt there needed to be greater contact between the University and placement area. A suggested way to improve communication was through better use of the Practice Assessment Record and Evaluation (PARE) system. PARE is where students and clinical practice staff document student/staff communications and assessments, progress and competence. However, names and email addresses of student's Academic assessors (University staff responsible for supporting students in the clinical practice environment) was not always clearly documented in PARE, making it difficult for students to access support from designated University staff when they were in the clinical practice environment. Furthermore, some clinical staff were not familiar with using PARE. One suggested solution was for there to be more training for students, clinical staff and Academic assessors on using PARE. Finally, participants also agreed there should be more PEFs in place to help support students and mentors/clinical practice supervisors/assessors in their role. Post-it note comments related to priority 2 and recommended solutions are summarised in table 10 (See Table 10 - Priority 2, Enhance student support in the clinical practice placement environment).

		Priority 2 - Enhance student support in the clinical practice placement environment				
Post-it note comments	let the Ward and mentor know that students are coming in advance	all ít a partnershíp rather than a mentorshíp	more education and how to help students and encourage students	incentives for staff to promote the advantages of mentoring a student	Add contact names and emails to the PARE document	
	Ensure all staff and previous mentors are familiar with the new roles. <u>I've</u> heard you can have a different practice supervisor for every shift	Have people in place to be mentors for the day so when you show up on placement it's not going to be like "who can we put you with?"	being able to work with a new person for each shift might provide valuable learning experience but it may mean the practise assessor does not get used to the student	more education on the additional needs of students and how to look out for struggling students	greater contact between university and placement	
	More training on PARE for all (students, clinical practice and academic staff)	on my first placement with the new practice supervisor/practice assessor system it did not feel any different to having a mentor and associate mentor. We all collaborated to make the most out of my placement for myself and then	ability to discuss problems through PARE	More PEFs to help support staff and students	there needs to be more of an outline of what is expected from year one, year two and year 3 students and what to expect from staff	
Summary of Solutions	 Ensure all clinical practice staff/previous mentors are familiar with their new roles as clinical practice supervisors Ensure a specific clinical practice supervisor is identified as a point of contact on the students first day in the clinical practice area Ensure clinical practice supervisors are educated in managing students with additional learning needs in the clinical practice area Ensure continuity of supervision and support by arranging for students identified as having neurodiversity to work alongside the same clinical practice supervisor throughout the clinical practice placement Ensure the Practice Assessment Record and Evaluation (PARE) document includes the contact name and detail of Academic assessors More training for students, clinical staff and Academic assessors on using PARE Employ more PEFs in practice to support students and mentors/clinical practice supervisors/assessors in their role 					

Table 10 - Priority 2, Enhance student support in the clinical practice placement environment

8.6 Priority 3 - Clarify and enhance the role of the Academic advisor

Clarifying and enhancing the role of the Academic advisor was a priority for course improvement identified in the key stakeholder event (Co-design event 1). Although no students had attended Co-design event 1, key stakeholders felt from the student journey map, the problem statements generated from the student and key stakeholder emotional touchpoint mapping event, and their own experience, that Academic advisors played an important part in students overall experience on the BSc Pre-registration (Adult) nursing course. Suggested solutions to clarifying and enhancing the role of the Academic advisor was through appointing an Academic advisor lead, offering Academic advisor training (particularly to new teaching staff) and devising clearer guidance for students and staff around the role and remit of an Academic advisor. A further solution was the introduction of a peer review and feedback process for the Academic advisor role, similar to the peer review process for teaching and learning, as this currently does not exist. As findings from this study suggest there was confusion around the variety of names used to denote the Academic advisor role including 'Academic advisor', Personal tutor' and 'Home group leader', key stakeholders suggested all material relating to the role should be updated to reflect one title. Post-it note comments related to priority 3 and recommended solutions are summarised in table 11 (See Table 11 - Priority 3, Clarify and enhance the role of the Academic advisor).

	Priority 3 - Clarify and enhance the role of the Academic advisor						
	Appoint an Academic advisor lead	l don't know how to contact my Academic advisor	Provide structured guidance to around running Academic advisor meetings	More support for being an Academic advisor			
	Provide Academic advisor training for the role	l am unaware of the Academic advisor role	Provide Academic advisor peer reviews and feedback	An academic advisor I would go to for university issues			
Post-it note comments	We need an Academic advisor definition and guidance	I wouldn't want to bother my Academic advisor with placement issues	l do not know how to contact my Academic advisor	Better guidance for students and staff about what the role involves			
Summary of Solutions	 Appoint an Academic advisor lead Offer Academic advisor training Clarify the purpose of the Academic advisor role Devise clearer guidance for students and staff around the role and remit of an Academic advisor Enhance student's awareness of the role of the Academic advisor via University intranet site Introduce a peer review and feedback process for the Academic advisor role similar to the peer review process for teaching and learning Ensure parity throughout the University around the title used to denote an Academic advisor, update all course/module materials and university webpages to use the same title for the Academic advisor Role 						

Table 11 - Priority 3, Clarify and enhance the role of the Academic advisor

8.7 Conclusion

In summary, this chapter has presented an overview of the two Co-design events conducted as part of this EBCD study, and the small co-design working groups where student and key stakeholder participants identified and collaboratively produced priority recommendations for improvements on the BSc Pre-registration (Adult) nursing course. Through a process of collaboration participants specifically identified and agreed three priorities including: facilitating and supporting student development of independent study skills, enhancing student support in the clinical practice placement environment and clarifying and enhancing the role of the Academic advisor. The activities in the two Co-design events served to synthesise some of the findings from the semi-structured interviews and emotional touchpoint mapping events in a solution focused way. The following chapter will discuss the overall findings from this study in relation to the wider theoretical discourse around student experience of a pre-registration nursing course, and the implication of this study for undergraduate nurse education and use of aspects of EBCD in higher education research.

CHAPTER NINE DISCUSSION

All principles by themselves are abstract, they become concrete only in the consequences which result from the application' (Dewey, 1938/2015, p.20)

9.1 Introduction to discussion

Chapters six and seven presented the results from the student and key stakeholder emotional touchpoint mapping events, whilst chapter eight described the three priority recommendations for course improvement. What follows is a discussion of the findings. The chapter is divided into two sections. The first section synthesises the overall findings from this study and situates the results from the semi-structured interviews, emotional touchpoint mapping events and Codesign events within the context of wider literature and theoretical insights on the student experience in higher education. The student experience of the BSc Pre-registration (Adult) nursing course will be contextualised by reviewing the aim and objective of this study in relation to the key findings and significant touchpoints that influenced the student experience. This is followed by an examination of the key findings and discussion on their contribution to existing knowledge and gaps in knowledge. Since a significant original aspect of this study is its use of EBCD as a research methodology to explore student experience, the second section of this chapter reflects on the features of EBCD that worked well, the challenges of using EBCD in a higher education setting and the implications of this study for EBCD use in future studies. Finally, implications of the study findings for Pre-registration nurse education and the limitations of this study will be discussed. Throughout this chapter the term 'this study' is utilised as self-referential phrase to indicate presentation of the findings from the research conducted as part of this thesis and does not refer to studies from the wider literature. Furthermore, as this study was conducted with a small number of students the findings cannot be generalised to all nursing students, therefore all references to 'students' in this chapter denote 'student participants' from this study and is not suggestive of the wider student population.

9.2 Contextualizing the findings on the student experience of the BSc Preregistration (Adult) nursing course

This Experienced Based Co-Design study has captured, explored and gained some understanding of the student nurse experience of the undergraduate Pre-registration (Adult) nursing course. At the beginning of this thesis the student experience was defined as a personal subjective phenomenon, elements of which could be understood when captured through personal narrative and the emotions described in that narrative (Bate & Robert, 2007a). This study has achieved the main objectives of identifying and exploring the emotional touchpoints that shaped student participants experience of the nursing course using semistructured interviews and emotional touchpoint mapping events with student and key stakeholder participants. Those emotional elements of the student experience serve to give the experience significance and quality, further influencing future experiences (Dewey, 1938/2015).

Exploration of the various emotional touchpoints along the student journey of the course revealed 6 of the 18 identified touchpoints were the most significant for both student and key stakeholder participants in this study including:

- The first clinical practice placement experience
- Mentor support in the clinical practice environment
- Facilitated learning sessions
- Module Assessment Group Support (MAGS)
- Academic advisor support
- Transition into year one/Welcome week

Overall, the core themes that emerged from these six touchpoints for the student participants related to their perceptions of 'support', and the 'influence of academic and clinical practice staff' on their experiences. Although some students describe experiencing good support throughout the course, a significant finding from this study is that most students have an expectation of much more support from academic and clinical practice placement staff than they report they had received. Expectations of more support were heightened during students transition to higher education study, transition into year two and year three of the course, and transition into the clinical practice environment. Unmet expectations around student support resulted in feelings of 'abandonment' as expressed by several students in this study.

Core themes that emerged from the key stakeholder event similarly related to issues around 'student support', as well as 'expectations of students' and 'expectations of themselves' as Academic advisors. By contrast key stakeholders perceived nursing students having access to lots of support mechanisms, but felt students were either unaware how to access available support or chose not to seek support. The findings in this study mirror that of Roberts et al. (2018) investigation of a range of institutional support provision that showed differences existed in students' awareness of support services, and non-use of student support services. This

suggests student support should be delivered in a meaningful and experience-enhancing way that ensures uptake (Smith 2007).

An overwhelming emotion expressed by most key stakeholders during the emotional touchpoint mapping event was 'sadness' that students felt 'abandoned' and unsupported'. Furthermore, some key stakeholders perceived a dichotomy with striking a balance between supporting students, particularly with their academic study skills, and encouraging students to become independent learners. Developing independent learning skills fosters higher order thinking (Roets & Maritz, 2017), which is essential for equipping nursing students for their graduate roles as patient advocates and evidence-based practitioners. Yet some students within this study appeared to locate their learning and development as an academic and clinical placement staff responsibility. Students' anxieties around independent learning related to the amount of independent learning, also referred to as self-directed study, required on the nursing course. Though the University provides support with independent study through Blackboard sites, assignment guidance, Module Assessment Group Support (MAGS) and central services such as WISER Academic support and Library services, several students indicated they still 'don't know how to learn!' and described themselves 'struggling with the same things' as a result. Although many students in this study honed their study skills quickly, those students who struggled to establish effective independent study skills early in the course continued to struggle well into year two and year three. As a result, several student narratives indicate those who either fail or perform more poorly than expected in at least one assessment considered leaving the course. Interestingly, many year two and year three students within this study expressed dissatisfaction with, and negative emotions related to, the amount of support they had received as the course progressed, irrespective of how well or poorly participants performed academically.

One interpretation of the findings above is that mechanisms for supporting students on the pre-registration nursing course have a critical influence on the overall student experience. Although support mechanisms are in place at the University some students do not know where or how to access the support, or how to engage with the support in a way that effectively helps them. Furthermore, other students perceive the support as insufficient, particularly as they progress through the course, and have an expectation of increased support to prepare them for professional qualification. On the other hand, key stakeholders have reported that students advancing through the course should typically require less support and exhibit more autonomy and independence as part of their professional development process. Dewey's (1938/2015) notion of 'experiential learning' explains these mismatched perceptions as arising out of the 'continuity and interaction' of the students and key stakeholders' experiences, culminating in

a positive or negative judgement about that experience as discussed in section 1.3 of this thesis. In other words, a current experience is not an isolated static event separated from other experiences, instead experience is a continuum with each new experience building on past experiences. Furthermore, interactions during a current situation will influence how individuals judge their present experience and their expectations of future experiences.

Therefore, many students within this study had previously attended Further Education (F.E.) colleges, or other learning environments, where they experienced different teaching and learning models to that encountered at university. Specifically, schools and colleges incorporate more guided learning and support (Money, et al., 2020) which can leave some students struggling to develop the higher order thinking skills necessary for independent study (Wingate, 2007). As students assimilate their previous experiences with their current experience (continuity) disparity between the two experiences becomes more evident. Within this study interactions experienced on the nursing course compounded this disparity further as participants described feeling 'abandoned' and unsupported, whilst key stakeholders believed striking the balance between 'support and empowerment' enabled students to 'think for themselves' and develop those necessary independent learning skills. Thus, a further interpretation of the findings from this study is that key stakeholders have an expectation students' take more active responsibility and control of their learning, something wider literature refers to as having more 'learner agency' (Mercer, 2012; Klemencic, 2015; Schoon, 2018; Vaughn, 2020). Whilst some student participants feel unprepared to manage their own learning independently and lacked, or perceived a lack of, 'learner agency' creating a feeling of dependency on the academic support of others for their learning experiences. A study by Titus and Roman (2019) purports learner agency as a critical component of positive learning experiences and the development of self-esteem for students on professional higher education courses.

It must be acknowledged that not all student nurses need, or expect, extensive support on nursing courses. Notably, some support needs are beyond the scope of university provision, potentially requiring students to take more personal responsibility. However, the findings above do suggest better support could improve students experiences on the nursing course overall. Moreover, as revealed in the introductory chapter of this thesis, following the introduction of student fees and increased competition across universities for student business there is a greater expectation that universities provide positive student experiences where possible. Therefore, the ensuing section will frame 'student support' in terms of understanding the concept of student support, establishing the student support needs identified in this study and identifying the mechanisms for supporting students on the pre-registration nursing course.

This is followed by a wider discussion of the mechanisms for supporting students as they relate to the findings from this study, therefore 'Support with transition onto the course', 'Support with transition through year two and year three of the course', 'Support with learning and assessment', 'Support with transition into the clinical practice placement environment', 'Clinical practice placement support through the mentorship model', 'Academic advisor support' and finally 'Supporting students develop learner agency'.

9.2.1 Framing student support

As numerous factors impact the nature of student support, understanding support from the perspective of students and support providers is crucial. Therefore, within this discussion student support is contextually framed through the lens of the student participants and key stakeholders in this study. Several students experienced effective support with some aspects of the nursing course, particularly the practical elements such as administrative support with the enrolment process. However, many students in this study perceived a shortfall in the support they had expected versus the support received with their academic studies on the course and with their learning and development in the clinical practice environment. Conceptualising what constitutes sufficient support that meets students' expectations is a challenge as it raises questions about what student's expectations and understanding of student support are, what influences those expectations and how students recognise support as it occurs? Added to this is the institutional challenge of offering generic support services that meets the individual needs of a diverse student population, whilst simultaneously striking the correct balance between support and 'hand-holding' (McInnis, 1998).

A further consideration when framing student support is the need to avoid positioning students as 'victims' that need help. Wider literature describes this emphasis on the shortcomings of the student with a problem to solve as a 'deficit model' of student support (Jacklin & Le Riche, 2009; Smit, 2012; Roberts et al., 2018). Smit (2012) argues the massification of higher education and increase in non-traditional students led to the growth of a school of thought that regards those students needing support as 'underprepared' and 'lacking' in the academic ability necessary to succeed. A danger of this belief is that ignores and deflects from the role higher education plays in perpetuating barriers to student success (Smit, 2012). To avoid a focus on perceived student inadequacies the remainder of this section seeks to frame student support through a positive lens with a focus on the essential features that constitute effective support experienced by students in this study, whilst recognising that they are a small sample of a very large cohort.

As espoused in chapter one of this thesis student experience is not restricted to the classroom but encompasses all aspects of student life, even extending beyond the university to include experiences within the clinical practice environment for participants in this study. Therefore, when seeking to discuss the study findings as they relate to the mechanisms for effective student support a model of student support that considers the whole student experience will be applied. In particular, Tait's (2000) 'Taxonomy of student support' remains instrumental in the development of concepts around student support (Jacklin & Le Riche, 2009), and provides a framework for supporting student's overall experience (Tait, 2015).

Tait (2000) proposes a taxonomy of student support that incorporates three interdependent and interrelated functions:

- Cognitive
- Affective
- Systemic

Within Tait's' taxonomy, cognitive student support relates to supporting students with their learning and assessment. Affective support relates to ensuring a supportive learning environment that generates student commitment and improves self-esteem (Tait, 2000). Systemic functions relate to ensuring efficient, student-friendly administrative processes (Tait, 2000). Tait's taxonomy mirrors some of the findings from this study. For instance, effective support for many students in this study incorporated cognitive elements, such as help with learning and assessment through the MAGS sessions. Systemic elements of support were evident through the expectation of administrative help for example with the process for applying for essay extensions. Affective student support was observable in students who reported gaining confidence when effectively supervised in clinical practice and supported academically. Affective student support seeks to develop student's well-being, self-confidence, self-esteem, personal agency and responsibility. Those students in this study who reported a lack of effective clinical supervision consistently described a reduction in confidence and overall well-being.

Findings from this study indicate a significant aspect of effective student support emerged out of the relationships and connections participants had developed whilst on the course. Jacklin and Le Riche (2009) report similar results in their study exploring students understanding of 'support' in which they identified interrelation support, where students had someone they could discuss their concerns with, as the most important type of support. Therefore, effective student support facilitates the formation of personal connections, interactions and communication with those in a position to offer support, i.e. with tutors, Academic advisors and peers. An example

from this study is found in several student accounts were having a 'one-on-one relationship' with their Academic advisor facilitated access to pastoral support during the more challenging times on the course. Notably, effective student support must also be accessible and available. Without accessibility and availability, cognitive, affective, systemic or relational student support cannot take place as was evident from the study findings some students did not know where or how to access support. One example from the findings of accessible and available student support was in the form of Home group meetings where Academic advisors would offer 'guidance' and 'address personal problems directly'. By contrast, student support was perceived as un-accessible, or unavailable, for some students when the approach to assessment support shifted from individual supervision to group supervision part way through the course. Tait's (2000) taxonomy of student support with the addition of relational support will form the framework around which the remainder of this discussion of the study findings is structured.

9.3 Cognitive student Support

Cognitive student support relates to supporting students with their learning (Jacklin & Le Riche, 2009). Cognitive support incorporates facilitating learning that helps students develop the skills and competencies necessary to meet their educational goals and pursue the postgraduate career (Kalamkarian & Karp, 2015). On the pre-registration nursing course student learning takes place in both an academic and clinical practice placement environment, with effective cognitive support necessary in both settings. For participants in this study the transition from one environment to another was perceived as a pivotal point (described as touchpoints in this study) on the course. Students who were supported well during these transitions reported more positive learning experiences overall. For example, one participant reports her first clinical placement as 'really amazing and such a good experience', stating 'they (the ward staff) made sure I was in the thick of things, and by the end of placement I felt very confident...my mentor was very busy but made sure I had two associate mentors if I wasn't with her'. The transition onto the course itself was a critical touchpoint for most participants in this study. One participant reported a lack of support during her transition onto the course as impacting her learning more negatively as she had felt 'there was a lack of support in some areas. If I didn't understand something you couldn't really contact somebody, you were a bit out on a limb really. I know its independent learning but coming into university is a big step, a different world and I just didn't feel like there was necessary support. The first year was the hardest... the adjustments? Four specific mechanisms to emerge from this study for supporting students during these pivotal touchpoints on the course were support with transition onto the course, support with transition through year two and year three of the

course, support with learning and assessment and support with transition into the clinical practice placement environment.

9.3.1 Support with transition onto the course

Study results suggest most students expected more support than they received during periods of significant transition through the nursing course. Within this study student transition denotes student movement or progression through the course marked by distinct changes or events, for example the movement to higher education. Many students experience changes to their personal circumstances, such as giving up full time work to study at higher education, and for some students' adjusting to the demands of university life and studies in the first year is difficult. Some student's transition into university is incremental requiring only small adjustments, for instance those students who have already studied at higher education level on a different course. Other students experience more radical transition, for example those students who have not studied at higher education level before or who have been 'out of education' for a long period of time. Within this study several students experience of the initial transition into higher education was like entering into a 'different world'. The current studies results echo findings from contemporary student transition literature which suggest all students experience some level of transition when entering higher education for the first time, although the level of that transition may be experienced differently by different students (Brinkworth et al., 2009, Gale & Parker, 2014; Cheng, 2015; O'Donnell et al., 2016; Turner et al., 2017).

Emotions associated with transition into year one for students in this study include feeling '*excited*', '*scared*', '*overwhelmed*' and even '*pure fear*'. These feelings are particularly evident when the first week (Welcome week) of the course is perceived as too '*busy*', '*hectic*' and '*chaotic*'. The study's findings are consistent with the themes identified in the literature review of this thesis, which suggest both the first weeks and first year at university are key periods requiring new students to adjust to intensified personal, lifestyle and social changes alongside the new academic demands during their transition into higher education (Birks et al., 2013; Smith et al., 2018). For some students the transition into higher education and adjustment to a new learning environment is difficult and challenging (Brinkworth et al., 2009), making it a critical point for risk of withdrawal (Turner et al., 2017). The results of this study show students who do not feel supported during this period struggle adjusting to year one on the course. Apparent lack of support leaves students with a sense of isolation and aloneness, as they are unsure who to approach for advice and support. Results from this study suggest student's

sense of support originates from their initial experience of Welcome week which will be explored in the following section.

The initial means for supporting early adjustment to university life and education at the time of this study was through students attending an induction and orientation week (referred to as Welcome week) at the beginning of the course. During Welcome week students meet key members of the course team such as their Course leader, Module leaders and Academic advisors. In addition, there is an emphasis on the significant points of contact for student support throughout the course, e.g. Study Skills Support and library services. For several students in this study too much information is relayed during Welcome week for them to process in any meaningful way. Evidently, it is difficult to recall some of the essential information around student support communicated during Welcome week when needed later in the course, resulting in students feeling unsure who to contact for help and support. Information overload occurs when an excessive amount of information is communicated resulting in information saturation whereby an individual is unable to process any further information (Khalid et al., 2016). Irrelevant, incomplete and inaccurate information exacerbates information overload subsequently leading to an information retrieval problem (Grimshaw, 2008), as expressed in some student narratives within this study. The information overload that occurs at the beginning of the course contributes to some students experiencing a more challenging transition. Ineffective transition early in the course leads to academic underachievement, student dissatisfaction and risk of student attrition (Vinson et al., 2010; Turner et al., 2017), whereas effective transition increases student experience, satisfaction and retention on the course (Burnett, 2007).

The findings around induction week above have been reflected, and debated, in the research for over 25 years (Carpmael et al., 1992; Gale & Parker, 2014; Barton, 2017). However, an unexpected finding from this study is how some student groups have a misconception about the purpose of the induction week, resulting in one student from the study choosing not to attend. Within the pre-registration nursing course the term 'Welcome week' has replaced what was traditionally called 'Freshers week' that takes place across most universities. Although Welcome week shares some similarities with Freshers week in that usually both are a period set aside for inducting students into university life, giving students an opportunity to register with academic services (i.e. course and library services), to enrol with different groups or societies of interest (i.e. sports clubs) and attend introductory lectures in order to meet key staff and other students on the course. Welcome week within Pre-registration nursing has minimal focus on social aspects of student life, whereas a major feature of Freshers week was the many organised social events students could attend. Notably the nursing course attracts

large numbers of non-traditional students including mature students and Black, Asian and minority ethnic (BAME) individuals who may practice abstinence from alcohol as part of their faith who, based on pre-existing assumptions of Freshers week, may feel Welcome week is not designed to meet their needs. One student participant in this study chose not to attend welcome reporting '*I thought it was just for people that drink, and I don't drink, so I wouldn't be able to get involved, so there's no point going*'. Students who choose not to attend Welcome week based on a misconception of its purpose forgo the chance to prepare for the rigours of the course and to form initial bonds with their peers and Academic advisors potentially lasting the next three years.

The need to change the labelling of Welcome week to reflect its importance emerged from the Key stakeholders' findings. Furthermore, this study highlights the need to ensure the information related to this pivotal week, such as the content of the week, is packaged and communicated to new students in a more effective way. Study results expose a potential lack of consideration for the needs of mature students and students from the BAME community when communicating about Welcome week. An examination of the wider literature reveals a distinct lack of research on the effectiveness of Welcome week in supporting students from the BAME community transition to higher education, and currently no published study specifically explores BAME students' perceptions and experiences of Welcome week, suggesting a gap in knowledge. Although the findings above are based on one student's experience and cannot be generalised to the general BAME population, this study does make a small but significant contribution towards the knowledge on both the role of Welcome week in supporting students transition into higher education and the potential perceptions and experiences of BAME students.

The use of a 'Home group' system during students' initial transition into higher education, and throughout the nursing course, appeared to be one pragmatic approach to student support. Home groups are field specific small groups of students (usually 25), apportioned from the larger nursing cohort. Home groups differ from other small class teaching approaches by having an Academic advisor who oversees the group and provides pastoral and academic support for the 3-year duration of the nursing course. This results in nursing students having a specific academic point of contact and a consistent group of peers throughout their education. For many students in this study regular Home group meetings held throughout the academic year acted as an accessible resource for student support where most Academic advisors would carry out welfare checks on their students by asking '*how students were doing*' and offering '*guidance*'. Jacklin and Robinson (2007) suggest student and Academic advisor contact is vital and emerged as one of the strongest supportive structures identified by

students in their study, alongside interpersonal relationships with their peers. For all but two students in this study their Home group experience fostered the formation of friendships early on the course, something perceived as quite difficult in class sizes of several hundred students. In turn those friendships became useful support networks that provided many pragmatic benefits academically, socially and emotionally as the course progressed. For example, several students described car sharing with their Home group peers and forming study groups during assessment times. Almost all students in this study acknowledged the importance of Home group peers' emotional support helping them continue through the course. One interpretation of this finding is that the concept of a 'home' group creates a sense of belonging in nursing students, with the Home group forming a base, or foundation, students always return to following time spent in large key lectures and in the clinical practice placement environment.

Helping students to develop a sense of belonging through peer support has also been reported in studies by Chapman and Orb (2001), Birks et al. (2013), O'Keeffe (2013), Grobecker (2016) and Freeman et al. (2017). O'Keeffe (2013) reports the development of a sense of belonging within higher education as a critical influence on student retention. For Grobecker (2016) a sense of belonging is a 'fundamental human need' that positively shapes students' confidence, motivation and learning. Research conducted as far back as Tinto (1987) acknowledges the significance of peer relationships on student's social integration and adjustment to higher education. As identified in chapter 2 of this study, a shared sense of belonging with peers appears to be a 'coping strategy' for nursing students, as they perceive that only student nurses can fully understand the experiences of other student nurses (Chapman & Orb, 2001) consequently making them a strong resource for support (Birks et al., 2013). Although lots of literature around the importance of supporting students develop 'a sense of belonging' exist, research exploring the use of a Home group approach to support students in higher education is limited making it difficult to compare the results of this study with the wider literature. Only one study by Juutilainen et al. (2018) explores the experiences of the Home group approach for supporting first year student teachers. Juutilainen et al. (2018) found that Home groups were perceived as 'emotionally safe' environment's where students develop their identity and agency. Although Juutilainen et al. (2018) research is focused on student teachers in Finland and the effect of Home groups for fostering student agency, their results do reflect some of the findings from this study. For example, Juutilainen et al. (2018, p.122) suggests as students "get to know each other at a profound level", that deep relationship facilitates emotional safety and a sense of freedom to engage more fully with learning on the course.

The positive effects of forming deep relationships with peers from their Home groups was evident within this study as seven student participants specifically identified the support they received from peers in their Home groups as instrumental in helping them emotionally and academically through the course. Several students frequently used comments like '*I* couldn't have done it without them', 'without our home group there's no way I would have got through this' and 'they become like a little family' to depict the significant role peers within Home groups played in supporting them on and through the nursing course. In summary the findings from the current study adds valuable insight into the benefit of Home groups for providing support to nursing students during their initial transition into higher education, particularly during the induction week, and throughout the nursing course. In particular, allocating students on courses with large student numbers, or class sizes, to smaller Home groups has been shown to have applied benefits, such as fostering useful friendships and facilitating a sense of belonging.

9.3.2 Support with transition through year two and year three of the course

Supporting students during the early transition period of their course has clearly been established in the previous section as pivotal to the student experience of, and satisfaction with, the nursing course. In order to effectively support students' universities must have some understanding of student transition, what it is, how it is experienced and the kind of support mechanisms that need to be in place to facilitate successful transition. However, a report by the Higher Education Academy (HEA) indicates current theories around transition to higher education fail to fully explain the transition process or develop sufficient evidence-based interventions for assisting new undergraduates through the process (O'Donnell et al., 2016). Furthermore, 'student transitions' as a field of study has failed to explicitly define 'transition', as such there remains no single definitional consensus, instead 'transition theory' relies on "implicitly held assumptions" around what transition is, why and how it occurs (O'Donnell et al., 2016, p.5).

Although there is a growing number of studies on transition of the general adult learner in higher education, much of the literature focuses extensively on the first-year experience as evidenced by Cheng (2015) whose report outlines six models of student transition, five of which relate exclusively to first year students transition early in the course. Whittle (2018) believes the concerns around student retention is what drives the current focus on the first year transition experiences. However, results from this study identified the transition to year two and year three as significant touchpoints for student participants, comparable to the transition into year one. This signifies student transition is a continuous process composed of

many phases across the whole course and highlights the need for a transition model focused on the experiences of year two and year three student nurse experience.

Some participants in this study found transition into year two of the nursing course a shock. For both second year and third year students the move to higher level study is more challenging than they expect, i.e. from level 4 to 5 and level 5 to 6. How well-prepared students feel for the elevated level of academic study positively or negatively affects transition through each year. Students who feel unprepared for year two and year three experience the most difficulty adjusting to the advanced level of study. Consequently, students who struggle in year two are most likely to contemplate leaving the course. Yet published literature currently focuses on first year students' preparedness for level 4 study, and the effect on academic performance, with few studies exploring challenges around the increased demands of studying at level 5 and level 6 (Moreton et al., 2017; Ayala & Manzano, 2018). CooperGibson Research (2018) suggests there is no standard definition of what constitutes level 4, 5 or 6 study in higher education, and no standard understanding of the nature of academic study at each level of education documented in the wider literature. Consequently, students entering consecutive years of study unequipped with clear knowledge of what to expect, or what is expected of them, anticipate support with adjusting to the advanced level of academic study. Unmet support needs appeared to have contributed to students' feelings of 'abandonment' as expressed by some participants in this study.

Several studies as far back as 1993 have cited a 'sense of abandonment' in student nurses, however this tended to relate to feeling abandoned by nurse academics in the clinical practice environment (Beck, 1993; Price et al., 2011; Heinonen et al., 2019). This study also found some students felt abandoned in the clinical practice placement area. For student participants abandonment was associated with 'left on their own' to manage poorly patients. However, this study appears to be the first to document students using the term 'abandonment' in relation to their academic studies in year two and year three, originating out of their anxiety around 'being left' to do more self-directed study without support the further they progressed through the course. An interesting study by Norrie and Dalby (2007) found nursing students exhibited less adult learner characteristics, such as being active and self-directed in their learning, as they progressed through their studies, something they term 'retrograde and ragogy'. Norrie and Dalby (2007, p.327) suggest third year students potentially need more support as they become "tired and jaded" with their studies, resulting in less willingness to engage in independent learning. The concept of 'retrograde and ragogy' could be used to explain why students' in the current study expected more support in year two and year three of the nursing course and experience a sense of abandonment when those expectations of support are not met.

However, the results from this study show many students struggle with independent learning from the first year of the course and some student persistently struggle throughout the remaining years of study, not just in their third year. Although this study was conducted with a small number of students and thus the findings cannot be generalised to the whole nursing student population, results could potentially have implications for the current curriculum constructed around assumptions of student's intrinsic andragogical characteristics. Furthermore, as the nursing curriculum, and undergraduate study in general, moves towards a more heutagogy instructional approach, the imperative to support students to become self-determined learners is more evident.

9.3.3 Support with learning and assessment

A finding from this study was that most student participants 'struggled' with the academic elements of the course at some time during their studies. For many students this is at the beginning of the course and throughout the first year of study, whilst some continue to struggle well into year two and year three. Interestingly, most students in this study expressed a level of dissatisfaction with the support for learning and assessment, including those students who had not identified themselves as struggling academically at any point in their studies. Students who articulate an expectation of individual supervision with assessments, and more taught contact time, express the most dissatisfaction with their experience of learning on the nursing course. Many student participants articulated struggling with the amount of independent study on the course with one participant commenting 'there is so much self-directed study, but if you don't know how to be doing that, that's a big thing!'. Without effective support and guidance some participants continue to struggle with the same issues around self-directed 'a lot of people don't know how to learn; I don't know how to learn! It lowers your confidence because I keep struggling with the same things because I don't know how to improve'. A critical review by Barker et al. (2016, p.22) suggests Pre-registration nursing students' inability to undertake independent study remains unaddressed within higher education due to academics "misguided assumption that all students (by the fact that they are adults) are prepared and motivated to direct their own learning".

The findings of the current study are consistent with the wider literature with several studies indicating many higher education students retain a dependent learning style developed during their earlier school experiences and prefer the more traditional didactic approach of teacher instruction over independent study (Timmins, 2008). Reasons for this include students need for physical and social interaction with their teacher and feeling unready or unable to self-regulate their own learning (Bingen et al., 2019). The more active learning pedagogy featuring independent study that has become a core tenet of undergraduate nurse education can lead

to increased student dissatisfaction. A study by Ferreri and O'Connor (2013) aimed at shifting the foundational content of their pharmacology course to independent study resulted in more negative comments in the course evaluations, even though the overall student grades had improved. Similar results were found in a systematic review by Betihavas et al. (2016, p.19) that showed two of the five studies reviewed reported negative comments and student dissatisfaction with the amount of "out-of-class preparation time". Other studies also reported a tension between academic performance and student satisfaction with their experience (Horn, 2013; Tune et al., 2013). Paradoxically, the increased amount of independent learning described in the current study may have improved some student grades whilst simultaneously reducing student's satisfaction with their experience of the nursing course. The results of this study suggest nurse academics can find themselves at an impasse when trying to both meet learner's needs to succeed in their studies and meet their expectations around the student experience. The challenge of balancing independent study with student satisfaction exists beyond nursing courses and requires academics to understand the benefits of and barriers to independent study.

9.3.3.1 Facilitating student development of independent study skills

Learning by oneself, or non-taught time, where students are expected to direct their own learning is a key component of all UK pre-registration nursing course, with various names designated to the activity including 'self-directed study', 'directed study time' (Barker et al., 2016), 'self-directed learning', 'independent learning' and 'independent study' (Tekkol & Demirel, 2018). Independent study correlates with upper-level thinking, problem solving, critical thinking and lifelong learning skills (Tekkol & Demirel, 2018), which are qualities required of student nurses as indicated in the NMC standards of proficiency (NMC, 2015). However, most student participants, and several key stakeholders, within this study felt there was 'too much self-directed study' on the nursing course. Though independent study is understood in the wider literature to improve individuals' self-confidence, autonomy, motivation and lifelong learning skills (Tekkol & Demirel, 2018), conversely for students in this study who 'don't know how to learn' the amount of independent study was perceived to '*lower confidence*'.

In keeping with the results of this study several factors were found by Barker et al. (2016) to influence students experience of independent study including ill-preparation for, orientation to and lack of the necessary skills to do independent study (Barker et al., 2016). Barker et al. (2016) believes the assumption student nurses possess the skills and motivation to effectively direct their own learning emanated from the move to higher education from schools of nursing. When 'nurse training' was based in schools of nursing located on NHS hospital sites, teacher-

centred methods of content delivery incorporating rote learning tended to be employed (Handwerker, 2012) with minimal apportioned independent learning time (Barker et al., 2016). Once Pre-registration nurse education transferred exclusively to higher education settings in the early 1990s, teaching and learning strategies were adapted to meet university requirements of more active learning, including the incorporation of independent study time (Barker et al., 2016). Thus, the move to higher education resulted in a paradigm shift, with the responsibility for learning transferring from external sources such as lecturers to an internal source, therefore the individual learner (Tekkol & Demirel, 2018). In their critical review, Barker et al. (2016) questions nurse lecturers' capacity to support student nurses developing independent study skills. Booth et al. (2016) suggests operating within two distinct disciplines 'education and nursing' does not result in expertise in both, which can explain why some nurse educators find it a challenge to effectively facilitate student independent study. Notably, findings from this study suggest students were mostly satisfied with their overall experience with academics on the nursing course, indicating that operating as a nurse and educationalist does not preclude expertise in both as stated by Booth et al. (2016) above. However, results show there was distinct dissatisfaction with support for independent study.

Although many nurse educators recognise the benefits of independent study, confusion exists regarding its use and application resulting in inappropriate and inconsistent use of independent study time on nursing courses (Timmins, 2008). Literature also reports the view learning on professional nursing courses differs from traditional adult learning as a result of the predetermined learning outcomes set by nursing's professional body, the NMC (Walsh, 2004; Timmins, 2008). Contemporary nurse education prepares students to meet the challenges of a dynamic and constantly changing healthcare environment requiring new graduates to be fit for purpose (Pryce-Miller & Serrant, 2019). The need for nurses to demonstrate their fitness for purpose by meeting specific NMC standards of proficiency requires a more didactic pedagogical approach than other non-professional university courses (Walsh, 2004). This can result in disparate approaches to facilitating learning and encouraging student independent study amongst some nurse educators. This was evident in the key stakeholder findings which suggest a lack of consistency among academics with some perceived as 'over delivering' and raising students 'expectations beyond what can be delivered. A perception of 'under delivery' of course content by some lecturers may contribute to students' feelings of dissatisfaction on the course. Study results indicate some lecturers find it hard to achieve the correct balance between 'delivering learning' and supporting students develop independent learning skills.

Facilitating student development of independent study skills was identified as an improvement priority on the nursing course during the Co-design event by student and key stakeholder participants in this study. Several recommendations were made that may help academics consistently support students develop independent study skills including, providing study skills guidelines and top-tips, academic study drop-in sessions, student led study support networks and academic study summer camps prior to students commencing their university studies. Wider literature suggests higher education institutions are still a long way from the shift to a self-directed learning paradigm (Guglielmino, 2013; Van Rensburg & Botma, 2015), and Morris (2018) suggests there is a distinct lack of studies investigating the extent to which selfdirected learning is facilitated within a higher education context. However, Bailey et al. (2019) recently published a collection of papers directly addressing self-directed learning pedagogies and provides examples of useful models for promoting and implementing self-directed study. Several papers (Bosch et al., 2019; Johnson & Johnson, 2019; Van Zyl & Mentz, 2019) advocate a collaborative approach to managing independent learning as proposed by the findings and priorities for course improvement identified in this study, suggesting the findings from this study are in line with contemporary literature around independent study in higher education. Example recommendations from this study of collaborative approaches to supporting independent learning include establishing student led support networks focused on study skills and independent learning and setting up academic study skills drop-in sessions where together students and staff can explore effective methods and practices for undertaking independent learning. Effective learning manifests when learners co-create their educational experiences by taking ownership of their learning (Gros & López, 2016). However, more research needs to be conducted exploring the effectiveness of implementing strategies for supporting independent study in nurse education.

Key stakeholder participants within this study indicated student expectations around independent learning were not being sufficiently managed. Several key stakeholders suggested the 'mismatch between staff perceptions and perception of students' of independent learning contributed to students' disappointment with support at the university, despite the 'significant efforts' by academic staff to support students with their learning. This notion of disparate expectations between students and academics is reflected in the wider published literature (Laing et al., 2005; Borghi et al., 2016; Dowell et al., 2016; Money et al., 2017). Both Laing et al. (2005) and Money et al. (2017) suggest students hold unrealistic expectations of teaching, learning and assessment, that derive from their prior learning experiences, life experiences and views of higher education (Laing et al., 2005). Dowell et al. (2016) indicates this disparity is contingent upon students' assumptions around the universities responsibility towards them compared with their sense of responsibility towards

198

their own learning experience. Adopting strategies for closing the gap between students and academics expectations is critical to support students transition better into higher education (Money et al., 2017). Most key stakeholders in this study agreed more work could be done with potential students' pre-arrival at university, for example through working closer with partner colleges to prepare students better for the transition to higher education and exploring potential students' perceptions of higher education study.

A final consideration when discussing the findings around independent learning in this study is the influence of student demographics, particularly the recruitment of more non-traditional students on the pre-registration nursing course. In a review of the research on student support in higher education Smith (2007) found non-traditional students were most likely to experience difficulties with independent study and the lack of supervision espoused in universities. Nontraditional students include students from low-income families, first generation students, mature students, students with a disability and students from minority ethnic groups (Cotton et al., 2017). The higher education widening participation agenda of recent decades seeks to close the gap in take-up of higher education between different traditional and underrepresented non-traditional student groups (Connell-Smith & Hubble, 2018). Within nurse education the need to meet the widening participation agenda along with the necessity to reflect a diverse healthcare workforce (Heaslip et al., 2017) has seen an increased recruitment of older students, men and Black and Minority Ethnic (BAME) students, alongside more students with non-traditional entry qualifications (Thomas & Duckworth, 2018). As all but all but one student identified as non-traditional within this study the findings related to students experience of independent study cannot be compared between the traditional and nontraditional student participants. Furthermore, of the three students who identified themselves as BAME in this study only one recounted struggling with independent study and a perceived 'lack of support', making it difficult to contrast their experiences of independent study with that of non-BAME students.

9.3.4 Support with transition into the clinical practice placement environment

The first clinical practice placement experience was a pivotal touchpoint on the journey through the course for students in this study. One participant recalls her first placement as 'quite a shock for me going into a general medical ward that was extremely busy, that was quite scary'. Although many participants in the present study expressed feeling 'daunted', 'scared' or 'shocked' by their first clinical placement, but a surprising finding was that both students with and without prior health care experience expressed feelings of anxiety around the first clinical practice placement on the course. Clearly having prior care experience does

not appear to alleviate fear or offer extra reassurance for students attending their clinical practice placement for the first time. A suggested reason for this is that 'experienced' carers perceive a distinct difference between their previous role of healthcare assistant and their new role as a student nurse.

A plethora of nursing literature has found the transition from university to clinical practice as the most challenging and stressful component of the pre-registration nursing course (Sharif & Masoumi, 2005; Andrews et al., 2006; Cooper et al., 2015). Research suggests the initial transition onto the first clinical experience is particularly associated with high levels of stress and anxiety (Andrew, 2013; Crombie et al., 2013; Hart & Swenty, 2016). The student experience of the first clinical practice placement is a key determinant of an undergraduate nurses' success on the course with the quality of the initial clinical experience setting the benchmark for subsequent experiences (Andrew, 2013). Cooper et al. (2015) concurs the first clinical placement experience as the most powerful and influential on undergraduate nurses. A catalyst for students leaving nursing education early in the programme can be a negative initial clinical experience (Andrew, 2013; Crombie et al., 2013; Cooper et al., 2015). Whereas successful transition to clinical practice is associated with a positive clinical experience and the movement towards becoming a competent nurse (Hart & Swenty, 2016). A project by Health Education England (HEE) (2018a) focused on reducing Pre-registration attrition and improving retention, suggest healthcare students' feeling unprepared for the clinical experience in the first year of study was amongst the top three reasons for leaving the course.

An effective way to facilitate a seamless transition to the clinical environment is through supporting students to feel better prepared for their first clinical practice placement. Although many participants in this study described feeling unprepared for placement, how quickly students overcame those feelings was dependent on the level of support students received whilst acclimatizing to the clinical practice placement environment. Stark differences were evident in the student participants experiences of support and supernumerary status. Students without prior nursing care experience gained confidence from working in a supernumerary capacity under close supervision from an allocated mentor, making them feel better prepared for subsequent clinical placements. Similarly, for some students a non-supernumerary status on the first clinical placement augments prior care experience, providing they are closely supported by a mentor or associate mentor. On the other hand, students who are expected to work in a non-supernumerary capacity without mentor support on their first clinical practice placement felt negatively overwhelmed by that experience. Juxtaposing these two experiences reveals the level of mentor support significantly influences the overall student experience of their first clinical practice placement irrespective of supernumerary status. This

200

finding strengthens the argument that mentor support in the clinical practice placement environment is a significant touchpoint on the pre-registration nursing course.

Notably, as students progressed through the course they appeared to favour more nonsupernumerary engagement in nursing practice but continued to value direction and support from mentors in the clinical practice environment. This was evident from several second year and third year nurses' narratives who described how being allowed a 'free rein to do more' alongside being 'strongly advised and guided' developed their understanding of patient care and confidence in their role. The findings above suggest there is no one size fits all approach to mentoring. Instead, this study proposes a model of mentorship that facilitates an individualised approach to managing student nurses in the clinical environment based on a students' year of study, previous health care experience and learning and development needs.

9.3.4.1 Preparing students for the clinical practice placement experience

A way to prepare students better for the clinical environment is by establishing environments and situations where students can develop their self-efficacy. Self-efficacy relates to student's belief about their own ability to initiate successfully actions when performing different tasks in specific contexts (Bandura, 1997). Facilitated classroom learning provides a setting where students can practice their nursing skills alongside developing their knowledge, and competence before being exposed to 'real life' clinical situations. Yet as previously indicated, students in this study described receiving far less taught contact time than they had expected, specifically clinical skills teaching on the nursing course. The fewer than expected number of taught skills sessions, also referred to as 'simulated practice' or 'simulation', appeared to contribute to student's anxiety around competency levels and feelings of being unprepared for the clinical practice placement environment. Key stakeholder participants also noted a reduction in simulation and taught clinical skills on the course over the last few years, suggesting the quality of the nursing programme has been 'diluted' as a result, and giving credence to students concerns. The findings above are in keeping with the wider literature which suggest hands-on practice contributes to the development of clinical competency and lack of preparation for clinical practice induces anxiety and worry (Strand et al., 2009; Onda, 2012). Preparation for practice begins with students acquiring generalised nursing knowledge and skills in the university academic environment that they prepare to transfer and consolidate in the clinical placement environment (Scully, 2011). Although, literature on nurse education has historically purported a gap between theory taught on nursing programmes and clinical practice (Scully, 2011), contemporary research suggests a closing of the theory-practice gap can be achieved by incorporating more situated learning strategies within nurse education (Jeppesen et al., 2017). Situated learning is employed throughout nursing curricula in the form of clinical practice placement experiences and 'simulation' or 'simulated clinical practice' in the university setting (Fetter, 2009; Onda, 2012; Jeppesen et al., 2017). Situated learning in the form of 'simulation' is viewed as a technique that replaces, or replicates, aspects of real-world experiences by fully immersing and guiding learners through interactive tasks or settings they would encounter in actual clinical practice environments (Gaba, 2007). Therefore, simulation enables students to connect theory with clinical practice experience in a controlled supervised environment such as a classroom setting or skills laboratory. As such, simulation is recognised as fundamental to the development of competency and preparation for the clinical practice placement environment (Fetter, 2009).

Essentially simulation facilitates student learning, skills practice and competency development in a safe environment guided by educators, without negatively impacting real patients (Aebersold, 2018; Camp & Legge, 2018; Palominos et al., 2019). A growing body of research indicates simulation increases students feeling of being more prepared for the clinical practice environment (Kuiper et al., 2008; Holland et al., 2013; Richardson & Claman, 2014). Furthermore, students value the opportunity to develop their cognitive, affective, and psychomotor skills in a safe, nonthreatening learning environment (Woolley & Jarvis, 2007; Strand et al., 2009; Labrague et al., 2019; Palominos et al., 2019). In a study by Jeppesen et al. (2017) student nurses perceived skills teaching through simulation to be more effective and meaningful, as it develops their critical thinking and situated cognition skills and stimulates motivation to learn, a sentient articulated by several participants in this study.

Simulation in various formats has been incorporated into nurse education for decades (Lavoie & Clarke, 2017). In 2007 the NMC (the governing body that regulate Pre-registration nursing) introduced a baseline standard for simulation which included a 300-hour cap on simulated practice learning within Pre-registration nursing programmes (NMC, 2007). However, that cap has since been removed allowing more flexibility for higher education institutes to determine the type and amount of simulation to include in their nursing programmes (NMC, 2018b). However, at the time of this study no system was in place to monitor the amount, quality, application or timing of simulated learning within nursing programmes once a course is validated by the NMC. Given that the study findings show student participants associated the amount of taught clinical skills sessions delivered on the course with a (perceived) level of clinical practice competency, and evidence from the wider literature suggests the inclusion of more simulated teaching can help prepare students better for the clinical learning environment, concerns around the amount and timing of simulation raised in this study cannot be ignored.

9.4 Relational student Support

The findings in this study suggest most students valued the relational interactive elements of the nursing course that enabled the formation of personal connections with peers, academic and clinical practice staff. These relationships played a significant role in students' socialisation and development of their professional identities. For example, several students described lecturer's 'knowledge', 'guidance' and 'approachability', and their mentor's 'quidance' and 'support' as preparing them well for the requirements of the course and registration as a qualified nurse. In addition, three student participants shared stories about specific occasions when their Academic advisors positively influenced their experience on the nursing course. A common theme in all the stories was the personal interactions that occurred with accessible and available academic and clinical practice staff at a time when each student needed them. Two students describe a sense of feeling 'cared for' when Academic advisors inquired about their well-being and offered practical help and advice with things they had been struggling with. Hawk (2017) notes a specific distinction between being 'cared for' and 'cared about' in the student-lecturer relationship. Being 'cared for' entails a level of engagement with or specific action by the one caring for, as described by the student participants above, this is only possible when students have a level of relationship and interaction with staff members.

In their final report exploring the factors impacting healthcare student attrition and retention HEE (2018b) use Tinto's 'model of student retention' to demonstrate a link between the students' relationship with the institution and their success on the course. Notably, an effective relationship between student and institution resulted in increased student retention, and consequently attrition was an indication of a poor relationship (HEE, 2018b). Similar findings are evident in the wider literature suggesting the greatest supportive structure in education is the development of interpersonal relationships, for example student to student, student to academic staff or student to mentor (Lowe, 2005; Jacklin & Robinson, 2007; Keeling, 2014; Guzzardo et al., 2021). Keeling (2014, p.142) specifically describes educational institutions as "structured communities of people" that require staff to "make and sustain responsive, empathic relationships with students". Similarly, Guzzardo et al. (2021) suggests alongside increased retention it those student-faculty relationships and interactions that are conducive to students' learning experiences. Two specific mechanisms to emerge from this study for facilitating relational student support is through the mentorship model in the clinical practice environment and Academic advisor support in the university environment.

9.4.1 Clinical practice placement support through the mentorship model

Mentors shape student learning and development and influenced future career decisions. Student participants express a mixture of both favourable and negative experiences. Notably, positive mentoring experiences resulted from mentors supporting students in the clinical practice environment by working alongside students and facilitating lots of learning opportunities. Good mentors were perceived as those who engage and motivate students in clinical learning. When unavailable to support students themselves, due to absence or workload constraints, good mentors ensured continuous support for students via other team members such as associate mentors as evident in one student narrative '(My mentor) was a Sister. She made sure I had two associate mentors so if I wasn't with her I was with my associate mentors and all the nurses on that unit were brilliant'. Conversely, participants associate negative mentoring experiences with unavailable or disinterested mentors who made students feel like a low priority or even a burden. Several participants described experiencing conflicts with mentors that left them feeling either 'intimidated and bullied' or that the clinical placement was 'untenable'. For example, one participant describes 'a strong clash with my mentor whilst on placement to the point where I felt intimidated and bullied! When I informed the PEF at the end of placement review they were shocked and said they would "sort it". When I returned to placement my mentor then made it evident that they had been informed about what I had said!

The findings above are consistent with the themes identified in the literature review of this thesis and add to the current body of knowledge which attest to the significance of a mentor on the student experience of clinical practice placement and the undergraduate nurse's journey (James & Chapman, 2010; Courtney-Pratt et al., 2012; Papastavrou et al., 2016). Numerous studies indicate the mentorship relationship is the most influential factor on student satisfaction, attrition and the development of clinical competency (Hamshire et al., 2012; Cooper et al., 2015; Dimitriadou et al., 2015; Papastavrou et al., 2016; Doyle et al., 2017). Andrew (2013) describes the mentoring experience as a 'make or break' component of students' first clinical placement (Andrew, 2013). Other studies illustrate a strong link between good mentoring experiences and student learning and success, satisfaction and self-confidence, psychological health and well-being and preparedness for registration (Ousey, 2009; Eby et al., 2010; Kaihlanen et al., 2013; Eller et al., 2014; Papastavrou et al., 2016).

The mentorship model of Pre-registration student nurse supervision was formally embedded in clinical practice during the 1990s (Rylance et al., 2017) and became a mandatory requirement within the UK in 2007 following the release of NMC (2006) '*Standards to support learning and assessment in practice*' (Ousey, 2009). Since then all undergraduate student nurses were allocated a named mentor, or associate mentor, whom they work alongside for a minimum of 40% of their time in the clinical practice placement environment (NMC, 2018c). Until recently all registered nurses were required to complete an approved mentorship training programme and fulfil the role of mentor (Clark & Casey, 2016). The NMC (2008, p.56) define a mentor as a registered nurse who "facilitates learning and supervises and assesses students in a practice setting". The key responsibilities of mentor include directly managing student learning in practice by supporting, guiding, observing and facilitating students' clinical practice, skills, knowledge development, learning and professional growth (NMC, 2008; Ousey, 2009).

To undertake the responsibilities of mentor the NMC (2008, p.39) suggest mentors "will need time, when undertaking work with a student, to be able to explain, question, assess performance and provide feedback to the student in a meaningful way". However, the NMC refrains from adopting a standard strategy or guideline for managing or funding protected time for mentors (Clark & Casey, 2016). Consequently, mentors are not allocated protected time or remuneration to fulfil the role, instead supporting students is additional to their normal workload (Clark & Casey, 2016). Hunt et al. (2016b) describe mentorship as a system reinforced by informal support that functions on the goodwill of mentors. This can result in inconsistency amongst mentorship experiences, which was a frustrating and common theme reported by student participants in this study.

Given the importance of mentors for supporting students with their transition into the clinical practice environment, facilitating student competency development and socialising students into the profession of nursing the abolition of 'mentors' which took effect from September 2019 appears to contrast with the evidence gathered from this study. Changes to the 'mentorship model' arose out of the necessity to usher in a new approach to support student learning and assessment in practice following a succession of negative national enguiries and reports on the quality of nurse education (Willis, 2015; Morley et al., 2019). New NMC standards has seen mentors replaced with 'practice supervisors' and 'practice assessors' (NMC, 2018a). The NMCs intention was to implement a multi-professional team approach to student learning with all members of the healthcare team assuming responsibility for a students' ongoing professional development irrespective of discipline (Morley et al., 2019). Although the new approach endorses a team approach to teaching and learning the practical need for clinical supervision and assessment remains. Therefore, under the new structure student supervision and assessment is divided into three distinct roles; the Practice supervisor, Practice assessor and Academic assessor (Heath, 2019) and the requirement for students to spend 40% of their time in clinical practice with their mentor has been removed (Royal-Fearn, 2019).
Currently it is too early to tell if these fundamental changes will benefit or impede the student experience of clinical practice learning and requires evaluation. Change processes in large organisations such as the NHS take time to roll out and embed. However, the findings from this study augment the need for students to consistently have access to a named person to advise, support and guide students into and through their clinical practice placement experience. The current danger is that students may feel disadvantaged by the new system whilst it is being embedded, and whilst mentors'/practice assessors/clinical supervisors get to grips with their new titles and responsibilities.

9.4.2 Academic advisor support

Findings from the present study suggest students have wide-ranging experiences with their Academic advisors both positive and negative. Students perceive positive experiences as those where Academic advisors take an active interest in student's personal welfare and academic progression. Familiarity and a personal connection with an Academic advisor results in students feeling comfortable approaching their Academic advisor for help and advice when needed as evident from the extract below:

My Academic advisor supported me when I went through a bit of a struggle at one point. I was thinking 'I don't think I can continue all of this at one time' and he was really good. He sat me down and asked if I needed any extensions on things and stuff, he was saying this is your future and it's affecting it so you need to take that opportunity, and he really helped me out at that point. If I hadn't had that Home group then I don't think I would have just gone to any lecturer and just said "I'm struggling", so the Home group is perfect for that really, just having that one-on-one relationship with a tutor (SP07)

By contrast factors that impede a good student experience include students' being unfamiliar with, or unaware of, their Academic advisor and Academic advisors showing disinterest in a student's welfare. Academic advising, also referred to as personal tutoring by students and key stakeholders in this study, is a widely used system for providing support, guidance and advice for students within higher education (latrellis et al., 2017; Chan et al., 2019) and is a vital function of higher education, with a direct link between effective advising, higher student retention and positive student satisfaction evident in the literature (Battin, 2014; Zarges et al., 2018; Larson et al., 2018).

Findings from this study indicate that fostering a good relationship between Advisor and Advisee is a prerequisite for engagement in honest and open dialogue, resulting in more effective student support. A positive and genuine relationship developed with an Academic advisor acts as a safeguard against some of the challenges faced by first year students (Yale, 2017). The large class sizes (many are over 400 students per class) on the nursing course is

a shock to many new students, Home group meetings with an Academic advisor alleviates fears and offers reassurance as it enables students to discuss their expectations and anxieties relating to the course. Results from this study show students who have a good early experience with Academic advisors feel comfortable throughout the three-year course approaching them for help and support. This finding is consistent with that of Yale (2017) who describe a good first encounter between a student and their Academic advisor as key to a successful relationship and students feeling comfortable approaching their tutor for help and support.

Further findings from this study show students are less likely to seek support when they feel a disconnect from their Academic advisor. A sense of detachment is exacerbated further when Academic advisors fail to remember their student's name as it implies disinterest in the student. This is particularly evident the closer a student is to completing the course as paradoxically this unfamiliar person is responsible for writing student's employment reference recommending their suitability for the role of qualified nurse; this leaves many students feeling bewildered and frustrated. A poor relationship and inadequate communication between an Academic advisor and student have been noted in the broader literature as having a direct link to negative experiences and low student satisfaction (Al-Ansari et al., 2015). Other studies have noted the importance of relationship building between Academic advisor and student during the initial transition into higher education to ensure student success later in the course (Thomas, 2012). Results from a UK project What works? Student retention and success reports one fifth of their respondents as unaware who their Academic advisor was, and 75% of students having a poor, very poor or no relationship with academic staff (Thomas, 2012). Thomas (2012) suggests those students who felt distant from academic staff and their Personal tutor were most likely to consider withdrawal from their course. Yale (2017, p.533) goes even further suggesting "poor personal tutoring is worse than not having a Personal tutor at all" and is a significant factor in students re-evaluating their decision to study at undergraduate level.

Comments from student participants in this study around their experiences with Academic advisors provoked strong negative emotions in the key stakeholders; many felt sad, challenged and frustrated by student's comments. Many key stakeholders were sad some students had poor experiences when they perceived themselves as working hard to provide a positive student experience. Although the disparity amongst Academic advisors in the same university was a source of frustration, several key stakeholders reflected on their own Academic advisor roles and acknowledged the role was often subsumed a lower priority than other academic responsibilities such as their teaching, module leadership and marking. The

207

findings suggest lack of time was the main reason for some academic advising not taking precedence over other responsibilities, and for one key stakeholder having too many (75) personal students to manage, some with '*difficult problems to sort out*' made it difficult for her to allocate time to less needy students.

The findings above contribute to the body of knowledge associated with the role of the Academic advisor in universities, with many current studies also concluding the Academic advisor role receives low priority and prominence, particularly within UK higher education institutes. For example, many higher education institutes within the US employ individuals in a full-time/part-time capacity exclusively as professional Academic advisors (White & Steele, 2016). By contrast academic advising/personal tutoring in the UK is perceived as a subsidiary aspect of academia, with personal tutoring delivered as a secondary part of an academic's role, often given lower priority than teaching and research (Myers, 2008). Furthermore, some personal tutors liken their role to that of a 'referral agent' as they signpost students to para-academic service providers such as counsellors for support (Macfarlane, 2011). As the literature reveals, this results in great variance in approaches and practices related to academic advising, not just between higher education institutes but across single campuses and departments within universities (White, 2015; Troxel et al., 2018; Zarges et al., 2018).

Variance with approaches and practices may result from a lack of universally agreed definition articulating the nature of academic advising (Grites et al., 2016; Troxel et al., 2018), with as many as 15 different definitions used to describe the concept of Academic advising (NACADA, 2006). Within this study the lack of a clear definition to denote the role and practice of academic advising could explain the confusion amongst student and key stakeholder participants about the role of the Academic advisor. However, a unique finding from this study is that confusion and misunderstanding around the role of the Academic advisor results predominately from the number of different terms designated to the same role. In particular, student participants within the present study referred consistently to an 'Academic advisor' as 'Personal tutor', and key stakeholder participants used the terms 'Academic advisor', 'Personal tutor' and 'Home group tutor' interchangeably throughout the emotional touchpoint mapping and Co-design events. At the time of writing this thesis the Academic advisor role within the whole University had undergone a major change and been replaced by 'Year tutors', each with responsibility for the pastoral and academic support of over 100 students in the nursing school, further adding to the confusion associated with the role.

Inconsistent use of one term designated to the role of the Academic advisor generates ambiguity and uncertainty around its meaning and purpose. Moreover, the use of the word 'Academic' within the name potentially misrepresents the role as it is suggestive of academic elements of the role, for instance giving advice relating to academic study, assessments and assignment writing; a remit usually associated with module supervisors. The description of an Academic advisor on the University intranet page within this study was "an academic member of staff whose role is to offer you academic advice and support by providing guidance or referral to appropriate services" (University of Central Lancashire, 2020), yet there was no explanation of the term 'academic advice and support' leaving students to interpret its meaning for themselves. Key stakeholders in the present study suggest this potentially misleads students giving rise to unmet expectations of extensive academic support. However, several Key stakeholders described the Academic advisor role as essentially focused on providing *'pastoral support'*. This perception of academic advising encompassing a more pastoral approach appears unique to the wider UK literature (Grey & Lochtie, 2016). Several studies have found a gap between a student's expectations of academic advising and what is then experienced, this can lead to student dissatisfaction with the course and disengagement in contact between an Academic advisor and student (Sutton & Sankar, 2011; Yale, 2017).

Clarifying and enhancing the role of the Academic advisor was identified as a course improvement priority during the Co-design event for this study. Students and key stakeholders specifically co-designed improvement suggestions that included devising clearer guidance for students and staff around the role and remit of an Academic Advisor, introducing a peer review and feedback process for the Academic advisor role, and ensuring parity throughout the University around the title used to denote an Academic advisor. However, since completion of this study all Academic advisor roles have been replaced by Year tutors throughout the university. At the time of writing this thesis the remit and application of the Year tutor role was evolving, and no clear guidance about the Year tutor role was available for students or staff to access. Therefore, a recommendation from this study is for the co-designed improvement suggestions above to be applied to the new role of Year tutor to avoid similar confusion to that associated with the previous role of Academic advisor. Specifically, clear guidance around the role and remit of Year tutors could be devised, and a peer review and feedback system for the Year tutor role could also be implemented.

9.5 Affective student Support

Affective student support considers student's emotional and well-being needs and seeks to develop student's self-confidence and self-esteem. Students' sense of self-efficacy and confidence is by strengthened by affective support (Kalamkarian & Karp, 2015). As previously indicated within this study good student support had an affective therapeutic effect of building

confidence and fostering well-being. Whereas poor support, such as ineffective clinical supervision, reduced self-efficacy and overall well-being for several students in this study. Keeling (2014) acknowledges the impact student's well-being has on their ability to learn, whilst Robinson and Snipes (2009) note a direct link between student's well-being, selfefficacy and development of their sense of agency. Bandura, an influential social cognitive theorist, defines self-efficacy as "people's beliefs about their capabilities to produce designated levels of performance that exercise influence over events that affect their lives" and proposes a person's self-efficacy beliefs impacts their feelings, thoughts, motivation and behaviour (Bandura, 1994, p.71). In other words, self-efficacy relates to an individuals perceived ability to accomplish success through the decisions and actions they take. Those with a strong self-efficacy deal with situations confidently and aptly apply their skills, resulting in higher levels of motivation and performance success or 'mastery' (Bandura, 1990). The aim of Pre-registration nursing education is to help students move from the place of a novice carer towards a level of mastery. As Bandura (1990) indicates, a person with a strong self-belief (self-efficacy) in their capabilities is more motivated and better prepared to learn and pursue their educational and occupational choices. However, central to a person's self-efficacy is their ability to exercise control (agency) over those events that affect their lives (Bandura, 1990). This suggests agency plays a critical role in students developing and maintaining self-efficacy beliefs that positively influences their success on the course and their overall learning experiences. Therefore, one way to develop and increase student's self-efficacy beliefs to emerge from the findings in this study was by supporting students develop learner agency as will be explored in the following section of this discussion.

9.5.1 Supporting students develop learner agency

A major finding from this study is that many participants lacked, or perceived a lack of, 'learner agency' creating a feeling of dependency on the academic support of others for their learning experiences. Learner agency, a term used interchangeably with student agency, refers to the level of students' self-reflective and intentional action and interaction with their environment (Klemencic, 2015). However, before a learner engages their agentic resources and chooses to exercise their agency in a particular learning context, they must hold a personal sense of agency – a belief that their behaviour can make a difference to their learning in that setting (Mercer, 2012). Originating out of social cognitive theory, learner agency correlates with an individual's self-efficacy (Vaughn, 2020), and ability to be self-reflective and intentional in their actions and decisions (Klemencic, 2015). In other words, students with agency are cognizant of their own capabilities, act with intent, and perform reflective self-examination of their actions and experiences. A study by Roman and Dison (2016) revealed a positive relationship between student agency and learning experiences, with students who exert agency being

describe as 'agentic'. Though students start their higher education courses with different levels of learner agency, an aim of higher education is for students to graduate with a greater capacity for agentic knowledge and action, something Case (2015) terms the 'morphogenesis of student agency'.

Learner agency requires individuals to engage with their environment (Schoon, 2018), set their own terms and conditions for learning and development (Zimmerman & Schunk, 2011) and take control of their learning experiences (Irvine et al., 2013). As individuals' experiences are shaped by their active participation in learning (Dewey, 1938/2015) students with learner agency are better able to influence and mediate their learning pathways and environments (Klemencic, 2017). Some participants in this study displayed learner agency early in the course and quickly established their own approach to independent study and time management. This is described as 'personal agency' by Bandura (2000) who presents three forms of agency including 'personal agency', 'collective agency' and 'proxy agency'. Other participants within this study developed a 'collective' form of learner agency as the course progressed, as was evident from those participants who described forming study groups with peers from their Home group. Collective agency sees individuals benefit from the shared efforts of various people to bring about a desired outcome or goal such as passing an assignment (Bandura, 2000). However, several participants displayed a 'proxy' form of learner agency whereby individuals remain reliant on others to act on their behalf exercising 'proxy agency' (Bandura, 2000). This was evident in those students who described expecting more, not less, support with their learning needs as they progressed through the course.

Proxy learner agency was particularly evident when participants transitioned into the clinical placement environment where they had expected to be closely supervised and directed by their mentors. Several participants had felt '*overwhelmed*' and out of their depth when left unsupervised, and some students felt they had '*missed a lot of learning opportunities*' they had expected their mentors to advise them about. Proxy agency is not uncommon where individuals have little or no direct control over their social and environmental circumstances (Bandura, 2018). In some circumstances individuals often turn to those perceived as experts with influence and power, such as mentors, to act on their behalf (Bandura, 2000), as appeared to be the case for students on their first clinical placement of the course.

Student agency is an important factor in empowering students (Roman & Dison, 2016), yet as humanly created social structures with the power to supress or support opportunities for personal agency (Bandura, 2000), some educational contexts inadvertently place constraints on learner agency (Klemencic, 2015). For example, the very nature of some clinical practice

areas alongside workload demands and staff shortages can impede student's freedom to choose and achieve their own learning goals. Furthermore, until students develop the competence to work safely in the workplace environment some mentors or clinical supervisors may be hesitant in enabling students to use their agency. Several participants in this study had expressed concerns about making mistakes in clinical practice that could be detrimental to their patients, colleagues or themselves, leading to anxiety around some of their actions and clinical decision-making. Additionally, suppressing learner agency lowered some student's confidence in their clinical practice over the three-year course, and left some students feeling unprepared for practice as registered nurses.

As previously established self-directedness and independent study skills featured recurrently within the student participant's and key stakeholder findings of this study. The development of learner agency for students undertaking a pre-registration nursing course is a fundamental mechanism for facilitating self-direction and independence. However, learner agency operates within humanly created social structures such as higher education settings and clinical practice environments, and as indicated in the discussion above these structures have the potential to constrain agency or conversely provide opportunity to practice agency. One framework for exploring learner agency, and specifically those experiences that either constrain or activate learner agency). Although Vaughn's model is rooted in promoting agency amongst school students in a learning context, the foundations of this model hold relevance and transferability for exploring agency in the wider context of a higher education setting and will be utilised to explore student experience in this study and the influence of learner agency on those experiences.



Figure 44 - A model of student agency (Vaughn, 2020)

Vaughn's model presents three dimensions of student agency; dispositional, motivational and positional. The dispositional dimension relates to a student's propensity to understand their goals and ideas, and act with intention to achieve their purpose (Vaughn, 2020). The motivational dimension relates to students' ability to regulate their actions (Bandura, 2001) through reflection on and perseverance with their decisions, regardless of perceived obstacles to achieving their goal (Vaughn, 2020). Whilst the positional dimension determines the social contexts and interactions that influence a student's experiences (Vaughn, 2020). The model suggests for learners to have agency, students must have a disposition to pursue their ideas, the motivation to act, and be positioned to engage in contexts and interactions with others conducive to achieving their goals. To develop learner agency students need a balance of support across all three dimensions (Vaughn, 2020), however the following sections will analyse each dimension separately in relation to the findings from this study.

9.5.2 Dispositional understanding of self, active, intentional, and possessing

purpose

The dispositional dimension relates to a student's self-awareness and proclivity to act with intention and purpose to achieve their goals (Vaughn, 2020). Students with a sense of agency use their initiative to purposely pursue and direct activities that create opportunities for learning (Vaughn, 2020). Although it was not the remit of this study to investigate student purposiveness, it is evident that all student participants exhibited dispositional tendencies to act with intention and purpose to achieve their goals and objectives. Most notably all students had achieved successful recruitment to the nursing course and at the point of interview were actively working towards their goal of becoming registered nurses. Nevertheless, many participants appeared to doubt their own ability or power to actively, and purposely, exercise agency and direct their own learning. Student's sense of agency is simultaneously grounded in past experiences and informs future experiences (Manyukhina & Wyse, 2019). Therefore, student nurses' beliefs about agentic capacities to act with intention and purpose may be negatively influenced by their past learning experiences with more traditional pedagogy.

Even when students possess the ability to act with intention and purpose, Manyukhina and Wyse (2019) suggest those with a sense of agency do not necessarily feel empowered to exercise that agency. Previous research has shown a sense of agency combined with the ability to exercise that agency as a prerequisite to students effectively developing independent study skills (Mercer, 2011). Furthermore, a direct link exists between students' agency and their capacity for autonomous learning (Manyukhina & Wyse, 2019). Yet findings within this study consistently revealed student participants '*struggled*' to manage their own learning autonomously and were dissatisfied with the amount of independent study on the course. Furthermore, students had expressed a sense of '*abandonment*' by academics and perceived a lack of support with developing their independent study skills. This perception was in direct contrast with key stakeholders' premise that their role is to encourage higher thinking skills traditionally associated with higher education that derive from more independent study. Academic's concerns around the distinction between supporting students to develop a strong sense of self-directedness and over-supporting can become a constraining factor for learner agency.

While conducting this study the nursing curriculum had moved to a more 'active learning' philosophy characterized by student led interactive learning necessitating more independent study than the previous curriculum. For both student and key stakeholder participants the impact of the curriculum change, and particularly of the reduction in content delivery via

classroom-based teaching sessions in year two and year three had left students feeling less prepared for 'the real world of nursing'. However, facilitating opportunities where students can see the inherent potential for learning can be achieved where educationalists understand the impact on learner agency of curriculum content and delivery (Manyukhina & Wyse, 2019). Therefore, academics have a key role to play in facilitating opportunities for students to recognise and understand the benefits of 'active learning' and support students to develop, and act on, a stronger sense of learner agency, something Vaugh et al. (2020) refers to as a 'dispositional shift'. One avenue for supporting students to experience a dispositional shift toward active learning and develop their learner agency identified in the Co-design event was through consistent and effective application of the Academic advisor role. However, at the point of writing this thesis there had been a shift from Academic advisors to Year tutors fulfilling the role of Academic advisors. Therefore, clarifying and enhancing the role of the Year tutor through staff training, regular peer reviews and clear guidance around the remit of the role could address the sense of 'abandonment' students described in this study, by providing a consistent point of contact throughout their three-year studies. As previously indicated, effective academic advising, particularly at the beginning of a course, has been shown to make students feel comfortable seeking help and support from their tutor and fosters student success later in the course. Thus, through effective application of the new role, Year tutors could be just as ideally positioned to shape students' beliefs and perceptions about their agentic disposition, build student confidence and provide real opportunities for students to effectively manage their own learning experiences.

9.5.3 Motivational beliefs, ability to act on beliefs, persisting and making choices

The motivational dimension relates to students' ability to reflect on and regulate, or manage, their own actions and ideas (Vaughn, 2020). Motivation creates, maintains and directs behaviour, and prevails as an important condition for learning (Adib et al., 2019). Students with agency exercise decision-making skills and persist with achieving their goals irrespective of perceived obstacles (Vaughn, 2020). Unsatisfactory experiences on the nursing course appeared to have little effect on the motivational beliefs of student participants within this study. Even those students who described feeling abandoned in year two and three of the course exhibited a strong motivation to persist with the course despite any perceived obstacles. For all students, the driver for completing the course was their desire to become qualified nurses. This was evident in one participant who had considered leaving the course after failing several assessments in year one. The demands of undertaking year one reassessments alongside year two academic work had felt unsustainable, forming a perceived

obstacle to completing the course as indicated in the participants own statement '*in second year it was like 'I'm just going to walk away*". For this participant analogically seeing '*the goal post*' was the motivational catalyst for acting on her disposition to successfully pass year one and two, and 'become a nurse'. For other students, the support of peers' or their Academic advisor had been pivotal in motivating them to persist with the course. These findings are consisted with a recent study by Adib et al. (2019) that showed high academic motivation amongst nursing students. Although goal orientation plays a part in motivating most university students to engage with and complete their studies, for nursing students the clinical practice placement experience and particularly the influence of mentors appears to be an additional factor within the motivational dimension.

Several participants regarded their mentors as influencing their motivation, describing them as 'a huge inspiration' who made them 'want to go and learn'. Yet poor clinical placement experiences, and specifically poor mentors, also had the ability to de-motivate students and erode their self-efficacy, ultimately constraining learner agency. As previously indicated selfefficacy relates to a person's belief about their ability to successfully perform in a specific context, task, or domain (Bandura, 1997). Well-developed self-efficacy in student nurses is essential for fostering confidence in clinical practice (Henderson et al., 2016). Participants who were apprehensive about their role as student nurses exhibited poorer self-efficacy and low confidence in the role of student nurse. Poor self-efficacy, demotivation and disengagement was evident in those participants who were left to work in a more HCA orientated role throughout their placements and when their mentors appeared indifferent or disinterested in helping/teaching them. Only one student indicated their mentors' lack of interest motivated them to seek out their own learning opportunities in the placement area. Clearly this student had a strong sense of learner agency and self-efficacy. However, for most participants in this study learner agency is activated when mentors take an interest in them and impel them to engage in deeper learning. Several examples articulated by participants in the findings include mentors 'challenging me to go and learn' and 'wanting me to do quite a lot of research'. In each case these prompts appeared to trigger participants motivation to engage in deeper learning and concurrently develop autonomous study skills.

The discussion above strengthens the argument for designating the role of mentor, or clinical supervisor, to those who have a predisposition to support students with their learning in the clinical practice area rather than the current system whereby all registered nurses are expected to engage in the role. Good mentoring results in students developing autonomous learning skills and confidence exerting agency as they work towards achieving their goal of becoming a nurse. Ramifications of poor mentoring for some students include reduced

motivation, confidence and self-efficacy. Student satisfaction with their learning journey is directly affected by their self-motivation and arises out of their sense of personal accomplishment (Walker et al., 2016). Furthermore, motivation and capacity for exercising agency will differ amongst students with some learners needing support to reach their maximum agentic potential (Manyukhina & Wyse, 2019). A study by Adib et al. (2019) revealed a direct correlation between academic motivation and self-directed learning suggesting educators have a significant role to play in increasing the academic motivation of students to promote their independent learning skills.

During the Co-design events student and key stakeholder participants identified 'facilitating and supporting student development of independent study skills' as one of the course improvement priorities. Several of the recommendations discussed in chapter eight of this thesis could provide opportunities to better motivate students with their academic studies and consequently improve their independent learning skills, for example introducing academic mentors (either staff member or peer) to support students with their academic study, implementing supplementary study skills in the form of drop-in sessions and step-by-step video guides to undertaking study skills/independent learning, and establishing student led support networks focused on study skills and independent learning. What is clear from the discussions above is that student's motivational beliefs, ability to decision make and intentionally act on beliefs is dependent on a level of self-efficacy which for student nurses is not only influenced by their experiences with mentors and academics but can also be directly affected by their positional experiences.

9.5.4 Positional experiences within social context and how situational variables inform agency interactions and negotiations

The positional dimension relates to individuals' interactions with different social contexts and learning environments (Vaughn, 2020). Vaughn (2014) proposes a sociocultural perspective of student agency where learning experiences are socially mediated and positioned within multifaceted learning contexts. As such learner agency is "not a personality characteristic, but a relational process that emerges through interaction with others" (Schoon, 2018, p.8) and is manifestly formed by the way students' interact with the wider socio-cultural context. In other words, learner agency is not innate but is generated through interaction with various social structures and environments (Larsen–freeman, 2019), and is co-constructed and influenced by the student learning was positioned within an academic university location as well as a clinical NHS setting, with learners moving between the two environments every few months.

How students choose to act, and make decisions, within these two contexts is influenced by their perceived level of agency within the positional dimension (Vaughn, 2018). Juxtaposing these two environments reveals disparate situational variables and interactions influencing learners' agency. As student nurses balance multiple shifting identities across various social contexts tensions arise between their professional nursing role and their identity as a university student (Jackson & Steven, 2020).

As was evident in this study the different expectations of students' agentic behaviour in clinical practice placement than when on university campus may contribute to identity confusion. For example, in the clinical practice environment many students appeared to depend on mentors to meet their learning needs. This behaviour is perpetuated when mentors were perceived as obstructing engagement in learning activities students had organised for themselves when placement areas where busy and required the student to work in an HCA capacity. On campus, academics expect students to engage with independent learning by actively seeking out valuable learning opportunities outside of the classroom setting. Frustrating learner agency in one context, whilst expecting students to exert agency in another context ultimately constrains confidence.

A further situational variable influencing student agency within the 'positional experience dimension' for the participants in this study relates to their capacity to exert influence within the context of their learning. Contemporary higher education thinking acknowledges students as coproducers of their knowledge and learning, and therefore members of the 'academic collective' where students are involved in the educational decision-making processes about the contents and delivery of their course (Klemencic, 2017). Effective application of the academic collective requires students to have a degree of autonomy (Klemencic, 2017). This was evident in the clinical practice setting for some students in this study. For example, several students report arranging specific learning opportunities in different clinical settings for themselves, often termed a SPOKE. However, the contents of the course in the university setting within this study is governed by the NMC that informs the nursing standards of proficiency, UK legislation, statutory regulations, clinical guidelines and local NHS hospital trust policies. Thus, student nurses may feel they have little say over the contents of the course or delivery methods impacting their learner agency. Furthermore, fear of contravening NMC and local NHS trust guidelines and policies can result in anxiety and hesitancy exerting agency, and dependence on mentors and academic staff (proxy agency) as appeared to be the case for those students in this study who reported 'missing out on learning opportunities' whilst waiting for their mentors, PEFs and academic staff to clarify what they were permitted to do as student nurses in the clinical practice environment. More focused support and clarity from mentors, PEFs and academic staff around the skills and activities sanctioned by the NMC could foster the confidence in students to exercise more agency with their learning needs.

In summary, the discussion above contributes to the argument for preparing student nurses better for clinical placement learning, specifically their understanding of their role and accepted practice. As student nurses work towards registration the process of developing professional characteristics, recognised as autonomous evidence-based decision-making skills, must be facilitated (NMC, 2018c). The findings from this study show facilitation of autonomy occurs when student nurses perceive a level of efficacy and agency, whereas feeling vulnerable and powerless reduces students' confidence, further impeding their efficacy. Accepting that students on pre-registration nursing course possess a disposition to act with intention and purpose, and the motivation to pursue their goal of becoming registrants, key stakeholders within the University and clinical practice environment have a responsibility to ensure students positional experiences, particularly those relating to individual interactions positively influence their progress through the course. A suggestion from the findings is that key stakeholders work collaboratively with students to establish environments where student nurses feel safe to disclose their lack of confidence and supported to develop and competently exert their agency without repercussions. Adopting an EBCD approach to future projects focused on the experiences of student nurses may be an effective way to accomplishing the greater collaboration espoused above.

A further suggestion from the study findings is that higher education institutions find ways to better meet students support needs, whilst ensuring current support mechanisms are more visible and accessible. Effective student support was shown in this study to be accessible and available whilst incorporating cognitive, relational, affective and systemic elements. Moreover, the discussion above has shown the elements of student support indicated above to be underpinned by wider literature. Therefore, tenets of the Tait's (2000) 'Taxonomy for supporting students' on pre-registration nursing courses to emerge from this study could be utilised by academic and clinical practice staff as a framework for supporting nursing students on pre-registration nursing courses.

9.6 Systemic student support

Systemic student support relates to supporting students with navigating administrative systems (Kalamkarian & Karp, 2015) and incorporates assisting students with registration, technical support, library access and learning, assessment and accreditation, needed for successful learning (Jung & Hong, 2014). For Tait (2000) efficient administrative support influences both student satisfaction and loyalty to an institution. Efficient systemic support

should be user-friendly, inclusive and provide opportunities for social networking, peer support and shared learning (Jung & Hong, 2014; Shikulo & Lekhetho, 2020). Participants in this study described positive experiences of systemic student support overall. For example, one participant reports administrative support as 'Overall, pretty good' because the Course Administration Service (CAS) team had always got back to her in a timely manner when she needed advice. Another participant describes struggling with the application process: 'When I'd gone onto the website they were saying about credits and I had no idea what credits were! I rang the university Course Enquiries and they were really helpful. They told me what I needed to do and how to go about it'. Applying for the BSc Pre-registration nursing course can be challenging, particularly when it relates to establishing the qualifications necessary for the nursing course or trying to make sense of unfamiliar terms and confusing jargon such as 'credits'. For potential candidates being able to speak directly to Course Enquiries about applying for the nursing course offered some relief and reassurance. However, the excerpt above suggests that despite the move towards an automated and technology-driven service some students still value being able to access an individual who can clarify and simplify information more clearly.

Despite the importance of efficient administrative support in ensuring a positive student experience, most participants in this study did not appear to know or understand the purpose or role of the administrative team. Several participants claimed to have had minimal experience of administrative support suggesting '*I* don't think I've had that much of a need for them' and 'I've not had much experience with Brook Hub (administrative support) apart from just basically calling in sick when I had my accident a few years ago'. Yet a clearer understanding of how effective administrative support ensures seamless delivery of the curriculum through the co-ordination of enrolment and induction, managing student data (e.g. attendance and module grades) and managing student progression through the course could significantly shape a better student experience overall.

Alongside ensuring efficient administrative assistance systemic student support fundamentally offers opportunities for social networking and peer support (Alias & Rahman, 2005; Jung & Hong, 2014; Shikulo & Lekhetho, 2020). Although peer support is more naturally situated within relational student support, Shikulo and Lekhetho (2020) argue institutions have a responsibility to provide opportunities for students to integrate into the university community through the use of social and multimedia as part of a systemic student support approach. Not least because early social integration and adjustment to higher education through peer support are intrinsically linked and can enhance the learning processes, interactions with other students and lecturers, and reduce attrition (Tinto, 1987; Irwin et al., 2012; Shikulo & Lekhetho,

2020). Many forms of social media exist within the university environment for example Twitter, WhatsApp groups, and TEAMS.

The main forum for seeking peer support for participants in the current study was through the School of nursing Facebook pages. Participants specifically described using Facebook to make connections with peers, form friendship groups and as a resource for information and advice early in the course. This was evident from comments such as "*I didn't know anybody whatsoever! Thankfully we had already built-up friends on the Facebook group*" and "*I like the Facebook page because other students comment on stuff you've got issues with. It was nice to see who was getting buses, who was driving in and who you could catch a lift with, very helpful*". By contrast, previous research had found on-line peer support was underutilised by most student participants in their study, with only those mature students (above 35 years of age) reporting seeking support from peers online (Alias & Rahman, 2005). This could indicate a shift in perceptions about the use of social media for support on an academic course.

Several participants in the current study reported specifically accessing the School of Nursing Facebook page specific to their cohort rather than the wider generic university Facebook pages. This may signify that some participants begin forming a sense of connectedness and belonging with peers on a course that fosters a sense of identity as a student nurse. However, an interesting finding from this study was that some participants who had found Facebook useful at the beginning of the course had found the use of Facebook counterproductive the further along the course they were. In particular, where Facebook had previously been a useful vehicle for information sharing and support, many participants describe the social space evolving into a place for 'moaning and bitching'. Furthermore, some participants perceived a level of unprofessionalism by some students posting their dissatisfaction with elements of the course, such as the lectures, even when knowing that academic staff and other students could view these comments. Evidently, some students appeared to use virtual worlds such as Facebook as a platform to vent their feelings and opinions seemingly with no apparent repercussions. An exert from the findings suggest this resulted in some participants leaving the Facebook group, "well I've left the Facebook group now because everything was just irritating me...people moaning and bitching about people, I just thought it's a bit unprofessional. I left it months ago".

The discussion above suggests student's view and use Facebook differently depending on how far along the journey through the nursing course they are. At the start of the course, students appear to use social media for guidance, help and advice, as well as for making friends and forming social networks. As students' progress through the course they appear to use Facebook as a medium for voicing their opinions and feelings, particularly if disgruntled with specific elements of the course. Some students approaching the final transition to qualified nurse began to acknowledge the qualities of professionalism and appear to dissociate with behaviours and mediums they perceive as unprofessional.

9.7 Implications of this study for pre-registration nursing education Systemic

This study contextualizes what it 'feels' like to undertake an undergraduate nursing course for some students in one UK university. Issues discovered around feeling prepared to study at higher education level, feeling abandoned on the course and feeling frustrated with perceived lack of support suggest more needs to be done within this university to increase support and to better highlight support mechanisms already in place. Though these issues may not be new in nurse education the narratives from students in this study draw attention to the heightened levels of stress and anxiety undergraduate nurses experience during the course, with anxiety levels increasing as students approach the registration point. During moments of reflection leading up to graduation some students teeter between excitement and anxiety around 'doing the job' of a registered nurse. Ultimately student nurses join a workforce renowned for high occupational stress (Sarafis et al., 2016) and burnout (Hetzel-Riggin et al., 2019); the impact of burnout on the nursing workforce is well documented in the literature as exacerbating nursing shortages within the UK. Without efficient strategies in place some newly qualified nurses will continue to enter the profession already experiencing the negative effects of stress and anxiety, subsequently hastening burnout in nurses who find themselves working in demanding environments. This study not only highlights the importance of supporting students prior to and during their transition to registrant but can contribute to existing 'health and wellbeing' policies within the organisation where this study was conducted.

Student transition is a persistent theme throughout this thesis, with many significant touchpoints occurring during pivotal transition phases. Findings indicate a variety of factors influence student's experience of these transition phases positively or negatively. Although student transition has been widely discoursed in the literature, transition theories focus predominantly on transition into year one and transition from student to registrant. Results from this study demonstrate students want more support, rather than less, as they progress through the course. Currently, institutions contribute significant investments into enhancing the first-year experience and improving year one retention rates, whilst major gaps continue to exist in the student retention literature beyond year one study in higher education (Sterling, 2018). The research presented in this thesis offers a unique perspective on the importance of understanding the effect of mid-course transitions on student experience and attrition; and

222

provides evidence for future research and investment on the year two and year three programme.

This study further highlights the need to develop a 'student model of transition' with an extensive focus on year two and year three of the pre-registration nursing course, alongside transition to clinical practice placement. Current theoretical frameworks explaining student transition focus principally on the general student populations' transition into year one. The few transition models related to undergraduate nursing similarly emphasise transition into year one alongside the graduation phase as individuals move from student to registrant. Findings from this study advances the argument for an 'undergraduate nursing transition model' that evaluates and explains how student nurses experience transition in year two and year three and offers effective measures to support students through transition. Any future 'undergraduate nursing transition model' must incorporate a distinct clinical practice component that considers the transition across environments, therefore from an academic to clinical practice placement setting.

Results from this study indicate mentors are pivotal to the student experience of the clinical practice environment. Mentors are in a unique position to support students with the development of knowledge, clinical competency and professional identity. Furthermore, mentors have a significant influence on student nurses' development of confidence and sense of preparedness for registration. Findings also suggest students value having access to a specified named person to advise, support and guide students through their clinical practice placement experience. With the new system of student supervision and assessment now underway findings from this study draw attention to the need for a systematic evaluation of the new NMC standards for supporting student learning and assessment in clinical practice. Meanwhile, strategies must be clearly identified and in place to support students and mentors whilst the new system is rolled out and embedded.

This study highlights the importance of nursing students developing effective independent study skills early in the course, with the support of key stakeholders. As previously established a fundamental mechanism for students developing independent study skills was through exercising learner agency. Those student participants who possess learner agency are less dependent on the academic support of others for their learning experiences. Learner agency operates within higher education and clinical practice social structures that have the potential to constrain or facilitate agency. This study demonstrates the effect on learner agency of student's disposition to pursue their ideas, motivation to act and position to engage with others to achieve their goals. A balance of support across all three dimensions is needed for students

223

to develop learner agency. This study not only highlights the importance of understanding the significance of learner agency in nurse education but was the first to utilise a model of student agency to explore factors that either constrain or activate learner agency for students on the BSc Pre-registration (Adult) nursing course. Priorities for course improvements identified during the Co-design event are potential mechanisms for supporting the development of learner agency, for example through clearer application of the Academic advisor model.

Although the university where this study was conducted has moved from an Academic advising approach to a Year tutor approach to supporting students pastorally across their three-year course, this study still adds to the debate on the purpose and role of the Academic advisor for those institutions across the UK that continue to employ an Academic advisor system. This study calls for a distinction to be noted between a Personal tutor and Academic advisor as currently a blurring of lines between academic advising and personal tutoring is contributing to confusion around the roles, for students and staff. This study also found inconsistency around the terminology used to denote an Academic advisor among staff, students, course documentation and the Universities website, with the addition of the new Year tutor role potentially adding to this confusion. Findings from this study can be used within the organisation where this study took place to help formulate a clear definition and guidelines framing the new Year tutor role.

9.8 Reflections on the use of Experience Based Co-design in a higher education setting

EBCD has been successfully used as a research methodology in a multitude of health care settings, yielding positive documented results around patient experience (Donetto et al., 2014). Tenets of EBCD were recognisably appealing for exploring the student experience, particularly as it enabled students to share their experiences of the BSc Pre-registration (Adult) nursing course, and both students and key stakeholders to engage in identifying and co-designing course improvements as was evident in the Co-design event. Fundamentally, this study is distinct from previously published research conducted on student experience of higher education as it is the first documented to use aspects of EBCD exclusively with student and staff participants in a university setting. Although EBCD has previously been used in healthcare settings, the search strategy indicated that no published study has demonstrated how EBCD could be used in a non-healthcare environment. This study reports an original use of aspects of EBCD and evidences the crossover of an EBCD approach from a healthcare setting to an educational setting, thus making an important contribution to knowledge on the challenges of using EBCD in a non-health care setting.

On reflection several aspects of the application of EBCD process could be improved. For example, no formal training in the use of an EBCD approach was undertaken prior to commencing this study. A recent report mapping the use of EBCD found around 50% of respondents had not experienced formal training in the use of EBCD (Donetto et al., 2014). This resulted in the development of an EBCD toolkit available for online access (Donetto et al., 2014), which formed the foundation of EBCD knowledge for this study alongside Bate and Robert's (2007a) published book and research articles identified as using EBCD. These resources were useful for understanding the overarching methodology of EBCD and several examples of how to conduct the emotional touchpoint mapping events were included. However, there was no distinct set of methods advocated for some aspects of the approach such as undertaking data analysis, as you expect from other research methodologies. Training in the use of EBCD may have been invaluable for addressing some of those issues and offered the chance to seek advice about the aspects of EBCD less explored in the literature.

The methodology section of this thesis outlined the use of an EBCD approach and posed several challenges. For example, as no previously published study had documented the use of an EBCD outside of a healthcare setting, specifically with students in a higher education environment, there was a distinct lack of available comparative data providing insight into the efficacy, strengths and weaknesses of using EBCD to explore the student experience of an undergraduate nursing course. The lack of previously published EBCD studies conducted in higher education offered a unique opportunity for this study to contribute to discourse around the use of aspects of EBCD outside of the healthcare setting.

Implementing an EBCD approach as part of a PhD created practical challenges with administering a realistic timeline for the study. As previously noted, EBCD projects in healthcare tend to have a short timeline with most projects completed in under 18 months, including setting up the project, gathering data, implementing priority recommendations and evaluating the changes (Donetto et al., 2014). The benefit of such a short timeline is that the research participants are more likely to be available, and willing, to take part in the whole study. However, this study took over six-years to complete due to the part-time nature of PhD work and the data collection phases lasting three-years. Therefore, the first two years were spent producing the research proposal, conducting a literature review and applying for ethics approval (setting up the project). The data collection and analysis process took three years due to the accessibility of student participants who only appeared available/willing to take part in the study during their theory blocks, and the workload responsibilities of the lead researcher (this included module leader, cohort lead, year lead and finally course leader responsibilities).

This meant student participants recruited to the earlier data collection events, such as the semi-structured interviews, had graduated from the nursing course and were unable to take part in the final Co-design events. Although a preferred aspect of EBCD is the collaborative inclusion of the same participants throughout the whole project a rolling recruitment approach where different participants are recruited to a study was adopted. This process can be seen in many other EBCD studies that encountered similar problems with long timelines (née Blackwell et al., 2017). On reflection, carefully planning a shorter timeline for data collection would enable some students to participate in all phases of data collection and the Co-design event for future PhD studies utilising an EBCD approach.

9.8.1 Implications of this study for Experience Based Co-design

Although toolkits exist offering step-by-step guidance and resources to support the implementation of EBCD, as noted above the process for analysing the data is not stipulated. As indicated in the methodology section of this thesis most published EBCD studies do not specify their adopted approach to data analysis, including the seminal work of Bate and Robert (2006). However, several studies since 2006 have identified thematic analysis as their method for analysing gathered data, and one study specifies using thematic framework analysis (née Blackwell et al., 2017). Within EBCD there is great focus on the practical stages of data collection and co-design, but little attention given to the analysis of any data gathered. Though the action of abstracting meaning and interpreting qualitative data is difficult to describe and illustrate, presenting the iterative process of data analysis in a series of steps is a useful tool for guiding readers and subsequent researchers through the decision-making process (Polit & Beck, 2009). Clearly documenting the analytical process potentially adds rigor and validates inferences and conclusions drawn in a report, increasing the credibility of a study amongst the research community. Therefore, this study makes an original and significant contribution to the growing knowledge on the application of EBCD as a research methodology by clearly identifying the 'Six phases of thematic analysis' framework (Braun & Clarke, 2006) as the applied method for analysing the raw data from this study.

Braun and Clarke's (2006) thematic analysis framework is cited in nearly 40,000 published studies; however no published research clearly documents its use in an EBCD study. Furthermore, this study systematically illustrates the relationship between themes identified in the raw data and the touchpoints that formed the student journey through the nursing course and presents this information in a framework that future researchers using EBCD can access and utilise (See Figure 45 - Framework for documenting the relationship between core themes and touchpoints).

Participant	Data	Code:	Theme:	Subtheme:	Touchpoint:
06	Group supervision overall, I've had a really positive experience, I've enjoyed them. I find it supportive because you're bouncing things off each other, and people think of things you might not have necessarily thought of yourself, and I have found them really supportive, I think they're a good idea to be honest.	Positive experience of group supervision	Navigating the academic journey	Engagement with academic assessment, support and feedback	Module Assessment Group Support (MAGS)
03	Group supervision I hate with a passion! I don't see how it was justified, I feel it really disadvantages our learning. The only thing I can think is the university was bringing it in right across the board and the school of nursing couldn't argue because it's a university initiative. I really struggle writing essays, and I feel right now like I've got no support	Feeling disadvantaged by group assessment supervision	Navigating the academic journey	Engagement with academic assessment, support and feedback	Module Assessment Group Support (MAGS)

Figure 45 - Framework for documenting the relationship between core themes and touchpoints

Consequently, this study addresses the gap in the literature for research focused on touchpoints within an HEI setting and presents a demonstrable model for analysing qualitative data using an EBCD approach.

9.9 Limitations of the study

This study had several potential limitations. Firstly, the length of time it took to complete the first five stages of the EBCD Cycle (Donetto et al., 2014) was six-years, from stage-one setting up the project to stage-five conducting the small Co-design events. This study formed a PhD undertaken in a part-time capacity for which six years is the acceptable norm, however most EBCD projects have a short timeline, with previously conducted studies documented as completing in under 18 months (Donetto et al., 2014; Locock et al., 2014). As previously noted, the six-year timeline impacted the timing of the data collection phases. All but one of the student participants who had contributed to the semi-structured interviews and emotional touchpoint mapping event had graduated from the course before the Co-design events took place. At the beginning of the project it had not been anticipated that data collection would take three years. A recommendation for future research EBCD projects focused on students as users of a service is to establish a shorter timeline for data collection that would enable students to participate in all phases of data collection and the Co-design event.

Potentiality, only completing five of the six staged EBCD cycle could be viewed as a limitation of the study. Therefore, stage-one to stage-five (setting up, engaging student and gathering experiences, engaging key stakeholder and gathering experiences, student and key stakeholder Co-design events and small working groups coproducing priority improvements and actions for the BSc pre-registration (Adult) nursing course) was completed. However, a deliberate decision was made early in the study to only complete stage-one to stage-five as that would successfully meet the aims and objectives of the study. Dimopoulos-Bick et al. (2019) notes variation in implementation of the EBCD methodology throughout the published literature, at present there is no critical evidence to suggest omitting a component of the process impacts the success of a project. A recommendation from this study is for stage-six (implementation of priority improvements and actions for the BSc pre-registration (Adult) nursing course and project evaluation) to be conducted as part of a post-doctoral project.

A further potential limitation of this study was the participant sample that represented a small proportion of the cohort numbers. Although the study aimed to use a purposive sampling strategy the number of participant responses was smaller than anticipated leading to a convenience sample being recruited and making maximum variation sampling unachievable. Reasons for the smaller than expected recruitment numbers could relate to the many curricula changes taking place at the time of the study (as alluded to several times in this thesis), alongside competing demands on student time academically and personally, leaving some students unwilling to take on extra activities such as engaging in research. Overall, the participant sample comprised of a diverse non-homogenous demographic which included Black, Asian and Minority Ethnic (BAME), male, female and mature students, however the sample was not representative of cohort and year of study. Therefore, the sample was skewed predominantly towards third year students from September cohorts. Notably, the sample size was large enough to generate rich descriptions based on informational need. However, when interpreting the results of this study it must be acknowledged that clinical practice representation was missing from the key stakeholder participant sample. Although PEFs were invited to take part in the study only one responded in the affirmative, but consequently attended on the wrong date (after the event was conducted). Therefore, this study does not claim to make generalisations, or to be statistically representative of the wider student and key stakeholder population.

Filmed narratives traditionally form part of the EBCD approach but within this study the decision was made to audio record rather than film all semi-structured interviews and emotional touchpoint mapping events, which could be perceived as a limitation. Filmed patient narratives are frequently used to trigger discussion in the Co-design events (Adams et al.,

2014). However, filming students for this study was carefully considered in relation to ethical issues and any future repercussions on study participants. Researchers go to great lengths to protect the interests of study participants by retaining anonymity and confidentiality whilst avoiding 'deductive disclosure'. Paradoxically filmed interviews make individuals identifiable potentially exposing them to future ramification. Thus, film clips used locally, for example in emotional touchpoint mapping events, risk participants being recognised by current peers and academics responsible for the teaching and assessment of student participants. On reflection this was the right decision as researchers have a great responsibility to act in the best interest of their study participants both during and after study completion. Several recent EBCD projects have also chosen not to film study participants in response to staff reluctance and patient nervousness (Cooper et al., 2016). Furthermore, audio recording and transcribing interviews verbatim has been shown to maintain fidelity to the data (DeJonckheere & Vaughn, 2019). Thus, all semi-structured interviews and emotional touchpoint mapping events conducted during the present study were audio recorded, transcribed and analysed for touchpoints and verbatim quotes related to each touchpoint incorporated into the student journey map to prompt discussion during the Co-design event in a similar way to film clips from other EBCD projects.

The inclusion of key stakeholders and student participants in the same Co-design event had the potential to limit the veracity of the findings. Though co-design centres on service users (in this case students) and key stakeholders working collaboratively to identify priorities for the redesign of services, within this study students may have perceived a power differential between the key stakeholders and themselves. Inherent power differential between educators and their students results in a disparity in power with academics having undue influence which can adversely affect student behaviours and responses during the study (Aycock & Currie, 2013). To avoid accentuating any perceived power differentials during the Co-design event participants were divided into three smaller groups with each group containing equal or more numbers of students; therefore two groups contained three students and one key stakeholder participants. Furthermore, students and key stakeholders were not asked to introduce themselves or their designation, therefore participants were not made aware of which students were first, second or third years and were not reminded about the role of each key stakeholder, for example Course leader or Year lead etc.

A final consideration within the limitation section is the issue of 'double agency'. Double agency arises when an individual simultaneously assumes two roles, for example academic and researcher, as was the case with this study (Edwards & Chalmers, 2002; Ferguson et al.,

2004). Concerns around double agency evolve where there is divided loyalty or a conflict-ofinterest such as when teachers engage their students in their research study (Edwards & Chalmers, 2002; Ferguson et al., 2004). Ferguson et al. (2004) identify academics unintentionally exerting pressure on students to participate in studies or students fearing repercussions from refusal to participate as issues relating to double agency. To avoid a conflict-of-interest arising in this study steps were taken to ensure ethical principles were adhered to throughout the briefing, recruitment and data collection phases of the study. Ethical approval was sought and gained from STEMH Ethics committee. Students were briefed about the study by impartial third parties not involved with the research such as module leaders, and student participants were informed they could withdraw from the study at any point up to their data being anonymized without the need for an explanation.

9.10 Dissemination of findings

Preliminary findings from this study have been disseminated centrally in the University via the UCLan Faculty of Health and Wellbeing PGRS forum, the UCLan Learning and Teaching Conference (June 2017), the Adult Division Team meeting (September 2018) and the UCLan School of Nursing Three Minute Thesis (December 2018). In addition, findings will be disseminated to students and key stakeholders in the academic and clinical practice environment through presentation at Student Staff Liaison Committee (SSLC) meetings, Programme Management meetings and Transforming our Practice Partnerships (TOPPS) meetings. Priority recommendations for course improvement identified during the Co-design events could also contribute to the next BSc Pre-registration (Adult) nursing programme Continuous Course Enhancement report. Wider dissemination of the study findings will be through paper publications in educational and professional journals such as Nurse Education Today, the Journal of Further and Higher Education and Nurse Education in Practice, conference presentations and webinar.

9.11 Conclusion

This study set out to capture, explore and understand the experiences of student nurses undertaking a BSc Pre-registration (Adult) nursing course in one university setting, and identify course improvement priorities that potentially enhance future student experience of the nursing course. The use of the innovative design methodology, Experienced Based Co-Design, enabled student and key stakeholder participants to reflect on the complete student journey through the nursing course rather than exclusively on one feature, time span or pathway on the course. Thus, the findings from the semi-structured interviews, emotional touchpoint mapping events and Co-design events helped formulate a meaningful portrayal of the whole student journey through the course for participants in this study. The substantive findings suggest students had varied experiences on the nursing course, both positive and negative, particularly of student support academically and in the clinical practice environment. Expectations of sufficient support was evident at the beginning, and throughout, the student's journey of the nursing course and appeared to align with students' sense of learner agency. Findings also revealed that for student support to be effective it must comprise of several key elements, it should be cognitive, affective, systemic and relational, whilst simultaneously accessible and available.

Insight from this study is particularly timely following the recent Covid 19 pandemic that resulted in more online learning and a reduction in face-to-face contact time with academics. Consequently, student support may feel less accessible and available as nursing students predominantly study nursing within a distance learning context. As a result, establishing student agency and fostering independent study skills early in the nursing course has taken on even more significance. As nursing courses are set to look very different in the foreseeable future, higher education institutes must adapt to these changes quickly and better support students through these changes. Universities that fail to adapt quickly and incorporate effective student support into curriculum design risk compromising the student experience. Thus, as the current global challenge to public health potentially marks a paradigm shift in delivering education to nursing students through a blended learning approach that incorporates more on-line and independent learning, this study aptly contributes to future discourse around providing higher education that considers and emphasises the importance of the student experience on nursing courses.

REFERENCES

- Adams, M., Maben, J., & Robert, G. (2014). *Improving patient-centred care through experience-based co-design (EBCD): an evaluation of the sustainability and spread of EBCD in a cancer centre.* King's College London.
- Adib, M., Ghiyasvandian, S., Varaei, S., & Roushan, Z. A. (2019). Relationship between Academic Motivation and Self-directed Learning in Nursing Students. *Journal of Pharmaceutical Research International*, 1-9.
- Aebersold, M. (2018). Simulation-Based Learning: No Longer a Novelty in Undergraduate Education. *OJIN: The Online Journal of Issues in Nursing*, 23(2).
- Ainley, P. (2008). The varieties of student experience–an open research question and some ways to answer it. *Studies in Higher Education*, 33(5), 615-624.
- Al-Amoudi, I., & Willmott, H. (2011). Where Constructionism and Critical Realism Converge: Interrogating the Domain of Epistemological Relativism. *Organization Studies*, 32(1), 27-46.
- Al-Ansari, A., El Tantawi, M., AbdelSalam, M., & Al-Harbi, F. (2015). Academic advising and student support: Help-seeking behaviors among Saudi dental undergraduate students. *The Saudi dental journal*, 27(2), 57-62.
- Alhojailan, M. I. (2012). Thematic analysis: a critical review of its process and evaluation. *West East Journal of Social Sciences*, 1(1), 39-47.
- Alias, N. A., & Rahman, N. S. N. A. (2005). The supportive distance learning environment: A study on the learning support needs of Malaysian online learners. *European Journal of Open, Distance and e-learning*, 8(2).
- Allsop, M. (2013). Competing paradigms and health research: design and process. In M. Saks, & J. Allsop (Eds.) *Researching health: qualitative, quantitative and mixed methods.* Sage.
- Andrew, N. (2013). Clinical imprinting: The impact of early clinical learning on career long professional development in nursing. *Nurse Education in Practice*, 13(3), 161-164.
- Andrew, N., McGuinness, C., Reid, G., & Corcoran, T. (2009). Greater than the sum of its parts: Transition into the first year of undergraduate nursing. *Nurse Education in Practice*, 9, 13–21.
- Andrew, N., Robb, Y., Ferguson, D., & Brown, J. (2011). 'Show us you know us': Using the Senses Framework to support the professional development of undergraduate nursing students. *Nurse Education in Practice*, 11, 356 - 359.
- Andrew, S., Salamonson, Y., Weaver, R., Smith, A., O'Reilly, R., & Taylor. C. (2008). Hate the course or hate to go: Semester differences in first year nursing attrition. *Nurse Education Today*, 28(7), 865–872.
- Andrews, G. J., Brodie, D. A., Andrews, J. P., Hillan, E., Thomas, B. G., Wong, J., & Rixon, L. (2006). Professional roles and communications in clinical placements: A qualitative study of nursing students' perceptions and some models for practice. *International Journal of Nursing Studies*, 43(7), 861-874.

- Archer, M., Bhaskar, R., Collier, A., Lawson, T., & Norrie, A. (Eds.) (2013). *Critical realism:* essential readings. Routledge.
- Aveyard, H. (2019). *Doing a literature review in health and social care: A practical guide* (4th ed.). Open University Press.
- Aveyard, H., & Sharp, P. (2017). A beginner's guide to evidence-based practice in health and social care (3rd ed.). McGraw-Hill Education.
- Ayala, J. C., & Manzano, G. (2018). Academic performance of first-year university students: the influence of resilience and engagement. *Higher Education Research & Development*, 37(7), 1321–1335. https://doi.org/10.1080/07294360.2018.1502258
- Aycock, D. M., & Currie, E. R. (2013). Minimizing Risks for Nursing Students Recruited for Health and Educational Research. *Nurse Educator*, 38(2), 56–60. https://doi.org/10.1097/nne.0b013e3182829c3a
- Bailey, R., Bergamin, P. B., Bosch, C., de Beer, J., du Toit, A., Goede, R., Golightly, A., Johnson, D.W., Johnson, R.T., Kruger, C. & Laubscher, D. (2019). Self-Directed Learning for the 21st Century: Implications for Higher Education. (p. 470). AOSIS.
- Bandura, A. (1990). Perceived self-efficacy in the exercise of personal agency. *Journal of applied sport psychology*, 2(2), 128-163.
- Bandura, A. (1997). Self-efficacy: The exercise of control. Freeman.
- Bandura, A. (2000). Exercise of human agency through collective efficacy. *Current directions in psychological science*, 9(3), 75-78.
- Bandura, A. (2001). Social cognitive theory: An agentic perspective. Annual review of psychology, 52(1), 1-26.
- Bandura, A. (2018). Toward a Psychology of Human Agency: Pathways and Reflections. *Perspectives on Psychological Science*, 13(2), 130–136. https://doi.org/10.1177/1745691617699280
- Bandura, A. (2018). Toward a psychology of human agency: Pathways and reflections. *Perspectives on Psychological Science*, 13(2), 130-136.
- Bandura. (1994). Self-efficacy. In *Encyclopedia of human behavior* (p. 71–81). Academic Press.
- Barker, C., King, N., Snowden, M., & Ousey, K. (2016). Study time within pre-registration nurse education: a critical review of the literature. *Nurse Education Today*, 41, 17-23.
- Barton, C. (2017). Exploring the experience of undergraduate students attending a library induction during Welcome Week at the University of Surrey. *Journal of Information Literacy*, 11(2), 105-117.
- Barton, T. D. (2007). Student nurse practitioners a rite of passage? The universality of Van Gennep's model. *Nurse Education in Practice*, 7, 338–347.
- Bate, P., & Robert, G. (2006). Experience-based design: from redesigning the system around the patient to co-designing services with the patient. BMJ quality & safety, 15(5), 307-310.

- Bate, P., & Robert, G. (2007a). Bringing user experience to healthcare improvement: the concepts, methods and practices of experience-based design. Abingdon: Radcliffe Publishing.
- Bate, P., & Robert, G. (2007b). Toward more user-centric OD: lessons from the field of experience-based design and a case study. *The Journal of Applied Behavioral Science*, 43(1), 41-66.
- Battin, J. R. (2014). Improving academic advising through student seminars: A case study. *Journal of Criminal Justice Education*, 25(3), 354-367.
- Baxter, H., Mugglestone, M. & Maher, L. (2009). *The EBD approach: experience-based design using patient and staff experience to design better healthcare services*, NHS Institute for innovation and improvement.
- Bazeley, P. (2009). Analysing qualitative data: More than 'identifying themes'. *Malaysian Journal of Qualitative Research*, 2(2), 6-22.
- Beck, C. T. (1993). Nursing students' initial clinical experience: a phenomenological study. *International Journal of Nursing Studies*, 30(6), 489-497.
- Becker, H. S., & Geer, B. (1958). The fate of idealism in medical school. *American* Sociological Review, 23(1), 50-56.
- Benckendorff, P., Ruhanen, L., & Scott, N. (2009). Deconstructing the student experience: A conceptual framework. *Journal of Hospitality and Tourism Management*, 16(01), 84-93.
- Benner, P. (1984). *From Novice to Expert: Excellence and Power in Clinical Nursing Practice*. California: Addison-Wesley.
- Bennington, K., & Laffoley, T. (2012). Beyond smiley sheets: Measuring the ROI of learning and development. UNC Executive Development.
- Betihavas, V., Bridgman, H., Kornhaber, R., & Cross, M. (2016). The evidence for 'flipping out': A systematic review of the flipped classroom in nursing education. *Nurse Education Today*, 38, 15-21.
- Bhaskar, R. (1978). A realist theory of science. Harvester.
- Bhaskar, R. (2013). The logic of scientific discovery. In M. Archer, R. Bhaskar, A. Collier, T. Lawson, & A. Norrie, (Eds.) *Critical realism: essential readings* (pp. 48-103). London: Routledge.
- Bingen, H. M., Steindal, S. A., Krumsvik, R., & Tveit, B. (2019). Nursing students studying physiology within a flipped classroom, self-regulation and off-campus activities. *Nurse Education in Practice*, 35, 55-62.
- Birks, M., Chapman, Y., Ralph, N., McPherson, C., Eliot, M., & Coyle, M. (2013). Undergraduate Nursing Studies: The First-Year Experience. *Journal of Institutional Research*, 18(1), 26-35.
- Blaikie, N. (2007). Approaches to Social Enquiry: Advancing Knowledge (2nd ed.). Polity Press.
- Boell, S. K., & Cecez-Kecmanovic, D. (2014). A Hermeneutic Approach for Conducting Literature Reviews and Literature Searches. *Communications of the Association for Information Systems*, 34(12), 257–286.

Bold, C. (2012). Analysing narrative data. Using Narrative in Research. SAGE.

- Booth, J., Tolson, D., Hotchkiss, R., & Schofield, I. (2007). Using action research to construct national evidence-based nursing care guidance for gerontological nursing. *Journal of Clinical Nursing*, 16, 945–953.
- Booth, T. L., Emerson, C. J., Hackney, M. G., & Souter, S. (2016). Preparation of academic nurse educators. *Nurse Education in Practice*, 19, 54-57.
- Borghi, S., Mainardes, E., & Silva, É. (2016). Expectations of higher education students: A comparison between the perception of student and teachers. *Tertiary Education and Management*, 22(2), 171-188.
- Bosch, C., Mentz, E. & Goede, R., (2019). Self-directed learning: A conceptual overview. In E. Mentz, J. De Beer & R. Bailey (Eds.). Self-Directed Learning for the 21st Century: Implications for Higher Education (pp. 1–36). (NWU Self-Directed Learning Series Volume 1).
- Boyd, H., McKernon, S., Mullin, B., & Old, A. (2012). Improving healthcare through the use of codesign. *The New Zealand medical journal*, 125(1357), 76-87.
- Braun, V., & Clarke, V. (2006). Using thematic analysis in psychology. Qualitative research in psychology, 3(2), 77-101.
- Braun, V., & Clarke, V. (2014). What can "thematic analysis" offer health and wellbeing researchers? *International journal of qualitative studies on health and well-being*, 9, 1-2.
- Braun, V., & Clarke, V. (2019). Reflecting on reflexive thematic analysis. *Qualitative Research in Sport, Exercise and Health*, 11(4), 589-597.
- Brimble, M. J. (2015). Does entry route really affect academic outcome? Academic achievement of traditional versus nontraditional entrants to BN (Hons) Pre-Registration nursing programmes. *Journal of Further and Higher Education*, 39(3), 379-398.
- Brinkworth, R., McCann, B., Matthews, C., & Nordström, K. (2009). First year expectations and experiences: Student and teacher perspectives. *Higher Education*, 58(2), 157-173.
- British Design Council. (2005). The design process: the 'double diamond' design process model. http://www.designcouncil.org.uk/about-design/how-designers-work/thedesign-process/. Acesso em, 11(12), 2013.
- Browne, J. (2010). Securing a Sustainable Future for Higher Education [Browne Report]. Online at: https://www. gov. uk/government/publications/the-browne-report-highereducationfunding-and-student-finance
- Brummett, B. (2011). Relativism and rhetoric. In R.A. Cherwitz, & H. W. Johnstone, (Eds). *Rhetoric and Philosophy* (pp. 79-98). Lawrence Erlbaum Associates Inc.
- Buchan, J., Charlesworth, A., Gershlick, B., & Seccombe, I. (2019). *A critical moment: NHS staffing trends, retention and attrition*. London: The Health Foundation.
- Burnett, L. (2007). Juggling first year student experiences and institutional changes: An Australian experience. In The *20th international conference on first year experience*. July, Hawaii. http://www98.griffith.edu.

- Butler-Kisber, L. (2010). Constant comparison inquiry. *Qualitative inquiry: Thematic, narrative, and arts-informed perspectives,* 24-49.
- Camp, S., & Legge, T. (2018). Simulation as a tool for clinical remediation: An integrative review. *Clinical Simulation in Nursing*, 16, 48-61.
- Cano, S., Collazos, C. A., Aristizábal, L. F., Gonzalez, C. S., & Moreira, F. (2018). Towards a methodology for user experience assessment of serious games with children with cochlear implants. *Telematics and Informatics*, 35(4), 993-1004.
- Carpmael, C., Morgan, S., & Nichols, J. (1992). Library orientation: a workable alternative?. *Library Review*. Vol. 41 No. 4.
- Carr, V. L., Sangiorgi, D., Büscher, M., Junginger, S., & Cooper, R. (2011). Integrating evidence-based design and experience-based approaches in healthcare service design. *HERD: Health Environments Research & Design Journal*, 4(4), 12-33.
- Case, J. M. (2015). A social realist perspective on student learning in higher education: the morphogenesis of agency. *Higher Education Research & Development*, 34(5), 841-852.
- Chan, Z. C., Chan, H. Y., Chow, H. C. J., Choy, S. N., Ng, K. Y., Wong, K. Y., & Yu, P. K. (2019). Academic advising in undergraduate education: A systematic review. *Nurse Education Today*, 75, 58-74.
- Chapman, R, & Orb, A. (2001). Coping strategies in clinical practice: The nursing students' lived experience. *Contemporary Nurse*, 11(1), 95-102.
- Charmaz, K. (2014). Constructing Grounded Theory (2nd ed.). Sage Publications Ltd.
- Cheng, M. (2015). *Transition Skills and Strategies: Transition Models and how Students Experience Change*. Glasgow. Scotland: Quality Assurance Agency for Higher Education.
- Clark, L., & Casey, D. (2016). Support for mentors—an exploration of the issues. *British Journal of Nursing*, 25(20), 1095-1100.
- Clarke, V., & Braun, V. (2013). Teaching thematic analysis: Overcoming challenges and developing strategies for effective learning. *The psychologist*, 26(2), 120-123.
- Cohen, L., Manion, L., & Morrison, K. (2011). *Descriptive statistics. Research Methods in Education Seventh Edition*. Routledge, 35, 622-640.
- Connell-Smith, A., & Hubble, S. (2018). Widening participation strategy in higher education in England. Briefing Paper, Number 8204, 24th January 2018.
- Cooper, J., Courtney-Pratt, H., & Fitzgerald, M. (2015). Key influences identified by first year undergraduate nursing students as impacting on the quality of clinical placement: A qualitative study. *Nurse Education Today*, 35(9), 1004-1008.
- Cooper, K., Gillmore, C., & Hogg, L. (2016). Experience-based co-design in an adult psychological therapies service. *Journal of Mental Health*, 25(1), 36-40.
- CooperGibson Research. (2018). Good practice in level 4 and 5 qualifications. Retrieved from https://www.voced.edu.au/content/ngv:81046

- Cotton, D. R., Nash, T., & Kneale, P. (2017). Supporting the retention of non-traditional students in Higher Education using a resilience framework. *European Educational Research Journal*, 16(1), 62-79.
- Courtney-Pratt, H., FitzGerald, M., Ford, K., Marsden, K., & Marlow, A. (2012). Quality clinical placements for undergraduate nursing students: A cross-sectional survey of undergraduates and supervising nurses. *Journal of Advanced Nursing*, 68(6), 1380-1390.
- Creswell, J. W. (2008). Educational research: Planning, conducting and evaluating quantitative and qualitative research (3rd ed.). Pearson Prentice Hall.
- Creswell, J. W. (2009). *Research design: Qualitative, quantitative and mixed methods approaches* (3rd ed.). Sage.
- Creswell, J. W. (2013). *Qualitative inquiry and research design: Five different approaches.* Sage
- Creswell, J. W. (2014). *Research design: Qualitative, quantitative and mixed methods approaches* (4th ed.). Sage.
- Crombie, A., Brindley, J., Harris, D., Marks-Maran, D., & Thompson, T. M. (2013). Factors that enhance rates of completion: What makes students stay?. *Nurse Education Today*, 33(11), 1282-1287.
- Crotty, M. (1998). The foundation of social research: Meaning and perspective in the research process. Sage.
- Crowe, M., & Sheppard, L. (2011). A review of critical appraisal tools show they lack rigor: alternative tool structure is proposed. *Journal of clinical epidemiology*, 64(1), 79-89.
- Danermark, B., Ekström, M., & Karlsson, J. C. (2019). *Explaining society: Critical realism in the social sciences.* Routledge.
- Data protection act (1998). Chapter 29. The Stationery Office (TSO)
- Davis, P., Scott, A., & Von Randow, M. (2013). Health research sampling methods. In M. Saks, & J. Allsop, (Eds.). *Researching health: qualitative, quantitative and mixed methods* (2nd ed.) (171-189). SAGE Publications Ltd.
- DeJonckheere, M., & Vaughn, L. M. (2019). Semistructured interviewing in primary care research: a balance of relationship and rigour. *Family Medicine and Community Health*, 7(2).
- Denzin, N. K., & Lincoln, Y. S. (2005). Introduction: the Discipline and practice of qualitative research. In N. K. Denzin, & Y. S. Lincoln, (Eds.). *The sage handbook of qualitative research* (3rd ed.) (pp.1-42). SAGE Publications, Inc.
- Denzin, N. K., & Lincoln, Y. S. (2011). Introduction: the Discipline and practice of qualitative research. In N. K. Denzin, & Y. S. Lincoln, (Eds.). *The sage handbook of qualitative research* (4th ed.) (pp.1-20). SAGE Publications, Inc.
- Denzin, N. K., & Lincoln, Y. S. (2013). *The SAGE Handbook of Qualitative Methods in Health Research.* 2013, 97126.
- Department of Health (DH). (2008). *Framing the Nursing and Midwifery Contribution:* Driving up the Quality of Care. Department of Health

- Dewar, B., Mackay, R., Smith, S., Pullin, S., & Tocher, R. (2010). Use of emotional touchpoints as a method of tapping into the experience of receiving compassionate care in a hospital setting. *Journal of Research in Nursing*, 15(1), 29-41.
- Dewey, J. (1923). *Democracy and education: An introduction to the philosophy of education*. Macmillan.
- Dewey, J. (1972). My pedagogic creed. In J. Dewey. (2013). The Early Works of John Dewey, 1882—1898, 15 vols:(1969—72) The Middle Works of John Dewey, 1899—1924, 5 vols (1976—88) The Later Works of John Dewey, 1925—1953, 17 vols. (1981—91). The Cambridge Companion to Pragmatism, 345.
- Dewey, J. (1984). Qualitative Thought. John Dewey: *The Later Works*, 5, 243-62. (Original work published 1930)
- Dewey, J. (2005). Art as experience. Penguin Group. (Original work published 1934)
- Dewey, J. (2015). Experience and Education. Free Press. (Original work published 1938)
- Dimitriadou, M., Papastavrou, E., Efstathiou, G., & Theodorou, M. (2015). Baccalaureate nursing students' perceptions of learning and supervision in the clinical environment. *Nursing & health sciences*, 17(2), 236-242.
- Dimopoulos-Bick, T. L., O'Connor, C., Montgomery, J., Szanto, T., Fisher, M., Sutherland, V., Baines, H., Orcher, P., Stubbs, J., Maher, L., Verma, R., & Palmer, V. J. (2019). "Anyone can co-design?": A case study synthesis of six experience-based co-design (EBCD) projects for healthcare systems improvement in New South Wales, Australia. *Patient Experience Journal*, 6(2), 93-104.
- Donaldson, J. H., McCallum, J., & Lafferty, P. (2010). Can we predict successful completion of the common foundation programme at interview? *Nurse Education Today*, 30(7), 649–656.
- Donetto, S., Pierri, P., Tsianakas, V., & Robert, G. (2015). Experience-based Co-design and Healthcare Improvement: Realizing Participatory Design in the Public Sector. *The Design Journal*, 18(2), 227-248.
- Donetto, S., Tsianakas, V., & Robert, G. (2014). Using Experience-based Co-design (EBCD) to improve the quality of healthcare: mapping where we are now and establishing future directions. King's College London.
- Donough G., & Van der Heever M. (2018). Undergraduate nursing students' experience of clinical supervision. *Curationis*. 41 (1) (pp e1-e8), 2018.
- Dowell, F. J., King, A., & Bain, M. M. (2016). *Managing Student Expectations During Transition*. Enlighten Publications.
- Downe, S., Simpson, L., & Trafford, K. (2007). Expert intrapartum maternity care: a metasynthesis. *Journal of Advance Nursing*, 257(2):127-40.
- Doyle, K., Sainsbury, K., Cleary, S., Parkinson, L., Vindigni, D., McGrath, I., & Cruickshank, M. (2017). Happy to help/happy to be here: Identifying components of successful clinical placements for undergraduate nursing students. *Nurse Education Today*, 49, 27-32.
- Duchscher, J. B., & Windey, M. (2018). Stages of transition and transition shock. *Journal for Nurses in Professional Development*, 34(4), 228-232.

- Eby, L. T., Butts, M. M., Durley, J., & Ragins, B. R. (2010). Are bad experiences stronger than good ones in mentoring relationships? Evidence from the protégé and mentor perspective. *Journal of Vocational Behavior*, 77(1), 81-92.
- Edwards, M., & Chalmers, K. (2002). Double agency in clinical research. *Canadian Journal* of Nursing Research Archive.
- El Ansari, W. (2002). Student nurse satisfaction levels with their courses: Part I effects of demographic variables. *Nurse Education Today*, 22, 159-170.
- Eller, L. S., Lev, E. L., & Feurer, A. (2014). Key components of an effective mentoring relationship: A qualitative study. *Nurse Education Today*, 34(5), 815-820.
- Eraut, M. (2000). Non-formal learning and tacit knowledge in professional work. *British Journal of Educational Psychology.* 70, 113–136.
- Faranda, W. T., & Clarke, I. (2004). Student observations of outstanding teaching: implications for marketing educators. *Journal of Marketing Education*, 26(3), 271-272.
- Ferguson, L. M., Yonge, O., & Myrick, F. (2004). Students' involvement in faculty research: Ethical and methodological issues. *International Journal of Qualitative Methods*, 3(4), 56-68.
- Ferreri, S. P., & O'Connor, S. K. (2013). Redesign of a large lecture course into a smallgroup learning course. *American journal of pharmaceutical education*, 77(1).
- Fetter, M. S. (2009). Curriculum strategies to improve baccalaureate nursing information technology outcomes. *Journal of Nursing Education*, 48(2), 78-85.
- Fielding, A. H., Dunleavy, P. J., & Langan, A. M. (2010) Effective use of the UK's National Student (Satisfaction) Survey (NSS) data in science and engineering subjects. *Journal of Further and Higher Education*, 33, 347–368.
- Fleming, S., & McKee, G. (2005). The mature student question. *Nurse Education Today,* 25, 230–237.
- Fletcher, A. J. (2017). Applying critical realism in qualitative research: methodology meets method. *International journal of social research methodology*, 20(2), 181-194.
- Frank, A. W. (2010). *Letting stories breathe: A socio-narratology*. University of Chicago Press.
- Frank, A. W. (2012). Practicing dialogical narrative analysis. *Varieties of narrative analysis*, 2010, 33-52.
- Frauenberger, C., Good, J., Fitzpatrick, G., & Iversen, O. S. (2015). In pursuit of rigour and accountability in participatory design. *International journal of human-computer studies*, 74, 93-106.
- Freeman, S., Theobald, R., Crowe, A. J., & Wenderoth, M. P. (2017). Likes attract: Students self-sort in a classroom by gender, demography, and academic characteristics. *Active Learning in Higher Education*, 18(2), 115-126.
- Gaba, D. M. (2007). The future vision of simulation in healthcare. *Simulation in Healthcare*, 2(2), 126-135.

- Gale, T., & Parker, S. (2014). Navigating change: a typology of student transition in higher education. *Studies in Higher Education*, 39(5), 734-753.
- Gasson, S. (2009). Employing a Grounded Theory Approach for MIS Research. In Y. K. Dwivedi, B. Lal, M. D. Williams, S. L. Schneberger, & M. Wade, (Eds.). Handbook of Research on Contemporary Theoretical Models in Information Systems (pp. 34-56). Hershey PA: IGI Publishing.
- Gibney, A., Moore, N., Murphy, F., & O'Sullivan, S. (2011). The first semester of university life; 'will I be able to manage it at all? *Higher Education*, 62, 351–366.
- Gorman, G. E., & Clayton, P. (1997). Writing qualitative research reports. *Qualitative research for the information professional: A practical handbook*, 222-239.
- Gough, D., Oliver, S., & Thomas, J. (2012). *An Introduction to Systematic Reviews*. Sage Publications Ltd.
- Grant, B. M., & Giddings, L. S. (2002). Making sense of methodologies: A paradigm framework for the novice researcher. *Contemporary nurse*, 13(1), 10-28.
- Grealish, L., & Trevitt, C. (2005). Developing a professional identity: Student nurses in the workplace. *Contemporary Nurse,* 19(1-2), 137-150.
- Grey, D., & Lochtie, D. (2016). Comparing personal tutoring in the UK and academic advising in the US. *Academic Advising Today*, 39(3).
- Grimshaw, D. J. (2008). Information Overload. *The International Encyclopedia of Communication.*
- Grites, T. J., Miller, M. A., & Voler, J. G. (Eds.). (2016). *Beyond foundations: Developing as a master academic advisor.* John Wiley & Sons.
- Grix, J. (2002). Introducing students to the generic terminology of social research. *Politics*, 22(3), 175-186.
- Grobecker, P. A. (2016). A sense of belonging and perceived stress among baccalaureate nursing students in clinical placements. *Nurse Education Today*, 36, 178-183.
- Gros, B., & López, M. (2016). Students as co-creators of technology-rich learning activities in higher education. *International Journal of Educational Technology in Higher Education*, 13(1), 1-13.
- Guba, E. G. (1990). The paradigm dialog. In *Alternative paradigms conference, mar, 1989, Indiana, school of education.* Sage Publications, Inc.
- Guba, E. G., & Lincoln, Y. S. (2005). Paradigmatic Controversies, Contradictions, and Emerging Confluences. In: N. K. Denzin, & Y. S. Lincoln, (Eds.). *The Sage handbook of qualitative research* (3rd ed.) (pp.1-42). SAGE Publications, Inc, 191-215.
- Guest, G., MacQueen, K. M., & Namey, E. E. (2012). *Applied thematic analysis*. SAGE Publications Ltd.
- Guglielmino, L. M. (2013). The case for promoting self-directed learning in formal educational institutions. *South African Education Journal* (SA e-DUC), 10(2).
- Guzzardo, M. T., Khosla, N., Adams, A. L., Bussmann, J. D., Engelman, A., Ingraham, N., Gamba, R., Jones-Bey, A., Moore, M.D., Toosi, N.R. & Taylor, S. (2021). "The

Ones that Care Make all the Difference": Perspectives on Student-Faculty Relationships. *Innovative Higher Education*, 46(1), 41-58.

- Halfpenny, P. (2014). *Positivism and Sociology (RLE Social Theory): Explaining Social Life*. Routledge.
- Hamshire, C., Barrett, N., Langan, M., Harris, E., & Wibberley, C. (2017). Students' perceptions of their *learning experiences: A repeat regional survey of healthcare students. Nurse Education Today*, 49, 168-173.
- Hamshire, C., Willgoss, T. G., & Wibberley, C. (2012). 'The placement was probably the tipping point'–The narratives of recently discontinued students. *Nurse Education in Practice*, 12(4), 182-186.
- Hamshire, C., Willgoss, T. G., & Wibberley, C. (2013). What are reasonable expectations? Healthcare student perceptions of their programmes in the North West of England. *Nurse Education Today*, 33(2), 173-179.
- Handwerker, S. M. (2012). Transforming nursing education: A review of current curricular practices in relation to Benner's latest work. *International Journal of Nursing Education Scholarship*, 9(1).
- Harris, A. D., McGregor, J. C., Perencevich, E. N., Furuno, J. P., Zhu, J., Peterson, D. E., & Finkelstein, J. (2006). The Use and Interpretation of Quasi-Experimental Studies in Medical Informatics. *Journal of the American Medical Informatics Association*, 13(1), 16–23.
- Hart, J. A., & Swenty, C. F. (2016). Understanding transitions to promote student success: A concept analysis. *Nursing forum*, 51(3), 180-185.
- Harvey, L., Burrows, A., & Green, D. (1992) *Total Student Experience: A First Report of the QHE National Survey of Staff and Students' Views of the Important Criteria for Assessing the Quality of Higher Education*. Quality of Higher Education.
- Hawk, T. F. (2017). Getting to know your students and an educational ethic of care. *Journal* of Management Education, 41(5), 669-686.
- Hazlett, A. (2015). The maturation of the Gettier problem. *Philosophical Studies*, 172(1), 1-6.
- Heale, R., & Twycross, A. (2015). Validity and reliability in quantitative studies. *Evidence-based nursing*, 18(3), 66-67.
- Heaslip, V., Board, M., Duckworth, V., & Thomas, L. (2017). Widening participation in nurse education: An integrative literature review. *Nurse Education Today*, 59, 66-74.
- Heath, S. (2019). Next steps for student supervision and assessment. *Practice Nursing*, 30(10), 496-500.
- Heinonen, A. T., Kääriäinen, M., Juntunen, J., & Mikkonen, K. (2019). Nursing students' experiences of nurse teacher mentoring and beneficial digital technologies in a clinical practice setting. *Nurse Education in Practice*, 40, 102631.
- Henderson, A., Rowe, J., Watson, K., & Hitchen-Holmes, D. (2016). Graduating nurses' self-efficacy in palliative care practice: An exploratory study. *Nurse Education Today*, 39, 141-146.
- Hetzel-Riggin, M. D., Swords, B. A., Tuang, H. L., Deck, J. M., & Spurgeon, N. S. (2019). Work engagement and resiliency impact the relationship between nursing stress and burnout. *Psychological reports*, 0(0), 1-19.
- Higher Education Statistics Agency (HESA). (2015). UKPIs Definitions. Retrieved from https://www.hesa.ac.uk/content/view/2379/
- Higher Education Statistics Agency. (2021, January 21). *Higher Education Student Statistics: UK, 2019/20 Subjects studied | HESA*. Https://Www.Hesa.Ac.Uk. https://www.hesa.ac.uk/news/27-01-2021/sb258-higher-education-student-statistics/subjects
- Holland, B., Landry, K., Mountain, A., Middlebrooks, M. A., Heim, D., & Missildine, K. (2013). Weaving the tapestry of learning: simulation, standardized patients, and virtual communities. *Nurse educator*, 38(6), 269-272.
- Holstein, J. A., & Gubrium, J. F. (Eds.). (2012). Varieties of narrative analysis. Sage Publications.
- Horn, M. B. (2013). The transformational potential of flipped classrooms. *Education Next*, 13(3), 78-79.
- Hu, X. (2018). Methodological implications of critical realism for entrepreneurship research. *Journal of Critical Realism*, 17(2), 118-139.
- Hunt, E. A., Duval-Arnould, J., Chime, N. O., Auerbach, M., Kessler, D., Duff, J. P., Shilkofski, N., Brett-Fleegler, M., Nadkarni, V., & Cheng, A. (2016a). Building consensus for the future of paediatric simulation: a novel 'KJ Reverse-Merlin'methodology. *BMJ Simulation and Technology Enhanced Learning*, 2(2).
- Hunt, L. A., McGee, P., Gutteridge, R., & Hughes, M. (2016b). Failing securely: The processes and support which underpin English nurse mentors' assessment decisions regarding under-performing students. *Nurse Education Today*, 39, 79-86.
- Huq, A., & Gilbert, D. (2017). All the world's a stage: transforming entrepreneurship education through design thinking. *Education+ Training*.
- Hyvarinen, M. (2008). *Analyzing narratives and story-telling.* The Sage handbook of social research methods, 1, 447-460.
- Iatrellis, O., Kameas, A., & Fitsilis, P. (2017). Academic advising systems: A systematic literature review of empirical evidence. *Education Sciences*, 7(4), 90.
- Information Commissioner's Office. (2018). Guide to the General Data Protection Regulation (GDPR). https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attac hment_data/file/711097/guide-to-the-general-data-protection-regulation-gdpr-1-0.pdf
- Irvine, V., Code, J., & Richards, L. (2013). Realigning higher education for the 21st century learner through multi-access learning. *Journal of Online Learning and Teaching*, 9(2), 172.
- Irwin, C., Ball, L., Desbrow, B., & Leveritt, M. (2012). Students' perceptions of using Facebook as an interactive learning resource at university. *Australasian Journal of Educational Technology*, 28(7), 1221–1232.

- Jacklin, A., & Le Riche, P. (2009). Reconceptualising student support: from 'support' to 'supportive'. *Studies in Higher Education*, 34(7), 735-749.
- Jacklin, A., & Robinson, C. (2007). What is meant by 'support' in higher education? Towards a model of academic and welfare support. *Journal of Research in Special Educational Needs*, 7(2), 114-123.
- Jackson, D. (2017). Developing pre-professional identity in undergraduates through workintegrated learning. *Higher Education*, 74(5), 833-853.
- Jackson, S., & Steven, A. (2020). Discourse tensions: Professional socialisation into nursing within the university education system. *Nurse Education Today*, 91, 104478.
- James, A., & Chapman, Y. (2010). Preceptors and patients the power of two: nursing student experiences on their first acute clinical placement. *Contemporary Nurse*, 34 (1), 34-47.
- Javadi, M., & Zarea, M. (2016). Understanding Thematic Analysis and its Pitfalls. *Journal of Client Care*, 1 (1), 33-39.
- Jeppesen, K. H., Christiansen, S., & Frederiksen, K. (2017). Education of student nurses–A systematic literature review. *Nurse Education Today*, 55, 112-121.
- Joffe, H. (2012). Thematic analysis. *Qualitative research methods in mental health and psychotherapy: A guide for students and practitioners*, 1, 210-223.
- Johnson, D.W. & Johnson, R.T., (2019). The impact of cooperative learning on self-directed learning. In E. Mentz, J. De Beer., & R. Bailey. (Eds.), *Self-Directed Learning for the 21st Century: Implications for Higher Education* (NWU Self-Directed Learning Series Volume 1), pp. 37–66
- Jones, R. (2018). The student experience of undergraduate students: towards a conceptual framework. *Journal of Further and Higher Education*, 42(8), 1040-1054.
- Jones, R., Hansen, M., Kaddoura, M., Schwab-McCoy, A., & Tocchini, K. (2018). The incidence of nursing students' perceived stress and burnout levels at a private university in California. *Journal of Nursing Education and Practice*, 8(10), 138-151.
- Judger, N. (2016). The thematic analysis of interview data: An approach used to examine the influence of the market on curricular provision in Mongolian higher education institutions. *Hillary Place Papers* (3 rd ed.), University of Leeds.
- Jung, I., & Hong, S. (2014). An elaborated model of student support to allow for gender considerations in Asian distance education. *International Review of Research in Open and Distributed Learning*, 15(2), 170-188.
- Juutilainen, M., Metsäpelto, R. L., & Poikkeus, A. M. (2018). Becoming agentic teachers: Experiences of the home group approach as a resource for supporting teacher students' agency. *Teaching and Teacher Education*, 76, 116-125.
- Kable, A. K., Pich, J., & Maslin-Prothero, S.E. (2012). A structured approach to documenting a search strategy for publication: a 12 step guideline for authors. *Nurse Education Today*, 32(8), 878–886.

- Kaihlanen, A. M., Lakanmaa, R. L., & Salminen, L. (2013). The transition from nursing student to registered nurse: The mentor's possibilities to act as a supporter. *Nurse Education in Practice*, 13(5), 418-422.
- Kalamkarian, H. S., & Karp, M. J. M. (2015). *Student attitudes toward technology-mediated advising systems*. Community College Research Center, Teachers College, Columbia University.
- Keeling, R. P. (2014). An ethic of care in higher education: Well-being and learning. *Journal* of College and Character, 15(3), 141-148.
- Khalid, S., Saeed, M., & Syed, S. (2016). Impact of Information Overload on Students' Learning: An Empirical Approach. *FWU Journal of Social Sciences*, 10(1), 58-66.
- King, N. (2004). Using templates in the thematic analysis of text. In C. Cassell & G. Symon (Eds.), Essential guide to qualitative methods in organizational research (pp. 257– 270). Sage
- Kivunja, C., & Kuyini, A. B. (2017). Understanding and applying research paradigms in educational contexts. *International Journal of higher education*, 6(5), 26-41.
- Klemencic, M. (2015). What is student agency? An ontological exploration in the context of research on student engagement. *Student engagement in Europe: Society, higher education and student governance*, 11-29.
- Klemencic, M. (2017). From Student Engagement to Student Agency: Conceptual Considerations of European Policies on Student-Centered Learning in Higher Education. *Higher Education Policy*, 30, 69-85.
- Kolb, D. (1984). *Experiential Learning, Experience as the Source of Learning and Development*. Prentice Hall.
- Kuiper, R., Heinrich, C., Matthias, A., Graham, M. J., & Bell-Kotwall, L. (2008). Debriefing with the OPT model of clinical reasoning during high fidelity patient simulation. *International Journal of Nursing Education Scholarship*, 5(1), 1-14.
- Kumaran, S., & Carney, M. (2014) Role transition from student nurse to staff nurse: Facilitating the transition period. *Nurse Education in Practice*, 14(6), 605-611.
- Kynigos, C. (2012). Niches for Constructionism: forging connections for practice and theory. In Proc. of Constructionism 2012 Conference, Athens, Greece.
- Labov, W., & Waletsky, J. (1967). Narrative analysis. In J. Helm (Ed.) *Essays on the Verbal and Visual Arts* (pp.12–44). University of Washington Press.
- Labrague, L. J., McEnroe-Petitte, D. M., Bowling, A. M., Nwafor, C. E., & Tsaras, K. (2019, July). High-fidelity simulation and nursing students' anxiety and self-confidence: A systematic review. *In Nursing forum*, 54(3), 358-368.
- Ladyman, J. (2007). Ontological, Epistemological, and Methodological Positions. In T. A. F. Kuipers, (Ed.), *General Philosophy of Science: Focal Issues* (pp. 303-376). North-Holland.
- Laing, C., Chao, K. M., & Robinson, A. (2005). Managing the expectations of non-traditional students: a process of negotiation. *Journal of Further and Higher Education*, 29(2), 169-179.

- Larsen–freeman, D. (2019). On language learner agency: A complex dynamic systems theory perspective. *The Modern Language Journal*, 103, 61-79.
- Larson, J., Johnson, A., Aiken-Wisniewski, S. A., & Barkemeyer, J. (2018). What is academic advising? An application of analytic induction. *NACADA Journal*, 38(2), 81-93.
- Last, L., & Fulbrook, P. (2003). Why do student nurses leave? Suggestions from a Delphi Study. *Nurse Education Today*, 23 (6), 449-458.
- Lavoie, P., & Clarke, S. P. (2017). Simulation in nursing education. *Nursing management*, 48(2), 16-17.
- Leeman, Y., & Wardekker, W. (2011). Redesigning vocational education: The possibilities of design-based research. *Journal of Curriculum Studies*, 43(3), 313-331.
- Levers, M. J. D. (2013). Philosophical paradigms, grounded theory, and perspectives on emergence. *Sage Open*, 3(4), 2158244013517243.
- Levett-Jones, T., & Lathlean, J. (2008). Belongingness: a prerequisite for nursing students' clinical learning. *Nurse Education in Practice*, 8, 103-111.
- Lewin, K. (1951). Field Theory in Social Sciences. Harper row.
- Lewis, C., & Ritchie, J. (2003). Generalising from qualitative research. In J. Ritchie, J. & Lewis, C, (Eds). *Qualitative research practice* (pp. 263-286). SAGE Publications Ltd.
- Liljedahl, M., Björck, E., Kalén, S., Ponzer, S., & Laksov, K.B. (2016). To belong or not to belong: nursing students' interactions with clinical learning environments an observational study. *BMC Medical Education*. 16, 197
- Lincoln, Y. S., & Guba, E. G. (1985). Establishing trustworthiness. *Naturalistic inquiry*, 289(331), 289-327.
- Lincoln, Y. S., & Guba, E. G. (1986). But is it rigorous? Trustworthiness and authenticity in naturalistic evaluation. *New directions for program evaluation*, 1986(30), 73-84.
- Locock, L., Robert, G., Boaz, A., Vougioukalou, S., Shuldham, C., Fielden, J., Ziebland, S., Gager, M., Tollyfieldd, R., & Pearcey, J. (2014). Testing accelerated experiencebased co-design: a qualitative study of using a national archive of patient experience narrative interviews to promote rapid patient centred service improvement. *Health Serves and Delivery Research*. 2(4), 1-150.
- Lowe, S. D. (2005). Responding to learner needs in distance education: Providing academic and relational support (PARS). Making distance education work: *Understanding learning and learners at a distance*, 73-87.
- Macfarlane, B. (2011). The morphing of academic practice: Unbundling and the rise of the para-academic. *Higher Education Quarterly*, 65(1), 59-73.
- Magnussen, L., & Amundson, M. J. (2003). Undergraduate nursing student experience. Nursing & Health Sciences, 5(4), 261-267.
- Maguire, M., & Delahunt, B. (2017). Doing a thematic analysis: A practical, step-by-step guide for learning and teaching scholars. *The All Ireland Journal of Teaching and Learning in Higher Education*, 9(3).

- Malouf, N., & West, S. (2015). "You'll be the RN soon": Separating performative roles of student and registered nurse. *Journal of Nursing Education and Practice*, 5(10), 120-128.
- Manyukhina, Y., & Wyse, D. (2019). Learner agency and the curriculum: a critical realist perspective. *The Curriculum Journal*, 30(3), 223-243.
- Mauss, I. B., & Robinson, M. D. (2009) Measures of emotion: A review, *Cognition and Emotion*, 23:2, 209-237.
- Maxwell, J. A. (2012). A Realist Approach for Qualitative Research. SAGE Publications.
- McInnis, C. (1998). Cultivating independent learning in the first year: New challenges in a changing context. In *Third Pacific Rim Conference of the First Year of Higher Education.*, Auckland.
- Mckendry, S., Wright, M., & Stevenson, K. (2014). Why here and why stay? Students' voices on the retention strategies of a widening participation university. *Nurse Education Today*, 34(5), 872-877.
- Mercer, S. (2011). Understanding learner agency as a complex dynamic system. *System*, 39(4), 427-436.
- Mercer, S. (2012). The complexity of learner agency. *Apples-Journal of Applied Language Studies*. 6(2), 41-59.
- Mills, A., Ryden, J., & Knight, A. (2020). Juggling to find balance: hearing the voices of undergraduate student nurses. *British Journal of Nursing*, 29(15), 897-903.
- Milton-Wildey, K., Kenny, P., Parmenter, G., & Hall, J. (2014). Educational preparation for clinical nursing: The satisfaction of students and new graduates from two Australian universities. *Nurse Education Today*, 34(4), 648-654.
- Mingers, J., & Willcocks, L. (2004). Social theory and philosophy for information systems. John Wiley series in information systems. John Wiley & Sons.
- Money, J., Nixon, S., & Graham, L. (2020). Do educational experiences in school prepare students for university? A teachers' perspective. *Journal of Further and Higher Education*, 44(4), 554-567.
- Money, J., Nixon, S., & Graham, L. (2020). Do educational experiences in school prepare students for university? A teachers' perspective. *Journal of Further and Higher Education*, 44(4), 554-567.
- Money, J., Nixon, S., Tracy, F., Hennessy, C., Ball, E., & Dinning, T. (2017). Undergraduate student expectations of university in the United Kingdom: What really matters to them?. Cogent Education, 4(1), 1301855.
- Moon, K., & Blackman, D. (2014). A guide to understanding social science research for natural scientists. *Conservation Biology*, 28(5), 1167-1177.
- Moreton, I. S., Newton, D. P., Newton, L. D., & Mathias, J. (2017). The selection of potential undergraduate students who lack customary academic qualifications: is a toolkit possible?. *Widening Participation and Lifelong Learning*, 19(3), 64-84.
- Morgan, M. (2012). *Improving the student experience: a practical guide for universities and colleges.* Routledge.

- Morley, D. A., Wilson, K., & Holbery, N. (2019). Introduction: taking steps towards a new vision of nursing and midwifery practice learning. In D. A. Morley, K. Wilson, & N. Holbery, (Eds.) Facilitating Learning in Practice: a research based approach to challenges and solutions, (pp. 1-20) Routledge.
- Morris, T. H. (2018). Vocational education of young adults in England: A systemic analysis of teaching–learning transactions that facilitate self-directed learning. *Journal of Vocational Education & Training*, 70(4), 619-643.
- Moule, P. (2015). *Making sense of research in nursing, health and social care*. Sage Publications Ltd.
- Mujtaba, U. (2012). Enhancing the student Experience: setting up a student experience unit. In G. Bell, J. Warwick, & P. Galbraith, (Eds.). *Higher education management and operational research: demonstrating new practices and metaphors* (pp.153-166). Rotterdam: Sense publishers.
- Mukerji, S., & Tripathi, P. (2014). *Handbook of Research on Transnational Higher Education.* IGI Global.
- Murphy, C. (1992). Feminist epistemology and the question of relativism. In J. E. Bayley, (Ed.). *Aspects of Relativism: Moral, Cognitive, and Literary* (pp.133-140). University Press of America.
- Myers, J. (2008). Is personal tutoring sustainable? Comparing the trajectory of the Personal tutor with that of the residential warden. *Teaching in Higher Education*, 13(5), 607-611.
- NACADA: The Global Community for Academic Advising. (2006). Concept of academic advising. Retrieved from https://www.nacada.ksu.edu/Resources/Pillars/Concept.aspx
- Nagel, J., San Juan, V., & Mar, R. A. (2013). Lay denial of knowledge for justified true beliefs. *Cognition*, 129(3), 652-661.
- National Audit Office (NAO) (2007) Staying the course: the retention of students in higher education. The Stationary Office
- née Blackwell, R. W., Lowton, K., Robert, G., Grudzen, C., & Grocott, P. (2017). Using experience-based co-design with older patients, their families and staff to improve palliative care experiences in the emergency department: a reflective critique on the process and outcomes. *International Journal Of Nursing Studies*, 68, 83-94.
- Neill K.M., McCoy A.K., Parry C.B., Cohran J., Curtis J.C., Ransom R.B. (1998). The clinical experience of novice students in nursing. *Nurse educator.* 23 (4), 16-21.
- Neves, J., & Hillman, N. (2016). The 2016 Student Academic Experience Survey. *Higher Education Academy.*
- Newhouse, R. P., Hoffman, J. J., Suflita, J., & Hairston, D.P. (2007). Evaluating an innovative programme to improve new nurse graduate socialization into the acute healthcare setting. *Nursing Administration Quarterly*, 31(1), 50–60.
- NHS Education for Scotland. (2013). The National Nursing and Midwifery Practice Education Facilitator Network: Their contribution to practice learning in Scotland. Available at https://www.nes.scot.nhs.uk/media/1887900/pef_review_web.pdf

- NHS Institute for Innovation and Improvement. (2009). *The EBD approach. Guide and tools.* NHS Institute for Innovation and Improvement.
- Nola, R., & Irzik, G. (2006). *Philosophy, science, education and culture* (Vol. 28). Springer Science & Business Media.
- Norrie, P., & Dalby, D. (2007). How adult are our learners? A cross-sectional exploration of the learning characteristics of nursing students in a United Kingdom University. *Journal of Research in Nursing*, 12(4), 319-328.
- Nowell, L. S., Norris, J. M., White, D. E., & Moules, N. J. (2017). Thematic Analysis: Striving to Meet the Trustworthiness Criteria. *International Journal of Qualitative Methods*, 16(1), 1609406917733847.
- Nursing and Midwifery Council. (2006a). *The Regulation of Health Care Support Workers: A Scoping Paper.* Nursing and Midwifery Council.
- Nursing and Midwifery Council. (2006b). *Standards to support learning and assessment in practice*. Nursing and Midwifery Council.
- Nursing and Midwifery Council. (2007). *Simulation and Practice Learning Project*. Nursing and Midwifery Council.
- Nursing and Midwifery Council. (2008). Standards to support learning and assessment in practice NMC standards for mentors, practice teachers and teachers. Nursing and Midwifery Council.
- Nursing and Midwifery council. (2010). *Standards for Pre-Registration nursing education*. Nursing and Midwifery Council.
- Nursing and Midwifery Council. (2015). *Equality and diversity report 1 March 2013 to 31 March 2014.* Nursing and Midwifery Council.
- Nursing and Midwifery Council. (2018a). *Realising professionalism: Standards for education and training. Part 2: Standards for student supervision and assessment.* Nursing and Midwifery Council.
- Nursing and Midwifery Council. (2018b). *Education standards consultation response*. Nursing and Midwifery Council.
- Nursing and Midwifery Council. (2018c). *Enabling professionalism in nursing and midwifery practice*. Nursing and Midwifery Council.
- Nursing and Midwifery Council. (2019). Supernumerary and protected learning time. Nursing and Midwifery Council.
- Nutt, L., & Bell, L. (2002). Divided loyalties, divided expectations: Research ethics, professional and occupational responsibilities. *Ethics In Qualitative Research*, 70-90.
- O'Brien, F., Keogh, B., & Neenan, K. (2009). Mature students' experiences of undergraduate nurse education programmes: The Irish experience. *Nurse Education Today*, 29 (6), 635–640.
- O'Donnell, V. L., Kean, M., & Stevens, G. (2016). Student transition in higher education. *Higher Education Academy*.

- O'Keeffe, P. (2013). A sense of belonging: Improving student retention. *College Student Journal*, 47(4), 605-613.
- Onda, E. L. (2012). Situated cognition: Its relationship to simulation in nursing education. *Clinical simulation in nursing*, 8(7), e273-e280.
- Ormston, Spencer, Barnard & Snape, (2014). The Foundations of Qualitative Research. In J. Ritchie, J. Lewis, C, Nicholls & R. Ormston, (Eds). *Qualitative research practice* (2nd ed.) (pp. 1-23). SAGE Publications Ltd.
- Ousey, K. (2009). Socialization of student nurses-the role of the mentor. *Learning in Health and Social Care*, 8(3), 175-184.
- Palmer, V. J., Chondros, P., Piper, D., Callander, R., Weavell, W., Godbee, K., Potiriadis, M., Richard, L., Densely, K., Herrman, H. & Gunn, J. (2015). The CORE study protocol: a stepped wedge cluster randomised controlled trial to test a co-design technique to optimise psychosocial recovery outcomes for people affected by mental illness in the community mental health setting. *BMJ open*, 5(3), e006688.
- Palominos, E., Levett-Jones, T., Power, T., & Martinez-Maldonado, R. (2019). Healthcare students' perceptions and experiences of making errors in simulation: An integrative review. *Nurse Education Today*, 77, 32-39.
- Pandey, S. C., & Patnaik, S. (2014). Establishing reliability and validity in qualitative inquiry: A critical examination. *Jharkhand journal of development and management studies*, 12(1), 5743-5753.
- Papastavrou, E., Dimitriadou, M., Tsangari, H., & Andreou, C. (2016). Nursing students' satisfaction of the clinical learning environment: a research study. *BMC nursing*, 15(1), 44.
- Parahoo, K. (2014). *Nursing research principles, process and issues* (3rd ed.). Palgrave Macmillan.
- Patterson, W. (2013). Narratives of events; La bovian narrative analysis and its limitations. *Doing narrative research*, 127.
- Petit, S. C., & Huault, I. (2008). From practice-based knowledge to the practice of research: Revisiting constructivist research works on knowledge. *Management learning*, 39(1), 73-91.
- Pham, L. T. M. (2018). Qualitative approach to research a review of advantages and disadvantages of three paradigms: Positivism, interpretivism and critical inquiry. *University of Adelaide*.
- Pickard, A., & Dixon, P. (2004). The applicability of constructivist user studies: How can constructivist inquiry inform service providers and systems designers. *Information research*, 9(3), 9-3.
- Pickles, J., Hide, E., & Maher, L. (2008). Experience based design: a practical method of working with patients to redesign services. *Clinical Governance: An International Journal*, 13(1), 51-58.
- Pigafetta, A., & Cachey, T. J. (2007). *The first voyage around the world, 1519-1522: an account of Magellan's expedition.* University of Toronto Press.

- Pitkethly, A., & Prosser, M. (2001). The First Year Experience Project: A model for university-wide change. *Journal of Higher Education Research & Development*, 20(2), 185–198.
- Pitt, V., Powis, D., Levett-Jones, T., & Hunter, S. (2012). Factors influencing nursing students' academic and clinical performance and attrition: an integrative literature review. *Nurse Education Today*, 32, 903-913.
- Plain, C. (2007). Build an affinity for KJ method. Quality Progress, 40(3), 88.
- Polit, D. F., & Beck, C. T. (2009). Essentials of nursing research: Appraising evidence for nursing practice. Lippincott Williams & Wilkins.
- Porteous, D. J., & Machin, A. (2018). The lived experience of first year undergraduate student nurses: A hermeneutic phenomenological study. *Nurse Education Today*, 60, 56-61.
- Price, L., Hastie, L., Duffy, K., Ness, V., & McCallum, J. (2011). Supporting students in clinical practice: Pre-registration nursing students' views on the role of the lecturer. *Nurse Education Today*, 31(8), 780-784.
- Pryce-Miller, M., & Serrant, L. (2019). Students' perceptions of self-direction in preregistration nurse education. *Nurse Education in Practice*, 40, 102626.
- Randle, J. (2003). Bullying in the nursing profession. *Journal of Advanced Nursing*, 43(4), 395 401.
- Richardson, K. J., & Claman, F. (2014). High-fidelity simulation in nursing education: A change in clinical practice. *Nursing Education Perspectives*, 35(2), 125-127.
- Riessman, C. K. (2008). Narrative methods for the human sciences. Sage.
- Robert, G. (2013). Participatory action research: using experience-based co-design to improve the quality of healthcare services. In S. Ziebland, A. Coulter, J. D. Calabrese & L. Locock (Eds), *Understanding and Using Health Experiences: Improving patient care*. Oxford University Press
- Roberts, P. A., Dunworth, K., & Boldy, D. (2018). Towards a reframing of student support: A case study approach. *Higher Education*, 75(1), 19-33..
- Robinson, C., & Snipes, K. (2009). Hope, optimism and self-efficacy: A system of competence and control enhancing African American college students academic well-being. *Multiple Linear Regression Viewpoints*, 35(2), 16-26.
- Roets, L., & Maritz, J. (2017). Facilitating the development of higher-order thinking skills (HOTS) of novice nursing postgraduates in Africa. *Nurse Education Today*, 49, 51-56.
- Roman, N. V., & Dison, A. (2016). Relationship between student preparedness, learning experiences and agency: Perspectives from a South African university. *African Journal of Health Professions Education*, 8(1), 30-32.
- Rosborough, J. (2018). Student experience management: everyone else is doing it so why can't we?. In: Chartered Association of Business Schools Learning, Teaching & Student Experience Conference (LTSE 2018), Radisson Blu Hotel, Glasgow, Scotland, 24-25 April 2018.
- Ross, T. (2012). A survival guide for health research methods. Open University Press.

- Royal-Fearn, J. (2019) Separating the mentor role brings risks and opportunities. *Nursing Times*, https://www.nursingtimes.net/opinion/separating-the-mentor-role-brings-risks-and-opportunities-24-06-2019/
- Ryan, G. (2018). Introduction to positivism, interpretivism and critical theory. *Nurse researcher*, 25(4), 41-49.
- Rychetnik, L., Frommer, M., Hawe, P., & Shiell, A. (2002). Criteria for evaluating evidence on public health interventions. *Journal of Epidemiology & Community Health*, 56(2), 119-27.
- Rylance, R., Barrett, J., Sixsmith, P., & Ward, D. (2017). Student nurse mentoring: an evaluative study of the mentor's perspective. *British Journal of Nursing*, 26(7), 405-409.
- Sarafis, P., Rousaki, E., Tsounis, A., Malliarou, M., Lahana, L., Bamidis, P., Niakas, D., & Papastavrou, E. (2016). The impact of occupational stress on nurses' caring behaviors and their health-related quality of life. *BMC nursing*, 15(1), p.56.
- Schoon, I. (2018). Conceptualising learner agency: a socio-ecological developmental approach. *Published by the Centre for Learning and Life Chances in Knowledge Economies and Societies.* www.llakes.ac.uk.
- Schumacher, R. (2016). What is Moral Relativism? Christian Apologetics & Research Ministry. https://carm.org/moral-relativism
- Scotland, J. (2012). Exploring the philosophical underpinnings of research: Relating ontology and epistemology to the methodology and methods of the scientific, interpretive, and critical research paradigms. *English language teaching*, 5(9), 9-16.
- Scully, N. J. (2011). The theory-practice gap and skill acquisition: An issue for nursing education. *Collegian*, 18(2), 93-98.
- Sharif, F., & Masoumi, S. (2005). A qualitative study of nursing student experiences of clinical practice. *BMC nursing*, 4(1), 1-7.
- Shikulo, L., & Lekhetho, M. (2020). Exploring student support services of a distance learning centre at a Namibian university. *Cogent Social Sciences*, 6(1), 1737401.
- Silverman, D. (2010) *Doing Qualitative Research: A Practical Handbook*. 3rd Edition, Sage Publications.
- Simpson, D. J. (2006). John Dewey (Vol. 10). Peter Lang.
- Smit, R. (2012). Towards a clearer understanding of student disadvantage in higher education: Problematising deficit thinking. *Higher Education Research & Development*, 31(3), 369-380.
- Smith M.R., Grealish L., & Henderson S. (2018). Shaping a valued learning journey: Student satisfaction with learning in undergraduate nursing programs, a grounded theory study. *Nurse Education Today*, 64, 175-179.
- Smith, G. (2015). A review of current student populations in respect of widening participation characteristics. University of Central Lancashire.

- Smith, R. (2007). An overview of research on student support: helping students to achieve or achieving institutional targets? Nurture or de-nature?. *Teaching in Higher Education*, 12(5-6), 683-695.
- Snape, D., & Spencer, L. (2003). The Foundations of Qualitative Research. In J. Ritchie, J. & Lewis, C, (Eds). *Qualitative research practice* (pp. 1-23). SAGE Publications Ltd.
- Spool, J. M. (2004). The KJ-technique: A group process for establishing priorities. User interface engineering.
- Sterling, A. J. (2018). Student experiences in the second year: Advancing strategies for success beyond the first year of college. *Strategic Enrollment Management Quarterly*, 5(4), 136-149.
- Stern, D. P. (2004). The Round Earth and Christopher Columbus. Educational Web Sites on Astronomy, Physics, Spaceflight and the Earth's Magnetism. http://wwwistp.gsfc.nasa.gov/stargaze/Scolumb.htm
- Strand, I., Nåden, D., & Slettebø, Å. (2009). Students Learning in a Skills Laboratory. *Nursing Science*, 29(3), 18–22.
- Sutton, K. L., & Sankar, C. S. (2011). Student satisfaction with information provided by academic advisors. *Journal of STEM Education: Innovations and Research*, 12(7).
- Tait, A. (2000). Planning student support for open and distance learning. *Open Learning: The Journal of Open, Distance and e-Learning,* 15(3), 287-299.
- Tait, A. (2015). Student success in open, distance and e-learning. The ICDE reports series, 1–10. International Council for Open and Distance Education. https://www.icde.org/s/studentsuccess-report.pd
- Tekkol, İ. A., & Demirel, M. (2018). An Investigation of Self-Directed Learning Skills of Undergraduate Students. *Frontiers in psychology*, 9, 2324. https://doi.org/10.3389/fpsyg.2018.02324
- Temkin, B. D. (2010). Mapping the customer journey. Forrester Research, 3, 20.
- Temple, P., Callender, C., Grove, L., & Kersh, N. (2014). *Managing the student experience in a shifting higher education landscape.* The Higher Education Academy.
- Ten Hoeve, Y., Castelein, S., Jansen, G., & Roodbol, P. (2017). Dreams and disappointments regarding nursing: Student nurses' reasons for attrition and retention. A qualitative study design. *Nurse Education Today*, 54, 28-36.
- The 1994 Group. (2007). Enhancing the Student Experience. Policy report. The 1994 Group. Retrieved from http://old.1994group.ac.uk/documents/public/SEPolicyReport.pdf
- Thomas, L. (2012). *What works? Student retention and success*. Higher Education Academy.
- Thomas, L., & Duckworth, V. (2018, April). Maintaining the diversity of the professional healthcare workforce through higher education qualification routes. *In 4th International Conference on Higher Education Advances.*
- Timmins, F. (2008). Take time to facilitate self-directed learning. *Nurse Education in Practice*, 8(5), 302.

- Tinto, V. (1987). *Leaving college: Rethinking the causes and cures of student attrition.* University of Chicago Press.
- Titus, S., & Roman, N. V. (2019). Predictors of student agency: the relationship between student agency, learning support and learning experiences in an interprofessional health science faculty. *Journal of interprofessional care*, 33(3), 308-312.
- Tollyfield, R. (2014). Facilitating an accelerated experience-based co-design project. *British Journal of Nursing*, 23(3), 136-141.
- Troxel, W. G., DiGregorio, G., Guertin, E., Harris, V., & Spight, D. B. (2018). Achieving Excellence and Assessing Impact of Academic Advising: A Landscape Analysis of Existing National Surveys 2018 Reinvention Collaborative National Conference November 8-10, 2018 Arlington, VA.
- Truncellito, D. A. (2016). Epistemology. Internet Encyclopedia of Philosophy: A peer reviewed Academic Resource. www.iep.utm.edu/epistemo/
- Tsianakas, V., Maben, J., Wiseman, T., Robert, G., Richardson, A., Madden, P., & Davies, E. A. (2012b). Using patients' experiences to identify priorities for quality improvement in breast cancer care: patient narratives, surveys or both? *BMC health services research*, 12(1), 271.
- Tsianakas, V., Robert, G., Maben, J., Richardson, A., Dale, C., & Wiseman, T. (2012a). Implementing patient-centred cancer care: using experience-based co-design to improve patient experience in breast and lung cancer services. *Supportive care in cancer*, 20(11), 2639-2647.
- Tune, J. D., Sturek, M., & Basile, D. P. (2013). Flipped classroom model improves graduate student performance in cardiovascular, respiratory, and renal physiology. *Advances in physiology education*, 37(4), 316-320.
- Turner, R., Morrison, D., Cotton, D., Child, S., Stevens, S., Nash, P., & Kneale, P. (2017). Easing the transition of first year undergraduates through an immersive induction module. *Teaching in Higher Education*, 22(7), 805-821.
- Ultanır, E. (2012). An epistemological glance at the constructivist approach: Constructivist learning in Dewey, Piaget, and Montessori. *International Journal of Instruction*, 5(2).
- Universities and Colleges Admissions Service (UCAS). (2018). March Deadline Analysis: Overview. https://www.ucas.com/file/159421/download?token=yfonUzjl
- Universities UK. (2014). *Trends in undergraduate recruitment*. Universities UK. Retrieved fromhttp://www.universitiesuk.ac.uk/highereducation/Documents/2014/TrendsInUn dergraduateRecruitment.pdf
- University of Central Lancashire (UCLan). (2015). *Data Protection Policy: Student regulations and policies*. University of Central Lancashire
- University of Central Lancashire. (2020). UCLan Study Support, https://www.uclan.ac.uk/students/support/study_support.php
- Vaismoradi, M., Turunen, H., & Bomdas, T. (2013). Content analysis and thematic analysis: Implications for conducting a qualitative descriptive study. *Nursing & health sciences*, 15(3), 398-405.

- Van Rensburg, G. H., & Botma, Y. (2015). Bridging the gap between self-directed learning of nurse educators and effective student support. *Curationis*, 38(2), 1-7.
- Van Zyl, S., & Mentz, E. (2019). Moving to deeper self-directed learning as an essential competency for the 21st century. In E. Mentz, J. De Beer, & R. Bailey, (eds.). Self-Directed Learning for the 21st Century: Implications for Higher Education (pp. 67-102). (NWU Self-Directed Learning Series Volume 1)
- Vaughn, M. (2014). The role of student agency: Exploring openings during literacy instruction. Teaching and Learning: *The Journal of Natural Inquiry & Reflective Practice*, 28(1), 4-16.
- Vaughn, M. (2018). Making sense of student agency in the early grades. *Phi Delta Kappan*, 99(7), 62-66.
- Vaughn, M. (2020). What is student agency and why is it needed now more than ever?. *Theory into Practice*, 59(2), 109-118.
- Vinson, D., Nixon, S., Walsh, B., Walker, C., Mitchell, E., & Zaitseva, E. (2010). Investigating the relationship between student engagement and transition. *Active learning in higher education*, 11(2), 131-143.
- von Glasersfeld, E. (1989). Facts and the self from a constructivist point of view. *Poetics*, 18(4-5), 435-448.
- Walker, S., Dwyer, T., Broadbent, M., Moxham, L., Sander, T., & Edwards, K. (2014). Constructing a nursing identity within the clinical environment: The student nurse experience. *Contemporary Nurse*, 49(1), 103-112.
- Walker, S., Rossi, D., Anastasi, J., Gray-Ganter, G., & Tennent, R. (2016). Indicators of undergraduate nursing students' satisfaction with their learning journey: An integrative review. *Nurse Education Today*, 43, 40-48.
- Walsh, D., & Downe, S. (2006). Appraising the quality of qualitative research. *Midwifery*, 22(2): 108-19.
- Walsh, M. (2004). *Lecturers understanding and experiences' of self-directed learning and its relevance to undergraduate nurse education*. University College Dublin (Doctoral dissertation, M. Sc. Thesis).
- Waltz, C. F., Jenkins, L. S., & Han, N. (2014). The use and effectiveness of active learning methods in nursing and health professions education: a literature review. *Nursing Education Perspectives*, 35(6), 392-400.
- Webster, S., Lewis, J., & Brown, A. (2013). Considerations in qualitative research. *Qualitative research practice: A guide for social science students and researchers*, 77.
- West, J., Fusari, G., Raby, E., Alwani, R., Meldaikyte, G., Wojdecka, A., & Matthews, E. (2017). Developing the Double Diamond process for implementation—insights from a decade of Inclusive Design projects. DESIGN4HEALTH, 310.
- White, E. R. (2015). Academic advising in higher education: A place at the core. *The Journal of General Education*, 64(4), 263-277.

- White, E. R., & Steele, G. E. (2016). Academic advising. In N. Zhang (Eds.) *Rentz's student affairs practice in higher education* (5th ed.) (pp.96-119). Charles C Thomas Publisher.
- Whittle, S. R. (2018). The second-year slump–now you see it, now you don't: Using DREEM-S to monitor changes in student perception of their educational environment. *Journal of Further and Higher Education*, 42(1), 92-101.
- Williams, V., Boylan, A., & Nunan, D. (2018). Critical appraisal of qualitative research: necessity, partialities and the issue of bias. *BMJ Evidence-Based Medicine*, bmjebm-2018.
- Willis Commission. (2012). Quality with compassion: the future of nursing education. *Royal College of Nursing*.
- Willis, J. W. (2007). Foundations of qualitative research: Interpretive and critical approaches. Sage Publications Ltd.
- Willis, L. (2015). Raising the bar. Shape of Caring: A Review of the Future Education and Training of Registered Nurses and Care Assistants: Health Education England.
- Willower, D. J. (1994). Dewey's theory of inquiry and reflective administration. *Journal of Educational Administration.*
- Wilson, A., Chur-Hansen, A., Marshall, A., & Air, T. (2011). Should nursing-related work experience be a prerequisite for acceptance into a nursing programme? A study of students' reasons for withdrawing from undergraduate nursing at an Australian university. *Nurse Education Today*, 31(5), 456-60.
- Wilson, C. (2013). Brainstorming and beyond: a user-centered design method. Newnes.
- Wingate, U. (2007). A framework for transition: supporting 'learning to learn' in higher education. *Higher Education Quarterly*, 61(3), 391-405.
- Woolley, N. N., & Jarvis, Y. (2007). Situated cognition and cognitive apprenticeship: A model for teaching and learning clinical skills in a technologically rich and authentic learning environment. *Nurse Education Today*, 27(1), 73-79.
- Yale, A. T. (2017). The personal tutor–student relationship: student expectations and experiences of personal tutoring in higher education. *Journal of Further and Higher Education*, 43 (4), 533-544.
- Yorke, M., & Longden, B. (2008). The first-year experience of higher education in the UK. York: *Higher Education Academy*.
- Zarges, K. M., Adams, T. A., Higgins, E. M., & Muhovich, N. (2018). Assessing the impact of academic advising: Current issues and future trends. *New Directions for Higher Education*, 2018 (184), 47-57.
- Zimmerman, B. J., & Schunk, D. H. (2011). *Handbook of self-regulation of learning and performance.* Routledge/Taylor & Francis Group.

BIBLIOGRAPHY

- Bowling, A. (2014). *Research methods in health: investigating health and health services*. McGraw-hill education (UK).
- Clandinin, D. J., & Connelly, F. M. (2004). *Narrative Inquiry: Experience and Story in Qualitative Research.* Jossey-Bass.
- Dewey, J., & Ross, R. (2008). *The Middle Works of John Dewey*, Volume 12, 1899-1924: Essays, Miscellany, and Reconstruction in Philosophy Published During 1920
- Dewey, J. (1986, September). Experience and education. *In The Educational Forum*, Vol. 50, No. 3, pp. 241-252. Taylor & Francis Group.
- Gray, D. E. (2014). Doing research in the real world (3rd ed.). SAGE Publications Ltd.
- Saks, M., & Allsop, J. (Eds.) (2013). *Researching health: qualitative, quantitative and mixed methods* (2nd ed.). SAGE Publications Ltd.
- Thomson, P., & Walker, M. (Eds.). (2010). *The Routledge doctoral student's companion: Getting to grips with research in education and the social sciences.* Routledge.

LIST OF APPENDICES

Appendix 1: 12-step guideline for structuring and documenting a literature search strategy

Step	Description						
1	Provide a purpose statement to describe the question to be addressed in the						
	literature search						
2	Document the databases or search engines used						
3	Specify the limits applied to the search e.g. dates, language, human studies						
	etc.						
4	List the inclusion criteria and exclusion criteria for the search						
5	List the search terms used						
6	Document the search process for each search engine including search engine,						
	terms and number retrieved on a search results table						
7	Assess retrieved articles for relevance using inclusion and exclusion criteria						
8	Document a summary table of included articles						
9	Provide a statement specifying the number of retrieved articles at the end of						
	the search process i.e. search results						
10	Conduct quality appraisal of retrieved literature						
11	Critical review of literature						
12	Check the reference list for accuracy						

Based on Kable, A. K., Pich, J., & Maslin-Prothero, S.E. (2012). A structured approach to documenting a search strategy for publication: a 12-step guideline for authors. *Nurse Education Today*, 32(8), 878–886.

Appendix 2: List of literature identified as eligible for review

	Article chosen for review
1	Andrew, N., McGuinness, C., Reid, G., & Corcoran, T. (2009). Greater than the sum of its parts: transition into the first year of undergraduate nursing. Nurse Education in Practice, 9(1), 13-21.
2	Birks, M., Chapman, Y., Ralph, N., McPherson, C., Eliot, M., & Coyle, M. (2013). Undergraduate Nursing Studies: The First-Year Experience. Journal of Institutional Research, 18(1), 26-35.
3	Donough, G., & Van der Heever, M. (2018). Undergraduate nursing students' experience of clinical supervision. Curationis, 41 (1) (pp e1-e8), 2018.
4	Grobecker, P. A. (2016). A sense of belonging and perceived stress among baccalaureate nursing students in clinical placements. Nurse Education Today, 36, 178-183.
5	Hamshire, C., Barrett, N., Langan, M., Harris, E., & Wibberley, C. (2017). Students' perceptions of their learning experiences: A repeat regional survey of healthcare students. Nurse Education Today, 49, 168-173.
6	Hamshire, C., Willgoss, T. G., & Wibberley, C. (2013). What are reasonable expectations? Healthcare student perceptions of their programmes in the Northwest of England. Nurse Education Today, 33(2), 173-179.
7	James, A., & Chapman, Y. (2010). Preceptors and patients - the power of two: nursing student experiences on their first acute clinical placement. Contemporary Nurse, 34 (1), 34-47.
8	Last, L., & Fulbrook, P. (2003). Why do student nurses leave? Suggestions from a Delphi sudy. Nurse Education Today, 23(6), 449-458.
9	Liljedahl, M., Björck, E., Kalén, S., Ponzer, S., & Laksov, K.B. (2016). To belong or not to belong: nursing students' interactions with clinical learning environments – an observational study. BMC Medical Education, 16, 197
10	Magnussen, L., & Amundson M.J. (2003). Undergraduate nursing student experience. Nursing & Health Sciences, 5(4), 261-267
11	Mckendry, S., Wright, M., & Stevenson, K. (2014). Why here and why stay? Students' voices on the retention strategies of a widening participation university. Nurse Education Today, 34(5), 872-877.
12	Mills, A., Ryden, J., & Knight, A. (2020). Juggling to find balance: hearing the voices of undergraduate student nurses. British Journal of Nursing, 29(15), 897-903.
13	Papastavrou, E., Dimitriadou, M., Tsangari, H., & Andreou, C. (2016). Nursing students' satisfaction of the clinical learning environment: a research study. BMC Nursing, 15(1), 44.
14	Porteous, D. J., & Machin, A. (2018). The lived experience of first year undergraduate student nurses: A hermeneutic phenomenological study. Nurse Education Today, 60, 56-61.
15	Smith, M.R., Grealish L., & Henderson, S. (2018). Shaping a valued learning journey: Student satisfaction with learning in undergraduate nursing programs, a grounded theory study. Nurse Education Today, 64 (pp 175-179)
16	Ten Hoeve, Y., Castelein, S., Jansen, G., & Roodbol, P. (2017). Dreams and disappointments regarding nursing: Student nurses' reasons for attrition and retention. A qualitative study design. Nurse Education Today, 54, 28-36.

Appendix 3: Criteria for appraising qualitative and mixed methods research studies

Stages	Essential criteria		
Scope and purpose	Clear statement of, and rationale for, research question/aims/purposes		
	Study thoroughly contextualised by existing literature		
Design	Method/design apparent, and consistent with research intent		
	Data collection strategy apparent and appropriate		
Sampling strategy	Sample and sampling method appropriate		
Analysis	Analytic approach appropriate		
Interpretation (findings)	Context described and taken account of in interpretation		
	Clear audit trail given		
	Data used to support interpretation		
Reflexivity	Researcher reflexivity demonstrated		
Ethical dimensions	Demonstration of sensitivity to ethical concerns		
Relevance and transferability	Relevance and transferability evident		

Appendix 4: Literature summary table

Andrew, N.,Investigate the experience of nursing students first clinical placement: how its parts: Transition intoMixed methods hixed methodsn = 418 First yearStudents appear to understand their role as nursing students, theirEthics approval obtained and ethical considerations clearly described.BAndrew, N., McGuinness, C., Reid, G., & Corcoran, T. (2009)Investigate the experience of nursing students first clinical placement: how prepared are studentsMixed methods Approachn = 418 First year nursing studentsStudents appear to understand their role as nursing students, theirEthics approval obtained and ethical considerations clearly described.BBEvaluative survey study usingnursing studentsin the first placement is less clear. There is also a lack of clarityClearly outlined. Findings comparative to wider literature.B	Author, Year & Setting	Scope and Purpose	Design and Analysis	Sampling Strategy	Interpretation (Findings)	Ethical dimensions, Relevance and Transferability	Grade
the first year of undergraduate nursing, Nurse Education in Practice (2009) 9, 13–21professionally and socially for year one of an undergraduate nursing programmequestionnaire of previous action research initiativePraposeful samplingregarding their role and that of the 	Andrew, N., McGuinness, C., Reid, G., & Corcoran, T. (2009) Greater than the sum of its parts: Transition into the first year of undergraduate nursing, <i>Nurse Education in</i> <i>Practice</i> (2009) 9, 13–21 UK Birks, M., Chapman, Y., Ralph, N., McPherson, C., Eliot, M., & Coyle, M. (2013). Undergraduate Nursing Studies: The First-Year Experience. Journal of Institutional Research, 18(1), 26-35. Australia	Investigate the experience of nursing students first clinical placement: how prepared are students professionally and socially for year one of an undergraduate nursing programme This study reports on a survey by a regional Australian university that aimed to identify the needs of beginning nursing students and develop strategies to reduce attrition and promote success in their undergraduate studies.	Mixed methods Approach Evaluative survey study using questionnaire of previous action research initiative Analysed using SPSS (Statistical package for the Social Services) Qualitative Online survey Thematic analysis	n = 418 First year nursing students Purposeful sampling n =112 First year nursing students	Students appear to understand their role as nursing students, their understanding of what is expected of in the first placement is less clear. There is also a lack of clarity regarding their role and that of the mentor in the achievement of clinical placement learning outcomes. Findings were presented as issues faced in adapting to the role of a university student in the first year of study, services or support mechanisms accessed to assist in transition to the role of university student, and services lacking that would have assisted in the transition to the role of university student. Issues faced by students beginning a program of tertiary study are varied and complex. Nursing students face additional challenges as a result of the requirement to participate in clinical placements	Ethics approval obtained and ethical considerations clearly described. Limitations/weaknesses of study clearly outlined. Findings comparative to wider literature. This article reported part of a wider project (the Enhance project) and findings within this article are described as one spiral of the whole project, aspects of which are transferable to the general student experience. Relevant study that discussed student experience of adapting to first year Higher education study 7% response rate. A low level of response can significantly impact on both the usefulness and transferability of the findings. Limitations/weaknesses of study not clearly outlined.	В

Donough, G., & Van der Heever, M. (2018). Undergraduate nursing students' experience of clinical supervision. Curationis. 41 (1) (pp e1- e8), 2018. South Africa	To explore the experiences of undergraduate nursing students on clinical supervision.	Qualitative Descriptive design with a qualitative approach using focus group interviews Thematic analysis	n=36 Nursing students (9 students from year-1, 9 students from year-2 and 9 students from year-3) Purposive sampling	Findings indicated both positive and negative experiences regarding clinical supervision. Positive experiences included the support that was received from supervisors. These were overshadowed by negative experiences concerning the behaviour and competencies of the supervisors. Students experienced differences in the clinical procedures demonstrated by various supervisors. Negative experiences that relate to abusive behaviour such as misuse of power were also found, as well the incongruence amongst clinical supervisors regarding clinical procedures.	Ethical considerations clearly described. Participants across all three years of nursing program included in study Conducted in South Africa, therefore some of the findings may not relate to the UK student experience.	В
Grobecker, P. A. (2016). A sense of belonging and perceived stress among baccalaureate nursing students in clinical placements. <i>Nurse Education Today</i> , 36, 178-183. USA	The purpose of this study was to describe the level of a sense of belonging and perceived stress among baccalaureate nursing students in their clinical placements.	Quantitative A descriptive correlational study using the National Student Nurses Association (NSNA) database and SurveyMonkey	n=1296 Nursing students (participant year of study not indicated)	The findings from this study revealed a statistically significant low inverse relationship between a sense of belonging and perceived stress among baccalaureate nursing students in their clinical placements. The results support the concept of a sense of belonging as a fundamental human need, having a positive influence and impact on students' learning, motivation and confidence. In contrast, perceived stress has negative consequences on the students' self-concept, learning skills and competence.	Large sample from across multiple Higher Education institutes The study utilised a convenience sample of student nurses which could result in the sample biases and reduce generalizability. Limitations/weaknesses of study not clearly outlined. Year of study on the nursing course not indicated	N/A

Hamshire, C., Barrett, N., Langan, M., Harris, E., & Wibberley, C. (2017). Students' perceptions of their learning experiences: A repeat regional survey of healthcare students. <i>Nurse Education Today</i> , 49, 168-173. UK	To determine similarities and differences between students' perceptions of their learning experiences between 2011 and 2015 in relation to campus- based learning, placement-based learning and personal circumstances.	Quantitative Survey	n=1983 Students on professional programmes (50% of respondents from the Adult nursing course)	Although students were positive overall about their experiences, a number were dissatisfied with some aspects of their experiences - particularly in relation to initial support on campus and whilst studying on placement. Four years on from the original survey, despite a considerable investment in improving students' experiences across the region, there appears to be little change in students' perceptions of their learning experiences	Ethics approval obtained and ethical considerations clearly described. As study included a multitude of students on professional course relevance and transferability was evidently not transferable to the general student population. Limitations/weaknesses of study not clearly outlined.	N/A
Hamshire, C., Willgoss, T. G., & Wibberley, C. (2013). What are reasonable expectations? Healthcare student perceptions of their programmes in the North West of England. <i>Nurse Education Today</i> , 33(2), 173-179. UK	The aim of this study was to determine how students in the North West of England perceived their studies and to identify the factors that could contribute to students' dissatisfaction	Mixed methods sequential exploratory study Phase 1 used narrative inquiry Thematic framework analysis Phase 2 online Survey Four-point Likert- type response format	Phase 1 N=24 Nursing and Midwifery students Phase 2 N=1080 Students on professional programmes (Including student nurses)	The majority of students reported positive experiences of their course. There was strong agreement in the importance of the role of the Personal tutor and of a supportive placement mentor. Some students indicated that the workload and academic level were a shock at first. Students identified difficulties around the cost of placement travel and the impact of placements on family life. Financial hardship was a frequently cited issue that students attributed to thoughts of discontinuation	Ethics approval obtained and ethical considerations clearly described. Limitations/weaknesses of study not clearly outlined. Aspects of this multi-professional study are transferable to the general student population.	В

James, A., & Chapman, Y. (2010). Preceptors and patients - the power of two: nursing student experiences on their first acute clinical placement. <i>Contemporary Nurse</i> , 34 (1), 34-47. Australia	Explore and describe the experiences of 2 nd year undergraduate nursing students undertaking their first acute clinical placement. Offer insight into how these experiences influence their journey as a student and their future career in nursing	Qualitative Heideggerian phenomenology Data analysed using Max van Manen's (1990) analysis technique	n = 6 Second year Undergraduate Nursing students Convenience sample	Three main themes emerged from the findings: being overwhelmed and confronted; patients as people and perceptions of preceptors. Findings inform why some students consider discontinuance of their studies and how the experience impacts on decisions about future nursing practice.	The research was confined to one campus of one school of nursing and midwifery in one university. Small sample size Participants were only drawn from one clinical context, that being medical/ surgical and from 1-year level. Results not representative of 2 nd year nursing students in general.	В
Last, L., & Fulbrook, P. (2003). Why do student nurses leave? Suggestions from a Delphi study. <i>Nurse Education Today</i> , 23(6), 449-458. UK	The purpose of this study was to establish a consensus view of the reasons why student nurses leave their Pre-Registration education programme.	Mixed methods triangulation Phase one - qualitative approach using focus groups and semi-structured one-to-one interviews. Phase two - Quantitative Delphi study	n=6 Nursing students across all three years (exact numbers not indicated) n= 12 Multi- professional focus group	A number of factors result in student nurses leaving. These include communication and operational factors between the university and clinical areas, feelings of not being valued, unmet expectations, and stress. These issues were of concern to students and appeared to have a cumulative effect that led them to question whether they should continue their education programme.	Limitations/weaknesses of study clearly outlined Number and designation of multi- professional participants included in study was vague	C
Liljedahl, M., Björck, E., Kalén, S., Ponzer, S., & Laksov, K.B. (2016). To belong or not to belong: nursing students' interactions with clinical learning environments –	The aim of this study was to explore the interdependence between affordances and engagement in clinical learning environments of	Qualitative Inductive thematic approach	n = 6/10 Nursing students/clinical supervisors	The results revealed students strived to fill out the role they were offered in an aspirational way but that they became overwhelmed when given the responsibility of care. When students' basic values did not align with those enacted by the	Limitations/weaknesses of study clearly outlined Number of participants from the study not clearly indicated Year of study on the nursing course not indicated	с

an observational study. BMC Medical Education. 16, 197 Sweden	nursing students. They posed the following research question: How are nursing students influenced in their interactions with clinical learning environments?		(participant year of study not indicated) Ethnography- inspired observational study	workplace, they were not willing to compromise their own values. Workplaces succeeded in inviting students into the community of nurses and the practice of care. Students demonstrated hesitance regarding their desire to belong to the workplace community.		
Magnussen, L., & Amundson M.J. (2003). Undergraduate nursing student experience. Nursing & Health Sciences, 5(4), 261-267 UK	To describe and explicate the experience of being a nursing student	Qualitative descriptive study Analysed using constant comparative method	n = 12 undergraduate nursing students Convenience sample (participant year of study not indicated)	Four major theme clusters emerged from the data: (i) meeting conflicting demands; (ii) feeling overworked; (iii) feeling unprepared and (iv) seeking respect and support from one's faculty. Student's stories provide insights into the current educational environment, which can assist faculties to understand the impact of their pedagogical approaches.	Focused on students' perceptions of their experience of being a student nurse Specific methodology/ research design not identified. The participants were all from the same school of nursing and represented a very limited sample of nursing students Year of study on the nursing course not indicated	С
Mckendry, S., Wright, M., & Stevenson, K. (2014). Why here and why stay? Students' voices on the retention strategies of a widening participation university. <i>Nurse Education Today</i> , 34(5), 872-877. UK	The study aimed to explore student motivations, experiences and support requirements during their first year to determine the efficacy of institutional retention initiatives.	Qualitative Focus groups Grounded theory and Thematic analysis	n=46 First year nursing and midwifery students.	Students utilise a range of support mechanisms before and during their studies in order to maintain their motivation and to juggle the many demands on their time. These include university staff, fellow students, friends and family and those they currently know within the profession.	This study clearly positions itself in relation to the prominent theoretical perspectives. Limitations/weaknesses of study clearly outlined.	В

Mills, A., Ryden, J., & Knight, A. (2020). Juggling to find balance: hearing the voices of undergraduate student nurses. British Journal of Nursing, 29(15), 897- 903. UK	To examine the health and wellbeing implications of undertaking a BSc nursing degree in the UK for first-year students.	Qualitative Narrative analysis of 100 written student reflections	n=100 First-year student nurses	Nursing students juggle multiple competing demands on their physical capabilities, personal resources, income and time. Students are constantly seeking to achieve balance and personal equilibrium through the use of a variety of coping strategies.	Limitations/weaknesses of study clearly outlined. Ethical considerations not identified. Findings from this study are not generalizable to the wider student population.	С
Papastavrou, E., Dimitriadou, M., Tsangari, H., & Andreou, C. (2016). Nursing students' satisfaction of the clinical learning environment: a research study. BMC nursing, 15(1), 44. Cyprus	The aim of the study was to investigate nursing students' satisfaction of the clinical settings as learning environments.	Quantitative descriptive, correlational design Descriptive statistics	n = 463 Student nurses (participant year of study not indicated)	Nursing students were highly satisfied with the clinical learning environment and their satisfaction has been positively related to the pedagogical atmosphere, the Ward Manager's leadership style, the premises of Nursing in the ward, the supervisory relationship (mentor) and the role of the Nurse Teacher. Students who had a named mentor reported more satisfied with the supervisory relationship.	Conducted across 3 university institutions Large number of participants Clear themes not identified Year of study on the nursing course not indicated	N/A
Porteous, D. J., & Machin, A. (2018). The lived experience of first year undergraduate student nurses: A hermeneutic phenomenological study. <i>Nurse Education Today</i> , <i>60</i> , 56-61. UK	This study aimed to understand how one group of undergraduate nursing students perceived their experiences of the transition into higher education and nursing profession.	Qualitative Hermeneutic phenomenology Data analysed using thematic analysis	n = 10 First year undergraduate student nurses Purposeful sampling	Five themes emerged from the data: uncertainty; expectations; learning to survive; seeking support; and moving forward. Findings identify that the participants had developed skills to survive however considerable variation in their experience, influenced motivation and behaviour. They developed their own skills of coping to deal with the demands of academic life and those of the practice setting. An	Results are in keeping with other published research on the experience of student nurses in higher education Findings from this study are not generalizable due to the interpretive paradigm utilised	В

				explanatory student journey model demonstrated that developing self- efficacy was key to their successful transition through the first year of undergraduate study.		
Smith, M.R., Grealish L., & Henderson, S. (2018). Shaping a valued learning journey: Student satisfaction with learning in undergraduate nursing programs, a grounded theory study. <i>Nurse</i> <i>Education Today</i> , 64, 175-179. Australia	To understand undergraduate nursing students' satisfaction with learning.	Qualitative Constructivist grounded theory Semi-structured interviews Symbolic interactionism interviews were analysed for action and meaning	n =17 undergraduate nursing students (6 students from year-1, 7 students from year-2 and 4 students from year-3)	Students are satisfied with learning when they shape a valued learning journey that accommodates social contexts of self, university and nursing workplace. The theory has three phases. Phase 1 - orienting self to valued learning in the pedagogical landscape; phase 2 - engaging with valued learning experiences across diverse pedagogical terrain; and phase 3 - recognising valued achievement along the way.	Limitations/weaknesses of study clearly outlined Themes not clearly identified but described in phases	В
Ten Hoeve, Y., Castelein, S., Jansen, G., & Roodbol, P. (2017). Dreams and disappointments regarding nursing: Student nurses' reasons for attrition and retention. A qualitative study design. <i>Nurse Education Today</i> , 54, 28- 36. The Netherlands	The main aim of this study was to examine which factors affect student nurses' decision to leave or complete their programme	Qualitative exploratory descriptive design, employing a phenomenologica I approach	n =17 Student nurses 3 rd year	Reasons for attrition were strongly related to the training programme and to their clinical placements, in particular the perceived lack of support from mentors and team. Feelings of being welcomed and working in a nice team proved to be more important reasons for completing the programme than the specific clinical field.	Contemporary focus with interesting results Results not statistically significant. Only 2 study participants left the course even though a key focus of this study was attrition.	В

Appendix 5: Critical appraisal of the quantitative surveys using the 'Critical Appraisal of a Survey' (CEBM) tool

Appraisal questions	Literature utilising quantitative survey methodology				
Answered – Yes, Can't tell, No	Grobecker (2016)	Papastavrou et al. (2016)	Hamshire et al. (2017)		
1. Did the study address a clearly focused question / issue?	Yes	Yes	Yes		
2. Is the research method (study design) appropriate for answering the research question?	Yes	Yes	Yes		
<i>3. Is the method of selection of the subjects (employees, teams, divisions, organizations) clearly described?</i>	Yes	Yes	Yes		
4. Could the way the sample was obtained introduce (selection) bias?	No	No	No		
5. Was the sample of subjects' representative with regard to the population to which the findings will be referred?	No	No	Yes		
<i>6. Was the sample size based on pre-study considerations of statistical power?</i>	Yes	Yes	Yes		
7. Was a satisfactory response rate achieved?	Yes	Yes	Yes		
8. Are the measurements (questionnaires) likely to be valid and reliable?	Yes	Yes	Yes		
9. Was the statistical significance assessed?	Yes	Yes	Yes		
10. Are confidence intervals given for the main results?	Yes	Yes	Yes		
11. Could there be confounding factors that haven't been accounted for?	Yes	Yes	Yes		
12. Can the results be applied to your organization?	Yes	Yes	Yes		

Appendix 6: STEMH Ethics Committee approval



Date: 16th January 2017

Rob Monks / Samantha Pollitt School of Nursing University of Central Lancashire

Dear Rob / Samantha

Re: STEMH Ethics Committee Application Unique Reference Number: STEMH 583

The STEMH ethics committee has granted approval of your proposal application 'Capturing, exploring and understanding the experiences of undergraduate student nurses undertaking a Pre-registration nursing degree course: An Experience Based Co-design study'. Approval is granted up to the end of project date* or for 5 years from the date of this letter, whichever is the longer. It is your responsibility to ensure that

- the project is carried out in line with the information provided in the forms you have submitted
- you regularly re-consider the ethical issues that may be raised in generating and analysing your data
- any proposed amendments/changes to the project are raised with, and approved, by Committee
- you notify <u>roffice@uclan.ac.uk</u> if the end date changes or the project does not start
- serious adverse events that occur from the project are reported to Committee
- a closure report is submitted to complete the ethics governance procedures (Existing
 paperwork can be used for this purposes e.g. funder's end of grant report; abstract for
 student award or NRES final report. If none of these are available use <u>e-Ethics Closure
 Report Proforma</u>).

Yours sincerely

Emma Sandon-Hesketh University Officer for Ethics STEMH Ethics Committee

Appendix 7: Participant Information Sheet (PIS) / Expression of interest form

INFORMATION SHEET FOR STUDENT PARTICIPANTS

Capturing, exploring and understanding the experiences of undergraduate student nurses undertaking a Pre-registration nursing degree course: An Experience Based Co-design study

There have been significant changes to undergraduate Pre-registration education in recent years including the move to higher education from schools of nursing and introduction of BSc and MSc only courses. Yet the experiences of adult student nurses remain under-researched in the UK. Therefore, you are invited to take part in a research study focused on your experiences on your course. Before deciding whether you want to take part, it is important for you to understand why the research is being done and what your participation will involve. Please take time to read the following information carefully and discuss it with others if you wish. Please contact the researchers if anything is unclear or if you would like more information. Their contact details can be found at the bottom of this sheet.

What is the purpose of the study?

The study aims to capture, explore and understand the experiences of student nurses undertaking the Pre-Registration Nursing (Adult) programme and to co-design improvements with key stakeholders who manage and deliver the programme.

Why have you been invited to participate?

All students currently enrolled on the Pre-registration nursing course (Adult field) within the school of nursing are being invited to participate and share their experiences.

Do you have to take part, and can you change my mind?

No. Taking part in this study is entirely your decision and your studies will not be affected in any way if you choose not to participate.

What if you wish to withdraw from this study?

You are free to withdraw at any time and without giving a reason. A decision to withdraw at any time will not affect your place on the course or impact any future grades or assessments. Please note that after each interview, focus group or Co-Design event the data will be anonymised, therefore you will not be able to withdraw your responses as it will not be possible to isolate your individual response as all responses will be anonymised after each interview/focus group.

What will you have to do if you decide to take part?

If you are interested in taking part in this research study and submit the expression of interest form attached you will be contacted by the research team and invited to either a one-to-one interview, focus group or both if you choose, at your convenience and at a time when you are already scheduled to be in university. The one-to-one interviews and focus group will be audio recorded to enable me to transcribe the discussion. You will be able to request a copy of your own transcript from the one-to-one interviews if you choose. Some volunteers may also be invited to an Experience Based Co-design event where staff and students meet together to discuss the findings generated from the interviews and focus groups and help produce recommendations for improving the future student experience on the nursing course and implementing solutions to problems identified. Please note if you agree to take part in an interview or focus group you are NOT under any obligation to take part in any further aspect of the study.

What are the possible benefits of taking part?

New knowledge generated from your participation in this study could enhance the design of future educational provision and contribute to future improvements on the course. Furthermore, participation in the research study can be included in your PDP (professional development portfolio). This may also be a useful experience for those planning to develop their own research portfolio after graduation.

What are the possible risks of taking part?

It is not anticipated that there will be any direct risks to you from participating in this study. If any issue is raised during the interview that you find upsetting the interview can be stopped at your request and support offered via your personal tutor or student support services. Anonymity and confidentiality will be ensured as far as is possible, however please note you may find yourself participating in a focus group with another student you are familiar with. If this is the case, you will have the option of either attending an alternative focus group or withdrawing your participation from the study without any negative consequence.

Will what you say in this study be kept confidential?

Pseudonyms or unique codes will be allocated to all audio digital recordings, and transcribed data, to retain anonymity and protect confidentiality. The consent forms and audio digital recordings will be stored in a secure cabinet, in a locked office, on-site at the University of Central Lancashire (UCLan, 2015). The audio digital recordings will be retained until after transcription of the data generated in phase four of the data collection process, after which all digital recordings will be deleted. Transcribed data will be stored electronically on a password protected database in accordance with the Data Protection Act (DPA) (1998) and the UCLan Data Protection Policy (2015). This data will be kept for 5 years from the end of the study in accordance with the University's policy on Academic Integrity, after which it will be deleted or destroyed.

What will happen to the results of the research study?

This results of this study will be presented at conferences and written up for publication in academic journals, retaining the anonymity of the participants. In addition, data from this study will be used to form the final thesis of a PhD study. Journal publications and the final PhD thesis will be available in the public domain for participants to view.

Who is organising and funding the research?

Sam Pollitt (lead researcher) is organising this study as part of a PhD programme within the University of Central Lancashire. The researcher is self-funding this project. Sam is being supervised by an experience supervisory team; Dr Dawne Gurbutt, Dr Nigel Harrison and Dr Rob Monks.

Who has reviewed the study?

The research has been approved by the University Research Ethics Committee for Science, Technology, Engineering, Medicine and Health (STEMH), and has been reviewed by a supervisory team.

What should you do if you want to take part?

If you are interested in taking part in this research study, please complete the expression of interest form attached to and submit it to Sam Pollitt in BB113 or email it to sjpollitt@uclan.ac.uk, within 7 days to express your interest (remember you can withdraw your participation at any time). You will then be contacted by Sam pollitt to discuss further and invited to attend a focus group, or interview arranged at your convenience over the next 6 months. Please note, if more students apply to take part than the study requires participants' will be selected based on maximum variation sampling. This could mean you may not be chosen to take part but your expression of interest is still appreciated.

Who do I contact if I have any complaints?

If you have any concerns or would like any further information about the study you can contact Sam Pollitt in the first instance. Concerns or complaints should be addressed to Dr Rob Monks as the study's Principal Investigator (<u>RMonks@uclan.ac.uk</u>) or the University Officer for Ethics at OfficerForEthics@uclan.ac.uk, information provided should include the study name or description (so that it can be identified), the researcher (Sam Pollitt), and the substance of the complaint.

Lead Researcher:	School of Nursing	Contact:
Sam Pollitt	Brook 113	E-mail: sjpollitt@uclan.ac.uk
	University of Central Lancashire	Tel: 01772 893624

Thank you for taking time to read the information sheet, I look forward to your response

EXPRESSION OF INTEREST FORM

Title of Research Study:

Capturing, exploring and understanding the experiences of undergraduate student nurses undertaking a Pre-registration nursing degree course: An Experience Based Co-design study

Name of Researcher: Sam Pollitt

Name (please PRINT)			
Email address:			
Contact number:			
Cohort:	Ye	'ear of study:	
Date of Birth:	A	\ge:	
Ethnicity (please indicate)	G	Gender (please circle)	Male / Female
Signature:			
Date:			

Please complete the information below...

If you are interested in taking part in the research study above please complete this expression of interest form, detach and either submit this hardcopy to Sam Pollitt in Brook Building 113 or copy and email the form to sjpollitt@uclan.ac.uk, within 7 days.

If you have any further questions or concerns about this study please contact the lead researcher, Sam Pollitt, on 01772 893624.

Thank-you

Appendix 8: Study flyer for student participants



9TH NOVEMBER 11:30AM, BB007 FOCUS GROUP

Lunch provided

This is your opportunity to take part in a study looking at your experiences of the undergraduate Pre-registration (Adult) nursing course so far. We want to know what your experiences have been of the welcome week, home groups, teaching, assessments, and clinical placements. The aim of the study is to explore ways to improve those experiences. Please consider sparing up to 40 min of your time between classes – lunch will be provided.



Research study -Understanding the experiences of Pre-registration undergraduate (Adult) student nurses

> Opportunity to explore your experiences

Chance to be part of the research process & include in your personal statement

> Certificate of Participation provided



SAM POLLITT School of Nursing BB349

Tel: 01772 893624 Email: sjpollitt@uclan.a.c.uk

Appendix 9: Student participant consent form for semi-structured interviews

PARTICIPANT CONSENT FORM - Interviews

Title of Research Study:

Capturing, exploring and understanding the experiences of undergraduate student nurses undertaking a Pre-registration nursing degree course: An Experience Based Co-design study

Name of Researcher: Sam Pollitt

	Please	tick
1.	I consent to take part in the above study	
2.	I confirm that I have read and understand the participant information sheet for the	
	above study	
3.	I understand I am a voluntary participant, and I am free to withdraw from the study at	
	any point	
4.	I understand and give consent for the interview to be audio recorded and kept on	
	record for up to five years	
5.	I understand that I will not be able to withdraw my data from the study after it has	
	been anonymised at the data transcription point	
6.	I DO / DO NOT (please delete as appropriate) give consent to the use of samples of my	
	audio recorded interview in the focus groups and Experience Based Co-design events	
	for the study	
7.	I give consent to the use of samples of my anonymised written quotes as bubble	
	graphics in the focus groups and Experience Based Co-design events for the study	
8.	I consent to the use of anonymised written quotes from me in publications / reports	
9.	I would like a copy of my one-to-one interview transcript when it is completed (please	
	tick this box only if you would like to receive the interview transcript of the results)	

Name of Participant (please PRINT)			
Signature			
Participant email address			
Participant contact number (optional)			
Date:	Cohort:	Year of study:	
Gender (please circle) Male / Female	Date of birth:	Age:	
Ethnicity (please Indicate)			
Name of Researcher	Sam Pollitt		
Date:	Signature:		

Participant Identification Code:

Appendix 10: Semi-structured interview topic guide

То	pics/questions	Prompts /probes/notes
•	Introduction and	Introduce myself and my nurse/researcher background
	background	My name is Sam Pollitt – thank-you for agreeing to be interviewed today
•	Re-state purpose of the study and interview (regain consent)	This study is trying to capture and understand the experiences of student nurses on the Pre-Registration Nursing (Adult) programme and to work with key stakeholders to make improvements on the programme.
•	Tell me about some of	
	your experiences of the	
	course	
•	Tell me about your	
	experiences in clinical	
	placement areas	
•	Tell me about your	
	experiences you're your	
	mentors	
•	What has your	
	experience been like with	
	lectures/admin staff	
•	Can you tell me about	
	your experiences with	
	Brook hub and placement	
	unit?	
•	How do you feel about	
	qualifying soon (or going	
	into your third year,	
	second year)?	
•	Can you tell me about	
	any particular highs on	
	the course?	
•	Can you tell me about	
	any particular lows on the	
	course?	
•	Is there anything you	Prompt participants to discuss what actions/support (if any) they feel could have
	think UCLan could have	made those events or experiences better
	done differently to make	
	that situation better for	
	you?	
•	Is there anything else you	
	would like to tell me	
	about or anything you	
	would like to add?	

Participant Identification Code.....Thank you for taking part in this study

Appendix 11: Student participant consent form for emotional touchpoint mapping event

PARTICIPANT CONSENT FORM - Experience Based Codesign Event

Title of Research Study:

Capturing, exploring, and understanding the experiences of undergraduate student nurses undertaking a Pre-registration nursing degree course: An Experience Based Co-design study

Name of Researcher: Sam Pollitt

		Please tick
1.	I consent to take part in the above study	
2.	I confirm that I have read and understand the participant information sheet	
	for the above study	
3.	I understand I am a voluntary participant, and I am free to withdraw from the	
	study at any point	
4.	I understand and give consent for the Experience Based Co-design Event to be	
	audio recorded and kept on record for up to five years	
5.	I understand that I will not be able to withdraw my audio recorded data from	
	the study due to the difficulties associated with isolating my voice and	
	comments from that of the other participants' in the Experience Based Co-	
	design Event	
6.	I consent to the use of my anonymised written data in publications / reports	

Name of Participant (please PRINT)	
Signature	
Date	
Participant email address	
Participant contact number (optional)	
Name of Researcher	Sam Pollitt
Signature	
Date	

Participant Identification Code:

Appendix 12: Key stakeholder participant information sheet (PIS) and consent form

INFORMATION SHEET FOR KEY STAKEHOLDER PARTICIPANTS

Capturing, exploring and understanding the experiences of undergraduate student nurses undertaking a Pre-registration nursing degree course: An Experience Based Co-design study

There have been significant changes to undergraduate Pre-registration education in recent years including the move to higher education from schools of nursing and introduction of BSc and MSc only courses. Yet the experiences of adult student nurses remain under-researched in the UK. Therefore, this research study is focused on capturing the student experiences of their course, and as a key stakeholder in the course you are considered as 'knowledgeable' in relation to the provision and delivery of the course. Please take time to read the following information carefully and decide whether you want to take part in this study. Please contact the researchers if anything is unclear or if you would like more information. Their contact details can be found at the bottom of this sheet.

What is the purpose of the study?

The study aims to capture, explore and understand the experiences of student nurses undertaking the Pre-Registration Nursing (Adult) programme and to co-design improvements with key stakeholders who manage and deliver the programme.

Why have you been invited to participate?

You are considered as a key stakeholder (i.e. course administrator, course year lead, module leader, lecturer, practice education facilitator or clinical placement co-ordinator) in relation to the provision of the course. Your experience and knowledge is valuable to offering relevant and realistic recommendations for improving the student experience.

Do you have to take part?

No. Taking part in this study is entirely voluntary and you are free to withdraw from the study at any time.

What if you wish to withdraw from this study?

You are free to withdraw at any time and without giving a reason. Please note that after the focus group/Experience Based Co-design event the data will be anonymised, therefore you will not be able to withdraw your responses as it will not be possible to isolate your individual response from the anonymised data.

What if you decide to take part?

Volunteers who agree to take part in this study will be invited to attend either a focus group, Experience Based Co-design event or both if you choose. Where staff and students meet together to discuss the findings generated from the student interviews and student focus group this will enable us to produce recommendations for improving the future student experience on the nursing course and implementing solutions to problems identified. Both the focus group and Experience Based Co-design event will be audio recorded to enable me to transcribe the discussion and prepare for later analysis. Please note that if you agree to take part in the focus group you are NOT under any obligation to take part in any further aspect of the study.
What are the possible benefits of taking part?

New knowledge generated from your participation in this study could enhance the design of future educational provision and contribute to future improvements of the Pre-Registration Nursing course.

What are the possible risks of taking part?

It is not anticipated that there will be any direct risks to you from participating in this study. Anonymity and confidentiality will be ensured as far as is possible, however please note you may find yourself participating in a focus group with other staff members you are familiar with.

Will what you say in this study be kept confidential?

Pseudonyms or unique codes will be allocated to all audio digital recordings, and transcribed data, to retain anonymity and protect confidentiality. The consent forms and audio digital recordings will be stored in a secure cabinet, in a locked office, on-site at the University of Central Lancashire (UCLan, 2015). The audio digital recordings will be retained until after transcription of the data generated in phase four of the data collection process, after which all digital recordings will be deleted. Transcribed data will be stored electronically on a password protected database in accordance with the Data Protection Act (DPA) (1998) and the UCLan Data Protection Policy (2015). This data will be kept for 5 years from the end of the study in accordance with the University's policy on Academic Integrity, after which it will be deleted or destroyed.

What will happen to the results of the research study?

The results of this study will be presented at conferences and written up for publication in academic journals, whilst also retaining the anonymity of the participants. In addition, data from this study will be used to form the final thesis of a PhD study. Journal publications and the final PhD thesis will be available in the public domain for participants to view.

Who is organising and funding the research?

Sam Pollitt (lead researcher) is organising this study as part of a PhD programme within the University of Central Lancashire. The researcher is self-funding this project. Sam is being supervised by an experience supervisory team; Dr Dawne Gurbutt, Dr Nigel Harrison and Dr Rob Monks.

Who has reviewed the study?

The research has been approved by the University Research Ethics Committee for Science, Technology, Engineering, Medicine and Health (STEMH), and has been reviewed by a supervisory team.

What should you do if you want to take part?

If you wish to take part in this research study please accept the invite to attend the event (remember you can withdraw your participation at any time).

Who do I contact if I have any complaints?

If you have any concerns or would like any further information about the study you can contact Sam Pollitt in the first instance. Concerns or complaints should be addressed to Dr Rob Monks as the study's Principal Investigator (<u>RMonks@uclan.ac.uk</u>) or the University Officer for Ethics at OfficerForEthics@uclan.ac.uk, information provided should include the study name or description (so that it can be identified), and the substance of the complaint.

Lead Researcher:	School of Nursing	Contact:
Sam Pollitt	Brook 349	E-mail: sjpollitt@uclan.ac.uk
	University of Central Lancashire	Tel: 01772 893624

Thank you for taking time to read the information sheet, I look forward to your response

PARTICIPANT CONSENT FORM - Emotional Touch Point Mapping event (Focus Group)

Title of Research Study:

Capturing, exploring, and understanding the experiences of undergraduate student nurses undertaking a Pre-registration nursing degree course: An Experience Based Co-design study

Name of Researcher: Sam Pollitt

		Please tick
1.	I consent to take part in the above study	
2.	I confirm that I have read and understand the participant information sheet	
	for the above study	
3.	I understand I am a voluntary participant, and I am free to withdraw from the	
	study at any point	
4.	I understand and give consent for the focus group to be audio recorded and	
	kept on record for up to five years	
5.	I understand that I will not be able to withdraw my audio recorded data from	
	the study due to the difficulties associated with isolating my voice and	
	comments from that of the other participants' in the focus group	
6.	I give consent to the use of samples of my anonymised written data in the	
	study	
7.	I consent to the use of my anonymised written data in publications / reports	

Name of Participant (please PRINT)	
Signature	
Participant email address	
Participant contact number (optional)	
Date:	
Gender (please circle) Male / Female	
Ethnicity (please Indicate)	
Name of Researcher	Sam Pollitt
Date:	Signature:

Participant Identification Code:

Appendix 13: Student and key stakeholder participant consent form for Codesign event

Student participant consent form - Co-design event

Title of Research Study:

Capturing, exploring and understanding the experiences of undergraduate student nurses undertaking a Pre-registration nursing degree course: An Experience Based Co-design study

Name of Researcher: Sam Pollitt

		Please tick
1.	I consent to take part in the above study	
2.	I confirm that I have read and understand the participant information sheet	
	for the above study	
3.	I understand I am a voluntary participant, and I am free to withdraw from the	
	study at any point	
4.	I give consent to the use of samples of my anonymised written data in the	
	study	
5.	I consent to the use of my anonymised written data in publications / reports	
6.	I consent for photographs of the Co-design event to be used within written	
	data in publications / reports	

Name of Participant (please PRINT)		
Signature		
Participant email address (optional)		
Participants contact number (optional)		
Date:	Cohort:	Year of study:
Gender (please circle): Male / Female	Date of birth:	Age:
Ethnicity (optional)		·
Name of Researcher		
Date:	Signature:	

Participant Identification Code:

Key stakeholder participant consent form - Co-design event

Title of Research Study:

Capturing, exploring and understanding the experiences of undergraduate student nurses undertaking a Pre-registration nursing degree course: An Experience Based Co-design study

Name of Researcher: Sam Pollitt

		Please tick
1.	I consent to take part in the above study	
2.	I confirm that I have read and understand the participant information sheet	
	for the above study	
3.	I understand I am a voluntary participant, and I am free to withdraw from the	
	study at any point	
4.	I give consent to the use of samples of my anonymised written data in the	
	study	
5.	I consent to the use of my anonymised written data in publications / reports	
6.	I consent for photographs of the Co-design event to be used within written	
	data in publications / reports	

Name of Participant (please PRINT)	
Signature	
Participant designation	
Participant email address	
Participant contact number (optional)	
Date:	
Gender (optional)	
Ethnicity (optional)	
Name of Researcher	Sam Pollitt
Date:	Signature:

Participant Identification Code:

Appendix 14: Problem statements tabulated from student and key stakeholder emotional touchpoint mapping events

Problem statement 1:

Some student participants within this study felt unsupported and even abandoned as they progressed through the course. Reasons for this include a perceived increase in self-directed learning linked with insufficient independent learning skills and decreased classroom-based contact time. Overall student participants indicated a need for more support and direction from academics with independent learning particularly in year three as the level of study became more challenging. Academics in this study were unsure how to achieve the right balance between supporting students with their academic studies and encouraging students to develop independent learning skills.

Question: How can we better facilitate independent learning skills?

Problem statement 2:

Student participants in this study expressed frustration with bad behaviour (incivility) from both academic staff and students in the classroom environment that made it difficult to concentrate and learn. Incivility included students talking in class, using mobile phones, 'scanning and scramming' and being generally disruptive. Some lecturers were perceived as failing to take control of disruptive classes or being overly controlling resulting in some students feeling intimidated and bullied.

Question: What can students and staff do to effectively manage incivility in the classroom environment?

Problem statement 3:

Some participants in this study felt unclear about the role of the academic advisor, others felt there were inconsistent approaches to the academic advisor role. This appeared to influence how well students felt supported by their academic advisor, and how comfortable they felt approaching their academic advisor or support.

Question: What would you like to see done to reduce perceived ambiguity associated with the role of the Academic advisor and improve the academic advisor role?

Problem statement 4:

Findings indicate support from mentors was amongst the most significant touchpoints that shaped students experiences on the course. Student participants reported mentors were pivotal in influencing their experience in the clinical practice area, both positively and negatively.

Question: How can we ensure and facilitate more positive mentoring experiences in clinical placements?

Problem statement 5:

Findings from this study indicate the nursing degree is a learning journey with an end result of professional registration, however key stakeholders have an expectation student nurses exhibit professional behaviour and attitudes from the start of the course and continue to develop their professionalism throughout the course. This expectation may contrast with student's self-perception as students whilst in a university setting and student nurses in the work-based learning environment. Having a clearer designation as a 'student nurse' may help students establish a distinct identity early in the course and facilitate the development of professionalism.

Question: how can students develop a distinct identity as a 'student nurse' whilst on the course and facilitate the development of professionalism?

Appendix 15: Map of the student journey through the course



Appendix 16: Example thematic analysis from Student Emotional touchpoint mapping event

Line	Student	Transcription from audio recorded interview following emotional mapping	Semantic coding:	Latent coding:	Description	Link to research question:	Theme:
1	10	I've got a lump in my throat, I feel upset. Because it highlights the difference between a good experience and a bad experience.	1. Upset 2. Good versus bad experience	 Extremes of experience Lack of parity between students' experiences of clinical placement 	After reading some of the positive comments other students have had in their clinical placement areas participant 010 becomes upset when reflecting on her own negative experiences. That difference has become stark through the process of emotional mapping.	Touchpoint - Clinical placement Emotion - upset	Clinical placement; disparity
2	10	When you discuss placements and you discuss how you got on with your mentoryou know with other peopleI get quite jealous that everybody's having this really, really good time.	1. Clinical placement 2. Mentor	1. Communication between peers uncovers the difference in experiences.	010 describes everyone else having a 'really good time' implying she is not 'having a good time', and feelings of jealousy. She does not describe feeling angry or confused that their experiences are different (better), but she is enviousness of their experiences	Touchpoint - Clinical placement Emotion - Jealousy of other students' experiences	Clinical placement; Mentor; disparity; peer communication
3	9	You're not on your own, I've had 2 out of 4 of my placements I've had a mentor that hasn't been as helpful as they could have been. But I think it's difficult because obviously hospitals are just so overwhelmed at the moment. But it's just difficult for the student because you don't want to intrude when they're busy with something. You're fighting to get your competencies signed off, or you just want to have a chat with them sometimes and it's just hard to find that time.	1. Clinical placement 2. Mentor 3. (clinical) workload	 Mentors Mentors unavailable Clinical /workload priorities (students lower on list) Student priorities versus staff priorities 	For student nurses having their competencies signed off is a mandatory requirement and key to them being able to qualify, 009 describes a dilemma she has whilst in placement - does she intrude (she justifies how qualified staff are busy and overwhelmed) or does she stand and do battle (fight) to have her own needs met? Finding 2 of her 4 mentors unhelpful based on the discussion that follows suggest unhelpful = unavailable. Mentors not investing time in students	Touchpoint - Clinical placement, completion of documentation Emotion - Contention (due to conflict of needs)	Clinical placement; Mentor
4	10	It's like the day before you leave the mentor says "oh come on we'll fill your book in now", "let's do this date, what date did you start?"	1. Completing documentation	 Last minute thought Devaluing the student / process 	010 describes having her mandatory documentation being left until the end of placement, with her mentor even back- dating some of the paperwork. This can leave a student feeling a low priority, devalued and even like a chore/burden to the mentor.	Touchpoint - Clinical placement, completion of documentation Emotion - devalued	Clinical placement; Mentor;
5	14	lt's very impersonal, you're made to feel like a burden.	1. Impersonal 2. Burden	 Mentor indifferent and disinterested Student a chore to be dealt with 	When something feels impersonal it can feel detached, lacking connection, emotion or warmth as opposed to something that feels personal (relate or belonging to). Thus 010 could be describing the way her paperwork is completed (last minute / back dated) as not truly relating directly to her (her performance / character / skills etc.)	Touchpoint - Clinical placement, completion of documentation Emotion - devalued / detached	Clinical placement; Mentor

Appendix 17: Post-it note comments from the student emotional touchpoint mapping event

Number of post-it notes	Touchpoint	Post-it notes comments	Negative Emotion words/phrases	Positive Emotion words/phrases
6	Mentor support	 2 out of 4 of my placements I have had a mentor who wasn't as helpful as they could have been. Some mentors don't have time for their students, and it can be a fight to get your PAD (Placement Assessment Document) signed (<i>Participant 009</i>) I was in a car accident with my backup mentor and informed her that I was going for physiotherapy she made it obvious by telling a patient in my presence that she thought I was wrong! The rest of the placement became untenable (<i>Participant 010</i>) Mixed mentors, sadly some of them have no interest in showing you anything or teaching you. I spent a whole placement finding my own things to see, learn and partake in (<i>Participant 011</i>) My mentors have been a huge inspiration and improved part of my learning. They communicated well, checking all areas needed to learn were covered and supported (<i>Participant 012</i>) I had a strong clash with my mentor whilst on placement review, they were shocked and said they would "sort it". When I returned to placement my mentor then made it evident that they had been informed about what I had said! Confidentiality? (<i>Participant 014</i>) I feel that there should be a better system in place to monitor mentorship skills and the placement are appropriate and integrated (<i>Participant 014</i>) 	 Unhelpful mentors Fight to have PAD signed Untenable placement Sad when mentors not interested in students Strong clash with mentor Feeling intimidated Feeling bullied 	 Inspirational mentor Mentors improved my learning
5	Academic staff	Lecturers - Some are awful! (<i>Participant 010</i>) It must be hard talking to a room full of people, but lecturers need to be stricter. Tell us to shut up if you need to, you're all too nicewhich is a good thing (<i>Participant 011</i>) Lecturers have mostly been approachable. However (lecturers) reading off a PowerPoint I feel it isn't as easy to take in the info. The lecturers who bring in experiences help (me) gain the understanding of the area (<i>Participant 012</i>)	 Awful lecturers Not easy to learn from PowerPoint presentations Challenging emotional journey 	 Nice lecturers Approachable lecturers Lecturers respected by students

		Lecture staff are very well respected. Their knowledge and guidance allow us to embrace this challenging emotional journey (<i>Participant 013</i>) Sometimes I wish lecturers would take more hold (control) of the room and speak up, because it is hard to hear and they sometimes just continue lectures, but I feel it is hard to learn in those circumstances (<i>Participant 015</i>)	4. Hard to learn when lecturers not in control of their class	
5	Facilitated learning sessions	 I am a more practical learner and feel personally more clinical skills time would be welcomed (Participant 009) Seminars are great in the 3rd year. Two home groups together, this allows diversity and extra opinions. The groups are still small enough to voice our opinions openly, as many wouldn't in the (large group) lectures (Participant 013) Important sessions are self-taught! Feel isolated and sometimes like I'm at an open university (Participant 013) I feel that taught sessions could have been timed differently to make sure relevant information is retained more effectively (Participant 014) Clinical skills I feel that there isn't time for practical (practice) because the lecturers discuss and talk for a long time. Those that learn practically struggle to listen for a long time, and clinical skills sessions are sometimes rushed (Participant 015) 	 Feel isolated doing directed study Struggles to listen during long sessions Feel clinical skills rushed 	1. Great seminars when combined with other Home groups
4	Academic advisor support	I have found my home group tutor to be a huge help to me, she even helps support me with my recent job application (<i>Participant 009</i>) Had a change of Personal tutor – (had) felt connected to the first tutor (<i>Participant 010</i>) When I've emailed with personal reasons my home group tutors been very understanding and supportive. Wish we had more home group meetings for questions or clarity, we are more open in them (<i>Participant 011</i>) I personal feel that home groups are more than sufficient. If I required a personal one-to-one, I would arrange this with my home group tutor (<i>Participant 013</i>)	-	 Academic advisor a huge helped Supported by Academic advisor Connection formed with Academic advisor Academic advisor Academic advisor understanding Academic advisor supportive

4	Administrative support	There has been a small number of occasions when I have emailed BrookHub regarding attendance and I haven't had a reply. It can become a concern where attendance is concerned (<i>Participant 009</i>) I was told by admin staff that I would have to produce my 'A' level certificate. I was really stressed and then spent nearly £60 on (replacement) certificates - the department didn't even look at them once I had them (<i>Participant 010</i>) There's been many times I've just been marked absent and I'm unsure why, maybe it's my card, they've (BrookHub) always been polite and sorted it for me, no problems (<i>Participant 011</i>)	 Concern about lack of communication from administrative staff Stressed about being asked to do something unnecessary Communication difficult with Placement 	1. Administrators polite and sort problems
		I felt until year 3, maybe end of year 2, the role for brook, they sorted all concerns out. Placement unit don't communicate well. Placement [unit] communication I find difficult <i>(Participant 012)</i>	unit	
4	Using the University Facebook pages	 Facebook is one of the biggest stressors at assignment time! (Participant 010) It's good when you first start to see/meet new people, however there are too many mean comments about lecturers/lessons. People give out wrong info and people worry more (Participant 011) Facebook should be banned, awful! (Participant 012) Facebook is very negative social site and opinions get voiced, especially regarding misinformation (Participant 013) 	 Facebook adds Stress Facebook awful should be banned Facebook makes people worry more Facebook a negative social site 	1. Facebook good for meeting people
4	Assessment Feedback	I don't feel that assignments are marked fairly. Lecturers' views can vary, and one person may have a different opinion to another (Lecturer) on assignment guidance (<i>Participant 009</i>) Every lecturer is different with feedback. Some give me hardly anything, some give you paragraphs! (<i>Participant 011</i>) I feel the feedback given depends on the marker. Talking to others (students) there doesn't feel much consistency (<i>Participant 012</i>) I found the style / amount of feedback to vary quite dramatically (<i>Participant 014</i>)	 Assignments marked unfairly Inconsistency feedback amongst lecturers Feedback varies dramatically 	-
4	Qualifying	I think placement experience can influence how prepared you feel about qualifying. I think the coming third year OSCEs will prepare me for qualifying, however I am very nervous (<i>Participant 009</i>) I feel that some newly qualified nurses have something to prove and can't understand how they can't remember that only a few months ago they were students! (<i>Participant 010</i>)	 Nervous about qualifying Jump from student nurse to qualified daunting 	1. Placement experience can prepare students for qualifying

		I think the jump from third year student nurse to qualified is daunting. I hope I have lots of support in my new job and I'm not expected to know it all straight away (<i>Participant 011</i>) As a 3rd year qualifying still feels a little away. The preparation Uni is offering doing the OSCE (NU3359), exams etc. I feel is ticking a lot of the boxes of the areas I was unsure about (<i>Participant 012</i>)		
4	Module Assessment Group Support (MAGS)	I don't find Group Assignment supervision not as supportive as individual supervision, sometimes I leave feeling more confused! (<i>Participant 009</i>) I've found that for quieter people, or people who are shy, Group supervision means they miss out (<i>Participant 010</i>) I feel the Group Assignment supervision are a waste of time. Difficult to get questions over as too many in the group. I feel this hasn't helped my development in assignments (<i>Participant 012</i>) I don't feel that group supervision is as supportive as individual supervision (<i>Participant 014</i>)	 Individual supervision more supportive than MAGS MAGS leave me feeling Confused Quieter people miss out during MAGS MAGS a waste of time Difficult to ask questions in large group Unhelpful with assignment development MAGS less supportive than individual supervision 	-
3	Module assessments	I am better in placements and learning hands-on. The amount of theory is overwhelming. I understand we need to know, and nurses are trained to do a lot more now, but sometimes I think I'm training to be a doctor (<i>Participant 011</i>) Anatomy and physiology units were a big eye opener. I felt it very difficult to understand, and in year 1 out of my depth! Things all fell into place in the 2nd year and my understanding developed (<i>Participant 012</i>) I thoroughly enjoyed the ethics lectures and found it an interesting part of the course. I got a good result for my assignment which reflected my enjoyment of the module (<i>Participant 012</i>)	 Amount of theory is overwhelming Anatomy and physiology difficult to understand In year 1 out of my depth 	 Things fell into place in the 2nd year / developed understanding Good assignment result reflected enjoyment of the module

3	Virtual learning environment (e.g. Blackboard)	No problems (with Blackboard) just wish PowerPoints were uploaded with lecturer talking over for extra information, better for revision (<i>Participant 011</i>) The structure on blackboard is easy to navigate. Wishful in thinking here – all PowerPoints to be uploaded prior to the session as this helps me focus on the content. Avoid 83 pages of pdf, we can't amend them (<i>Participant 013</i>) Blackboard is scary to look at and it isn't always clear where you need to go, but once you've played around (with it) it's really beneficial and helpful. The materials on Blackboarddepending on the lecturer (who has uploaded it)can be more helpful, so upload more resources (<i>Participant 015</i>)	 Blackboard is scary Blackboard is unclear 	 Blackboard is easy to navigate Thoroughly enjoyed the ethics lectures Material on Blackboard can be helpful
3	Home group contact / meetings	 When I've emailed with personal reasons my home group tutors been very understanding and supportive. Wish we had more home group meetings for questions or clarity, we are more open in them (<i>Participant 011</i>) I personal feel that home groups are more than sufficient. If I required a personal one-to-one, I would arrange this with my home group tutor (<i>Participant 013</i>) I feel well supported in home group meetings. However, I feel that a taught session may develop a stronger relationship as there are not many home group sessions over the course (<i>Participant 014</i>) 	-	 Academic advisor understanding and supportive is scary I feel well supported
3	Course enquires / Interview and selection process	In comparison to other interviews UCLan offered a far better personal experience, took away some of the anxiety due to the group session (group interviews) this made me feel relaxed and able to contribute as me easier (<i>Participant 012</i>) The interview was structured and professional, extremely compassionate towards all potential students (<i>Participant 013</i>) I only got an interview at UCLan, but it was my first choice, so I was pleased. I'd changed course last minute so was happy I'd got on the course I'd wanted. When I came for my interview, they weren't expecting me as my name wasn't on the list. I was a little embarrassed (<i>Participant 015</i>)	1. Not expected for interview so felt a little embarrassed	 Took away some of the anxiety Made me feel relaxed Pleased to get an interview at UCLan Happy I'd got on the course I'd wanted
3	Transition into year one - Welcome week	The huge lecture room was very intimidating. Very difficult to concentrate, so many students. However, I was here something I felt a huge achievement for (<i>Participant 012</i>) Welcome event was absolutely amazing (<i>Participant 013</i>) The huge lecture was a little daunting for me and what I was used to, which was small groups in classes. It put me off a little bit, but as the week went on and we met our home groups it was easier and more welcoming (<i>Participant 015</i>)	 Huge lecture room intimidating Difficult to concentrate with so many students Huge lecture daunting 	 Welcome event was absolutely amazing Thoroughly enjoyed the ethics lectures

				3. Meeting our home groups made it easier and more welcoming
2	Transition into year two	I found the beginning of year 2 quite a shock! Year 1 seemed so easy (Participant 009) Second year is a huge jump compared to year 1, not prepared for it either. 3rd year I do feel very much left on your own. I'm disappointed with this as this is when you need additional support (Participant 012)	 The beginning of year 2 quite a shock! Not prepared for year 1 Feel very much left on your own Disappointed with year 3 	
2	Transition into year three	I feel Competent and achieved reaching my goal (Participant 013) I feel abandoned in year 3 (Participant 014)	1. I feel abandoned	1. I feel Competent
2	First clinical placement (subsequent clinical placements)	I felt at a disadvantage in placement being older! (<i>Participant 010</i>) Wish placements were selected better. I had orthopaedics twice, then rehab which was patients from orthopaedics! I'm going into year 3 and just getting my first medical ward. (<i>Participant 011</i>)	 Feeling disadvantaged for being a mature student 	-
0	Practice Education Facilitator (PEF) support	No post-it note comments	-	-

Appendix 18: Comments' student participants strongly agree with

Touchpoint	Comment	Strongly agree	Total number of sticker stickers
Academic staff	I love it when we have a strict lecturer or one that goes "shhh" because it's not fair on the people who want to take notes and learn. There are certain lecturers who I know we will get a good lecture today because you know people won't get away with it.	6	6
Facilitated learning sessions	Clinical skills are so valuable, and we feel there should be more in programme. I know it's difficult as there is so much theory for us to go through but if more clinical skills could be built in that would be fabulous	5	5
Course enquires / Interview and selection process	I came for my interview and got accepted - it was very scary in the interview because it's so nerve-wracking. Obviously, I did like the system because you have your personal one-to-one interview, and you have a large group (interview) and that's good if you've got a strong character - but we had one really strong character, and other people couldn't get a word in.	5	5
Transition into year three	It's been a struggle for sure. There's been a few times where I've thought of quitting. I think everybody has at some point.	5	5
Transition into year two	You get through the first year and you feel like you're on top of the world and then the second year knocks you back down because it is really hard, and you have to stay focused	4	4
Qualifying	Nervous! Very excited, can't wait but very nervous at the same time because all of a sudden, you've got that full responsibility.	4	8
Qualifying	It's exciting to think that I've done two years I'm coming to the end because I know it goes really quickly. Everyone says third year you blink, and you missed it!	4	
Facebook	Well I've left the Facebook group now because everything was just irritating me! It went through a big thing of people moaning and bitching about people talking in lectures, then there's been other bits of things on there as well, and I just thought it's a bit unprofessional for people to be talking that way in a group were lecturers and other people in your course can see	4	4
Welcome week	It was brilliant, absolutely brilliant. It was hard going but I think that's because I'm not used to sitting in lecture halls. They had planned so much, and it was so informative, but it was a really good idea. I felt a bit blagged by the end of it and my head was a bit!!!	4	4

Appendix 19: Comments' student participants strongly disagree with

Touchpoint	Comment	Strongly disagree
Academic staff	For example, there's a lecturer who is very grumpy - I call him the grumpy-saurus-rex because he's very moody - and there's been some incidences during lectures when he's been absolutely out of line, and I get very, very frustrated when there are incidences where I think if I had behaved like you, I would be in so much trouble, and I wouldn't get away with it, but you have	3
Mentor support	My mentor was fantastic, she was always challenging me, not challenging me in a nasty way, but the way that she was with me it made me want to go and learn	2
Module Assessment Group Support	I've also found as well the group supervision for the essays, I've found that supportive as well.	2

Number of post- it notes	Touch point	Post-it notes comments	Negative Emotion words/phrases	Positive Emotion words/phrases
8	Transition into year one - Welcome week	If students can't commit to week one, it's disappointing! As a professional in training they need to commit to that mindset from day one! KS001 (KS007 added to this post-it note - However, is this more widely perceived? Do we need to change the labelling of the welcome week to reflect the importance?) Concerned [about] students view! May need to involve admin team in welcome week KS002 I'm annoyed that some students attending a professional program in a community like University don't want to engage in all the events offered [Welcome week] KS003 Just how important is welcome week? Over the years welcome weeks have been diluted somewhat, but it's cleared here how we need to get it right! What we class as transition? Where does it begin and end? How do we support students other than welcome week? KS004 Very proud to hear how welcome week was amazing, it really is a crucial time for university to get it right KS004 (KS006 added to this post-it note - Absolutely agree, first impressions, starting positive) We need to make sure students know what is involved in welcome week, encourage to attend KS005 Large lecture theatre – difficult to hear how it makes students feel daunted/fearful KS006 This shows the importance of a good welcome week. We show ourselves During the interviews And we must carry that on in welcome week! KS007	 Its disappointing Concerned about student views I'm annoyed [student's don't turn up to welcome week] Difficult to hear students feel daunted / fearful 	1. Proud hear how welcome week was amazing
5	Academic staff	It's not being strict, it's firm but fair, to create a learning environment it's a professional course and whilst adults should the nursing want to learn social spaces of the socializing. I like lots of discussion not chat about the night before. It is learning professional values that transfer to the ward environment KS002 I'm glad that a majority of students appreciate the difficulty academic staff experience in keeping everyone engaged KS003 Disgusting! Grumpy lecturers – really? Does not seem appropriate for this to be evident for students to pick up on – this will impact on the learning KS004	1. Disgusting, grumpy lecturers!	 Nice to have validation of a job well done Glad students appreciate academic staff

Appendix 20: Post-it note comments from the key stakeholder emotional touchpoint mapping event

		Professional course – parent to parent relationship – don't want to have a parent-child relationship – respect both ways – student comments about lecturers to be more "stricter" – ground rules – engaging students interesting sessions KS006 It's really nice to have validation of a job well done KS007		
5	Academic advisor support	Consistency of support from staff is disappointing! Over delivering can raise expectations beyond what can be delivered, and we need to take care that we get the balance between support and empowerment KS001 Disappointing to hear the disparity in support/connection with personal tutor. This role can often contribute to building students resilience and ability for success KS002 How sad when students do not know their personal tutor, it does happen but the impact of it is fast, very sad! KS004 It's a shame that the students did not feel supported by the home group tutor. Academic adviser support, the outdoor education days, especially when held near the start of the course encourages bonding and	 Inconsistent support from staff disappointing Disparity of staff support disappointing Sad when students do not know their personal tutor, very sad A shame students feel unsupported 	-
		strong group identities KS005 Importance of home group tutors by students – ensuring basics right, knowing each student's name – difficulties encountered when having three home groups, two is enough – some students need additional support – need to factor in time when students need it KS006	5. Difficult when Academic advisors have too many Home groups	
4	Mentor support	I think mentors need to be supported with more than annual updates in terms of development. They are not necessarily teachers as they are qualified nurses – it's a skill (teaching) KS002 Upsetting to read about poor mentors, kind of makes me angry – I was a good mentor so it's sad when students do not get good experience! This is their dream, to be a nurse – we need brilliant mentors! KS004 (KS002 added to this post-it note - Totally agree) Sad feeling regarding some mentors not being supportive – placement support – perceptions of student support in placement – exploring the bigger picture KS006 Sad to hear that a mentor could be lax! I will be interested to hear in what way? KS007	 Upsetting to read about poor mentors Makes me angry It's sad Sad feeling Sad to hear 	-
4	Administrative support	Very disappointing to hear that placement communication was difficult. More disappointing that this was not escalated to staff to allow this to be addressed to improve experience KS001	1. Disappointing to hear	

		Concerned [about] students view! May need to involve admin team in welcome week KS002 I'm frustrated that the admin role and support available is not clear enough to students, not necessarily anything that can be easily rectified as it covers so many elements KS003 Students have no knowledge of what the admin team actually do and that makes me feel undervalued and unimportant KS003	 2. Concerned [about] students view 3. Frustrated the admin role not clear to students 4. Feel undervalued and unimportant 	-
4	Using the University Facebook pages	Good to hear the student voice re-Facebook and that their frustrations are the same as ours KS001 I'm glad that students agree regarding consequences of use of social media inappropriately in terms of where to get academic answers KS002 I'm relieved to hear that some people are aware that Facebook can do more harm than good and is not the best place to seek support KS003 Digital forums have a place but it can create confusion with so many students voicing their opinions – I can see why students opt out! KS004	-	 Good to hear the student voice Glad students agree regarding social media Relieved students aware of harm related to Facebook use
4	Assessment Feedback	Need to focus on the purpose of feedback – maybe we need to rephrase what it's called. It's disappointing that so much commitment to someone's attainment is viewed so negatively KS001 The new marking standards will follow learning outcomes/the feedforward should address some of these issues KS002 I think students' expectations of how much feedback is given is difficult to judge for markers KS003 Sad to hear that we might not be managing expectations enough KS007	 Disappointing feedback viewed negatively Students' expectations difficult to judge Sad expectations not being managed 	-
4	Module assessments	Individual supervision does not aid progression, sad that the message of the effect on attainment has not been taken on board KS001 More work may be needed before students transition to higher education around what it will be like. New services such as academic coach could help with this KS002 Some students seem ignorant of how much of the actual job relies on academic ability KS003	1. Sad	-

		Having previous care experience isn't always a good thing. Think of a clean slate and not having any bad habits KS005		
4	Transition into year three	Sad that student's feelings about support and quitting in terms of transition into year three are as such. Reminders of why and the work of a degree seem to be lost when??? KS001 Really sad to hear students felt alone and unsupported, maybe students don't always see support until its really needed rather than knowing it's there. Also this could be around their expectations, especially if previous education 'spoon fed' method of teaching and learning! KS002 Abandoned in year three! Very sad to hear this, too much directed study and workbooks and a lack of timetabled sessions isn't good! KS004 Year three, good to hear this feedback but there are lots of support mechanisms in place. This is when you should be coming into your own! Perhaps a reminder pre-year three of who is there to help KS005	 Sad students felt like quitting Really sad students feel unsupported Very sad students feel abandoned in year 3 	1. Good to hear this student feedback
2	Transition into year two	I'm concerned that students going into second year feel unprepared. Throughout year one they are made aware that year two will be far more challenging KS003 Concerned we may not be supporting our mature students enough academically. But what constitutes 'enough' support in adult learning? KS007	 Concerned students feel unprepared Concerned about the support of mature students 	-
2	Facilitated learning sessions	Affirmation of what I know! Students always love skills – they see the need for the job, they struggle with the fact they need to know the AMP to underpin some skills and nursing management. Also, if they want to develop post-qualifying i.e. two advanced clinical practitioner they need firm foundations they can build on may help this KS002 It's nice to hear honest feedback from students about lectures. You don't always feel that the feedback you get is completely honest KS007	-	 Affirmation Nice to have student feedback
2	Virtual learning environment (e.g. Blackboard)	Experiential learning – developing skills on blackboard – encourages students to use every opportunity KS006 Importance of communication between home group tutor/personal tutor/academic advisors and students – issues with Facebook – approachable staff so students can ask the questions post a discussion on boards KS006	-	-

1	Qualifying	Changing mindsets in the NHS is needed – sad to hear that upon qualifying students already feel unhappy and intimidated – no wonder we have 40,000 vacancies! KS004	 Sad students feel unhappy and intimidated when qualifying 	-
1	Module Assessment Group Support (MAGS)	Group supervision should be supportive and successful – makes you think about year one again and question are we really embedding the skills needed to study independently! KS004		
1	Course enquires / Interview and selection process	Happy to hear that UCLan provided a more positive experience for interviewing compared to other universities KS007	-	1. Happy UCLan provided a positive experience
1	First clinical placement (subsequent clinical placements)	First weeks in placement are about putting everything you have learnt in unit into use KS005	-	-
1	Practice Education facilitator support	Gap in comments related to PEFs is significant - but relates directly to the role of the PEF KS001	-	-
0	Home group contact / meetings		-	-

Appendix 21: Comments key stakeholder participants strongly agreed with

Touchpoint	Comment	Strongly agree	Total number of sticker stickers
Course enquires / Interview and selection process	In comparison to other interviews UCLan offered a far better personal experience, took away some of the anxiety due to the group session (group interviews) this made me feel relaxed and able to contribute as me easier.	3	3
MAGS	I went to group supervision with her and she really broke it down for me, and afterwards it made so much more sense	3	3
Qualifying	Nervous! Very excited, can't wait but very nervous at the same time because all of a sudden you've got that full responsibility.	2	8
Qualifying	I'm a little bit scared to go into being a qualified nurse but you know I do feel quite prepared	2	-
Qualifying	As a third year qualifying still seems a little away. The preparation Uni is offering during the OSCE exams etc I feel is ticking a lot of the boxes of the areas I was unsure about	2	
Qualifying	I think our coming OSCE will prepare me for qualifying. However, I am very nervous!	2	-
Academic staff	I love it when we have a strict lecturer, one that goes "shush" because it's not fair on the people who want to take notes and learn. There are lecturers who I know we will get a good lecture today because you know people won't get away with it	2	4
Academic staff	I think the lecturers are all very approachable and you can go to any of them if you need to talk to them. You know they want you to do well	2	-
Welcome week	Welcome event was absolutely amazing	2	2
Course enquires / Interview and selection process	The interview was structured and professional, extremely compassionate towards all potential students	2	2
Virtual learning environment	On the whole blackboard is quite easy to use	2	2
Facilitated learning sessions	I quite enjoy clinical skill sessions because it feels like you're allowed to make mistakes, you know you're learning in an environment where everyone is the same	2	2
Mentor support	You're not on your own, I've had 2 out of 4 of my placements I've had a mentor that hasn't been as helpful as they could have been. But I think it's difficult because obviously hospitals are just so overwhelmed at the moment. But it's just difficult for the student because you don't want to intrude when they're busy with something. You're fighting to get your competencies signed off, or you just want to have a chat with them sometimes and it's just hard to find that time	2	2

Appendix 22: Comments key stakeholder participants strongly disagree/felt saddened by

Touchpoint	Comment	Strongly disagree / saddened by	Total number of sticker
Transition into year three	It does feel like an open university at the moment, that's what I'll say. It feels like an open university, self- taught teaching right now. And if I wanted to do an open university course I would have applied for one! That's our opinion from studying at college to this. I do feelwe are older adults aren't webut we do need extra knowledge and understanding and more depth in what we're doing	4	4
Transition into year three	I feel abandoned in year three	4	-
Mentor support	I was in a car accident with my backup mental and informed her that I was going for physio. She made it obvious by telling a patient in my presence that she thought I was wrong. The rest of the placement became untenable!	4	4
Clinical placements	I felt at a disadvantage in placement being older!	3	3
Home group tutor	I've never had a personal tutor say "is everything alright?" (Participant 010)	3	5
	I don't even think my personal tutor knows my name (012 turns to participant next to her and asks) does your personal tutor no your name? (Participant 012)		
Home group tutor	One of the things that I think could possibly be done differently is with the home group, I personally feel like I have maybe seen my home group tutor twice. I highly doubt she knows who I am from anybody else, I don't think she really knows any of us by name	2	
Academic staff	For example, there's a lecturer who is very grumpy - I call him the grumpy-saurus-rex because he's very moody - and there's been some incidences during lectures when he's been absolutely out of line, and I get very, very frustrated when there are incidences where I think if I had behaved like you I would be in so much trouble, and I wouldn't get away with it, but you have	2	2
Welcome week	I didn't come to welcome week, I suppose in my eyes I thought it was just for people that drink, and I don't really drink, so I just thought I wouldn't be able to get involved, so I just thought 'well there's no point! So when it came to the large group lecture days I did fell like 'Oh God' I wish I had come to welcome week really!	2	2
Clinical placements	Wish placements were selected better. I had orthopaedics twice, then rehab which was patients from orthopaedics! I'm going into year 3 and just getting my first medical ward. Mixed mentors, sadly some of them have no interest in showing you anything or teaching you. I spent a whole placement finding my own things to see, learn and partake in	2	6
Clinical placements	The first couple of weeks I learned as much of my placement as what I'd done those first few months at uni!	2	-
Clinical placements	You're not on your own, I've had 2 out of 4 of my placements I've had a mentor that hasn't been as helpful as they could have been. But I think it's difficult because obviously hospitals are just so overwhelmed at the moment. But it's just difficult for the student because you don't want to intrude when they're busy with something. You're fighting to get your competencies signed off, or you just want to have a chat with them sometimes and it's just hard to find that time	2	
Qualifying	I feel that some newly qualified nurses have something to prove, and I can't understand how they can't remember that only a few months ago they were students!	2	2
Transition into year three	It's been a struggle for sure. There's been a few times where I've thought of quitting and that's just honest. I think everybody has at some point	2	2
Assessment feedback	I don't feel that assignments are marked fairly. Lecturer's views can vary, and one person may have a different opinion to another on assignment guidance	2	4
Assessment feedback	I got that one back and there was SO much feedback on it, it literally felt like 'oh my gosh how have I even managed to write the title' And as well with the feedback, probably more the way that I've interpreted the feedback, it felt very negative to me. It wasn't like they could have spun it with a little bit of a positive, they could have said I'd used the right font or something like that (student laughs). I just felt that it was just like 'this is bad, that's bad' that type of thing 'this isn't right, that isn't right' and it was all stuff like referencing and things like that, which on my first one I'd had no comments about. It made me cry. It made me question if I should still be doing the course or not if my academic writing was that bad that I was getting everything wrong from grammar to structure to referencing. All these different things wrong?	2	